UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

FORM 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2008

Commission File Number 001-08106

Mastec, INC.

(Exact name of registrant as specified in Its charter)

Florida	65-0829355
(State or other jurisdiction of	(I.R.S. Employer
incorporation or organization)	Identification No.)
800 S. Douglas Road, 12 th Floor, Coral Gables, FL	33134
(Address of principal executive offices)	(Zip Code)

Registrant's telephone number, including area code: (305) 599-1800

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. Yes \square No o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer or a smaller reporting company. See the definitions of "large accelerated filer," "accelerated filer," and "smaller reporting company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer \square	Accelerated filer o	Non-accelerated filer o (Do not check if a smaller reporting company)	Smaller reporting company o
-----------------------------------	---------------------	---	-----------------------------

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes o No 🗵

As of July 25, 2008, MasTec, Inc. had 67,215,973 shares of common stock, \$0.10 par value, outstanding.

MASTEC, INC. FORM 10-Q QUARTER ENDED JUNE 30, 2008

TABLE OF CONTENTS

PART 1	. FINANCIA	AL INFORMATION	3
	<u>Item 1.</u>	Financial Statements	3
		Condensed Unaudited Consolidated Statements of Operations	3
		Condensed Consolidated Balance Sheets	Z
		Condensed Unaudited Consolidated Statements of Cash Flows	5
		Notes to the Condensed Unaudited Consolidated Financial Statements	6
	<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	18
	<u>Item 3.</u>	Quantitative and Qualitative Disclosures about Market Risk	25
	<u>Item 4.</u>	Controls and Procedures	26
PART I	<u>I. OTHER II</u>	NFORMATION	26
	<u>Item 1.</u>	Legal Proceedings	26
	<u>Item 1A.</u>	Risk Factors	26
	<u>Item 2.</u>	Unregistered Sales of Equity Securities and Use of Proceeds	28
	<u>Item 4.</u>	Submission of Matters to a Vote of Security Holders	28
	<u>Item 5.</u>	Other Information	28
	<u>Item 6.</u>	<u>Exhibits</u>	29
<u>SIGNA</u>	<u>TURES</u>		30
		2	

PART 1. FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

MASTEC, INC. CONDENSED UNAUDITED CONSOLIDATED STATEMENTS OF OPERATIONS

(In thousands, except per share amounts)

	For the Three Months Ended June 30,		For the Six M June	
	2008	2007	2008	2007
Revenue	\$305,034	\$256,284	\$567,026	\$497,280
Costs of revenue, excluding depreciation	259,561	213,327	486,405	424,348
Depreciation	6,017	4,082	10,805	7,862
General and administrative expenses, including non-cash stock compensation expense of \$1,105 and \$1,949, respectively, in 2008 and \$1,500 and \$3,467,				
respectively, in 2007	19,966	20,234	40,012	39,482
Interest expense, net of interest income	3,656	2,120	6,152	4,915
Other (income) expense, net	(394)	(573)	(545)	(4,057)
Income from continuing operations before provision for income taxes and				
minority interest	16,228	17,094	24,197	24,730
Provision for income taxes	407	_	440	_
Minority interest	<u> </u>	1,035	<u> </u>	1,652
Income from continuing operations	15,821	16,059	23,757	23,078
Loss from discontinued operations	(85)	(158)	(240)	(5,507)
Net income	\$ 15,736	\$ 15,901	\$ 23,517	\$ 17,571
Basic net income (loss) per share:				
Continuing operations	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.35
Discontinued operations	<u> </u>	<u> </u>	<u> </u>	(0.08)
Total basic net income per share	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.27
Basic weighted average common shares outstanding	67,207	65,854	67,197	65,634
Diluted net income (loss) per share:				
Continuing operations	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.34
Discontinued operations	—	—	—	(0.08)
Total diluted net income per share	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.26
Diluted weighted average common shares outstanding	68,182	67,431	68,123	67,075

The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.

MASTEC, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

(In thousands, except share and per share amounts)

		December 31, 2007 (Audited)	
Assets	(Unaudited)	(ruuneu)	
Current assets:			
Cash and cash equivalents, including restricted cash of \$18,050 at June 30, 2008 and December 31, 2007	\$ 46,955	\$ 74,288	
Securities available for sale	2,000	44,360	
Accounts receivable, unbilled revenue and retainage, net	198,278	160,089	
Inventories	23,976	32,402	
Deferred tax asset	4,176	26,467	
Prepaid expenses and other current assets	30,470	29,801	
Total current assets	305,855	367,407	
Property and equipment, net	121,477	81,939	
Goodwill and other intangibles, net	230,395	202,829	
Deferred taxes, net	52,677	30,386	
Securities available for sale	26,888	_	
Other assets	25,545	28,188	
Total assets	\$ 762,837	\$ 710,749	
Liabilities and Shareholders' Equity			
Current liabilities:			
Current maturities of debt	\$ 12,538	\$ 2,694	
Accounts payable and accrued expenses	130,106	133,861	
Accrued legal settlement charges	25,056	21,269	
Accrued insurance	13,341	16,645	
Other current liabilities	31,226	29,126	
Total current liabilities	212,267	203,595	
Other liabilities	30,955	32,310	
Long-term debt	181,492	160,279	
Total liabilities	424,714	396,184	
Commitments and contingencies			
Shareholders' equity:			
Preferred stock, \$1.00 par value; authorized shares — 5,000,000; issued and outstanding shares — none			
Common stock, \$0.10 par value; authorized shares — 100,000,000; issued and outstanding shares — 67,208,645 and			
67,174,171 shares at June 30, 2008 and December 31, 2007, respectively	6,721	6,717	
Capital surplus	554,507	552,491	
Accumulated deficit	(216,059)	(239,576	
Accumulated other comprehensive loss	(7,046)	(5,067	
Total shareholders' equity	338,123	314,565	
Total shareholders' equity			
Total liabilities and shareholders' equity	\$ 762,837	\$ 710,749	

The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.

MASTEC, INC. CONDENSED UNAUDITED CONSOLIDATED STATEMENTS OF CASH FLOWS

(In thousands)

		Aonths Ended e 30,
	2008	2007
Cash flows from operating activities:		
Net income	\$ 23,517	\$ 17,571
Adjustments to reconcile net income to net cash provided by operating activities:		
Depreciation and amortization	11,606	8,134
Impairment of goodwill and assets		572
Non-cash stock and restricted stock compensation expense	1,949	3,467
Gain on sale of fixed assets	(323)	(3,670
Provision for doubtful accounts	1,787	1,705
Minority interest and income from equity investment	—	1,533
Changes in assets and liabilities, net of assets acquired:		
Accounts receivable, unbilled revenue and retainage, net	(29,470)	8,898
Inventories	(6,172)	8,364
Other assets, current and non-current portion	(2,193)	10,507
Accounts payable and accrued expenses	9,165	(21,944)
Other liabilities, current and non-current portion	4,030	(7,882
Net cash provided by operating activities	13,896	27,255
Cash flows (used in) investing activities:		
Cash paid for acquisitions, net of cash acquired	(59,186)	(11,213)
Capital expenditures	(16,104)	(14,813)
Investments in unconsolidated companies	—	(1,025
Investments in life insurance policies	(569)	(539)
Net proceeds from sale of assets	1,619	3,544
Purchases of securities available for sale	(16,437)	(285,523)
Proceeds from sales of securities available for sale	29,924	296,575
Net cash used in investing activities	(60,753)	(12,994
Cash flows provided by financing activities:		
Proceeds from other borrowings, net	20,638	149,187
Payments of capital lease obligations	(1,178)	(942
Payments of senior subordinated notes	_	(121,000
Proceeds from issuance of common stock pursuant to stock option exercises	71	3,749
Payments of financing costs		(3,794
Net cash provided by financing activities	19,531	27,200
Net (decrease) increase in cash and cash equivalents	(27,326)	41,461
Net effect of currency translation on cash	(7)	9
Cash and cash equivalents — beginning of period	74,288	35,282
Cash and cash equivalents — end of period	\$ 46,955	\$ 76,752
Cash paid during the period for:		
Interest	\$ 6,264	\$ 6,416
Income taxes	\$ 551	\$ 0,410
Supplemental disclosure of non-cash information:	φ 251	¢ /4
Equipment acquired under capital lease	¢ 000	¢ = 017
	\$ 880 \$ 22 280	\$ 5,317 \$ 11,132
Accruals for inventory at quarter-end	\$ 22,280	\$ 11,13

The accompanying notes are an integral part of these condensed unaudited consolidated financial statements.

Notes to the Condensed Unaudited Consolidated Financial Statements

Note 1 — Nature of the Business

MasTec, Inc. (collectively, with its subsidiaries, "MasTec" or the "Company") is a leading specialty contractor operating mainly throughout the United States and across a range of industries. The Company's core activities are the building, installation, maintenance and upgrade of communications and utility infrastructure, including but not limited to, fiber deployment, electrical utility transmission and distribution, water and sewer, and wind farm and natural gas pipeline construction. MasTec's primary customers are in the following industries: communications (including satellite television and cable television), utilities and government. The Company provides similar infrastructure services across the industries it serves. Customers rely on MasTec to build and maintain infrastructure and networks that are critical to their transport and delivery of voice, video and data communications, electricity and other energy resources.

Note 2 — Basis for Presentation

The accompanying condensed unaudited consolidated financial statements have been prepared in accordance with accounting principles generally accepted in the United States for interim financial information and with the instructions for Form 10-Q and Rule 10-01 of Regulation S-X. Accordingly, these financial statements do not include all information and notes required by accounting principles generally accepted in the United States for complete financial statements and should be read in conjunction with the audited consolidated financial statements and notes thereto included in the Company's Form 10-K for the year ended December 31, 2007. In management's opinion, all adjustments necessary for the fair presentation of the financial position, results of operations and cash flows for the periods presented have been included.

The preparation of financial statements in conformity with accounting principles generally accepted in the United States requires management to make estimates and assumptions that affect the amounts reported in the financial statements and accompanying notes. Key estimates include the recognition of revenue, allowance for doubtful accounts, accrued self-insured claims, the fair value of goodwill, intangible assets and securities available for sale, asset lives used in computing depreciation and amortization, including amortization of intangibles, and accounting for income taxes, contingencies and litigation. While management believes that such estimates are fair when considered in conjunction with the consolidated financial position and results of operations taken as a whole, actual results could differ from those estimates and such differences could be material to the financial statements.

Note 3 — Significant Accounting Policies

(a) Principles of Consolidation

The accompanying financial statements include MasTec, Inc. and its subsidiaries. For the three and six month periods ended June 30, 2007, GlobeTec Construction, LLC ("GlobeTec") was consolidated as the Company had a 51% controlling interest in this entity. Other parties' interest in GlobeTec was reported as minority interest in the condensed unaudited consolidated financial statements for such period. During 2007, the Company acquired an additional 45% ownership interest in GlobeTec, and during the first quarter of 2008, the remaining 4% interest in GlobeTec was acquired bringing MasTec's ownership interest in this entity to 100%. All intercompany accounts and transactions have been eliminated in consolidation.

(b) Comprehensive Income

Comprehensive income is a measure of net gain (loss) and all other changes in equity that result from transactions other than with shareholders. Comprehensive income consists of net income, foreign currency translation adjustments, and unrealized gains and losses on securities available for sale.

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

Comprehensive income consisted of the following (in thousands):

		For the Three Months Ended June 30,				Six Months Ended June 30,	
	2008	2007	2008	2007			
Net income	\$ 15,736	\$ 15,901	\$ 23,517	\$ 17,571			
Foreign currency translation gain (loss)	(7)		(7)	9			
Unrealized gain (loss) from securities available for sale	822		(1,972)				
Comprehensive income	\$ 16,551	\$ 15,901	\$ 21,538	\$ 17,580			

(c) Basic and Diluted Net Income Per Share

The following table sets forth the computation of basic and diluted net income per share from continuing operations for the three and six months ended June 30, 2008 and 2007 (in thousands):

	For the Three Months Ended June 30,		For the Six Months Endec	
	2008	2007	2008	2007
Basic net income from continuing operations per share computation:				
Net income from continuing operations	\$ 15,821	\$ 16,059	\$ 23,757	\$ 23,078
Weighted average common shares outstanding	67,207	65,854	67,197	65,634
Basic net income per share from continuing operations	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.35
Diluted net income from continuing operations per share computation:				
Net income from continuing operations	\$ 15,821	\$ 16,059	\$ 23,757	\$ 23,078
Weighted average common shares outstanding	67,207	65,854	67,197	65,634
Incremental shares attributable to the assumed exercise of outstanding options and unvested restricted stock (common share equivalents)	975	1,577	926	1,441
Total diluted weighted average shares	68,182	67,431	68,123	67,075
Diluted net income per share from continuing operations	\$ 0.23	\$ 0.24	\$ 0.35	\$ 0.34

(d) Valuation of Goodwill and Intangible Assets

In accordance with Statement of Financial Accounting Standards ("SFAS") No. 142, "*Goodwill and Other Intangible Assets*" ("SFAS 142"), the Company conducts, on at least an annual basis, a review of reporting entities to determine whether the carrying values of goodwill exceed the fair market value using a discounted cash flow methodology for each entity. Should this be the case, the value of its goodwill may be impaired and written down. Goodwill acquired in a purchase business combination and determined to have an infinite useful life is not amortized, but instead tested for impairment at least annually in accordance with provisions of SFAS 142. In addition, acquired intangible assets are recognized and amortized over their useful lives if the benefit of the asset is based on contractual or legal rights.

During the three and six month periods ended June 30, 2008, the Company recorded approximately \$24.9 million and \$28.4 million, respectively, of goodwill and other intangible assets in connection with acquisitions made (see Note 4 — Acquisition of Pumpco, Inc.), of which \$2.9 million and \$5.7 million, respectively, is related to earn-out obligations.

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

During the six months ended June 30, 2007, the Company recorded \$29.4 million of goodwill and other intangible assets in connection with the acquisition of the remaining 51% interest in an equity method investment, and wrote-off \$0.4 million in goodwill in connection with the decision to sell all of the Company's Canadian net assets.

(e) Insurance Reserves

MasTec maintains insurance policies subject to per claim deductibles of \$1 million for its workers' compensation policy, \$2 million for its general liability policy and \$2 million for its automobile liability policy. The Company has excess umbrella coverage up to \$100 million per claim and in the aggregate. The Company also maintains an insurance policy with respect to employee group health claims subject to per employee deductibles of \$350,000. MasTec actuarially determines liabilities for unpaid claims and associated expenses, including incurred but not reported losses, and reflects those liabilities in the balance sheet as other current and non-current liabilities. The determination of such claims and expenses and appropriateness of the related liability is reviewed and updated quarterly. Accruals are based upon known facts, historical trends and a reasonable estimate of future expenses. However, a change in experience or actuarial assumptions could nonetheless materially affect results of operations in a particular period. Known amounts for claims that are in the process of being settled, but have been paid in periods subsequent to those being reported, are also recorded in such reporting period.

The Company is periodically required to post letters of credit and provide cash collateral to its insurance carriers. As of June 30, 2008 and December 31, 2007, such letters of credit amounted to \$66.2 million and \$64.8 million, respectively, and cash collateral posted amounted to \$3.3 million as of the end of both periods, which is included in other assets.

(f) Stock Based Compensation

The Company has granted to employees and others restricted stock and options to purchase common stock. The Black-Scholes valuation model is used to estimate the fair value of options to purchase our common stock, and MasTec uses the ratable method (an accelerated method of expense recognition under SFAS No. 123R, *"Share-based Payment"*) to amortize compensation expense over the vesting period of the option grant. Total non-cash stock compensation expense for grants of restricted stock and options for the three months ended June 30, 2008 and 2007 was \$1.1 million and \$1.5 million, respectively, and for the six months ended June 30, 2008 and 2007 was \$1.9 million and \$3.5 million, respectively. Non-cash stock compensation expense is included in general and administrative expense in the condensed unaudited consolidated statements of operations. During the six months ended June 30, 2008 and 2007, there were no stock options granted.

The Company also grants restricted stock, which is valued based on the market price of MasTec's common stock on the date of grant. Compensation expense arising from restricted stock grants with graded vesting is recognized using the ratable method over the vesting period. Those issued with cliff vesting are amortized on a straight line basis over the vesting period. Unearned compensation for performance-based options and restricted stock is shown as a reduction of shareholders' equity in the condensed unaudited consolidated balance sheets. Through June 30, 2008, 874,576 shares of restricted stock have been issued, valued at approximately \$8.5 million which is being expensed over vesting periods ranging from one to five years. Total unearned compensation related to restricted stock grants as of June 30, 2008 is approximately \$4.3 million. Restricted stock expense for the three and six months ended June 30, 2008 was approximately \$0.5 million and \$0.8 million, respectively. Restricted stock expense for the three and six months ended June 30, 2007 was approximately \$0.4 million and \$1.0 million, respectively. These costs are included in general and administrative expenses in the condensed unaudited consolidated statements of operations.

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

(g) Cash and cash equivalents

All short-term investments with original maturities of three months or less are considered to be cash equivalents. Restricted cash related to collateral of letters of credit is also included in cash and cash equivalents.

(h) Fair value of financial instruments

Effective January 1, 2008, the Company adopted SFAS No. 157, "*Fair Value Measurements*" (SFAS 157). SFAS 157 defines fair value, establishes a framework for measuring fair value in accordance with accounting principles generally accepted in the United States, and expands disclosure requirements about fair value measurements. In accordance with Financial Accounting Standards Board ("FASB") Staff Position No. FAS 157-2, "*Effective Date of FASB Statement No. 157*" (FSP 157-2), the Company will defer the adoption of SFAS 157 for our nonfinancial assets and nonfinancial liabilities, except those items recognized or disclosed at fair value on an annual or more frequent recurring basis, until January 1, 2009 and is currently evaluating the impact of adoption. The adoption of SFAS 157 for our financial assets and liabilities did not have a material impact on our fair value measurements.

The fair market value of financial instruments is estimated through the use of public market prices, quotes from financial institutions and other available information. Judgment is required in interpreting data to develop estimates of market value and, accordingly, amounts are not necessarily indicative of the amounts that could be realized in a current market exchange. Short-term financial instruments, including cash and cash equivalents, accounts and notes receivable, accounts payable and other liabilities, consist primarily of instruments without extended maturities, the fair value of which, based on management's estimates, equals their carrying values. At June 30, 2008 and December 31, 2007, the fair value of the Company's outstanding senior notes was approximately \$130 million and \$142 million, respectively.

(i) Securities Available for Sale

Securities available-for-sale are accounted for in accordance with the provisions of SFAS No. 115, "Accounting For Certain Debt and Equity Securities." Securities available-for-sale are recorded at fair value in accordance with SFAS 157, and temporary unrealized holding gains and losses are recorded as a separate component of accumulated other comprehensive income (loss). Unrealized losses are charged against net earnings when a decline in fair value is determined to be other-than-temporary. In accordance with FASB Statement of Position FAS 115-1 and 124-1, "The Meaning of Other-Than-Temporary Impairment and Its Application to Certain Investments," several factors are reviewed to determine whether a loss is other-than-temporary. These factors include but are not limited to: i) the length of time a security is in an unrealized loss position, ii) the extent to which fair value is less than cost, iii) the financial condition and near term prospects of the issuer and, iv) the Company's intent and ability to hold the security for a period of time sufficient to allow for any anticipated recovery in fair value.

The Company's securities available for sale consist of auction-rate securities which represent interests in corporate debt obligations and student loans guaranteed by the U.S. government under the Federal Family Education Loan Program. These auction-rate securities are accounted for as securities available for sale at fair value and unrealized gains and losses are included in accumulated other comprehensive income as a separate component of shareholders' equity. The Company's estimate of fair value is sensitive to market conditions and management judgment and can change significantly based on the assumptions used. Factors that may impact the Company's valuation include changes to credit ratings of the securities as well as to the underlying assets supporting those securities, rates of default of the underlying assets, underlying collateral value, discount rates, counterparty risk and ongoing strength and quality of market credit and liquidity. See Note 5 — Securities Available for Sale.



MasTec, Inc. Notes to the Condensed Unaudited Consolidated Financial Statements — continued

(j) Income taxes

As a result of the Company's recent history of operating losses, valuation allowances aggregating \$32.6 million and \$47.9 million have been recorded as of June 30, 2008 and December 31, 2007, respectively, to reduce certain of our net deferred federal, foreign and state tax assets to their estimated net realizable value. Management anticipates that the Company will generate sufficient pretax income in the future to realize the deferred tax assets based on continuing operations and feasible tax planning strategies available. On January 1, 2007, MasTec adopted FASB Interpretation No. 48, "Accounting for Uncertainty in Income Taxes — an interpretation of FASB Statement 109," ("FIN 48") and recorded the cumulative effect of applying FIN 48 of \$1.9 million as an adjustment to the balance of deferred tax assets, and an offset to the valuation allowance on that deferred tax asset. There are no open Federal tax years under audit.

(k) Reclassifications

Certain reclassifications were made to the prior period financial statements in order to conform to the current period presentation. Also, auction rate securities previously classified as "Cash and cash equivalents" have been reclassified to "Securities available for sale" in the condensed consolidated balance sheets and statements of cash flows for all periods presented.

Note 4 — Acquisition of Pumpco, Inc.

In May 2008, MasTec acquired all of the issued and outstanding capital stock of Pumpco, Inc. ("Pumpco") for a purchase price of \$44 million, paid in cash, plus the retirement and assumption of certain indebtedness and earn-out payments payable over a five-year period equal to fifty percent of Pumpco's earnings before taxes above significant specified thresholds. The earn-out is payable in cash, MasTec common stock or a combination thereof. In connection with the acquisition, the Company entered into a \$22.5 million equipment term loan and used the proceeds to pay off \$8.7 million of Pumpco indebtedness with the balance used to pay a portion of the acquisition purchase price. The equipment term loan is secured by most of Pumpco's existing equipment. The acquisition is effective as of May 1, 2008, and, accordingly, Pumpco's earnings have been consolidated as of that date.

Pumpco, headquartered in Giddings, Texas, has been in business for over 25 years and specializes in midstream oil and gas pipeline construction. The acquisition of Pumpco continues MasTec's diversification and growth strategy and expands its presence and capabilities in servicing gas pipeline customers.

The allocation of purchase price to the fair value of the tangible and intangible assets and the useful lives of these assets remains preliminary as management continues to assess the valuation of the acquired assets and liabilities. The purchase price to acquire Pumpco including transaction costs has been allocated on a preliminary basis to the assets acquired and liabilities assumed at the effective date of the acquisition based on estimated fair values, as summarized below.

Current Assets	\$ 11,349
Property and equipment	34,655
Customer contracts and relationships	5,200
Non-compete agreement	1,740
Trade name	2,400
Goodwill	12,943
Total assets acquired	68,287
Current liabilities	(3,655)
Other liabilities	(3,464)
Debt	(9,539)
Total liabilities assumed	(16,658)
Net assets	\$ 51,629

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

The portion of intangible assets related to existing customer contracts is amortized over the remaining term of these contracts. The remaining portion of the intangible assets related to customer relationships is amortized on an accelerated basis over its ten-year useful life. Intangible assets related to customer contracts and relationships, the non-compete agreement with the seller and the Pumpco trade name have a weighted average useful life of 13 years.

Inclusion of proforma results of operations as if the acquisition had been completed at the beginning of the current period would not have a material impact on the results of operations as presented.

In connection with the acquisition, the Company entered into lease agreements for office space and land with the selling shareholder. The impact of these agreements is not material to the results of operations to the Company.

Note 5 — Securities Available For Sale

The Company's securities available for sale consist of highly rated auction rate securities that represent interests in pools of corporate debt obligations and student loans guaranteed by the U.S. government under the Federal Family Education Loan Program. Liquidity for these auction-rate securities is typically provided by an auction process that resets the applicable interest rate at pre-determined intervals, usually every 7, 28 or 35 days. In the current period, due to disruptions in the credit markets, these auctions have not had sufficient bidders to allow investors to complete a sale, indicating that immediate liquidity at par is unavailable. All of these securities carry AAA ratings from one or more of the major credit rating agencies, and the Company continues to earn and collect a market rate of interest on these securities. Management has the intent and believes the Company has the ability to hold these securities until they can be sold at par value. Management is uncertain at this time as to when the liquidity issues associated with these investments will improve, and as a result of this uncertainty, has reclassified \$26.9 million of the book value of these securities to long-term assets at June 30, 2008. The remaining \$2.0 million of these securities is classified in current assets as these securities will be distributed at par value as a portion of an acquisition earn out payment in the third quarter of 2008. Management is uncertain at this time as to when the Company will be able to exit the remainder of these investments at their par value or whether additional temporary or other than temporary impairment related to these investments will be incurred in the future. Management continues to monitor market conditions, and any future failed auctions will be evaluated based on the most relevant and timely information available to us.

The fair value of auction rate securities representing interests in corporate debt obligations is based primarily upon indicative bids on these securities in this inactive market (Level 2 inputs in accordance with SFAS 157). At June 30, 2008, there was insufficient observable market data to determine the fair value of auction rate securities backed by student loans. Therefore, the fair value of these investments was estimated by incorporating assumptions market participants would use in their estimates of fair value such as reset interest rates, final stated maturities, collateral values and credit quality (Level 3 inputs under SFAS 157). The Company's valuation is sensitive to market conditions and management's judgment and can change significantly based on the assumptions used. Factors that may impact the Company's valuation include changes to credit ratings of the securities as well as to the underlying assets supporting those securities, rates of default of the underlying assets, underlying collateral value, discount rates, counterparty risk and ongoing strength and quality of market credit and liquidity.

As of June 30, 2008, the fair value of the securities available for sale totaled \$28.9 million. During the three and six months ended June 30, 2008, the Company recorded an unrealized gain of \$0.8 million and an unrealized loss of \$2.0 million, respectively, in Other Comprehensive Income in addition to the \$4.8 million in unrealized losses recorded for these securities during the year ended December 31, 2007. Management believes this temporary impairment is primarily attributable to the limited liquidity of these investments.

The cost basis, gross unrealized gains and losses and fair value for these securities are as follows:

		June 30, 2008			December 31, 2007			
		Gross			Gross			
	Cost	Unrealized	Fair	Cost	Unrealized	Fair		
	Basis	Losses	Value	Basis	Losses	Value		
Auction rate securities	\$ 35,650	\$ (6,762)	\$ 28,888	\$ 49,150	\$ (4,790)	\$ 44,360		

The contractual maturity of the auction rate securities available for sale at June 30, 2008 ranges from 8 to 39 years.

Note 6 — Other Assets and Liabilities

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

Prepaid expenses and other current assets as of June 30, 2008 and December 31, 2007 consist of the following (in thousands):

	J	fune 30, 2008	D	ecember 31, 2007
Non-trade receivables	\$	9,974	\$	17,081
Other receivables		5,431		4,346
Prepaid expenses and deposits		12,809		5,708
Deferred project costs		1,147		907
Other		1,109		1,759
Total prepaid expenses and other current assets	\$	30,470	\$	29,801

Other current liabilities consist of the following as of June 30, 2008 and December 31, 2007 (in thousands):

	J	June 30, 2008	De	cember 31, 2007
Current liabilities:				
Billings in excess of costs	\$	9,620	\$	6,142
Accrued amounts related to discontinued operations		1,887		9,882
Obligations related to acquisitions		5,873		5,919
Accrued losses on contracts		67		364
Other		13,779		6,819
Total other current liabilities	\$	31,226	\$	29,126

Note 7 — Debt

Debt is comprised of the following at June 30, 2008 and December 31, 2007 (in thousands):

	June 30, 2008	De	cember 31, 2007
Revolving credit facility	\$ _	\$	_
7.625% senior notes due February 2017	150,000		150,000
Equipment term loan	22,186		
Capital lease obligations	11,991		12,289
Notes payable for equipment, at interest rates from up to 8.0% due in installments through the year 2011	9,853		684
Total debt	 194,030		162,973
Less current maturities	(12,538)		(2,694)
Long-term debt	\$ 181,492	\$	160,279

In connection with the acquisition of Pumpco (see Note 4), the Company entered into an equipment term loan in the aggregate principal amount of \$22.5 million with an interest rate of 7.05%, payable in sixty monthly installments, and maturing in 2013. This loan is secured by most of Pumpco's existing equipment. Proceeds from this loan were used to pay off \$8.7 million of Pumpco's indebtedness with the remaining balance used to pay a portion of the acquisition purchase price.

Notes to the Condensed Unaudited Consolidated Financial Statements - continued

Note 8 — Discontinued Operations

On April 10, 2007, substantially all of the net assets of the Company's Canadian operations were sold. Accordingly, the operations in Canada have been accounted for as discontinued operations for all periods presented.

The following table summarizes the results of the discontinued Canadian operations (in thousands):

	I	For the Three Jun	Months Er ie 30,	nded	For the Six Months Ended June 30,		
		2008 2007			2008		2007
Revenue	\$	_	\$	7	\$ _	\$	675
Cost of revenue		—		(1)	_		(823)
Operating and other expenses		(85)		(164)	(240)		(931)
Loss from operations before benefit for income taxes		(85)		(158)	 (240)		(1,079)
Benefit for income taxes		—			—		
Net loss	\$	(85)	\$	(158)	\$ (240)	\$	(1,079)

On February 14, 2007, the state Department of Transportation related projects and net assets were sold.

The following table summarizes the results of operations for the six months ended June 30, 2007 for the state Department of Transportation related projects and assets that are discontinued (in thousands). There were no material results for this discontinued operation in the three months ended June 30, 2007 or for any period in 2008.

	Mon	the Six ths Ended e 30, 2007
Revenue	\$	5,663
Cost of revenue		(6,311)
Operating and other expenses		(3,780)
Net loss	\$	(4,428)

n 4 6'

Note 9 — Commitments and Contingencies

In 2005, former employees filed a Fair Labor Standards Act ("FLSA") collective action against MasTec in the Federal District Court in Tampa, Florida, alleging failure to pay overtime wages as required under the FLSA. While MasTec denies the allegations underlying the lawsuit, in October 2007 we agreed to a settlement to avoid significant legal fees, the uncertainty of a jury trial, other expenses and management time that would have to be devoted to protracted litigation. The settlement covers MasTec's current and former install-to-the home employees who were employed by MasTec from October 2001 through September 2007 in California, Florida, Georgia, Maryland, New Jersey, New Mexico, North Carolina, South Carolina, Texas and Virginia. Based on the members of the purported class that have opted in, the maximum amount to be paid in connection with this settlement is \$8.4 million. In April 2008, the settlement was approved by the court and the Company paid \$8.0 million in connection with this settlement in July 2008.

We contracted to construct a natural gas pipeline for Coos County, Oregon in 2003. Construction work on the pipeline ceased and we declared a breach of contract and brought an action for breach of contract against Coos County in Federal District Court in Oregon seeking payment for work done and interest. In April 2004, Coos County announced it was terminating the contract and seeking another company to complete the project. Coos County subsequently counterclaimed against us in the Federal District Court action seeking damages in excess of \$15 million for breach of contract for alleged failures to properly construct the pipeline and for alleged environmental and labor law violations, and other causes.

In April 2008, we entered into a definitive settlement agreement to settle our dispute with Coos County, which provides for a \$4.35 million payment to Coos County on the later of June 1, 2008 or ten days after the entry of a

Notes to the Condensed Unaudited Consolidated Financial Statements - continued

judgment in the Corps of Engineers matter, included in Other current liabilities, and a \$4.35 million payment to Coos County on June 1, 2009 with 3% interest accruing beginning June 1, 2008, included in Other liabilities, at June 30, 2008 and December 31, 2007. The settlement agreement is subject to MasTec not being penalized greater than \$1.5 million in the Corps of Engineers matter described below. No assurances can be given that this condition will be met.

In connection with the Coos County pipeline project, the United States Army Corps of Engineers, or "Corps of Engineers", and the Oregon Department of Environmental Quality issued cease and desist orders and notices of non-compliance to Coos County and to us with respect to the project. While we did not agree that the notices were appropriate or justified, we cooperated with the Corps of Engineers and the Oregon Division of State Land, Department of Environmental Quality to mitigate any adverse impact as a result of construction. On March 30, 2006, the Corps of Engineers brought a complaint in a federal district court against us and Coos County and are seeking damages in excess of \$16 million. The matter went to trial in February 2008 and we are awaiting a verdict.

In April 2006, we settled, without payment to the plaintiffs by us, several complaints for purported securities class actions that were filed against us and certain officers in the second quarter of 2004. As part of the settlement, our excess insurance carrier retained the rights to seek reimbursement of up to \$2.0 million from us based on its claim that notice was not properly given under the policy. We were also seeking reimbursement of expenses incurred by us which we believe are reimbursable by our excess insurance carrier. We believed the claims of the excess insurance carrier were without merit and vigorously pursued this action. Although a district court ruled in March 2007 that there is no coverage provided under the excess policy, we appealed. The appellate court, however, affirmed the district court's ruling and we paid \$2.0 million to the insurance carrier to resolve such dispute. We are now vigorously pursuing claims against the insurance broker for the losses arising from the same issue involving notice.

In June 2005, we posted a \$2.3 million bond in order to pursue the appeal of a \$2.0 million final judgment entered against us for damages plus attorney's fees resulting from a break in a Citgo pipeline that occurred in 1999. We are seeking a new trial and a reduction in the amount of damages awarded. We will continue to contest this matter in the appellate court, and on subsequent retrial, if any.

MasTec filed suit against Con Edison in May 2002, alleging that Con Edison directly interfered with MasTec's work for Telergy, and that this interference resulted in the bankruptcy of Telergy and resulted in Con Edison obtaining MasTec's work on the Telergy project without paying for it. MasTec seeks in excess of \$40 million from Con Edison.

We provided telecommunication infrastructure services to Adesta Communications, Inc. in 2000 and 2001. Adesta filed for bankruptcy in 2001. At June 30, 2008, we were seeking to recover amounts in excess of \$4 million from the Adesta bankruptcy trustee from the proceeds of the sale of Adesta's assets. Based on our understanding of the current status of the bankruptcy trustee's sales negotiations with respect to these assets, we have reflected \$1.3 million in other current assets on our condensed unaudited consolidated balance sheet at June 30, 2008 related to Adesta.

On January 24, 2008, we entered into a settlement agreement with the buyer of our state Department of Transportation projects and assets to settle previously disclosed warranty, indemnification and other claims primarily relating to work we had performed on the state Department of Transportation projects we sold. In connection with the settlement agreement, the parties also agreed to further amend and restate the Amended Asset Purchase Agreement between the parties effective as of January 24, 2008, which we refer to as the revised sale agreement.

Under the terms of the settlement agreement, we paid \$6 million in cash, which was previously accrued, and obtained from the buyer a covenant not to sue and general release from nearly all obligations owed by us to the buyer under the purchase agreement, including warranty and other indemnification obligations. The revised sale agreement, among other things, deleted substantially all of our representations and warranties and indemnification obligations set forth in the Amended Asset Purchase Agreement, reduced the term of our covenants against competition and solicitation of customers, suppliers and other third parties (other than the buyer's employees) from the five year period ending February 13, 2012 to the four year period ending February 13, 2011 and released us from our covenant not to compete in the following states: Arizona, Nevada, Colorado, Oklahoma, New Mexico, Missouri and Minnesota. See Part II. Other Information — Item IA. Risk Factors.

Notes to the Condensed Unaudited Consolidated Financial Statements ---- continued

Accrued aggregate liabilities related to the matters discussed above and other litigation matters amounted to \$25.1 million at June 30, 2008 and \$25.6 million at December 31, 2007. A charge of \$1.6 million and \$3.2 million was recorded in the three and six months ended June 30, 2008, respectively, related to these matters.

Other Commitments and Contingencies. We are required to provide payment and performance bonds for some of our contractual commitments related to projects in process. At June 30, 2008, the cost to complete on these projects, for which \$359.3 million in performance and payment bonds are outstanding, was \$80.6 million.

Note 10 — Concentrations of Risk

MasTec provides services to its customers in the following industries: communications, utilities and government.

Revenue for customers in these industries is as follows (in thousands):

		Three Months Ended June 30,		ths Ended e 30,
	2008	2007	2008	2007
Communications	\$182,274	\$189,217	\$364,747	\$367,700
Utilities	104,643	51,939	172,052	103,053
Government	18,117	15,128	30,227	26,527
	\$305,034	\$256,284	\$567,026	\$497,280

The Company grants credit, generally without collateral, to its customers. Consequently, it is subject to potential credit risk related to changes in business and economic factors. However, the Company generally has certain lien rights with respect to these services rendered, and concentrations of credit risk are limited due to the diversity of the customer base. Management believes the billing and collection policies are adequate to minimize potential credit risk. During the three months ended June 30, 2008, 43% of our total revenue was attributed to two customers. Revenue from these two customers accounted for 35% and 8%, respectively, of the total revenue for the three months ended June 30, 2007, two customers accounted for 54% of our total revenue. Revenue from these two customers accounted for 43% and 11%, respectively, of the total revenue from these two customers accounted for 40% and 9%, respectively, of the total revenue for the six months ended June 30, 2008. During the six months ended June 30, 2007, two customers accounted for 55% of our total revenue. Revenue from these two customers accounted for 44% and 11%, respectively, of the total revenue for the six months ended June 30, 2007.

The Company maintains an allowance for doubtful accounts of \$15.9 million and \$15.3 million as of June 30, 2008 and December 31, 2007, respectively, for both specific customers and as a reserve against other uncollectible accounts receivable. As of June 30, 2008, remaining receivables from customers undergoing bankruptcy reorganization totaled \$1.6 million, of which \$0.3 million is specifically reserved. Should additional customers file for bankruptcy or experience financial difficulties, or should anticipated recoveries in existing bankruptcies and other workout situations fail to materialize, the Company could experience reduced cash flows and losses in excess of the current allowance.

Note 11 — Related Party Transactions

Management believes amounts paid in related party transactions are equivalent to the payments that would have

Notes to the Condensed Unaudited Consolidated Financial Statements — continued

been made between unrelated parties for similar transactions acting on an arms-length basis, as described below.

MasTec purchases, rents and leases equipment used in its business from a number of different vendors, on a non-exclusive basis, including Neff Corp. ("Neff"), in which Jorge Mas, Chairman of the Board of Directors, and Jose Mas, the Company's President and Chief Executive Officer, were directors and owners of a controlling interest through June 4, 2005. Juan Carlos Mas, the brother of Jorge and Jose Mas, was the Chairman, Chief Executive Officer, a director and a shareholder of Neff until May 31, 2007 when he sold a portion of his Neff shares and resigned as its chief executive officer. Juan Carlos Mas remains as chairman of the Neff board of directors. During the three months ended June 30, 2008 and 2007, the Company paid Neff approximately \$0.1 million and \$0.5 million, respectively and paid \$0.3 million and \$0.9 million, during the six months ended June 30, 2008 and 2007, respectively.

The Company charters aircrafts from a third party who leases two of its aircraft from entities in which Jorge Mas, Chairman of our Board of Directors, and Jose Mas, the Company's President and Chief Executive Officer, have an ownership interest. MasTec paid this unrelated chartering company approximately \$46,000 and \$66,000 during the three and six month periods ended June 30, 2008, respectively, and \$0.2 million and \$0.6 million during the three and six month periods ended June 30, 2008, respectively.

During the three months ended June 30, 2008 and 2007, the Company had an arrangement with a customer whereby it leased employees to that customer and charged approximately \$0.1 million each quarter to the customer. Charges to this customer totaled \$0.2 million during the six month periods ended June 30, 2008 and 2007. Jorge Mas, Chairman of the Board of Directors, and Jose Mas, the Company's President and Chief Executive Officer, are minority owners of this customer.

MasTec has entered into split dollar agreements with key executives and former executives, and with the Chairman of the Board of Directors. During the three months ended June 30, 2008 and 2007, the Company paid approximately \$0.3 million in each quarter in premiums in connection with these split dollar agreements. During the six months ended June 30, 2008 and 2007, the Company paid \$0.6 million and \$0.5 million, respectively, in premiums in connection with these split dollar agreements.

The Company paid Irma Mas, the mother of Jorge Mas, the Chairman and Jose Mas, the Company's President and Chief Executive Officer, \$12,000 in each of the quarters ended June 30, 2008 and 2007, for the lease of certain property located in Florida and paid \$24,000 and \$52,000 during the six month periods ended June 30, 2008 and 2007, respectively.

Note 12 — Recent Accounting Pronouncements

On February 15, 2007, the FASB issued SFAS No. 159, "*The Fair Value Option for Financial Assets and Financial Liabilities* — *Including an Amendment of FASB Statement No. 115*" ("SFAS 159"). This standard permits an entity to measure financial instruments and certain other items at estimated fair value. Most of the provisions of SFAS 159 are elective; however, the amendment to SFAS No. 115, "*Accounting for Certain Investments in Debt and Equity Securities*," applies to all entities that own trading and available-for-sale securities. The fair value option created by SFAS 159 permits an entity to measure eligible items at fair value as of specified election dates. The fair value option (a) may generally be applied instrument by instrument, (b) is irrevocable unless a new election date occurs, and (c) must be applied to the entire instrument and not to only a portion of the instrument. The adoption of SFAS 159 as of January 1, 2008 did not have a material impact on the Company's consolidated financial statements.

On January 1, 2008, MasTec adopted EITF Issue No. 06-11, "Accounting for Income Tax Benefits of Dividends on Share-Based Payment Awards." (EITF 06-11). EITF 06-11 requires that a realized income tax benefit from dividends or dividend equivalent units paid on unvested restricted shares, restricted share units and stock options be reflected as an increase in capital surplus and reflected as an addition to the company's excess tax benefit pool, as defined under SFAS No. 123(R). Because MasTec did not declare any dividends during the first half of 2008 and does not currently

Notes to the Condensed Unaudited Consolidated Financial Statements --- continued

anticipate declaring dividends in the near future, EITF 06-11 did not have any impact during the first half of 2008, and is not expected to have a material impact in the near term, on MasTec's consolidated financial statements.

In December 2007, the FASB issued No. 141(R), "Business Combinations" (SFAS 141(R)) and SFAS No. 160 "Accounting and Reporting of Noncontrolling Interests in Consolidated Financial Statements, an amendment of ARB No. 51" (SFAS 160). SFAS 141(R) and SFAS 160 significantly change the accounting for and reporting of business combination transactions and noncontrolling (minority) interests. SFAS 141(R) and SFAS 160 are effective for the fiscal years beginning after December 15, 2008. SFAS 141(R) and SFAS 160 are effective prospectively; however, the reporting provisions of SFAS 160 are effective retroactively. SFAS 141(R) is required to be adopted concurrently with SFAS 160 and is effective for business combination transactions for which the acquisition date is on or after the beginning of the first annual reporting period beginning on or after December 15, 2008. The Company will apply SFAS 141(R) prospectively to business combinations with an acquisition date on or after January 1, 2009. The Company is currently evaluating SFAS 160 and does not expect it will have material impact on the consolidated financial statements, as the Company currently does not have any noncontrolling interests.

On March 19, 2008 the FASB issued SFAS No. 161, "*Disclosures About Derivative Instruments and Hedging Activities*" (SFAS 161). This statement is intended to improve financial reporting about derivative instruments and hedging activities by requiring enhanced disclosures. This statement is effective for financial statements issued for fiscal years and interim periods beginning after November 15, 2008, with early application encouraged. The Company is currently evaluating the impact that SFAS 161 will have on the consolidated financial statements.

In April 2008, the FASB issued FASB Staff Position No. FAS 142-3, "*Determination of the Useful Life of Intangible Assets*" (FSP 142-3). FSP 142-3 amends the factors that should be considered in developing renewal or extension assumptions used to determine the useful life of a recognized intangible asset under SFAS No. 142, "*Goodwill and Other Intangible Assets*" (SFAS 142) to improve the consistency between the useful life of a recognized intangible asset under SFAS 142 and the period of expected cash flows used to measure the fair value of the asset under SFAS No. 141, *Business Combinations*, and other U.S. GAAP. FSP 142-3 is effective for financial statements issued for fiscal years beginning after December 15, 2008, and interim periods within those fiscal years. Early adoption is prohibited. The Company is currently evaluating the impact of FSP 142-3 on the consolidated financial statements and will adopt FSP 142-3 effective January 1, 2009.

In May 2008, the FASB issued SFAS No. 162, "*The Hierarchy of Generally Accepted Accounting Principles*" (SFAS 162). This statement identifies the sources of accounting principles and the framework for selecting the principles to be used in the preparation of financial statements of nongovernmental entities that are presented in conformity with generally accepted accounting principles (GAAP) in the United States. This Statement shall be effective 60 days following the SEC's approval of the Public Company Accounting Oversight Board (PCAOB) amendments to AU Section 411, "*The Meaning of Present Fairly in Conformity With Generally Accepted Accounting Principles*". The Company does not believe the adoption of SFAS 162 will have a material impact on the consolidated financial statements.

Note 13 — Subsequent Events

The Company entered into an amended and restated Senior Secured Credit Facility effective July 29, 2008, which expires May 10, 2013 (the "Credit Facility"). The Credit Facility has a maximum amount of available borrowing of \$210.0 million, subject to certain restrictions. The maximum available borrowing may be increased to \$260.0 million if certain conditions are met.

Notes to the Condensed Unaudited Consolidated Financial Statements - continued

As in the past, the amount the Company can borrow at any given time is based upon a formula that takes into account, among other things, eligible billed and unbilled accounts receivable, equipment, real estate and eligible cash collateral, which can result in borrowing availability of less than the full amount of the Credit Facility. The Credit Facility is collateralized by a first priority security interest in substantially all of our assets and the assets of our wholly-owned subsidiaries and a pledge of the stock of certain of our operating subsidiaries. Interest under the Credit Facility accrues at variable rates based, at the Company's option, on the agent bank's base rate plus a margin of between 0.5% and 1.00%, or at the LIBOR rate (as defined in the Credit Facility) plus a margin of between 1.50% and 2.50%, depending on certain financial thresholds. The current margin over LIBOR is 2.0%. The Credit Facility includes an unused facility fee ranging from 0.25% to 0.375% based on usage.

The Credit Facility contains customary events of default (including cross-default) provisions and covenants related to our operations that prohibit, among other things, making investments and acquisitions in excess of specified amounts, incurring additional indebtedness in excess of specified amounts, creating liens against Company assets, prepaying other indebtedness excluding our 7.625% senior notes, making acquisitions in excess of specified amounts, and engaging in certain mergers or combinations without the prior written consent of the lenders. The Credit Facility also limits our ability to make certain distributions or pay dividends. In addition, the Company is required to maintain a minimum fixed charge coverage ratio of 1.20 to 1.00, as defined in the Credit Facility. Any deterioration in the quality of billed and unbilled receivables, reduction in the value of equipment or an increase in lease expense related to real estate, could reduce availability under the Credit Facility.

ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

Forward-Looking Statements

This report contains forward-looking statements within the meaning of the Securities Act of 1933 and the Securities Exchange Act of 1934, as amended by the Private Securities Litigation Reform Act of 1995. These forward-looking statements are not historical facts but are the intent, belief, or current expectations, of our business and industry, and the

assumptions upon which these statements are based. Words such as "anticipates", "expects", "intends", "will", "could", "would", "should", "may", "plans", "believes", "seeks", "estimates" and variations of these words and the negatives thereof and similar expressions are intended to identify forward-looking statements. These statements are not guarantees of future performance and are subject to risks, uncertainties, and other factors, some of which are beyond our control, are difficult to predict, and could cause actual results to differ materially from those expressed or forecasted in the forward-looking statements. These risks and uncertainties include those described in "Management's Discussion and Analysis of Financial Condition and Results of Operations," and elsewhere in this report and in the Company's Annual Report on Form 10-K for the year ended December 31, 2007, including those described under "Risk Factors" in the Form 10-K as updated by Item 1A "Risk Factors" in this report and other of our SEC filings. Forward-looking statements that were true at the time made may ultimately prove to be incorrect or false. Readers are cautioned to not place undue reliance on forward-looking statements, which reflect our management's view only as of the date of this report. We undertake no obligation to update or revise forward-looking statements to reflect changed assumptions, the occurrence of unanticipated events or changes to future operating results.

Overview

We are a leading specialty contractor operating mainly throughout the United States and across a range of industries. Our core activities are the building, installation, maintenance and upgrade of communications and utility infrastructure, including but not limited to, fiber deployment, electrical utility transmission and distribution, water and sewer, and wind farm and natural gas pipeline construction. Our primary customers are in the following industries: communications (including satellite television and cable television), utilities and government. We provide similar infrastructure services across the industries we serve. Our customers rely on us to build and maintain infrastructure and networks that are critical to their delivery of voice, video and data communications, electricity and other energy resources.

We, or our predecessor companies, have been in business for over 70 years. We offer our services under the MasTec service mark and operate through a network of approximately 200 locations and approximately 7,700 employees as of June 30, 2008. Providing services to communication industries, utility industries and government markets, we have consistently ranked among the top specialty contractors by Engineering News-Record.

We serve a diversified domestic customer base including DIRECTV[®], Verizon, AT&T, Energy Transfer Company, Tetra-Tech, M.A. Mortenson Company, EMBARQ, Qwest, Pecan Pipeline and Progress Energy. For the three and six month periods ended June 30, 2008, approximately 72% of our revenues were from our ten largest customers. For the three and six month periods ended June 30, 2007, approximately 79% of our revenues were from our ten largest customers. We have longstanding relationships with many customers and often provide services under multi-year master service agreements and other service agreements.

We have continued our diversification and expansion strategy through several recent acquisitions. During 2007 and 2008, we acquired the remainder of GlobeTec Construction, LLC ("GlobeTec") to bring our ownership of this entity to 100%. GlobeTec is involved in the construction and maintenance of water and sewer pipelines and projects. In October 2007, we acquired Three Phase Line Construction, Inc., which is involved in the construction and maintenance of transmission and distribution utility systems, substation and storm restoration in several northern states. In December 2007, we acquired certain assets of Power Partners, LLC, which is an electrical utility contractor specializing in wind farm construction.

In May 2008, we acquired Pumpco, Inc. ("Pumpco"), a private company headquartered in Giddings, Texas, specializing in midstream oil and gas pipeline construction. Pumpco has been in business for over twenty five years and had annual revenue of approximately \$70 million in 2007. This acquisition expands our presence and capabilities in the growing area of gas pipeline construction and maintenance.

Revenue

We provide services to our customers, which are companies in the communications and utilities industries, as well as government customers.

Revenue for customers in these industries (in thousands) and the percent of our total revenue earned from such customers is as follows:

		Three Months Ended June 30,				Six Months Ended June 30,			
	2008		2007	7	2008	_	20	07	
Communications	\$182,274	60%	\$189,217	74%	\$364,747	64%	\$367,700	74%	
Utilities	104,643	34%	51,939	20%	172,052	30%	103,053	21%	
Government	18,117	6%	15,128	6%	30,227	6%	26,527	5%	
	\$305,034	100%	\$256,284	100%	\$567,026	100%	\$497,280	100%	

A significant portion of our revenue is derived from projects performed under service agreements. We also provide services under master service agreements which are generally multi-year agreements. Certain of our master service agreements are exclusive up to a specified dollar amount per work order for each defined geographic area. Work performed under master service and other agreements is generated by work orders, each of which is performed for a fixed fee. The majority of these services are of a maintenance nature and, to a lesser extent, upgrade services. These master service agreements and other service agreements are frequently awarded on a competitive bid basis, although customers may negotiate contract extensions beyond their original terms without re-bidding. Our master service agreements and other service agreements have various terms, depending upon the nature of the services provided and are typically subject to termination on short notice.

The remainder of our work is generated pursuant to contracts for specific projects that may require the construction and installation of specified units within an infrastructure system or an entire infrastructure system. Customers are billed with varying frequency: weekly, monthly or upon attaining specific milestones. Such contracts generally include retainage provisions under which 2% to 15% of the contract price is withheld from us until the work has been completed and accepted by the customer.

Revenue by type of contract (in thousands) and the percent of our total revenue from such contracts is as follows:

		Three Months Ended June 30,				Six Months En	ded June 30,	
	2008		2007	,	2008	3	20	07
Master service and other								
service agreements	\$185,079	61%	\$190,195	74%	\$378,246	67%	\$377,758	76%
Installation/construction								
projects agreements	119,955	39%	66,089	26%	188,780	33%	119,522	24%
	\$305,034	100%	\$256,284	100%	\$567,026	100%	\$497,280	100%

Litigation and Contingencies

Litigation and contingencies are reflected in our condensed unaudited consolidated financial statements based on our assessments of the expected outcome. If the final outcome of any litigation or contingencies differs significantly from our current expectations, a charge to earnings could result. See Note 9 to our condensed unaudited consolidated financial statements in this Form 10-Q for updates to our description of legal proceedings and commitments and contingencies.



Results of Operations

Comparison of Quarterly Results

The following table reflects our consolidated results of operations in dollar (in thousands) and percentage of revenue terms for the periods indicated. Our consolidated results of operations are not necessarily comparable from period to period due to the impact of recent acquisitions. All periods presented reflect our Canadian and state Department of Transportation operations as discontinued operations;

	Three Months Ended June 30,		Six Months Ended June 30,					
	2008		2002	7	200	8	200	7
Revenue	\$305,034	100.0%	\$256,284	100.0%	\$567,026	100.0%	\$497,280	100.0%
Costs of revenue, excluding								
depreciation	259,561	85.1%	213,327	83.2%	486,405	85.8%	424,348	85.3%
Depreciation	6,017	2.0%	4,082	1.6%	10,805	1.9%	7,862	1.6%
General and administrative								
expenses	19,966	6.5%	20,234	7.9%	40,012	7.1%	39,482	7.9%
Interest expense, net of								
interest income	3,656	1.2%	2,120	0.8%	6,152	1.1%	4,915	1.0%
Other (income) expense, net	(394)	(0.1)%	(573)	(0.2)%	(545)	(0.1)%	(4,057)	(0.8)%
Income from continuing								
operations before								
provision for income taxes								
and minority interest	16,228	5.3%	17,094	6.7%	24,197	4.3%	24,730	5.0%
Provision of income taxes	407	0.1%		0.0%	440	0.1%		0.0%
Minority interest	_	_	1,035	0.4%		—	1,652	0.3%
Income from continuing								
operations	15,821	5.2%	16,059	6.3%	23,757	4.2%	23,078	4.6%
Loss from discontinued								
operations	(85)	(0.0)%	(158)	(0.1)%	(240)	(0.0)%	(5,507)	(1.1)%
Net income	\$ 15,736	5.2%	\$ 15,901	6.2%	\$ 23,517	4.2%	\$ 17,571	3.5%

Three Months Ended June 30, 2008 Compared to Three Months Ended June 30, 2007

Revenue. Revenue was \$305.0 million for the three months ended June 30, 2008, compared to \$256.3 million for the same period in 2007, representing an increase of \$48.7 million or 19%. This increase was driven primarily by gas pipeline, wind farm and other energy projects with the recent acquisitions of Pumpco, Power Partners and Three Phase Line Construction. Our install-to-the-home business revenue declined 7.6% in the second quarter of 2008 compared to 2007 as this business is our most sensitive to the overall negative economic sentiment.

Costs of Revenue. Costs of revenue were \$259.6 million or 85.1% of revenue for the three months ended June 30, 2008, compared to \$213.3 million or 83.2% of revenue for the corresponding period in 2007. The \$28.3 million increase is driven largely by our recent acquisitions. As a percentage of revenue, cost of revenue increased 190 basis points as higher fuel, materials, and maintenance costs more than offset savings in labor. Fuel costs as a percent of revenue increased 80 basis points in the second quarter of 2008 compared to 2007 as higher fuel rates more than offset decreased usage. Increases in materials costs are largely due to wind farm projects with the balance related to the rise in overall material costs on the Company's other projects.

Depreciation. Depreciation was \$6.0 million for the three months ended June 30, 2008, compared to \$4.1 million for the same period in 2007, representing an increase of \$1.9 million or 46%. The increase was due primarily to depreciation associated with Pumpco and Three Phase Line Construction with the balance resulting from capital expenditures and capital lease agreements executed during 2007 and 2008 to finance machinery and equipment to support our growth.

General and administrative expenses. General and administrative expenses were \$20.0 million or 6.5% of revenue for the three months ended June 30, 2008, compared to \$20.2 million or 7.9% of revenue for the same period in 2007, representing a decrease of \$0.2 million or 140 basis points as a percent of revenue. This decrease was primarily due to lower employee related costs and professional legal fees partially offset by charges totaling \$1.6 million related to legal settlements and higher amortization expense resulting from recent acquisitions.

Interest expense, net. Interest expense, net of interest income was \$3.7 million for the three months ended June 30, 2008, compared to \$2.1 million for the same period in 2007, representing an increase of \$1.6 million or 76%. This increase was the result of a decline in interest income due to reduced interest rates and lower cash balances as well as higher interest expense on higher debt balances, driven by amounts paid in connection with recent acquisitions.

Other income, net. Other income, net was \$0.4 million for the three months ended June 30, 2008, compared to \$0.6 million for the three months ended June 30, 2007, representing a decrease of \$0.2 resulting from lower gains on sale of property and equipment.

Provision for income taxes. The provision for income taxes in 2008 represents state and local taxes for recently acquired companies in jurisdictions in which MasTec does not have an offsetting net operating loss position.

Minority interest. For the three months ended June 30, 2007, the minority interest expense for GlobeTec was \$1.0 million as we owned 51% of this entity. During the three months ended June 30, 2008, we owned 100% of this entity. As such there was no minority interest charge during this period.

Discontinued operations. The loss on discontinued operations for the three months ended June 30, 2008 was \$0.1 million compared to \$0.2 million for the same period in 2007 related to our disposed Canadian operations.

Six Months Ended June 30, 2008 Compared to Six Months Ended June 30, 2007

Revenue. Revenue was \$567.0 million for the six months ended June 30, 2008, compared to \$497.3 million for the same period in 2007, representing an increase of \$69.7 million or 14%. This increase was primarily related to gas pipeline, wind farm and other energy projects with the recent acquisitions of Pumpco, Power Partners and Three Phase Line Construction and higher revenue from DIRECTV[®] due to the acquisition of the remainder of our equity method investment, effective February 2007, as well as organic growth in this business partially offset by lower install-to-the-home revenue in the first half of 2008.

Costs of Revenue. Costs of revenue were \$486.4 million or 85.8% of revenue for the six months ended June 30, 2008, compared to \$424.3 million or 85.3% of revenue for the corresponding period in 2007. As a percentage of revenue, cost of revenue increased 50 basis points as higher fuel, maintenance and materials costs more than offset savings in labor. Fuel costs as a percent of revenue increased 80 basis points in the first half of 2008 compared to 2007 as higher fuel rates more than offset decreased usage. Increases in materials costs are largely due to wind farm projects with the balance related to the rise in overall material costs on the Company's other projects.

Depreciation. Depreciation was \$10.8 million for the six months ended June 30, 2008, compared to \$7.9 million for the same period in 2007, representing an increase of \$2.9 million or 37%. The increase was due primarily to depreciation associated with Pumpco and Three Phase Line Construction with the balance resulting from capital expenditures and capital lease agreements executed during 2007 and 2008 to finance machinery and equipment to support our growth.

General and administrative expenses. General and administrative expenses were \$40.0 million or 7.1% of revenue for the six months ended June 30, 2008, compared to \$39.5 million or 7.9% of revenue for the same period in 2007, decreasing by 80 basis points as a percent of revenue. This decrease was primarily due to lower employee related costs and professional legal fees partially offset by charges totaling \$3.2 million related to legal settlements.

Interest expense, net. Interest expense, net of interest income was \$6.2 million for the six months ended June 30, 2008, compared to \$4.9 million for the same period in 2007, representing an increase of \$1.3 million. This increase was the result of a decline in interest income due to reduced interest rates and lower cash balances as well as higher interest expense on higher debt balances, driven by cash paid in connection with recent acquisitions.

Other income, net. Other income, net was \$0.5 million for the six months ended June 30, 2008, compared to \$4.1 million for the six months ended June 30, 2007, representing a decrease of \$3.6 million primarily due to lower gains on sale of property and equipment. We recognized gains on asset sales of approximately \$0.3 million during the six months ended June 30, 2008, compared to \$3.7 million for the six months ended June 30, 2007, including a \$2.5 million gain on the sale of the property.

Provision for income taxes. The provision for income taxes in 2008 represents state and local taxes for recently acquired companies in jurisdictions in which MasTec does not have an offsetting net operating loss position.

Minority interest. For the six months ended June 30, 2007, the minority interest expense for GlobeTec was \$1.7 million as we owned 51% of this entity. During the six months ended June 30, 2008, we owned 100% of this entity. As such there was no minority interest charge during this period.

Discontinued operations. The loss on discontinued operations for the six months ended June 30, 2008 was \$0.2 million compared to \$5.5 million for the same period in 2007. The first half of 2007 included \$4.4 million in losses related to our state Department of Transportation projects and assets and \$1.1 million in losses related to our Canadian operations, both of which were disposed of during 2007.

Financial Condition, Liquidity and Capital Resources

Our primary sources of liquidity are cash flows from continuing operations, availability under our Credit Facility, capital lease arrangements, proceeds from sales of assets and investments, and our cash balances.

We amended and restated our Senior Secured Credit Facility effective July 29, 2008, expiring May 10, 2013 (the "Credit Facility"). The Credit Facility has a maximum amount of available borrowing of \$210.0 million, subject to certain restrictions. The maximum available borrowing may be increased to \$260.0 million if certain conditions are met.

As in the past, the amount that we can borrow at any given time is based upon a formula that takes into account, among other things, eligible billed and unbilled accounts receivable, equipment, real estate and eligible cash collateral, which can result in borrowing availability of less than the full amount of the Credit Facility. At June 30, 2008 and December 31, 2007, net availability under the Credit Facility totaled \$59.3 million and \$44.0 million, respectively, which includes outstanding standby letters of credit aggregating \$89.4 million and \$86.4 million in each period, respectively. These letters of credit mature at various dates and most have automatic renewal provisions subject to prior notice of cancellation. The Credit Facility is collateralized by a first priority security interest in substantially all of our assets and the assets of our wholly-owned subsidiaries and a pledge of the stock of certain of our operating subsidiaries. At June 30, 2008 and December 31, 2007, we had no outstanding cash draws under the Credit Facility. Interest under the Credit Facility accrues at variable rates based, at our option, on the agent bank's base rate plus a margin of between 0.5% and 1.00%, or at the LIBOR rate (as defined in the Credit Facility) plus a margin of between 1.50% and 2.50%, depending on certain financial thresholds. The current margin over LIBOR is 2.0%. The Credit Facility includes an unused facility fee ranging from 0.25% to 0.375% based on usage.

The Credit Facility contains customary events of default (including cross-default) provisions and covenants related to our operations that prohibit, among other things, making investments and acquisitions in excess of specified amounts, incurring additional indebtedness in excess of specified amounts, creating liens against our assets, prepaying other indebtedness excluding our 7.625% senior notes, making acquisitions in excess of specified amounts, and engaging in certain mergers or combinations without the prior written consent of the lenders. The Credit Facility also limits our ability to make certain distributions or pay dividends. In addition, we are required to maintain a minimum fixed charge coverage ratio of 1.20 to 1.00, as defined in the Credit Facility. Any deterioration in the quality of billed and unbilled receivables, reduction in the value of our equipment or an increase in our lease expense related to real estate, would reduce availability under the Credit Facility.

Based upon the current availability under our Credit Facility, liquidity and projections for 2008, we believe we will be in compliance with the Credit Facility's terms and conditions and the minimum availability requirements for the remainder of 2008. We are dependent upon borrowings and letters of credit under this Credit Facility to fund operations. Should we be unable to comply with the terms and conditions of the Credit Facility, we would be required to obtain modifications to the Credit Facility or another source of financing to continue to operate. We may not be able to achieve our 2008 projections and this may adversely affect our ability to remain in compliance with the Credit Facility's minimum net availability requirements and minimum fixed charge ratio in the future.

On January 31, 2007, we issued \$150.0 million of 7.625% senior notes due February 2017 with semi-annual interest payments of approximately \$5.7 million. On March 2, 2007, we used \$121.8 million of the proceeds from the senior note offering to redeem all of our remaining 7.75% senior subordinated notes plus interest. We are using the remaining net proceeds from the senior note offering for working capital, acquisitions of assets and businesses and other general corporate purposes.

In connection with the acquisition of Pumpco (see Notes 4 and 7 in the Notes to the Condensed Unaudited Consolidated Financial Statements), we entered into an equipment term loan in the aggregate amount of \$22.5 million at 7.05% interest, payable in sixty monthly installments, maturing in 2013. The proceeds from this equipment term loan were used to pay off \$8.7 million of Pumpco indebtedness, with the remaining balance funding a portion of the acquisition purchase price.

Our primary liquidity needs are for working capital, capital expenditures, insurance collateral in the form of cash and letters of credit, earn out obligations and debt service. In addition to ordinary course working capital requirements, we estimate we will spend between \$30 million and \$45 million per year on capital expenditures. We will continue to evaluate lease versus buy decisions to meet our equipment needs and based on this evaluation, our capital expenditures may increase from this estimate in the future. We expect to continue to sell older vehicles and equipment as we upgrade to new equipment, and we expect to generate proceeds from these sales. Additionally, the Company has made certain acquisitions and has agreed to pay certain of the sellers earn-out payments generally based on the future performance of the investment or acquired business. During the six months ended June 30, 2008 and 2007, we paid \$5.7 million and \$0, respectively, related to such earn-out obligations.

The Company needs working capital to support seasonal variations in our business, primarily due to the impact of weather conditions on external construction and maintenance work, including storm restoration work, and the corresponding spending by customers on their annual capital expenditure budgets. Our business is typically slower in the first and fourth quarters of each calendar year and stronger in the second and third quarters. Accordingly, we generally experience seasonal working capital needs from approximately April through September to support growth in unbilled revenue and accounts receivable, and to a lesser extent, inventory. Our billing terms are generally net 30 to 60 days, and some of our contracts allow our customers to retain a portion (from 2% to 15%) of the contract amount until the job is completed according to the terms and conditions therein. We maintain inventory to meet the material requirements of certain of our contracts. Certain of our customers pay us in advance for a portion of the materials we purchase for their projects, or allow us to pre-bill them for materials purchases up to specified amounts. Our vendors generally offer us terms ranging from 30 to 90 days. Our agreements with subcontractors usually contain a "pay-when-paid" provision, whereby our payments to subcontractors are made only after we are paid by our customers.

We anticipate that funds generated from continuing operations, the remaining net proceeds from our senior note offering completed in the first quarter of 2007, borrowings under our Credit Facility, proceeds from sales of assets and investments, and our cash balances will be sufficient to meet our working capital requirements, anticipated capital expenditures, insurance collateral requirements, earn-out obligations, letters of credit and debt service obligations for at least the next twelve months.

As of June 30, 2008, we had \$93.6 million in working capital, defined as current assets less current liabilities, compared to \$163.8 million as of December 31, 2007. This decrease was driven by lower cash balances and the reclassification of the majority of out auction rate securities to long-term assets. Cash and cash equivalents, including approximately \$18.1 million of restricted cash, decreased from \$74.3 million at December 31, 2007 to \$47.0 million at June 30, 2008 mainly due to \$59.2 million paid for acquisitions and contingent considerations during the first half of 2008 partially offset by net proceeds of \$13.5 million from the sale of auction rate securities classified as securities available for sale in our balance sheet.

Net cash provided by operating activities was \$13.9 million for the six months ended June 30, 2008 and was primarily related to improved earnings, and timing of cash payments to vendors offset by increased receivables. Net cash provided by operating activities was \$27.3 million during the six months ended June 30, 2007 and was primarily related to the timing of cash payments to vendors, improvement in earnings as well as the management of inventory and other assets.

Net cash used in investing activities was \$60.8 million for the six months ended June 30, 2008 and was driven by \$59.2 million used in connection with acquisitions and contingent considerations, net of cash acquired, and \$16.1 million used for capital expenditures, partially offset by net proceeds of \$13.5 million from the sales and purchases of auction rate securities. Net cash used in investing activities during the six months ended June 30, 2007 was \$13.0 million and was primarily related to \$11.2 million used in connection with acquisitions made net of cash acquired, and \$14.8 million used for capital expenditures offset by \$11.1 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales and purchases of auction rate securities and \$3.5 million in net proceeds from sales of assets.

Net cash provided by financing activities was \$19.5 million for the six months ended June 30, 2008 compared to \$27.2 million net cash provided by financing activities for the six months ended June 30, 2007. Net cash provided by financing activities in the six months ended June 30, 2008, consisted primarily of proceeds from borrowings under the equipment term loan in connection with the acquisition of Pumpco (see Note 4 in the Notes to Condensed Unaudited Consolidated Financial Statements). Net cash provided by financing activities in the six months ended June 30, 2007, consisted primarily of the proceeds from the issuance of \$150.0 million 7.625% senior notes in January 2007, partially offset by the redemption of \$121.0 million 7.75% senior subordinated notes in March 2007 and \$3.8 million in payments of financing costs.

As of June 30, 2008, \$150.0 million of our 7.625% senior notes due in February 2017 were outstanding. The notes contain default (including cross-default) provisions and covenants restricting many of the same transactions as under our Credit Facility. The indenture which governs our senior notes allows us to incur the following additional indebtedness among others: credit facilities over a defined threshold, renewals to existing debt permitted under the indenture plus an additional \$50 million of indebtedness, and further indebtedness if our fixed charge coverage ratio, as defined, is at least 2:1. In addition, the indenture prohibits incurring additional capital lease obligations in excess of 5% of our consolidated net assets at any time the senior notes remain outstanding.

We hold a variety of highly rated interest bearing auction rate securities that represent interests in pools of either corporate debt obligations or student loans guaranteed by the U.S. government under the Federal Family Education Loan Program. None of our holdings are mortgage backed securities. These auction rate securities provide liquidity via an auction process that resets the applicable interest rate at predetermined calendar intervals, usually every 7, 28, or 35 days. This mechanism allows existing investors either to rollover their holdings, whereby they would continue to own their respective interest in the auction rate security, or to gain immediate liquidity by selling such interests at par. In the current period, due to disruptions in the credit markets, these auctions have not had sufficient bidders to allow investors to complete a sale, indicating that immediate liquidity at par is unavailable. However, all of these securities carry AAA ratings from one or more of the major credit rating agencies, and the Company continues to earn and collect a market rate of interest on these securities. As of June 30, 2008, we hold \$35.7 million in par value of these auction rate securities, with a fair value and carrying value of \$28.9 million, net of a \$6.8 million unrealized loss. While the investments are of a high credit quality, at this time we are uncertain when the liquidity issues associated with these investments will improve and when we will be able to exit these investments at their par value. We currently anticipate holding these securities until we can realize their par value and believe our existing cash resources will be sufficient to meet our anticipated needs for working capital and capital expenditures to execute our current business plan. We are continuing to monitor this situation. See Note 5, "Securities Available for Sale" in the notes to condensed unaudited consolidated financial statements.

Some of our contracts require us to provide performance and payment bonds, which we obtain from a surety company. If we were unable to meet our contractual obligations to a customer and the surety paid our customer the amount due under the bond, the surety would seek reimbursement of such payment from us. At June 30, 2008, the cost to complete on our \$359.3 million performance and payment bonds was approximately \$80.6 million.

Seasonality

Our operations are historically seasonally slower in the first and fourth quarters of the year. This seasonality is primarily the result of client budgetary constraints and preferences and the effect of winter weather on network activities. Some of our clients, particularly the incumbent local exchange carriers, tend to complete budgeted capital expenditures before the end of the year and defer additional expenditures until the following budget year.

Impact of Inflation

The primary inflationary factor affecting our operations currently is rising fuel costs. We are also affected by changes in fuel costs which continue to increase significantly. In both the three and six month periods ended June 30, 2008, fuel costs increased 80 basis points as a percent of revenue. We are also affected by inflationary pressures in materials and labor costs.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

7	5
	J

We are exposed to market risk related to changes in interest rates. Our variable rate Credit Facility exposes us to interest rate risk. However, we had no cash borrowings under the Credit Facility at June 30, 2008.

Interest Rate Risk

Less than 1% of our outstanding debt at June 30, 2008 was subject to variable interest rates. The remainder of our debt has fixed interest rates. Our fixed interest rate debt includes \$150.0 million (face value) in senior notes. The fair market value of these senior notes at June 30, 2008 was \$130 million. Based upon debt balances outstanding at June 30, 2008, a 100 basis point (i.e., 1%) addition to our weighted average effective interest rate for variable rate debt would not have a material impact on our consolidated results of operations.

Foreign Currency Risk

Previously, we had an investment in a subsidiary in Canada and sold our services into this foreign market. On April 10, 2007, we sold substantially all of our Canadian operations. Accordingly, the operations in Canada have been accounted for as discontinued operations for all periods presented.

ITEM 4. CONTROLS AND PROCEDURES

As of the end of the period covered by this report, we carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and our Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures (as defined in Rule 13a-15(e) under the Securities Exchange Act of 1934, as amended). Based upon that evaluation, we concluded that as of June 30, 2008, our disclosure controls and procedures are effective to ensure that information required to be disclosed in reports that we file or submit under the Exchange Act are recorded, processed, summarized and reported within the time periods specified in the rules and forms of the Securities and Exchange Commission and that such information is accumulated and communicated to our management, including our Chief Executive Officer and Chief Financial Officer, to allow for timely decisions regarding required disclosures.

Changes in Internal Control over Financial Reporting

There were no changes in our internal control over financial reporting identified in connection with the evaluation required by paragraph (d) of Exchange Act Rules 13a-15 or 15d-15 that occurred during the period covered by this report that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

PART II. OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Refer to Note 9 to our consolidated financial statements of this Quarterly Report on Form 10-Q for a discussion of any recent material developments related to our legal proceedings since the filing of our most recent Annual Report on Form 10-K as updated by our subsequent Quarterly Reports on Form 10-Q.

ITEM 1A. RISK FACTORS

Except as set forth below, there have been no material changes to any of the risk factors disclosed in our most recently filed Annual Report on Form 10-K.



We derive a significant portion of our revenue from a few customers, and the loss of one of these customers or a reduction in their demand, the amount they pay or their ability to pay, for our services could impair our financial performance.

In the three months ended June 30, 2008, we derived approximately 35% and 8% of our revenue from DIRECTV[®] and Verizon, respectively. During the six months ended June 30, 2008, we derived approximately 40% and 9% of our revenue from DIRECTV[®] and Verizon, respectively. Because our business is concentrated among relatively few major customers, our revenue could significantly decline if we lose one or more of these customers or if the amount of business from any of these customer reduces significantly, which could result in reduced profitability and liquidity.

We have agreed to keep certain liabilities related to the state Department of Transportation related projects and assets that were sold in February 2007.

In connection with the sale of our state Department of Transportation related projects and assets and the related settlement explained in Note 9 in the condensed unaudited consolidated financial statements, we agreed to keep certain liabilities, mainly related to the cost to maintain and continue certain performance and payment bonds, certain obligations under leases between the parties and certain other litigation matters. We may also be unable to recover any losses we incur as a result of any third party claims to the extent any third parties seek payment from us directly and we are unable to recover such losses from the buyer pursuant to the indemnification obligations contained in the revised sale agreement; including, in the event the buyer were financially unable to meet certain obligations, any losses resulting from creditor claims.

Under the terms of the revised sale document, the buyer is no longer required to issue a standby letter of credit in our favor in February 2008 to cover any remaining exposure related to our bonded obligations. Instead, pursuant to the terms of the settlement agreement, the buyer entered into indemnity agreements directly with certain surety bonding companies in connection with our bonded obligations. Therefore, if the buyer is unable to meet its contractual obligations, the surety bonding company can seek its remedies from the buyer under the indemnity agreement. If the surety bonding company, however, pays the amounts due under the bonds, the surety bonding company will seek reimbursement of such payment from us. Accordingly, we may incur losses in the future related to these contingent liabilities if the buyer does not complete the bonded contracts and we are unable to recover such losses from the buyer pursuant to the indemnification provisions contained in the revised sale agreement. At June 30, 2008, we estimate that the cost to complete, on the \$159.7 million in performance and payment bonds related to these projects and assets, was \$6.6 million.

We recorded an unrealized loss in 2007 and 2008 to reduce the carrying value of certain auction rate securities we hold, and we may incur additional impairment charges with respect to auction rate securities in future periods.

The current overall credit concerns in capital markets may affect our ability to liquidate certain securities that we classify as securities available for sale on our balance sheet. As of June 30, 2008, all of our securities available for sale, or \$35.7 million in par value of auction rate securities, had insufficient bidders at the scheduled rollover dates. As a result, we have recorded an aggregate unrealized loss of \$6.8 million as of June 30, 2008. The Company's valuation is sensitive to market conditions and management's judgment and can change significantly based on the assumptions used. Factors that may impact the Company's valuation include changes to credit ratings of the securities as well as to the underlying assets supporting those securities, rates of default of the underlying assets, underlying collateral value, discount rates, counterparty risk and ongoing strength and quality of market credit and liquidity. At this time we are uncertain when the liquidity issues associated with these investments will improve, when we will be able to exit these investments at their par value or whether we will incur any additional losses as a result of these investments and as a result of this uncertainty, we have reclassified \$26.9 million of the book value of these securities to long-term assets at June 30, 2008. See Note 5 in the notes to condensed unaudited consolidated financial statements.

ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS

On May 30, 2008, MasTec acquired all of the issued and outstanding capital stock of Pumpco, Inc., which we refer to as Pumpco, for a purchase price of \$44 million, paid in cash, plus the retirement and assumption of certain liabilities and earn-out payments payable over a five-year period based on Pumpco's future performance. See Note 4 to our condensed unaudited consolidated financial statements in this Form 10-Q for additional information concerning the acquisition. The earn-out is payable in cash, MasTec common stock or a combination thereof. Pursuant to the terms of the Pumpco Purchase Agreement, MasTec may, at its option, issue up to 672,088 shares of its common stock, which we refer to as the Earn-Out Shares, to the Seller in connection with the earn-out. MasTec has offered to potentially issue the Earn-Out Shares to the Seller in reliance on the exemption from registration provided by Section 4(2) of the Securities Act of 1933, as amended. The Purchase Agreement only allows MasTec to issue the Earn-Out Shares, if it has first registered for resale such Earn-Out Shares pursuant to an effective registration statement that has been filed with the SEC.

ITEM 4. SUBMISSION OF MATTERS TO A VOTE OF SECURITY HOLDERS

We held our 2008 Annual Meeting of Shareholders on May 29, 2008 at which time the holders of a majority of our issued and outstanding common stock (61,936,983 out of the total issued and outstanding of 67,177,056) were present and voted to approve the election of our Class I directors noted in the chart below. The following Class II and Class III directors' terms of office continued after the meeting: Carlos M. de Cespedes, Austin Shanfelter, John Van Heuvelen, Robert Dwyer, Frank Jaumot and Jose Sorzano. As previously disclosed, in July 2008 Mr. de Cespedes resigned from our Board of Directors for personal reasons.

Set forth below are the results of the election of directors:

Class I Directors (term to expire in 2011)	Votes for	Votes Against	Votes Withheld
Ernst Csiszar	61,335,165		601,820
Julia Johnson	61,568,117	—	368,868
Jorge Mas	61,429,500	—	507,485
Jose Mas	61,485,903	—	451,082

In addition, the shareholders voted to re-approve the Section 162(m) of the Internal Revenue Code material terms of the 2003 Employee Stock Incentive Plan with 59,606,387 voting for this proposal, 2,052,367 voting against, and 278,229 withholding their vote.

ITEM 5. OTHER INFORMATION

The Company entered into an amended and restated Senior Secured Credit Facility with Bank of America, N.A., as collateral and administrative agent and General Electric Capital Corporation, as syndication agent, effective July 29, 2008, which expires May 10, 2013 (the "Credit Facility"). The Credit Facility has a maximum amount of available borrowing of \$210.0 million, subject to certain restrictions. The maximum available borrowing may be increased to \$260.0 million if certain conditions are met.

As in the past, the amount the Company can borrow at any given time is based upon a formula that takes into account, among other things, eligible billed and unbilled accounts receivable, equipment, real estate and eligible cash collateral, which can result in borrowing availability of less than the full amount of the Credit Facility. The Credit Facility is collateralized by a first priority security interest in substantially all of our assets and the assets of our wholly-owned subsidiaries and a pledge of the stock of certain of our operating subsidiaries. Interest under the Credit Facility accrues at variable rates based, at the Company's option, on the agent bank's base rate plus a margin of between 0.5% and 1.00%, or at the LIBOR rate (as defined in the Credit Facility) plus a margin of between 1.50% and 2.50%, depending on certain financial thresholds. The current margin over LIBOR is 2.0%. The Credit Facility includes an unused facility fee ranging from 0.25% to 0.375% based on usage.

The Credit Facility contains customary events of default (including cross-default) provisions and covenants related to our operations that prohibit, among other things, making investments and acquisitions in excess of specified amounts,

incurring additional indebtedness in excess of specified amounts, creating liens against Company assets, prepaying other indebtedness excluding our 7.625% senior notes, making acquisitions in excess of specified amounts, and engaging in certain mergers or combinations without the prior written consent of the lenders. The Credit Facility also limits our ability to make certain distributions or pay dividends. In addition, the Company is required to maintain a minimum fixed charge coverage ratio of 1.20 to 1.00, as defined in the Credit Facility. Any deterioration in the quality of billed and unbilled receivables, reduction in the value of equipment or an increase in lease expense related to real estate, could reduce availability under the Credit Facility.

ITEM 6. EXHIBITS

Exhibit No.	Description
3.1	Fourth Amended and Restated Bylaws of MasTec, Inc, amended and restated as of May 29, 2008 filed as Exhibit 3.1 to our Form 8-K filed with the SEC on June 3, 2008 and incorporated by reference herein.
10.1	Amendment to Employment Agreement dated April 14, 2008, between MasTec, Inc. and Austin J. Shanfelter filed as Exhibit 10.1 to our Form 8-K filed with the SEC on April 18, 2008 and incorporated by reference herein.
10.2	MasTec Deferred Compensation Plan effective as of June 1, 2008 filed as Exhibit 99.1 to our Form 8-K filed with the SEC on April 4, 2008 and incorporated by reference herein.
10.3	Stock Purchase Agreement executed on May 30, 2008 and dated as of May 1, 2008, between MasTec North America, Inc., as buyer, and Alan B. Roberts, as seller filed as Exhibit 10.1 to our Form 8-K filed with the SEC on June 5, 2008 and incorporated by reference herein.
10.4	Form of Promissory Note for the Equipment Term Loan dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation filed as Exhibit 10.2 to our Form 8-K filed with the SEC on June 5, 2008 and incorporated by reference herein.
10.5	Master Security Agreement dated May 30, 2008 between Pumpco, Inc. and General Electric Capital Corporation filed as Exhibit 10.3 to our Form 8-K filed with the SEC on June 5, 2008 and incorporated by reference herein.
10.6	Corporate Guaranty dated May 30, 2008 from MasTec, Inc. to General Electric Capital Corporation filed as Exhibit 10.4 to our Form 8-K filed with the SEC on June 5, 2008 and incorporated by reference herein.
10.7	Second Amended and Restated Loan and Security Agreement dated July 29, 2008, by and among MasTec, Inc., certain of its subsidiaries, Bank of America, N.A., as collateral and administrative agent and General Electric Capital Corporation, as syndication agent.
31.1	Certification of Chief Executive Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
31.2	Certification of Chief Financial Officer, pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.
32.1	Certification of Chief Executive Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.
32.2	Certification of Chief Financial Officer, pursuant to 18 U.S.C. Section 1350, as Adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

MASTEC, INC.

Date: July 30, 2008

/s/ Jose R. Mas

Jose R. Mas President and Chief Executive Officer (Principal Executive Officer)

/s/ C. Robert Campbell

C. Robert Campbell Chief Financial Officer (Principal Financial and Accounting Officer)

SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT

THIS SECOND AMENDED AND RESTATED LOAN AND SECURITY AGREEMENT is made on July 29, 2008, by and among MASTEC, INC., a Florida corporation (individually and, in its capacity as the representative of the other Borrowers pursuant to Section 4.4 hereof, "MasTec"), a Florida corporation with its chief executive office and principal place of business at 800 Douglas Road, North Tower, 12th Floor, Coral Gables, Florida 33134; its subsidiaries named as "Borrowers" on <u>Annex A</u> hereto, MasTec, and each other Borrower being referred to collectively as "Borrowers," and individually as a "Borrower"); its subsidiaries party hereto as "Guarantors"; the various financial institutions listed on the signature pages hereof (together with their respective successors and permitted assigns, the "Lenders"); BANK OF AMERICA, N.A., a national banking association, in its capacity as collateral and administrative agent for the Lenders pursuant to Section 13 hereof (together with its successors in such capacity, "Agent"), and GENERAL ELECTRIC CAPITAL CORPORATION, a Delaware corporation, as Syndication Agent (together with its successors in such capacity, the "Syndication Agent"). Capitalized terms used in this Agreement have the meanings assigned to them in Section 1.

Recitals:

MasTec, its Affiliates party thereto as borrowers, certain financial institutions (the "Original Lenders") and Fleet Capital Corporation ("FCC"), as Administrative Agent (the "Original Agent"), were parties to a certain Revolving Credit and Security Agreement dated January 22, 2002 (as at any time amended, restated, modified or otherwise supplemented, the "Original Loan Agreement"), pursuant to which the Original Lenders made certain revolving credit loans and letter of credit accommodations to Borrowers.

Prior to the date of the Existing Loan Agreement (as defined below), (i) FCC assigned all of its interests in the loans under the Original Loan Agreement to Bank of America, N.A., (ii) FCC resigned as Administrative Agent under the Original Loan Agreement and (iii) Bank of America, N.A. was appointed as successor Administrative Agent by the Lenders party to the Original Loan Agreement.

MasTec, its Affiliates party thereto as borrowers, certain financial institutions (the "Existing Lenders") and Agent, are parties to a certain Amended and Restated Loan and Security Agreement dated May 10, 2005 (as at any time amended, restated, modified or otherwise supplemented, and as in effect on the date hereof, the "Existing Loan Agreement"), pursuant to which the Existing Lenders made certain revolving credit loans and letter of credit accommodations to Borrowers.

Borrowers have requested that the Existing Loan Agreement be amended and restated in its entirety to become effective and binding on the Obligors pursuant to the terms hereof, and the Lenders (including the Existing Lenders that are parties hereto) have agreed, subject to the terms of this Agreement, to amend and restate the Existing Loan Agreement in its entirety to read as set forth herein, and it has been agreed by the parties hereto that (a) the commitments which the Existing Lenders that are parties hereto extended to Borrowers under the Existing Loan Agreement and the commitments of new Lenders that become parties hereto shall be extended or advanced upon the amended and restated terms and conditions contained in this Agreement, (b) the Loans and other Secured Obligations outstanding under (and as defined in) the Existing Secured Obligations are and shall continue to be (and all Obligations incurred pursuant hereto shall be) secured by the Loan Documents, which for purposes of this Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Secured Obligations are and shall continue to be (and all Obligations incurred pursuant hereto shall be) secured by the Loan Documents, which for purposes of this Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall include the Loan Documents under (and as defined in) the Existing Loan Agreement shall

hereafter amended, modified, supplemented or restated in connection with the credit facility under this Agreement.

Obligors have requested that the Lenders establish a revolving credit facility pursuant to which revolving credit loans may be made to Borrowers and letters of credit may be issued for the account of Borrowers, which shall be used by Borrowers to finance their mutual and collective enterprise of providing construction, design, installation, maintenance and upgrade services to providers of telecommunications, broadband, energy services, traffic control and homeland security services. In order to utilize the financial powers of each Borrower in the most efficient and economical manner, and in order to facilitate the financing of each Borrower's needs, Lenders will, at the request of any Borrower, make loans to all Borrowers under the revolving credit facility on a combined basis and in accordance with the provisions hereinafter set forth. Borrowers' business is a mutual and collective enterprise and Borrowers believe that the consolidation of all revolving credit loans under this Agreement will enhance the aggregate borrowing powers of each Borrower and ease the administration of their revolving credit loan relationship with Lenders, all to the mutual advantage of Borrowers. Lenders' willingness to extend credit to Borrowers and to administer each Borrower's collateral security therefor, on a combined basis as more fully set forth in this Agreement, is done solely as an accommodation to Borrowers and at Borrowers' request in furtherance of Borrowers' mutual and collective enterprise.

Each Borrower has agreed to be liable for loans and all outstanding other obligations under this Agreement and to guarantee the obligations of each of the other Borrowers under this Agreement and each of the other Loan Documents.

NOW, THEREFORE, for Ten Dollars (\$10.00) and other good and valuable consideration, the parties hereto, intending to be bound hereby, agree that the Existing Loan Agreement is hereby amended and restated in its entirety by this Agreement:

SECTION 1. DEFINITIONS; RULES OF CONSTRUCTION

1.1. Definitions.

As used in this Agreement, the following terms shall have the following meanings ascribed to them (terms used in the singular to have the same meaning when used in the plural, and *vice versa*):

<u>Account</u> — shall have the meaning given to the term "account" in the UCC or the PPSA, as applicable, and shall include any and all rights of an Obligor to payment for goods sold or leased or for services rendered that are not evidenced by an Instrument or Chattel Paper, whether or not they have been earned by performance.

Account Debtor — a Person who is or becomes obligated under or on account of an Account, Chattel Paper or General Intangible.

<u>Accounts Formula Amount</u> — on any date of determination thereof, an amount equal to the lesser of (i) the Revolver Commitments on such date or (ii) 85% (or such lesser percentage as Agent may in its reasonable credit judgment determine from time to time) of the net amount of Eligible Accounts on such date. As used herein, the phrase "net amount of Eligible Accounts" shall mean the face amount of such Accounts on any date less any and all returns, rebates, discounts (which may, at Agent's option, be calculated on shortest terms), credits, allowances or Taxes (including sales, excise or other taxes) at any time issued, owing, claimed by Account

-2-

Debtors, granted, outstanding or payable in connection with, or any interest accrued on the amount of, such Accounts at such date.

<u>Ace LC</u> — a Letter of Credit issued for the account of MasTec for the benefit of ACE American Insurance Company ("<u>ACE</u>") in the stated amount of \$18,000,000 with respect to MasTec's insurance coverage for claims under worker's compensation laws or similar legislation effected through ACE from time to time.

<u>Acquisition</u> — the acquisition of a Business Unit or of Equity Interests in another Person sufficient to give the acquiring Person control of such other Person, in each case whether by purchase, exchange, issuance of stock or other securities, or by merger, reorganization or any other method.

<u>Acquisition Earn-Out Payments</u> — earn-out payments required to be paid by Borrowers relating to any Permitted Acquisition pursuant to the terms of the acquisition documentation with the applicable seller(s), to the extent paid by Borrowers in cash.

<u>Adjusted EBITDA</u> — for any fiscal period of Borrowers and their Subsidiaries (other than DirectStar), an amount equal to the sum for such period of (i) Adjusted Net Earnings, <u>plus</u> (ii) provision for taxes based on income and for state or provincial franchise taxes, to the extent deducted in the calculation of Adjusted Net Earnings, <u>plus</u> (iii) interest expense, to the extent deducted in the calculation of Adjusted Net Earnings, <u>plus</u> (iv) depreciation and amortization to the extent deducted in the calculation of Adjusted Net Earnings, <u>plus</u> (iv) depreciation and algustments that are as required by FASB 141 and 142, <u>plus</u> (vi) non-cash charges (including inventory adjustments, loss on job contract accruals, litigation and legacy accounts receivable charges initially recognized for the third Fiscal Quarter of 2007, expenses relating to equity award compensation and write down of assets and the cumulative effect of changes in accounting principles under GAAP) either (A) as approved by Agent or (B) from discontinued operations to the extent deducted in the calculation of Adjusted Net Earnings for such period, all calculated on a Consolidated basis, <u>plus</u> (vii) without duplication, any cash Distributions made by DirectStar to any Borrower, all calculated on a Consolidated basis; <u>provided that</u>, solely for the purpose of determining the Leverage Ratio in connection with the calculation of the Applicable Margin, (x) Adjusted EBITDA shall be calculated on a Consolidated basis for Borrowers and their Subsidiaries (excluding those discontinued operations of Borrowers and Guarantors reported to Agent in writing by Borrowers on July 31, 2007), and (y) Adjusted EBITDA shall include, with respect to any Permitted Acquisition, the Adjusted EBITDA of the acquired Person for the twelve month period immediately preceding the Permitted Acquisition.

<u>Adjusted LIBOR Rate</u> — for any Interest Period, with respect to LIBOR Loans, the rate of interest per annum determined pursuant to the following formula:

LIBOR Rate = Offshore Base Rate

1.00 - Eurodollar Reserve Percentage

Where,

"<u>Offshore Base Rate</u>" means the rate per annum appearing on Telerate Page 3750 (or any successor page) as the London interbank offered rate for deposits in Dollars at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period for a term comparable to such

-3-

Interest Period. If for any reason such rate is not available, the Offshore Base Rate shall be, for any Interest Period, the rate per annum appearing on Reuters Screen LIBO Page as the London interbank offered rate for deposits in Dollars at approximately 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period for a term comparable to such Interest Period; provided, however, if more than one rate is specified on Reuters Screen LIBO Page, the applicable rate shall be the arithmetic mean of all such rates. If for any reason none of the foregoing rates is available, the Offshore Base Rate shall be, for any Interest Period, the rate per annum determined by Agent as the rate of interest at which Dollar deposits in the approximate amount of the applicable LIBOR Loan would be offered by BofA's London Branch to major banks in the offshore Dollar market at their request at or about 11:00 a.m. (London time) two Business Days prior to the first day of such Interest Period for a term comparable to such Interest Period.

"<u>Eurodollar Reserve Percentage</u>" means, for any day during any Interest Period, the reserve percentage (expressed as a decimal, rounded upward to the next 1/8th of 1%) in effect on such day applicable to member banks under regulations issued from time to time by the Board of Governors for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to Eurocurrency funding (currently referred to as "Eurocurrency liabilities" under Regulation D of the Board of Governors). The Offshore Rate for each outstanding LIBOR Loan shall be adjusted automatically as of the effective date of any change in the Eurodollar Reserve Percentage.

<u>Adjusted Net Earnings</u> — with respect to any fiscal period, the net earnings (or loss) for such fiscal period of Borrowers and their Subsidiaries, all as reflected on the financial statement of Borrowers and their Subsidiaries supplied to Agent and Lenders pursuant to **Section 10.1.3**, but excluding: (i) any gain or loss arising from the sale of capital assets; (ii) any gain arising from any write-up of assets during such period; (iii) earnings of any Subsidiary accrued prior to the date it became a Subsidiary; (iv) earnings of any Person, substantially all the assets of which have been acquired in any manner by a Borrower or Subsidiary, realized by such Person prior to the date of such acquisition; (v) net earnings of any entity (other than a Subsidiary of a Borrower) in which a Borrower has an ownership interest unless such net earnings have actually been received by a Borrower in the form of cash Distributions; (vi) any portion of the net earnings of any Subsidiary which for any reason is unavailable for payment of Distributions to a Borrower; (vii) the earnings of any Person to which any assets of a Borrower shall have been sold, transferred or disposed of, or into which a Borrower shall have merged, or been a party to any consolidation or other form of reorganization, prior to the date of such transaction; (viii) any gain arising from the acquisition of any Equity Interests of a Borrower; and (ix) any gain arising from extraordinary or non-recurring items, all as determined in accordance with GAAP on a Consolidated basis.

<u>Affiliate</u> — a Person: (i) which directly or indirectly through one or more intermediaries controls, or is controlled by, or is under common control with, another Person; (ii) which beneficially owns or holds 10% or more of any class of the Equity Interests of a Person; or (iii) 10% or more of the Equity Interests with power to vote of which is beneficially owned or held by another Person or a Subsidiary of another Person. For purposes hereof, "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of any Equity Interest, by contract or otherwise.

-4-

<u>Agency Appointment as to Vehicle Titles</u> — as defined in **Section 8.4.4** of this Agreement.

<u>Agent Indemnitees</u> — Agent and all of Agent's present and future officers, directors, employees, agents and attorneys.

<u>Agent Professionals</u> — attorneys, accountants, appraisers, business valuation experts, environmental engineers or consultants, turnaround consultants and other professionals or experts retained by Agent.

Agreement — this Second Amended and Restated Loan and Security Agreement and all Annexes, Exhibits and Schedules thereto.

Anti-Terrorism Laws — any laws relating to terrorism or money laundering, including Executive Order No. 13224 and the USA Patriot Act.

<u>Applicable Law</u> — all laws, rules and regulations applicable to the Person, conduct, transaction, covenant, Loan Document or Material Contract in question, including all applicable common law and equitable principles; all provisions of all applicable state, provincial, federal and foreign constitutions, statutes, rules, regulations and orders of Governmental Authorities; and all orders, judgments and decrees of all courts and arbitrators.

<u>Applicable Margin</u> — a percentage equal to 0.50% with respect to Revolver Loans that are Base Rate Loans and 2.00% with respect to Revolver Loans that are LIBOR Loans; <u>provided</u>, <u>that</u> commencing on the first day of the calendar month immediately succeeding the third Business Day (each an "Adjustment Date") after Agent's receipt of the applicable financial statements and corresponding Compliance Certificate for each Fiscal Quarter ending on or after June 30, 2009, the Applicable Margin shall be increased or (if no Default or Event of Default exists) decreased, on a quarterly basis according to the performance of Borrowers as measured by the Leverage Ratio for the immediately preceding Fiscal Quarter of Borrowers, as follows:

Level	Leverage Ratio	Applicable LIBOR Margin	Applicable Base Rate Margin					
Ι	≥ 4.00 to 1.00	2.50%	1.00%					
II	≥ 3.00 to 1.00 but < 4.00 to 1.00	2.25%	0.75%					
III	≥ 2.00 to 1.00 but < 3.00 to 1.00	2.00%	0.50%					
IV	≥ 1.50 to 1.00 but < 2.00 to 1.00	1.75%	0.50%					
V	≥ 1.00 to 1.00 but < 1.50 to 1.00	1.625%	0.50%					
VI	< 1.00 to 1.00	1.50%	0.50%					
	Except as otherwise set forth herein, any such increase or reduction in the Applicable							

-5-

Margin shall be subject to receipt by Agent of the applicable financial statements and corresponding Compliance Certificate. If the financial statements and the Compliance Certificate of Borrowers setting forth the Leverage Ratio are not received by Agent by the date required pursuant to Section 10.1.3 of this Agreement, the Applicable Margin shall be determined as if the Leverage Ratio exceeds 4.00 to 1.00 until such time as such financial statements and Compliance Certificate are received and any Event of Default resulting from a failure to timely deliver such financial statements or Compliance Certificate is waived in writing by Agent and Lenders; provided, however, that Agent and Lenders shall be entitled to accrue and receive interest at the Default Rate to the extent authorized by Section 3.1.5 of this Agreement and, on each date that the Default Rate accrues on any Loan, the Applicable Margin on such date for such Loan shall be the Applicable Margin that would apply if the Leverage Ratio exceeded 4.00 to 1.00 (without regard to the actual Leverage Ratio). For the final Fiscal Quarter of any Fiscal Year of Borrowers, Borrowers may provide the unaudited financial statements of Borrowers, subject only to year-end adjustments, for the purpose of determining the Applicable Margin; provided, however, that if, upon delivery of the annual audited financial statements required to be submitted by Borrowers to Agent pursuant to Section 10.1.3(i) of this Agreement, Borrowers have not met the criteria for reduction of the Applicable Margin pursuant to the terms hereinabove for the final Fiscal Quarter of the Fiscal Year of Borrowers then ended, then (a) such Applicable Margin reduction shall be terminated and, effective on the first day of the month following receipt by Agent of such audited financial statements, the Applicable Margin shall be the Applicable Margin that would have been in effect if such reduction had been implemented based upon the audited financial statements of Borrowers for the final Fiscal Quarter of the Fiscal Year of Borrowers then ended, and (b) Borrowers shall pay to Agent, for the Pro Rata benefit of the Lenders, on the first day of the month following receipt by Agent of such audited financial statements, an amount equal to the difference between the amount of interest that would have been paid using the Applicable Margin determined based upon such audited financial statements and the amount of interest actually paid during the period in which the reduction of the Applicable Margin was in effect based upon the unaudited financial statements for the final Fiscal Quarter of the Fiscal Year of Borrowers then ended.

<u>Approved Account Debtor</u> — on any date of determination (i) each Account Debtor whose corporate credit rating or senior debt rating (secured or unsecured), or any of them, by Moody's and S&P is at least Baa3 and BBB-, respectively, and (ii) each Account Debtor that is named on the list of proposed Approved Account Debtors delivered to Agent by Borrower Agent with the Borrowing Base Certificate most recently delivered prior to such date of determination and that Agent has approved in writing as an Approved Account Debtor (which approval shall be effective until the next Borrowing Base Certificate is timely delivered to Agent by Borrower Agent). Approved Account Debtors as of the date of this Agreement are listed on <u>Annex B</u>.

<u>Approved Credit Enhancement</u> — in Agent's sole discretion and at its option, either (i) an irrevocable letter of credit that is in form and substance acceptable to Agent, issued or confirmed by a bank acceptable to Agent, and payable in Dollars at a place of payment within the United States that is acceptable to Agent, which letter of credit is assigned to Agent for the benefit of the Secured Parties (with such assignment acknowledged by the issuing or confirming bank) or, if so requested by Agent, duly transferred to Agent for the benefit of the Secured Parties (together with sufficient documentation to permit direct draws under any such letter of credit by Agent for the benefit of the Secured Parties) or (ii) credit insurance that is issued by a credit insurance company acceptable to Agent and is in form and substance acceptable to Agent (which credit insurance shall be payable to Agent for the benefit of the Secured Parties in Dollars).

-6-

<u>Asset Disposition</u> — a sale, lease, license, consignment or other transfer or disposition of assets (real or personal, tangible or intangible) of an Obligor, including a termination of rights of any Obligor under any lease, license agreement or other contract or a disposition of Property in connection with a sale-leaseback transaction or synthetic lease.

<u>Assignment and Acceptance</u> — an assignment and acceptance entered into by a Lender and an Eligible Assignee and accepted by Agent, in the form of **Exhibit G**.

AT&T — collectively, AT&T Corp. and its Subsidiaries and Affiliates.

AT&T Concentration Percentage — 40%, or such lesser percentage as Agent may in its reasonable credit judgment determine from time to time.

<u>Availability</u> — on any date, the amount that Borrowers are entitled to borrow as Revolver Loans on such date, such amount being the difference derived when the sum of the principal amount of Revolver Loans then outstanding (including any amounts that Agent or Lenders may have paid for the account of Obligors pursuant to any of the Loan Documents and that have not been reimbursed by Borrowers and any outstanding Swingline Loans) is subtracted from the Borrowing Base on such date. If the amount outstanding is equal to or greater than the Borrowing Base, Availability is zero.

<u>Availability Reserve</u> — on any date of determination thereof, an amount equal to the sum of the following (without duplication): (i) the Rent Reserve; (ii) any amounts which any Obligor is obligated to pay to Agent, Lenders or other Persons pursuant to the provisions of any of the Loan Documents that Agent or any Lender elects to pay for the account of such Obligor in accordance with authority contained in any of the Loan Documents; (iii) the LC Reserve; (iv) the aggregate amount of reserves established by Agent from time to time in its discretion in respect of Banking Relationship Debt; (v) the aggregate amount of Royalties that have accrued and are unpaid, whether or not then due and payable by an Obligor; (vi) the aggregate amount of all liabilities and obligations that are secured by Liens upon any of the Collateral that are senior in priority to Agent's Liens if such Liens are not Permitted Liens (provided that the imposition of a reserve hereunder on account of such Liens shall not be deemed a waiver of the Event of Default that arises from the existence of such Liens) or are Permitted Liens under **Section 10.2.5(iii)** of this Agreement; (vii) the DirecTV Account Reserve; (viii) the Dilution Reserve; and (ix) such additional reserves, in such amounts and with respect to such matters, as Agent in its reasonable credit judgment may elect to impose from time to time.

<u>Average Days Sales Outstanding</u> — with respect to Accounts for which a specified Person is the Account Debtor, as of the last day of any calendar month, that number of days derived (a) by calculating the average amount of Accounts of Borrowers and their Subsidiaries for which such specified Person is the Account Debtor for the last 3 months ending with such month by dividing the sum of the Accounts of Borrowers and their Subsidiaries for which such specified Person is the Account Debtor as of the last day of each such month, determined on a consolidated basis in accordance with GAAP, for such 3 months by 3, (b) by calculating the average amount of net revenues of Borrowers and their Subsidiaries attributable to such specified Account Debtor for the same 3 months by dividing the sum of the net revenues of Borrowers and their Subsidiaries attributable to such specified Person, determined on a consolidated basis in accordance with GAAP, for such 3 months by 3, (c) by dividing the amount resulting from the calculation described in (b), and (d) by multiplying the result of the calculation described in (c) by 30.

-7-

<u>Average Revolver Loan Balance</u> — for any period, the amount obtained by adding the unpaid balance of Revolver Loans and LC Obligations at the end of each day for the period in question and by dividing such sum by the number of days in such period.

<u>Bank Products</u> — any one or more of the following types of products, services or facilities extended to any Obligor (other than any Canadian Obligor) by any Lender or any Affiliate of any Lender: (i) commercial credit cards; (ii) merchant card services; (iii) products or services under Cash Management Agreements; (iv) Hedging Agreements; (v) interstate depository network services; and (vi) such other banking products or services provided by any Lender or any Affiliate of any Lender as may be requested by any Obligor (whether requested on behalf of itself or its Subsidiaries), other than Letters of Credit and Bank Products Exclusions.

<u>Bank Products Exclusions</u> — banking products or services provided by any Lender or any Affiliate of any Lender that consist of financings of Equipment with Permitted Purchase Money Debt or providing operating lease financing of Equipment other than the financing provided under this Agreement or the other Loan Documents.

<u>Banking Relationship Debt</u> — Debt or other obligations of an Obligor (other than any Canadian Obligor) to any Lender (or any Affiliate of any Lender) arising out of or relating to Bank Products.

Bankruptcy Code — title 11 of the United States Code.

<u>Base Rate</u> — the rate of interest announced or quoted by BofA from time to time as its prime rate. The prime rate announced by BofA is a reference rate and does not necessarily represent the lowest or best rate charged by BofA. BofA from time to time makes loans or other extensions of credit at, above or below its announced prime rate. If the prime rate is discontinued by BofA as a standard, a comparable reference rate designated by BofA as a substitute therefor shall be the Base Rate.

Base Rate Loan — a Loan, or portion thereof, during any period in which it bears interest at a rate based upon the Base Rate.

Blocked Person — as defined in Section 9.1.27(ii) of this Agreement.

Board of Governors — the Board of Governors of the Federal Reserve System.

BofA — Bank of America, N.A., a national banking association, and its successors and assigns.

BofA Indemnitees — BofA and all of its present and future officers, directors, employees agents and attorneys.

<u>Borrower</u> — each of MasTec and each Subsidiary of MasTec listed as a "Borrower" on **Annex A** hereto and each Subsidiary of MasTec that becomes a "Borrower" after the Effective Date pursuant to **Section 10.1.11** of this Agreement.

Borrower Agent — as defined in Section 4.4 of this Agreement.

-8-

<u>Borrowing</u> — a borrowing consisting of Loans of one Type made on the same day by Lenders (or by BofA in the case of a Borrowing funded by Swingline Loans) or a conversion of a Loan or Loans of one Type from Lenders on the same day.

<u>Borrowing Base</u> — on any date of determination thereof, an amount equal to the lesser of: (a) the aggregate amount of the Revolver Commitments on such date <u>minus</u> the Availability Reserve on such date, or (b) an amount equal to (i) the sum of (A) the Accounts Formula Amount on such date <u>plus</u> (B) the Eligible Unbilled Accounts Formula Amount on such date <u>plus</u> (C) the Fixed Assets Formula Amount on such date <u>plus</u> (D) the Real Estate Formula Amount on such date <u>plus</u> (E) the Eligible Cash Collateral Amount on such date <u>minus</u> (ii) the Availability Reserve on such date, or (c) the sum of (i) the amount on such date that constitutes Indebtedness permitted under Section 4.08(b)(1) of the Indenture as in effect on the Closing Date, plus (ii) the amount of the Indenture as in effect on the Closing Date.

<u>Borrowing Base Certificate</u> — a certificate, in the form requested by Agent, by which Borrowers shall certify to Agent and Lenders, with such frequency as Agent may request or as otherwise required by Section 8.6 of this Agreement, the amount of the Borrowing Base as of the date of the certificate and the calculation of such amount.

<u>Business Day</u> — any day excluding Saturday, Sunday and any other day that is a legal holiday under the laws of the State of Georgia or is a day on which banking institutions located in such state are closed; <u>provided</u>, <u>however</u>, that when used with reference to a LIBOR Loan (including the making, continuing, prepaying or repaying of any LIBOR Loan), the term "Business Day" shall also exclude any day on which banks are not open for dealings in Dollar deposits on the London interbank market.

<u>Business Interruption Insurance Assignment</u> — the Collateral Assignment of Business Interruption Insurance to be executed by each Obligor and pursuant to which such Obligor shall assign in favor of Agent, for the benefit of Secured Parties, all of such Obligor's rights under any business interruption insurance policy owned by or in favor of such Obligor, as security for the Obligations.

Business Unit — assets constituting a business, whether all of the assets of any Person or the assets of a division or operating unit of any Person.

<u>Cdn\$</u> and <u>Canadian Dollars</u> — each means freely transferable Canadian dollars.

<u>Canadian Obligor</u> — Phasecom Systems Inc., an Ontario (Canada) corporation, and Integral Power & Telecommunications Corporation Ltd., a Canadian corporation, and both Wholly Owned Subsidiaries of MasTec and each other direct or indirect Wholly Owned Subsidiary organized under the laws of Canada or any province thereof that becomes an Obligor, singly or collectively.

<u>Canadian Obligor Guarantee</u> — each Guaranty executed by a Canadian Obligor.

<u>Canadian Security Agreement</u> — collectively, each General Security Agreement, Deed of Hypothec and other security instrument with respect to Property of a Canadian Obligor executed and delivered by a Canadian Obligor, pursuant to which such Canadian Obligor grants in favor of Agent for the benefit of the Secured Parties a first priority Lien upon its personal Properties.

-9-

<u>Capital Adequacy Regulation</u> — any guideline, request or directive of any central bank or other Governmental Authority, or any other law, rule or regulation, whether or not having the force of law, in each case regarding capital adequacy of any bank or of any corporation controlling a bank.

<u>Capital Expenditures</u> — expenditures made or liabilities incurred by an Obligor for the acquisition of any fixed assets or improvements, replacements, substitutions or additions thereto which have a useful life of more than one year, including the total principal portion of Capitalized Lease Obligations.

<u>Capitalized Lease Obligation</u> — any Debt represented by obligations under a lease that is required to be capitalized for financial reporting purposes in accordance with GAAP.

<u>Cash Collateral</u> — cash, and any interest or other income earned thereon, that is deposited with Agent in accordance with this Agreement for the Pro Rata benefit of Lenders to Cash Collateralize any LC Obligations.

<u>Cash Collateral Account</u> — a demand deposit, money market or other account established by Agent at such financial institution as Agent may select in its discretion, which account shall be in Agent's name and subject to Agent's Liens for the benefit of Secured Parties.

<u>Cash Collateralize</u> — with respect to LC Obligations arising from Letters of Credit outstanding on any date or Obligations arising under Hedging Agreements on such date, the deposit with Agent of immediately available funds into the Cash Collateral Account in an amount equal to 105% of the sum of the aggregate Undrawn Amounts of such Letters of Credit and other LC Obligations, all Obligations existing under such Hedging Agreements, and all related fees and other amounts due or to become due in connection with such LC Obligations and Hedging Agreements.

<u>Cash Equivalents</u> — (i) marketable direct obligations issued or unconditionally guaranteed by the United States government and backed by the full faith and credit of the United States government; (ii) domestic certificates of deposit and time deposits having maturities of not more than 12 months from the date of acquisition, bankers' acceptances having maturities of not more than 12 months from the date of acquisition and overnight bank deposits, in each case issued by any commercial bank organized under the laws of the United States, any state thereof or the District of Columbia, which at the time of acquisition are rated A-1 (or better) by S&P or P-1 (or better) by Moody's, and (unless issued by a Lender) not subject to offset rights in favor of such bank arising from any banking relationship with such bank; (iii) repurchase obligations with a term of not more than 30 days for underlying securities of the types described in clauses (i) and (ii) entered into with any financial institution meeting the qualifications specified in clause (ii) above; (iv) commercial paper having at the time of investment therein or a contractual commitment to invest therein a rating of A-1 (or better) by S&P or P-1 (or better) by Moody's, and (v) shares of any money market fund that (a) has substantially all of its assets invested continuously in the types of investments referred to in clauses (i) — (iv), (b) has net assets not less than \$500,000,000 and (c) has the highest rating obtainable from either Moody's or S&P.

<u>Cash Management Agreements</u> — any agreement entered into from time to time between any Obligor or any of its Subsidiaries, on the one hand, and a Lender or any of its Affiliates or any other banking or financial institution, on the other, in connection with cash management services for operating, collections, payroll and trust accounts of such Obligor or its Subsidiaries

-10-

provided by such banking or financial institution, including automatic clearinghouse services, controlled disbursement services, electronic funds transfer services, information reporting services, lockbox services, stop payment services and wire transfer services.

CERCLA — the Comprehensive Environmental Response Compensation and Liability Act, 42 U.S.C. § 9601 et seq. and its implementing regulations.

<u>Change of Control</u> — the occurrence of any of the following events after the date of this Agreement: (a) MasTec shall cease to own, directly, or indirectly through a Subsidiary, 100% of the Equity Interests in any Obligor (other than as a result of a merger, reorganization, consolidation or amalgamation permitted under **Section 10.2.1** of this Agreement), or (b) any "Change of Control," "Change in Control" or similar event or circumstance, however defined or designated, under the Indenture (as in effect on the date of this Agreement) shall occur.

Chattel Paper — shall have the meaning given to the term "chattel paper" in the UCC or the PPSA, as applicable.

<u>Clearing Bank</u> — BofA, Wachovia Bank, N.A., each bank listed on **Schedule 1.1C** - **Clearing Banks**, and any other U.S. banking institution with which a Controlled Account has been established pursuant to a Control Agreement.

<u>Closing Date</u> — the date on which all of the conditions precedent in **Section 11** of this Agreement are satisfied or waived and the initial Loans are made under this Agreement.

<u>Collateral</u> — all of the Property and interests in Property described in **Section 7** of this Agreement; all Property described in any of the Security Documents as security for the payment or performance of any of the Obligations; and all other Property and interests in Property that now or hereafter secure (or are intended to secure) the payment and performance of any of the Obligations.

<u>Commercial Tort Claim</u> — shall have the meaning given to the term "commercial tort claim" in the UCC.

<u>Commitment</u> — at any date for any Lender, the aggregate amount of such Lender's Revolver Commitment on such date, and "<u>Commitments</u>" means the aggregate amount of all Revolver Commitments.

<u>Commitment Termination Date</u> — the date that is the soonest to occur of (i) the last day of the Term; (ii) the date on which either Borrowers or Agent terminates the Commitments pursuant to **Section 6.2** of this Agreement; or (iii) the date on which the Commitments are automatically terminated pursuant to **Section 12.2** of this Agreement.

<u>Compliance Certificate</u> — a Compliance Certificate to be provided by Borrowers to Agent in accordance with, and in the form annexed as <u>Exhibit E</u> to, this Agreement, and the supporting schedules to be annexed thereto.

<u>Consolidated</u> — the consolidation in accordance with GAAP of the accounts or other items as to which such term applies.

<u>Contingent Obligation</u> — with respect to any Person, any obligation of such Person arising from any guaranty, indemnity or other assurance of payment or performance of any Debt, lease,

-11-

dividend or other obligation ("primary obligations") of any other Person (the "primary obligor") in any manner, whether directly or indirectly, including (i) the direct or indirect guaranty, endorsement (other than for collection or deposit in the Ordinary Course of Business), co-making, discounting with recourse or sale with recourse by such Person of the obligation of a primary obligor, (ii) the obligation to make take-or-pay or similar payments, if required, regardless of nonperformance by any other party or parties to an agreement, (iii) any obligation of such Person, whether or not contingent, (A) to purchase any such primary obligation or any Property constituting direct or indirect security therefor, (B) to advance or supply funds (1) for the purchase or payment of any such primary obligations or (2) to maintain working capital or equity capital of the primary obligor or otherwise to maintain the net worth or solvency of the primary obligor, (C) to purchase Property, Securities or services primarily for the purpose of assuring the owner of any such primary obligation against loss in respect thereof. The amount of any Contingent Obligation shall be deemed to be an amount equal to the stated or determinable amount of the primary obligation with respect to which such Contingent Obligation is made (or, if less, the maximum amount of such primary obligation for which such Person may be liable pursuant to the terms of the instrument evidencing such Contingent Obligation) or, if not stated or determinable, the maximum reasonably anticipated liability with respect thereto (assuming such Person is required to perform thereunder), as determined by such Person in good faith.

<u>Control</u> or <u>controlled by</u> or <u>under common control</u> — possession, directly or indirectly, of the power to direct or cause the direction of management or policies (whether through ownership of voting securities, by contract or otherwise, but not solely by being an officer or director of that Person); <u>provided</u>, <u>however</u>, that in any event any Person which beneficially owns, directly or indirectly, 10% or more (in number of votes) of the Equity Interest having ordinary Voting Power with respect to another Person shall be conclusively presumed to control such Person.

<u>Control Agreement</u> — an agreement among an Obligor, Agent and a Clearing Bank, in form and substance satisfactory to Agent, concerning the collection and transfer of payments which represent the proceeds of Accounts or of any other Collateral, and which gives Agent, for the benefit of the Secured Parties, control (within the meaning of Section 9-104 of the UCC) over the deposit accounts of such Obligor maintained at such Clearing Bank.

Controlled Account — a demand deposit account of an Obligor maintained by it with a Clearing Bank that is subject to a Control Agreement.

<u>CWA</u> — the Clean Water Act (33 U.S.C. §§ 1251 <u>et seq</u>.).

<u>Debt</u> — as applied to a Person means, without duplication: (i) all items which in accordance with GAAP would be included in determining total liabilities as shown on the liability side of a balance sheet of such Person as of the date as of which Debt is to be determined, including Capitalized Lease Obligations; (ii) all Contingent Obligations of such Person; (iii) all reimbursement obligations in connection with letters of credit or letter of credit guaranties issued for the account of such Person; and (iv) in the case of an Obligor (without duplication), the Obligations. The Debt of a Person shall include any recourse Debt of any partnership or joint venture in which such Person is a general partner or joint venturer.

<u>Default</u> — an event or condition the occurrence of which would, with the lapse of time or the giving of notice, or both, become an Event of Default.

<u>Default Rate</u> — on any date, a rate per annum that is equal to (i) in the case of each Revolver Loan outstanding on such date, 2% in excess of the rate otherwise applicable to such Loan on such date, and (ii) in the case of any of the other Obligations outstanding on such date, 2% <u>plus</u> the highest Applicable Margin for Base Rate Loans.

Deposit Account — shall have the meaning given to the term "deposit account" in the UCC.

<u>Dilution Reserve</u> — at any date of determination, an amount equal to the excess of (i) non-cash reductions to Obligors' Accounts during a 12-month period prior to the date of determination as established by Obligors' records or by a field examination conducted by Agent's employees or representatives, expressed as a percentage of Obligors' Accounts outstanding during the same period, as the same may be adjusted by Agent in the exercise of its reasonable credit judgment, over (ii) 5%, multiplied by the aggregate amount of Eligible Accounts as of the date of determination.

<u>DirecTV</u> — collectively, DirecTV, Inc. and its Affiliates and Subsidiaries.

<u>DirecTV Account Reserve</u> — a reserve, in such amount as shall be determined from time to time by Agent in its sole discretion as sufficient to protect against the risk of offset by DirecTV with respect to any Accounts for which DirecTV is the Account Debtor.

<u>DirecTV Concentration Percentage</u> — (a) at any time that (i) DirecTV's corporate credit rating or senior debt rating (secured or unsecured) by Moody's is Ba2 and by S&P is BB, and (ii) Average Days Sales Outstanding on Accounts for which DirecTV is the Account Debtor is 50 or fewer days, 40%; (b) at any time that DirecTV's corporate credit rating or senior debt rating (secured or unsecured) by Moody's is Ba3 or by S&P is BB-, 25%; (c) at any time that DirecTV's corporate credit rating or senior debt rating (secured or unsecured) by Moody's is B1 or by S&P is B+, 15%; (d) at any time that Average Days Sales Outstanding on Accounts for which DirecTV is the Account Debtor is more than 50 days, 15%; and (e) irrespective of the Average Days Sales Outstanding on Accounts for which DirecTV is the Account Debtor, or DirecTV's rating, such lesser percentage as Agent may in its reasonable credit judgment determine from time to time.

DirectStar — DirectStar TV, LLC, a North Carolina limited liability company, and its Subsidiaries.

<u>DirectStar Purchase Agreement</u> — that certain Membership Interest Purchase Agreement dated February 6, 2007, by and among MasTec North America, MasTec, DirectStar, Red Ventures, LLC, a North Carolina limited liability company, and Ricardo Elias, Daniel S. Feldstein and Mark Brodsky, pursuant to which MasTec North America has agreed to purchase the remaining fifty-one percent (51%) of the issued and outstanding membership shares in DirectStar.

<u>Distribution</u> — in respect of any entity, (i) any payment of any dividends or other distributions on Equity Interests of the entity (except distributions in such Equity Interests) and (ii) any purchase, redemption or other acquisition or retirement for value of any Equity Interests of the entity or any Affiliate of the entity unless made contemporaneously from the net proceeds of the sale of Equity Interests.

Document — shall have the meaning given to the term "document" in the UCC.

-13-

Dollars and the sign § — lawful money of the United States of America.

<u>Domestic Subsidiary</u> — a Subsidiary of a Borrower (other than a Subsidiary that is a Borrower) that is incorporated under the laws of a state of the United States or the District of Columbia.

<u>Dominion Account</u> — a special account of Agent established by Borrowers at BofA or another bank selected by Borrowers, but acceptable to Agent in its sole discretion, and over which Agent shall have exclusive access and control for withdrawal purposes.

Electronic Chattel Paper — shall have the meaning given to the term "electronic chattel paper" in the UCC.

Eligible Account — an Account which arises in the Ordinary Course of Business of an Obligor from the sale of goods or rendition of services, is payable in Dollars (or, with respect to Accounts that arise from a sale to an Account Debtor located in Canada, in Canadian Dollars), is subject to Agent's duly perfected Lien, and is deemed by Agent, in its reasonable credit judgment, to be an Eligible Account. Without limiting the generality of the foregoing, no Account shall be an Eligible Account if: (i) it arises out of a sale made by an Obligor to a Subsidiary or an Affiliate of an Obligor, a Person controlled by an Affiliate of an Obligor or a Blocked Person; (ii) it is unpaid for more than 60 days after the original due date shown on the invoice; (iii) it is due or unpaid more than 90 days after the original invoice date (or as to any Approved Account Debtor, 120 days after original invoice date); (iv) 50% or more of the Accounts from the Account Debtor are not deemed Eligible Accounts hereunder; (v) the total unpaid Accounts of the Account Debtor or group of affiliated Account Debtors (excluding any affiliated Account Debtor that satisfies the criteria of clause (i) in the definition of Approved Account Debtor) exceed 7.5% (or (A) as to any Approved Account Debtor (other than DirecTV, Verizon or AT&T), 15%, (B) as to DirecTV, the DirecTV Concentration Percentage, (C) as to Verizon, the Verizon Concentration Percentage, and (D) as to AT&T, the AT&T Concentration Percentage) of the aggregate amount of all Eligible Accounts or exceed a credit limit established by Agent for such Account Debtor, in each case, to the extent of such excess; (vi) any covenant, representation or warranty contained in this Agreement with respect to such Account has been breached in any respect deemed material by Agent; (vii) the Account Debtor is also such Obligor's creditor or supplier, or the Account Debtor has disputed liability with respect to such Account, or the Account Debtor has made any claim with respect to any other Account due from such Account Debtor to such Obligor, or the Account otherwise is or may become subject to any right of setoff, counterclaim, recoupment, reserve, defense or chargeback, provided that, the Accounts of such Account Debtor shall be ineligible only to the extent of such dispute or right of offset, counterclaim, recoupment, reserve, defense or chargeback; (viii) an Insolvency Proceeding has been commenced by or against the Account Debtor (other than Accounts arising under and in accordance with the terms of post-petition contracts with such Account Debtor approved by the court in which the Insolvency Proceeding of the Account Debtor is being heard or otherwise permitted in such case) or the Account Debtor has failed, suspended or ceased doing business; (ix) the Account Debtor is not or has ceased to be Solvent; (x) it arises from a sale to an Account Debtor that is organized under the laws of any jurisdiction outside of the United States or Canada or that has its principal office or principal place of business outside the United States or Canada except to the extent that the sale is supported or secured by an Approved Credit Enhancement; (xi) it arises from a sale to the Account Debtor on a bill-and-hold, guaranteed sale, sale-or-return, sale-on-approval, consignment or any other repurchase or return basis; (xii) the Account Debtor is the United States of America or Canada any department, agency, Crown corporation or instrumentality thereof, unless the applicable Obligor is not prohibited from assigning the Account and does assign its

-14-

right to payment of such Account to Agent, in a manner satisfactory to Agent, so as to comply, in the case of the United States, with the Assignment of Claims Act of 1940 (31 U.S.C. §3727 and 41 U.S.C. §15), and, in the case of Canada, with the Financial Administration Act, R.S.C. c. F-11 or is a state, province, county or municipality, or a political subdivision or agency thereof and Applicable Law disallows or restricts an assignment of Accounts on which it is the Account Debtor; (xiii) the Account Debtor is located in any jurisdiction which imposes conditions on the right of a creditor to collect accounts receivable unless the applicable Obligor has either qualified to transact business in such jurisdiction as a foreign entity or filed a Notice of Business Activities Report or other required report with the appropriate officials in those jurisdictions for the then current year; (xiv) the Account Debtor is located in a jurisdiction in which such Obligor is deemed to be doing business under the laws of such jurisdiction and which denies creditors access to its courts in the absence of qualification to transact business in such jurisdiction or of the filing of any reports with such jurisdiction, unless such Obligor has qualified as a foreign entity authorized to transact business in such jurisdiction or has filed all required reports; (xv) the Account is subject to a Lien other than a Permitted Lien; (xvi) the goods giving rise to such Account have not been delivered to and accepted by the Account Debtor or the services giving rise to such Account have not been performed by such Obligor and accepted by the Account Debtor or the Account otherwise does not represent a final sale; (xvii) the Account is evidenced by Chattel Paper or an Instrument of any kind, or has been reduced to judgment; (xviii) the Account represents a billing in excess of cost or a retainage or arises from a sale on a cash-on-delivery basis; (xix) such Obligor has made any agreement with the Account Debtor for any deduction therefrom, except for discounts or allowances which are made in the Ordinary Course of Business for prompt payment and which discounts or allowances are reflected in the calculation of the face value of each invoice related to such Account; (xx) such Obligor has made an agreement with the Account Debtor to extend the time of payment thereof; (xxi) the Account arises from the performance of services under a contract in respect of which any performance bond or surety bond has been issued or under or related to any warranty obligation of an Obligor; (xxii) the Account represents, in whole or in part, a billing for interest, fees or late charges, provided that such Account shall be ineligible only to the extent of the amount of such billing; or (xxiii) it arises from a retail sale of Inventory to a Person who is purchasing the same primarily for personal, family or household purposes.

<u>Eligible Assignee</u> — a Person that is a Lender, a U.S. based Affiliate of a Lender or an Approved Fund (as defined below); a commercial bank, finance company, or other financial institution, in each case that is organized under the laws of the United States or any state, has total assets in excess of \$5 billion, extends asset-based lending facilities of the type contemplated herein in the Ordinary Course of Business and whose becoming an assignee would not constitute a prohibited transaction under Section 4975 of ERISA or any other Applicable Law, is acceptable to Agent and, unless a Default or an Event of Default exists, Borrowers (such approval by Borrowers, when required, not to be unreasonably withheld or delayed and to be deemed given by Borrowers if no objection is received by the assigning Lender and Agent from Borrowers within 2 Business Days after notice of such proposed assignment has been provided by the assigning Lender as set forth in **Section 14.3** of this Agreement); and, at any time that an Event of Default exists, any other Person acceptable to Agent in its sole discretion. The term "<u>Approved Fund</u>" means any Person (other than a natural person) that is engaged in making, purchasing, holding or investing in bank loans and similar extensions of credit in the Ordinary Course of Business of such Person and that is administered or managed by (i) a Lender, (ii) an Affiliate of a Lender or (iii) an entity or an Affiliate of an entity that administers or manages a Lender.

<u>Eligible Cash Collateral</u> — cash and Cash Equivalents of MasTec that (i) are on deposit in the LC Collateral Account, (ii) are subject to the perfected, first priority security interest and Lien

-15-

in favor of Agent, on behalf of the Secured Parties, upon such cash and Cash Equivalents, and BofA (or the other applicable depositary institution or intermediary at which the LC Collateral Account is maintained) has executed a deposit account control agreement in favor of Agent, for the benefit of the Secured Parties, or such other documentation as may be required by Agent with respect to the LC Collateral Account and the cash and Cash Equivalent maintained therein, in each case in form and substance satisfactory to Agent in all respects, and (iii) are not subject to any Lien, claim or other interest in favor of any Person other than Agent.

<u>Eligible Cash Collateral Amount</u> — on any date of determination, an amount equal to 100% of the amount of Eligible Cash Collateral on such date; <u>provided</u>, <u>that</u>, it is the intent of the parties that (i) the amount of eligible Cash Collateral maintained in the LC Collateral Account shall be equal to the undrawn amount of the Letters of Credit that are Cash Collateralized on such date, (ii) Borrowers shall not be required on any date to maintain Eligible Cash Collateral in an amount greater than the undrawn amount of the Letters of Credit that are Cash Collateral Account exceeds the undrawn amount of Letters of Credit that are Cash Collateralized on such date, BofA or its Affiliate at which the LC Collateral Account is maintained shall, at the request of Borrowers, pay the amount of Eligible Cash Collateral in excess of undrawn amount of the Letters of Credit that are Cash Collateral 2.3.3.

<u>Eligible Fixed Assets</u> — Equipment owned by an Obligor, subject to the Lien in favor of Agent and to no other Lien, that is located within the continental United States and has been included in the most recent Net Orderly Liquidation Value Appraisal accepted by Agent.

<u>Eligible Real Estate</u> — each parcel of Real Estate which Agent, in its reasonable credit judgment, determines to be Eligible Real Estate. Without limiting the discretion of Agent to establish other criteria of eligibility, Eligible Real Estate (i) shall be owned by an Obligor, (ii) shall be subject to the Lien in favor of Agent and to no other Lien (other than those Liens or encumbrances, if any, which are expressly permitted in the Mortgages applicable to such Real Estate), (iii) shall be located within the continental United States, (iv) shall be in compliance with all of the representations, warranties, covenants and agreements set forth in the Mortgage(s) applicable thereto and in **Sections 7.3.1, 9.1.30** and **10.1.17** hereof, and (v) shall have been included in a fair market value appraisal of the Real Estate, and shall be covered by an environmental report, in each case which have been accepted by and are satisfactory to Agent.

<u>Eligible Unbilled Accounts</u> — an amount which, when an invoice is issued with respect thereto, will be an Eligible Account, and in respect of which an invoice is issued within 30 days (or 45 days with respect to an Account Debtor principally engaged in the power distribution and transmission business or communications business) after such amount is first included as an eligible unbilled account on any Borrowing Base Certificate.

<u>Eligible Unbilled Accounts Formula Amount</u> — on any date of determination thereof, an amount equal to the lesser of (i) 70% (or such lesser percentage as Agent may in its reasonable credit judgment determine from time to time) of the aggregate amount of Eligible Unbilled Accounts on such date, or (ii) an amount equal to the product of (x) the sum of the Accounts Formula Amount on such date, the amount derived pursuant to subsection (i) of this definition, the Fixed Assets Formula Amount, and the Real Estate Formula Amount multiplied by (y) 15%.

<u>Enforcement Action</u> — action taken or to be taken by Agent, during any period that an Event of Default exists, to enforce collection of the Obligations or to realize upon the Collateral

-16-

(whether by judicial action, under power of sale, by self-help repossession, by notification to Account Debtors, or by exercise of rights of setoff or recoupment).

<u>Environmental Agreement</u> — the Agreement Regarding Environmental Matters to be executed by Obligors in favor of Agent on or about the Closing Date and by which each Obligor shall, among other things, indemnify Agent and Lenders from liability for such Obligor's failure to comply with any Environmental Laws.

<u>Environmental Laws</u> — all federal, state, provincial local and foreign laws, rules, regulations, codes, ordinances, orders and consent decrees (together with all programs, permits and guidance documents promulgated by regulatory agencies, to the extent having the force of law), now or hereafter in effect, that relate to public health (but excluding occupational safety and health, to the extent regulated by OSHA) or the protection or pollution of the environment, whether new or hereafter in effect, including CERCLA, RCRA and CWA.

<u>Environmental Notice</u> — a notice (whether written or oral) from any Governmental Authority or any other Person of a possible or alleged noncompliance with or liability under any Environmental Laws, including any complaints, citations, demands or requests from any Governmental Authority or any other Person for correction or remediation of any asserted violation of any Environmental Laws or any investigations concerning any asserted violation of any Environmental Laws.

Environmental Release — a release as defined in CERCLA or under any other applicable Environmental Laws.

<u>Equipment</u> — shall have the meaning given to the term "equipment" in the UCC or the PPSA, as applicable, and shall include all of each Obligor's machinery, apparatus, equipment, fittings, furniture, fixtures, motor vehicles and other tangible personal Property (other than Inventory) of every kind and description, whether now owned or hereafter acquired by such Obligor and wherever located, and all parts, accessories and special tools therefor, all accessions thereto, and all substitutions and replacements thereof.

<u>Equity Interest</u> — the interest of (i) a shareholder in a corporation, (ii) a partner (whether general or limited) in a partnership (whether general, limited or limited liability), (iii) a member in a limited liability company, or (iv) any other Person having any other form of equity security or ownership interest.

ERISA — the Employee Retirement Income Security Act of 1974 and all rules and regulations from time to time promulgated thereunder.

Event of Default — as defined in Section 12 of this Agreement.

<u>Executive Order No. 13224</u> — Executive Order No. 13224 on Terrorist Financing, effective September 24, 2001, as the same has been, or shall hereafter be, renewed, extended, amended or replaced.

Existing Lenders — the Lenders under (and as defined in) the Existing Loan Agreement as in effect on the Closing Date.

Existing Letters of Credit — as defined in Section 2.3.1 of this Agreement.

-17-

Extraordinary Expenses — all costs, expenses, fees (including fees incurred to Agent Professionals) or advances that Agent may suffer or incur, whether prior to or after the occurrence of an Event of Default, and whether prior to, after or during the pendency of an Insolvency Proceeding of an Obligor, on account of or in connection with (i) the audit, inspection, repossession, storage, repair, appraisal, insuring, completion of the manufacture of, preparing for sale, advertising for sale, selling, collecting or otherwise preserving or realizing upon any Collateral; (ii) the defense of Agent's Lien upon any Collateral or the priority thereof or any adverse claim with respect to the Loans, the Loan Documents or the Collateral asserted by any Obligor, any receiver or trustee for any Obligor or any creditor or representative of creditors of any Obligor; (iii) the settlement or satisfaction of any Liens upon any Collateral (whether or not such Liens are Permitted Liens); (iv) the collection or enforcement of any of the Obligations, whether by Enforcement Action or otherwise; (v) the negotiation, documentation, and closing of any restructuring or forbearance agreement with respect to the Loan Documents or Obligations; (vi) amounts advanced by Agent pursuant to Sections 8.1.3 or 15.10 of this Agreement; (vii) the enforcement of any of the provisions of any of the Loan Documents; or (viii) any payment under a guaranty, indemnity or other payment agreement provided by Agent, which is reimbursable to Agent by Borrowers pursuant to Section 3.4.2 of this Agreement. Such costs, expenses and advances may include transfer fees, taxes, storage fees, insurance costs, permit fees, utility reservation and standby fees, legal fees, appraisal fees, brokers' fees and commissions, auctioneers' fees and commissions, accountants' fees, environmental study fees, wages and salaries paid to employees of any Obligor or independent contractors in liquidating any Collateral, travel expenses, all other fees and expenses payable or reimbursable by Borrowers or any other Obligor under any of the Loan Documents, and all other fees and expenses associated with the enforcement of rights or remedies under any of the Loan Documents, but excluding compensation paid to employees (including inside legal counsel who are employees) of Agent.

<u>Federal Funds Rate</u> — for any period, a fluctuating interest rate per annum equal for each date during such period to the weighted average of the rates on overnight federal funds transactions with members of the Federal Reserve System arranged by federal funds brokers, as published for such day (or, if such day is not a Business Day, for the next preceding Business Day) in Atlanta, Georgia by the Federal Reserve Bank of Atlanta, or if such rate is not so published for any day which is a Business Day, the average of the quotations for such day on such transactions received by Agent from 3 federal funds brokers of recognized standing selected by Agent.

Fee Letter — the fee letter agreement between Agent and Borrowers.

FEIN — with respect to any Person, the Federal Employer Identification Number of such Person.

Fiscal Quarter — each of the 4 consecutive 3-month periods beginning on the first day of a Fiscal Year.

<u>Fiscal Year</u> — the fiscal year of Borrowers and their Subsidiaries for accounting and tax purposes, which ends on December 31 of each year and when preceded by the designation of a calendar year (e.g., 2008 Fiscal Year) means the fiscal year of Borrowers and their Subsidiaries ending on December 31 of such designated calendar year.

<u>Fixed Assets Formula Amount</u> — on any date of determination thereof, the lesser of (i) \$100,000,000 or (ii) an amount equal to eighty percent (80%) of the Net Orderly Liquidation Value of Eligible Fixed Assets, <u>provided however</u>, that the amount calculated under this clause

-18-

(ii) shall be adjusted (either upward or downward, as applicable) by an amount equal to the lesser of (A) \$10,000,000 (in the case of an upward adjustment) and (B) the result of (I) eighty percent (80%) of the aggregate cost of all Equipment that has been acquired by Obligors since the date of the most recent Net Orderly Liquidation Value Appraisal, <u>minus</u> (II) an amount, as determined by Agent, equal to the aggregate amount of the fair market value or book value, whichever is more, of all Equipment that has been disposed of by Obligors (other than in accordance with **Section 8.4.2(ii)**) since the date of the most recent Net Orderly Liquidation Value Appraisal of the Equipment.

<u>Fixed Charge Coverage Ratio</u> — for any period, the ratio of (a) Adjusted EBITDA for such period <u>minus</u> Net Capital Expenditures (excluding Capital Expenditures financed by Permitted Purchase Money Debt) for such period, <u>minus</u> Distributions made during such period, <u>minus</u> cash Taxes paid during such period, <u>minus</u> cash Investments made by Borrowers in DirectStar, <u>minus</u> any portion of the Earn-Out Payments under (and as defined in) the DirectStar Purchase Agreement made by Borrowers in cash, <u>minus</u> all other Acquisition Earn-Out Payments, to (b) the sum of all Fixed Charges for such period, all calculated for Borrowers and their Subsidiaries (other than DirectStar) on a Consolidated basis.

<u>Fixed Charges</u> — for any fiscal period, the sum of (i) interest expense (other than interest payable-in-kind to the extent not paid in cash and interest expense arising from this Agreement or the other Loan Documents that is deferred into future periods in accordance with GAAP) for such period <u>plus</u> (ii) current maturities of Funded Debt (including Capitalized Lease Obligations) as of the last day of such period.

<u>FLSA</u> — the Fair Labor Standards Act of 1938.

<u>Foreign Lender</u> — any Lender that is organized under the laws of a jurisdiction other than the laws of the United States, any state thereof or the District of Columbia.

<u>Foreign Subsidiary</u> — a Subsidiary that is not a Domestic Subsidiary.

<u>Full Payment</u> — with respect to any of the Obligations, the full, final and indefeasible payment in full, in cash and in Dollars, of all of such Obligations, including all interests, fees and other charges payable in connection therewith under any of the Loan Documents, whether such interests, fees or other charges accrue or are incurred prior to or during the pendency of an Insolvency Proceeding and whether or not any of the same are allowed or recoverable in any Bankruptcy Case pursuant to Section 506 of the Bankruptcy Code or otherwise; with respect to any LC Obligations represented by undrawn Letters of Credit and Banking Relationship Debt, the depositing of cash with Agent, as security for the payment of such Obligations, not to exceed 105% of the aggregate undrawn amount of such Letters of Credit and 100% of Agent's good faith estimate of the amount of Banking Relationship Debt due and to become due; and with respect to any Obligations that are contingent in nature (other than Obligations consisting of LC Obligations or Banking Relationship Debt), such as a right of Agent or a Lender to indemnification by any Obligor, the depositing of cash with Agent in an amount equal to 100% of such Obligations or, if such Obligations are unliquidated in amount and represent a claim which has been overtly asserted (or is reasonably probable of assertion) against Agent or a Lender and for which an indemnity has been provided by Obligors in any of the Loan Documents, in an amount that is equal to such claim or Agent's good faith estimate of such claim. None of the Loans shall be deemed to have been paid in full until all Commitments related to such Loans have expired or been terminated.

-19-

<u>Funded Debt</u> — with respect to Borrowers and their Subsidiaries, the sum, without duplication, of (i) the aggregate amount of Debt of Borrowers and their Subsidiaries relating to (a) the borrowing of money or the obtaining of credit (other than trade payables incurred in the Ordinary Course of Business), including the Obligations and any other notes or bonds, (b) the deferred purchase price of assets (other than trade payables incurred in the Ordinary Course of Business), and (c) any Capitalized Lease Obligations, <u>plus</u> (ii) Debt of the type referred to in clause (i) of another Person guaranteed by a Borrower or Subsidiary, in each case as determined on a Consolidated basis.

GAAP — generally accepted accounting principles in the United States of America in effect from time to time.

GECC — General Electric Capital Corporation, a Delaware corporation, and its successors and assigns.

<u>General Intangibles</u> — shall have the meaning given to the term "general intangibles" in the UCC and shall include each Obligor's choses in action, causes of action, company or other business records, inventions, blueprints, designs, patents, patent applications, trademarks, trademark applications, trade names, trade secrets, service marks, goodwill, brand names, copyrights, registrations, licenses, franchises, customer lists, permits, tax refund claims, computer programs, operational manuals, internet addresses and domain names, insurance refunds and premium rebates, all claims under guaranties, security interests or other security held by or granted to such Obligor to secure payment of any of any of such Obligor's Accounts by an Account Debtor, all rights to indemnification and all other intangible property of such Obligor of every kind and nature (other than Accounts).

<u>Goods</u> — shall have the meaning given to the term "goods" in the UCC or the PPSA.

<u>Governmental Approvals</u> — all authorizations, consents, approvals, licenses and exemptions of, registrations and filings with, and reports to, all Governmental Authorities.

<u>Governmental Authority</u> — any federal, state, provincial, municipal, national, foreign or other governmental department, commission, board, bureau, court, agency or instrumentality or political subdivision thereof or any entity or officer exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to any government or any court, in each case whether associated with a state of the United States, the District of Columbia or a foreign entity or government.

<u>Guarantor</u> — each Canadian Obligor and each other Person who guarantees payment or performance of the whole or any part of the Obligations.

<u>Guaranty</u> — each guaranty agreement now or hereafter executed by a Guarantor in favor of Agent with respect to any of the Obligations, including each Canadian Obligor Guarantee.

<u>Hedging Agreement</u> — any interest rate protection agreement, foreign currency exchange agreement, forward contract, currency swap agreement, commodity price protection agreement or other interest or currency exchange rate or commodity price hedging arrangement.

<u>Impermissible Qualification</u> — any qualification or exception to the opinion or certification of any independent public accountant as to any financial statement of Borrowers which (i) is of a "going concern" or similar nature, (ii) relates to the limited scope of examination

-20-

of matters relevant to such financial statements, or (iii) relates to the treatment or classification of any item in such financial statement in which, a condition to its removal, would require an adjustment to such item the effect of which would be to cause the occurrence of an Event of Default.

Increase Effective Date — as defined in Section 2.1.6(ii) of this Agreement.

Indemnified Amount — in the case of Agent Indemnitees, the amount of any loss, cost, expenses or damages suffered or incurred by Agent Indemnitees and against which Lenders or any Obligor have agreed to indemnify Agent Indemnitees pursuant to the terms of this Agreement or any of the other Loan Documents; in the case of Lender Indemnitees, the amount of any loss, cost, expenses or damages suffered or incurred by Lender Indemnitees and against which Lenders or any Obligor have agreed to indemnify Lender Indemnitees pursuant to the terms of this Agreement or any of the other Loan Documents; and, in the case of BofA Indemnitees, the amount of any loss, cost, expenses or damages suffered or incurred by BofA Indemnitees and against which Lenders or any Obligor have agreed to indemnify BofA Indemnitees pursuant to the terms of this Agreement or any of the other Loan Documents; or any Obligor have agreed to indemnify BofA Indemnitees pursuant to the terms of this Agreement or any of the other Loan Documents.

<u>Indemnified Claim</u> — any and all claims, demands, liabilities, obligations, losses, damages, penalties, actions, judgments, suits, awards, remedial response costs, expenses or disbursements of any kind or nature whatsoever (including reasonable attorneys', accountants', consultants' or paralegals' fees and expenses), whether arising under or in connection with the Loan Documents, any Applicable Law (including any Environmental Laws) or otherwise, that may now or hereafter be suffered or incurred by any Indemnitee (whether suffered or incurred in or as a result of any investigation, litigation, arbitration or other judicial or non-judicial proceeding or any appeals related thereto) and that is covered by an indemnity of an Obligor under any of the Loan Documents.

Indemnitees — the Agent Indemnitees, the Lender Indemnitees and the BofA Indemnitees.

Indenture — the Indenture, among MasTec, its Subsidiaries and the trustee named thereunder, as Trustee, governing the Senior Notes.

Initial Lenders — BofA, GECC and PNC Bank, National Association, each in its capacity as a "Lender" under this Agreement on the Closing Date.

<u>Insolvency Proceeding</u> — any action, case or proceeding commenced by or against a Person under any state, provincial, federal or foreign law, or any agreement of such Person, for (i) the entry of an order for relief under any chapter or section of the Bankruptcy Code, the Bankruptcy and Insolvency Act of Canada, the Companies' Creditors Arrangement Act of Canada, or under any other bankruptcy , insolvency or debt adjustment (whether state, provincial, federal or foreign), (ii) the appointment of a receiver (or administrative receiver), interim receiver, assignee, trustee, liquidator administrator, conservator sequestrator, monitor or other custodian for such Person or any part of its Property, (iii) an assignment or trust mortgage for the benefit of creditors of such Person, or (iv) the liquidation, dissolution or winding up of the affairs of such Person.

<u>Instrument</u> — shall have the meaning given to the term "instrument" in the UCC.

<u>Insurance Cash Collateral Account</u> — each demand deposit account in existence on the date of this Agreement established by and held in the name of an insurance company or insurance

-21-

service company at a depository bank, as described on **Schedule 1.1B** attached hereto, to establish loss reserves with respect to policies insuring MasTec issued by such insurance company or through such insurance service company.

<u>Intellectual Property</u> — all intellectual and similar Property of a Person of every kind and description, including inventions, designs, patents, patent applications, copyrights, trademarks, service marks, trade names, mask works, trade secrets, confidential or proprietary information, know-how, software and databases and all embodiments or fixations thereof and related documentation, registrations and franchises, all books and records describing or used in connection with the foregoing and all licenses, or other rights to use any of the foregoing.

<u>Intellectual Property Claim</u> — the assertion by any Person of a claim (whether asserted in writing, by action, suit or proceeding or otherwise) that any Obligor's ownership, use, marketing, sale or distribution of any Inventory, Equipment, Intellectual Property or other Property is violative of any ownership or right to use any Intellectual Property of such Person.

Interest Period — shall have the meaning ascribed to it in **Section 3.1.3** of this Agreement.

<u>Inventory</u> — shall have the meaning given to the term "inventory" in the UCC or the PPSA, as applicable, and shall include all goods intended for sale or lease by an Obligor, or for display or demonstration; all work in process, all raw materials and other materials and supplies of every nature and description used or which might be used in connection with the manufacture, printing, packing, shipping, advertising, selling, leasing or furnishing such goods or otherwise used or consumed in such Obligor's business (but excluding Equipment).

<u>Investment Property</u> — shall have the meaning given to "investment property" in the UCC and shall include all Securities (whether certificated or uncertificated), security entitlements, securities accounts, commodity contracts and commodity accounts.

<u>Issuing Bank</u> — BofA or an Affiliate of BofA, including Fleet National Bank, as issuer of the Letters of Credit, and, with respect to the Existing Letters of Credit, Fleet National Bank.

Issuing Bank Indemnitees — Issuing Bank and all of its present and future officers, employees, directors and agents.

<u>Joinder Agreement</u> — a Joinder Agreement to be executed and delivered by each new Subsidiary that pursuant to **Section 10.1.11** of this Agreement becomes a Borrower hereunder, in accordance with, and in the form of **Exhibit J** annexed to this Agreement.

<u>LC Application</u> — an application by any or all Borrowers to Issuing Bank, pursuant to a form approved by Issuing Bank, for the issuance of a Letter of Credit, that is submitted to Issuing Bank at least 5 Business Days prior to the requested issuance of such Letter of Credit.

<u>LC Collateral Account</u> — a Deposit Account, securities account or other account established by MasTec at BofA or an Affiliate of BofA and over which Agent has control within the meaning of the UCC; <u>provided</u>, <u>that</u>, any interest, interest equivalent or other income that accrues, from time to time, on the cash or financial assets maintained or deposited in such account shall be the property of Borrowers and paid by BofA or its Affiliate to Borrowers from time to time.

-22-

LC Conditions — the following conditions, the satisfaction of each of which is required before Issuing Bank shall be obligated issue a Letter of Credit: (i) each of the conditions set forth in **Section 11** of this Agreement has been and continues to be satisfied, including the absence of any Default or Event of Default; (ii) after giving effect to the issuance of the requested Letter of Credit and all other unissued Letters of Credit for which an LC Application has been signed by a Borrower and approved by Agent and Issuing Bank, no Out-of-Formula Condition would exist, and, if no Revolver Loans are outstanding, the LC Obligations do not, and would not upon the issuance of the requested Letter of Credit, exceed the Borrowing Base; (iii) such Letter of Credit has an expiration date that is no more than 365 days from the date of issuance and such expiration date is at least 30 days prior to the last Business Day of the Term unless otherwise agreed by Agent in its discretion; (iv) the currency in which payment is to be made under the Letter of Credit is Dollars; and (v) the form of the proposed Letter of Credit is satisfactory to Agent and Issuing Bank in their discretion, provides for sight drafts only and does not contain any language that automatically increases the amount available to be drawn under the Letter of Credit.

<u>LC Documents</u> — any and all agreements, instruments and documents (including an LC Application) required by Issuing Bank to be executed by Borrowers or any other Person and delivered to Issuing Bank for the issuance, amendment or renewal of a Letter of Credit.

LC Facility — the subfacility for Letters of Credit established as part of the Revolver Commitments pursuant to Section 2.3 of this Agreement.

<u>LC Obligations</u> — on any date, an amount (in Dollars) equal to the sum of (without duplication) (i) all amounts then due and payable by any Obligor on such date by reason of any payment that is made by Issuing Bank under a Letter of Credit and that has not been repaid to Issuing Bank, <u>plus</u> (ii) the aggregate undrawn amount of all Letters of Credit which are then outstanding or for which an LC Application has been delivered to and accepted by Issuing Bank and approved by Agent, <u>plus</u> (iii) all fees and other amounts due or to become due in respect of Letters of Credit outstanding on such date.

LC Request — a Letter of Credit Request from Borrowers to Issuing Bank in the form of Exhibit I annexed hereto.

<u>LC Reserve</u> — at any date, the aggregate of all LC Obligations on such date, other than (i) LC Obligations that Borrowers shall Cash Collateralize on or prior to such date and (ii) during any period that no Default or Event of Default exists, the portion of LC Obligations described in clause (iii) of the definition thereof.

Lender Indemnitees — Lenders and all of their respective present and future officers, directors, employees, agents and attorneys.

<u>Lenders</u> — shall have the meaning given to it in the preamble to this Agreement and shall include BofA (whether in its capacity as a provider of Loans under **Section 2** of this Agreement or as the provider of Swingline Loans under **Section 4.1.3** of this Agreement) and any other Person who may from time to time become a "Lender" under this Agreement.

Letter of Credit — any standby or documentary letter of credit issued by Issuing Bank for the account of any Borrower.

Letter-of-Credit Right — shall have the meaning given to the term "letter-of-credit-right" in the UCC.

<u>Leverage Ratio</u> — with respect to any fiscal period, the ratio of (i) the amount of total Funded Debt (including Senior Funded Debt, Capitalized Lease Obligations, LC Obligations (other than those LC Obligations that are Cash Collateralized) and Subordinated Debt, if any) as of the last day of such fiscal period <u>minus</u> the amount of domestic unrestricted balance sheet cash as of the last day of such fiscal period, to (ii) Adjusted EBITDA for such fiscal period, all calculated for Borrowers and their Subsidiaries on a Consolidated basis.

<u>LIBOR Lending Office</u> — with respect to a Lender, the office designated as a LIBOR Lending Office for such Lender on the signature page hereof (or on any Assignment and Acceptance, in the case of an assignee) and such other office of such Lender or any of its Affiliates that is hereafter designated by written notice to Agent.

<u>LIBOR Loan</u> — a Loan, or portion thereof, during any period in which it bears interest at a rate based upon the applicable Adjusted LIBOR Rate.

<u>License Agreement</u> — any agreement between an Obligor and a Licensor pursuant to which such Obligor is authorized to use any Intellectual Property in connection with the manufacturing, marketing, sale or other distribution of any Inventory of such Obligor.

<u>Licensor</u> — any Person from whom an Obligor obtains the right to use (whether on an exclusive or non-exclusive basis) any Intellectual Property in connection with such Obligor's manufacture, marketing, sale or other distribution of any Inventory.

Lien — any interest in Property securing an obligation owed to, or a claim by, a Person other than the owner of the Property, whether such interest is based on common law, statute or contract. The term "Lien" shall also include reservations, exceptions, encroachments, easements, rights-of-way, covenants, conditions, restrictions, leases and other title exceptions and encumbrances affecting Property and, additionally with respect to any Canadian Obligor, any other lien, charge, privilege, secured claim, hypothec, prior claim, title retention right, garnishment right, deemed trust, encumbrance or other right affecting any of the property of such Canadian Obligor, choate or inchoate, arising by any statute, act of law of any jurisdiction at common law or in equity or by agreement . For the purpose of this Agreement, an Obligor shall be deemed to be the owner of any Property which it has acquired or holds subject to a conditional sale agreement or other arrangement pursuant to which title to the Property has been retained by or vested in some other Person for security purposes.

Lien Waiver — an agreement duly executed in favor of Agent, in form and content acceptable to Agent, by which (i) for locations leased by an Obligor, an owner or mortgagee of premises upon which any Property of an Obligor is located agrees to waive or subordinate any Lien it may have with respect to such Property in favor of Agent's Lien therein and to permit Agent to enter upon such premises and remove such Property or to use such premises to store or dispose of such Property, or (ii) for locations at which any Obligor places Inventory with a warehouseman or a processor, such warehouseman or processor agrees to waive or subordinate any Lien it may have with respect to such Property in favor of Agent's Lien therein and to permit Agent to enter upon such premises and remove such premise to enter upon such premises and remove such Property or to use such premises to store or dispose of such Property.

Loan — a Revolver Loan (and each Base Rate Loan and LIBOR Loan comprising such Loan).

Loan Account — the loan account established by each Lender on its books pursuant to Section 5.8 of this Agreement.

Loan Documents — this Agreement, the Other Agreements and the Security Documents.

<u>Loan Year</u> — a period commencing each calendar year on the same month and day as the date of this Agreement and ending on the same month and day in the immediately succeeding calendar year, with the first such period (<u>i.e.</u> the first Loan Year) to commence on the date of this Agreement.

Lockbox — each U. S. Post Office Box specified in a Lockbox Agreement.

<u>Lockbox Agreement</u> — each agreement between a Borrower and a Clearing Bank concerning the establishment of a Lockbox for the collection of Accounts.

Margin Stock — shall have the meaning ascribed to it in Regulation U and of the Board of Governors.

<u>Material Adverse Effect</u> — the effect of any event, condition, action, omission or circumstance, which, alone or when taken together with other events, conditions, actions, omissions or circumstances occurring or existing concurrently therewith, (i) has a material adverse effect upon the business, operations, Properties, prospects or condition (financial or otherwise) of Obligors, taken as a whole; (ii) has or could be reasonably expected to have any material adverse effect whatsoever upon the validity or enforceability of this Agreement or any of the other Loan Documents; (iii) has any adverse effect upon the value of the whole or any material part of the Collateral, the Liens of Agent with respect to the Collateral or the priority of any such Liens; (iv) impairs the ability of any Obligor to perform its obligations under this Agreement or any of the other Loan Documents, including repayment of any of the Obligations when due; or (v) impairs the ability of Agent or any Lender to enforce or collect the Obligations or realize upon any of the Collateral in accordance with the Loan Documents and Applicable Law.

<u>Material Contract</u> — an agreement to which an Obligor is a party (other than the Loan Documents) (i) which is deemed to be a material contract as provided in Regulation S-K promulgated by the SEC under the Securities Act of 1933 or (ii) for which breach, termination, cancellation, nonperformance or failure to renew could reasonably be expected to have a Material Adverse Effect.

<u>Maximum Rate</u> — the maximum non-usurious rate of interest permitted by Applicable Law that at any time, or from time to time, may be contracted for, taken, reserved, charged or received on the Debt in question or, to the extent that at any time Applicable Law may thereafter permit a higher maximum non-usurious rate of interest, then such higher rate. Notwithstanding any other provision hereof, the Maximum Rate shall be calculated on a daily basis (computed on the actual number of days elapsed over a year of 365 or 366 days, as the case may be).

<u>Money Borrowed</u> — as applied to any Obligor, without duplication, (i) Debt arising from the lending of money by any other Person to such Obligor; (ii) Debt, whether or not in any such case arising from the lending of money by another Person to such Obligor, (A) which is represented by notes payable or drafts accepted that evidence extensions of credit, (B) which constitutes obligations evidenced by bonds, debentures, notes or similar instruments, or (C) upon which interest charges are customarily paid (other than accounts payable) or that was issued or

-25-

assumed as full or partial payment for Property; (iii) Debt that constitutes a Capitalized Lease Obligation; (iv) reimbursement obligations with respect to letters of credit or guaranties of letters of credit; and (v) Debt of such Obligor under any guaranty of obligations that would constitute Debt for Money Borrowed under clauses (i) through (iii) hereof, if owed directly by such Obligor.

Moody's — Moody's Investors Services, Inc.

<u>Mortgages</u> — the mortgages, deeds of trust and/or deeds to secure debt to be executed by a Borrower in favor of Agent and pursuant to which such Borrower shall grant and convey to Agent, for the benefit of Secured Parties, Liens upon the Real Estate of such Borrower as security for the payment of the Obligations.

<u>Multiemployer Plan</u> — has the meaning set forth in Section 4001(a)(3) of ERISA.

<u>Net Capital Expenditures</u> — for any period, the sum of all Capital Expenditures made during such period <u>less</u> the amount of net cash proceeds received by Borrowers or their Subsidiaries from sales of Equipment made during such period in accordance with the terms of this Agreement, all calculated for Borrowers and their Subsidiaries on a Consolidated basis.

<u>Net Disposition Proceeds</u> — with respect to a Permitted Asset Disposition, proceeds (including cash receivable (when received) by way of deferred payment) received by an Obligor in cash from such Asset Disposition, net of: (i) the reasonable and customary costs and expenses actually incurred in connection with such Asset Disposition (including legal fees and sales commissions); (ii) amounts applied to repayment of Debt (other than the Obligations) secured by a Permitted Lien on such Collateral disposed of that is senior in priority to Agent's Liens; (iii) transfer or similar taxes; and (iv) reserves for escrows and indemnities, until such reserves are no longer required and such reserves or escrows are released to an Obligor.

<u>Net Orderly Liquidation Value</u> — on any date of determination and with respect to any Eligible Fixed Assets, the value of such Eligible Fixed Assets expected to be realized at an orderly, negotiated sale of such Eligible Fixed Assets that is held within a reasonable period of time, as such value is determined by Agent from the most recent Net Orderly Liquidation Value Appraisal received by Agent on or before such date.

<u>Net Orderly Liquidation Value Appraisal</u> — an appraisal of the orderly liquidation value of Equipment performed by an appraiser satisfactory to Agent, which may be a desktop appraisal performed by an employee or agent of Agent, which appraisal shall include, as a factor in the determination of orderly liquidation value, all costs and expenses projected to be incurred in the conduct of any liquidation of all or any portion of the Equipment.

Notes — each Revolver Note and any other promissory note executed by Borrowers at Agent's request to evidence any of the Obligations.

<u>Notice of Borrowing</u> — as defined in **Section 4.1.1(i)** of this Agreement.

Notice of Conversion/Continuation — as defined in Section 3.1.2(ii) of this Agreement.

<u>Obligations</u> — in each case, whether now in existence or hereafter arising, (i) the principal of, and interest and premium, if any, on, the Loans; (ii) all LC Obligations of any Obligor to Agent or Issuing Bank arising in connection with the issuance of any Letter of Credit; (iii) all Indemnified Amounts; (iv) all Extraordinary Expenses; and (v) all other Debts, covenants, duties

-26-

and obligations (including Contingent Obligations) now or at any time or times hereafter owing by any Obligor to Agent or any Lender under or pursuant to this Agreement or any of the other Loan Documents, or owing by any Obligor to Agent or any Lender (or any Affiliate of any Lender) with respect to Banking Relationship Debt, in each case, whether evidenced by any note or other writing, whether arising from any extension of credit, opening of a letter of credit, acceptance, loan, guaranty, indemnification or otherwise and whether direct or indirect, absolute or contingent, due or to become due, primary or secondary, or joint or several, including all interest, charges, expenses, fees or other sums chargeable to any or all Obligors under this Agreement or under any of the other Loan Documents.

<u>Obligor</u> — each Borrower and each Guarantor, and any other Person that is at any time liable for the payment of the whole or any part of the Obligations or that has granted in favor of Agent a Lien upon any of such Person's assets to secure payment of any of the Obligations.

<u>Ordinary Course of Business</u> — with respect to any transaction involving any Person, the ordinary course of such Person's business, as conducted by such Person in accordance with past practices and undertaken by such Person in good faith and not for the purpose of evading any covenant or restriction in any Loan Document.

<u>Organic Documents</u> — with respect to any Person, its charter, certificate or articles of incorporation, bylaws, articles of organization, limited liability agreement, operating agreement, members agreement, shareholders agreement, partnership agreement, certificate of partnership, certificate of formation, voting trust, or similar agreement or instrument governing the formation or operation of such Person.

<u>OSHA</u> — the Occupational Safety and Hazard Act of 1970.

<u>Other Agreements</u> — the Notes, the Fee Letter, each Lien Waiver, each Letter of Credit and any and all other agreements, instruments and documents (other than this Agreement and the Security Documents), heretofore, now or hereafter executed by any Obligor, any other Obligor or any other Person and delivered to Agent or any Lender, or otherwise executed by Agent or any Lender in favor of any Person on behalf or for the account of an Obligor, in each case in respect of the transactions contemplated by this Agreement or other Loan Documents.

Out-of-Formula Condition — as defined in Section 2.1.2 of this Agreement.

<u>Out-of-Formula Loan</u> — a Revolver Loan made or existing when an Out-of-Formula Condition exists or the amount of any Revolver Loan which, when funded, results in an Out-of-Formula Condition.

<u>Participant</u> — as defined in **Section 14.2.1**.

<u>Participating Lender</u> — as defined in **Section 2.3.2(i)**.

<u>Payment Account</u> — an account maintained by Agent to which all monies from time to time deposited to a Dominion Account shall be transferred and all other payments shall be sent in immediately available federal funds.

Payment Intangible — shall have the meaning given to the term "payment intangible" in the UCC.

<u>Payment Items</u> — all checks, drafts, or other items of payment payable to an Obligor, including those evidencing or constituting proceeds of any of the Collateral.

<u>PBA</u> — the Pension Benefits Act of Ontario or any other Canadian federal or provincial statute in relation to Plans and all regulations thereunder as amended from time to time, and any successor legislation.

<u>Pending Revolver Loans</u> — at any date, the aggregate principal amount of all Revolver Loans which have been requested in any Notice of Borrowing received by Agent but which have not theretofore been advanced by Agent or Lenders.

Permitted Acquisition — an Acquisition by an Obligor or any Subsidiary of an Obligor in which each of the following conditions is satisfied:

(a) no Default or Event of Default exists before or would exist immediately after giving effect thereto;

(b) the Acquisition is of (i) Equity Interests of any other Person organized under the laws of the United States of America or any state thereof or of Canada or any province thereof sufficient to give such Obligor or Subsidiary control of such other Person or (ii) all or substantially all of the assets of a Business Unit located in the United States or Canada, and such Person or Business Unit is engaged in a business that is substantially similar, related or incidental to the business conducted by Obligors;

(c) the Purchase Price of such Acquisition does not exceed \$100,000,000, and the cash portion of such Purchase Price does not exceed \$80,000,000;

(d) Availability after giving pro forma effect to such Acquisition would exceed \$25,000,000;

(e) MasTec or the applicable Obligor has made available to Agent, not later than 14 days prior to the proposed date of such Acquisition, the results of any due diligence investigation of the target performed by or on behalf of such Obligor or its Subsidiaries, environmental assessment reports if any real property is to be acquired, copies of the Acquisition documents, and historical financial statements of the target since inception but no longer than the 3 previous years;

(f) Agent shall have received evidence satisfactory to it that no Default of Event of Default has occurred and is continuing or would exist after giving effect to such transaction and of the Obligors' continued compliance with the provisions of this Agreement and the other Loan Documents, including the provisions of **Sections 10.1.14, 10.2.22,** and, on a *pro forma* basis after giving effect to such Acquisition Borrowers shall have a Fixed Charge Coverage Ratio of not less than 1.20 to 1.0 for the 12 calendar month period ending on the date of the Acquisition;

(g) to the extent financed with Debt other than Loans, such Debt is Subordinated Debt payable to the seller,

-28-

(h) such Acquisition is not "hostile" or contested;

(i) Agent shall have received evidence reasonably satisfactory to it demonstrating on a *pro forma* basis that Adjusted EBITDA (calculated by MasTec and approved by Agent as described below in this definition) of the target for the period of 12 consecutive calendar months ended nearest to the date of determination, is at least equal to the sum of interest expense and scheduled principal payments on any Debt incurred in connection with payment of the Purchase Price (including Loans);

(j) if requested by Agent or the Required Lenders, any new Subsidiary shall have executed and delivered a Subsidiary Guaranty and a Subsidiary Security Agreement, or, at Agent's election, a Joinder Agreement, and in either case shall have delivered or caused to be delivered as to such Subsidiary the items referred to in **Sections 11.1.4, 11.1.5** and **11.1.7** and an opinion of counsel for such Subsidiary as to such matters in connection with the transactions contemplated by the Subsidiary Guaranty and Subsidiary Security Agreement or Joinder Agreement as Agent may reasonably request; and

(k) financial statements shall have been delivered to Agent and the Lenders for the most recently completed Fiscal Quarter in compliance with the provisions of **Section 10.1.3.**

A determination made for purposes of this definition on a *pro forma* basis shall be based upon Borrowers' actual results of operations and the actual results of operations of the target for the same period of 12 months ended prior to the date of determination, as if such Acquisition had occurred (and any related Debt had been incurred) on the first day of such 12-month period, as adjusted with the approval of Agent to reflect verifiable, adequately documented severance payments and reductions in officer and employee compensation, insurance expenses, interest expense and rental expense that will be realized effective upon completion of such Acquisition.

Notwithstanding any provision of this Agreement to the contrary, in connection with any merger (or other distribution of the assets) of a Subsidiary that is not an Obligor with and into (or to) an Obligor, or any Acquisition by an Obligor, whether by purchase of stock, merger, or purchase of assets, and whether in a single transaction or series of related transactions, Agent shall have the right to determine in its reasonable credit judgment (based on standards and methodologies similar to those applied to Borrowers' then existing Accounts and Equipment to the extent that the Accounts and Equipment so acquired are similar to such then existing Accounts and Equipment), whether any Accounts or Equipment so acquired shall be included in the Borrowing Base (subject to the other applicable provisions of this Agreement). In connection with such determination, Agent may obtain, at Borrowers' expense, such appraisals, commercial finance exams and other assessments of such Accounts and related Inventory, Equipment and Real Estate as Agent may deem desirable.

<u>Permitted Asset Disposition</u> — an Asset Disposition that (i) consists of the sale of Inventory of an Obligor in the Ordinary Course of Business; (ii) is a disposition of Equipment permitted by **Section 8.4.2(ii)**; (iii) is a transfer of Property to a Borrower by another Obligor or a Subsidiary; (iv) is an Asset Disposition for cash (the Net Disposition Proceeds from which are

-29-

promptly remitted to Agent for application as provided in **Section 5.2.3**), no Event of Default exists at the time of or after giving effect to the disposition, the purchaser or other transferee is not an Affiliate of the Obligor effectuating such Asset Disposition, and the Property that is the subject of the Asset Disposition consists solely of (a) Equipment which, when aggregated with all other Equipment disposed of during any Fiscal Year, has a fair market value or book value, whichever is more, that does not exceed \$10,000,000; or (b) real Property and the aggregate Net Disposition Proceeds from all Asset Dispositions pursuant to this clause (iv)(b) does not exceed \$2,500,000 in any Fiscal Year or \$5,000,000 from and after the Closing Date; or (c) other Property and the aggregate Net Disposition Proceeds from all Asset Dispositions pursuant to this clause (iv)(c) does not exceed \$5,000,000 from and after the Closing Date; (v) consists solely of a termination of a lease of real or personal Property that is not necessary to the conduct of an Obligor's business in the ordinary course, could not reasonably be expected to have a Material Adverse Effect and does not result from an Obligor's default or failure to perform under such lease; (vi) consists of a deposit to secure an obligation to the extent such deposit is permitted under **Section 10.2.5**; or (vii) is any other Asset Disposition that has been consented to in writing by Agent and the Required Lenders.

<u>Permitted Contingent Obligations</u> — Contingent Obligations (a) arising from endorsements of Payment Items for collection or deposit in the Ordinary Course of Business; (b) arising from Hedging Agreements entered into in the Ordinary Course of Business pursuant to this Agreement or with Agent's prior written consent; (c) of any Borrower and its Subsidiaries existing as of the Closing Date, including extensions and renewals thereof that do not increase the amount of such Contingent Obligations as of the date of such extension or renewal; (d) incurred in the Ordinary Course of Business with respect to surety bonds, appeal bonds, performance bonds and other similar obligations; (e) arising under indemnity agreements to title insurers to cause such title insurers to issue to Agent title insurance policies; (f) with respect to customary indemnification obligations in favor of purchasers in connection with dispositions of Equipment permitted under **Section 8.4.2** of this Agreement; (g) consisting of reimbursement obligations from time to time owing by any Borrower to an Issuing Bank with respect to Letters of Credit (but in no event to include reimbursement obligations at any time owing by a Borrower to any other Person that may issue letters of credit for the account of Borrowers); (h) of MasTec arising from any guaranty, indemnity or other assurance of payment or performance of any equipment lease for which any other Obligor is the primary obligor; (i) arising from the Earn-Out Payments under (and as defined in) the DirectStar Purchase Agreement and subject to the terms and conditions of any subordination agreement or other tri-party agreement required by Agent, (j) of DirectStar arising from a Funding Obligation Loan under (and as defined in) the DirectStar Purchase Agreement and other payments of Net Available Cash required under (and as defined in) the DirectStar Purchase Agreement and other such Funding Obligation Loan remains outstanding, and (k) other than those Contingent Obligations described in the foregoi

<u>Permitted DirectStar Investments</u> — cash Investments in DirectStar in amounts determined by the Borrowers <u>so long as</u> (i) the aggregate amount of Borrowers' cash Investments in DirectStar does not at any time exceed \$3,000,000, (ii) no Default or Event of Default exists at the time of or will exist immediately after giving effect to any such Investment, and (iii) Availability before and immediately after giving effect to any such Investment equals or exceeds \$25,000,000.

<u>Permitted Distribution</u> — any Distribution by MasTec so long as: (a) no Default or Event of Default exists before or after giving effect to such Distribution (including any Loans made hereunder to finance such Distribution); (b) MasTec is Solvent before and after giving effect to

-30-

such Distribution (including any Loans made hereunder to finance such Distribution); (c) such Distribution does not contravene MasTec's Organic Documents or violate, or cause MasTec to be in breach of or default under, (i) any provision of any Applicable Law, order, writ, judgment, injunction, decree, determination or award in effect having applicability to MasTec, or (ii) any indenture or loan or credit agreement or any other agreement, lease or instrument to which MasTec or any of its Subsidiaries is a party or by which it or its Properties may be bound or affected, including the Indenture; (d) after giving effect to such Distribution (including any Loans made hereunder to finance such Distribution), Availability is greater than \$25,000,000; and (e) on the date of such Distribution, a Senior Officer delivers to Agent a certificate as to each of the foregoing conditions and containing such pro forma balance sheets and other financial statements as Agent may reasonably require in order to confirm that MasTec is Solvent before and after giving effect to such Distribution (including any Loans made hereunder to finance such Distribution).

Permitted Lien — a Lien of a kind specified in **Section 10.2.5** of this Agreement.

<u>Permitted Purchase Money Debt</u> — Purchase Money Debt of Borrowers and their Subsidiaries that is secured by no Lien or only by a Purchase Money Lien, <u>provided that</u> the aggregate amount of Purchase Money Debt outstanding at any time does not exceed the sum of (i) \$50,000,000 plus (ii) 5% of MasTec's Consolidated Net Assets (as defined under the Indenture). For the purposes of this definition, the principal amount of any Purchase Money Debt consisting of capitalized leases shall be computed as a Capitalized Lease Obligation.

<u>Person</u> — an individual, partnership, corporation, limited liability company, limited liability partnership, joint stock company, land trust, business trust, or unincorporated organization, or a Governmental Authority.

<u>Plan</u> — an employee pension benefit plan that is covered by Title IV of ERISA, the PBA or other applicable law of any jurisdiction or subject to the minimum funding standards under Section 412 of the Internal Revenue Code and that is either (i) maintained by any Obligor for employees or (ii) maintained pursuant to a collective bargaining agreement or any other arrangement under which more than one employer makes contributions and to which such Obligor is then making or accruing an obligation to make contributions or has within the preceding 5 years made or accrued such contributions.

<u>Pledge Agreement</u> — each Pledge Agreement pursuant to which an Obligor pledges to Agent as collateral for the Obligations, shares of capital stock or other equity interests in any Subsidiary.

<u>PPSA</u> — collectively, the Personal Property Security Act of the Province of Ontario and any similar or equivalent Canadian (federal or provincial) legislation which is required as a result thereof to be applied in connection with the creation, validity, perfection or enforcement of security interests or hypothecs, including the Civil Code of Quebec, as in effect in any applicable jurisdiction, and all regulations thereunder, as amended from time to time, and any successor legislation.

<u>Pro Rata</u> — for any Lender on any date, a percentage (expressed as a decimal, rounded to the ninth decimal place) arrived at by dividing the amount of the total Commitments of such Lender on such date by the aggregate amount of the Commitments of all Lenders on such date (regardless of whether or not any of such Commitments have been terminated on or before such date).

-31-

<u>Projections</u> — (i) prior to the Closing Date and thereafter until Agent and Lenders receive new projections pursuant to **Section 10.1.5** hereof, the projections of Borrowers' financial condition, results of operations, cash flow and projected Availability, prepared on a monthly basis for the Fiscal Year ending December 31, 2008, and on an annual basis for the Fiscal Years ending 2009, 2010, 2011, 2012 and 2013; and (ii) thereafter, the projections most recently received by Agent and Lenders pursuant to and as required by **Section 10.1.5** hereof.

<u>Properly Contested</u> — in the case of any Debt of an Obligor (including any Taxes) that is not paid as and when due or payable by reason of such Obligor's bona fide dispute concerning its liability to pay same or concerning the amount thereof, (i) such Debt is being properly contested in good faith by appropriate proceedings promptly instituted and diligently conducted; (ii) such Obligor has established appropriate reserves as shall be required in conformity with GAAP; (iii) the non-payment of such Debt will not have a Material Adverse Effect and will not result in a forfeiture or sale of any assets of such Obligor; (iv) no Lien is imposed upon any of such Obligor's assets with respect to such Debt unless such Lien is at all times junior and subordinate in priority to the Liens in favor of Agent (except only with respect to property taxes that have priority as a matter of Applicable Law) and enforcement of such Lien is stayed during the period prior to the final resolution or disposition of such dispute; (v) if the Debt results from, or is determined by the entry, rendition or issuance against an Obligor or any of its assets of a judgment, writ, order or decree, enforcement of such judgment, writ, order or decree is stayed pending a timely appeal or other judicial review; and (vi) if such contest is abandoned, settled or determined adversely (in whole or in part) to such Obligor, such Obligor forthwith pays such Debt and all penalties, interest and other amounts due in connection therewith.

<u>Property</u> — any interest in any kind of property or asset, whether real, personal or mixed and whether tangible or intangible.

Protective Advances — as defined in Section 2.1.5 of this Agreement.

<u>Purchase Money Debt</u> — means and includes (i) Debt (other than the Obligations) for the payment of all or any part of the purchase price of any fixed assets, (ii) any Debt (other than the Obligations) incurred at the time of or within 10 days prior to or after the acquisition of any fixed assets for the purpose of financing all or any part of the purchase price thereof, and (iii) any renewals, extensions or refinancings (but not any increases in the principal amounts) thereof outstanding at the time.

<u>Purchase Money Lien</u> — a Lien upon fixed assets which secures Purchase Money Debt, but only if such Lien shall at all times be confined solely to the fixed assets acquired through the incurrence of the Purchase Money Debt secured by such Lien and such Lien constitutes a purchase money security interest under the UCC or the PPSA, as applicable.

<u>Purchase Price</u> — for purposes of the definition of "Permitted Acquisition" means an amount equal to the total consideration paid for such Acquisition, including all cash payments (whether classified as purchase price, noncompete payments, consulting payments, "earn out" or otherwise and without regard to whether such amount is paid in whole or in part at the closing of the Acquisition or over time thereafter, but excluding any finance charges attributable to deferred payments, any salary or other employment compensation paid to a seller for the purpose of retaining such seller's services as an active employee of a Borrower or a Subsidiary and, upon Agent's determination to such effect, any "earn out" based upon achievement of a material improvement in financial performance of the target), the remaining principal amount of all Debt of the Acquisition target and of any Subordinated Debt owing to the seller, and the value (as

-32-

determined by the board of directors of MasTec, including pursuant to the applicable purchase agreement between the relevant Obligor or Subsidiary of an Obligor and the seller, in the case of any property, the fair value of which is not readily ascertainable) of all other property, other than capital stock or options to acquire such capital stock of MasTec, transferred by MasTec to the seller. For purposes of determining the Purchase Price of the DirectStar Acquisition, such Purchase Price shall be calculated as set forth above, <u>provided</u>, <u>that</u>, the amount of the "earn-out" portion of such Purchase Price will be based on the projected calculation thereof provided by MasTec to Agent on or prior to the closing date of the DirectStar Acquisition.

RCRA — the Resource Conservation and Recovery Act (42 U.S.C. §§ 6991-6991i) and all rules and regulations promulgated pursuant thereto.

<u>Real Estate</u> — all right, title and interest of a Borrower (whether as owner, lessor or lessee) at any time or times held by such Borrower in any real Property or any buildings, structures, parking areas or other improvements thereon, including the real Property and improvements thereon owned by a Borrower and located at 2801 SW 46th Avenue, Davie, Florida, SR540 Lakeland, Florida, 4250 North Powerline Road, Pompano, Florida, 7221 Dr. Martin Luther King Boulevard East, Tampa, Florida, 209 Art Bryant Drive, Asheboro, North Carolina, Highway #2 East, Shevlin, Minnesota, and 2700, 2701 and 2716 E. 5th Street and 2808 Industrial Terrace, Austin, Texas.

<u>Real Estate Formula Amount</u> — on any date of determination thereof, the lesser of (i) \$15,000,000 or (ii) an amount equal to the product of sixty percent (60%) <u>multiplied</u> by the fair market value of Eligible Real Estate set forth in the most recent fair market value appraisal of the Real Estate which has been accepted by and is satisfactory to Agent; <u>provided</u>, <u>that</u> the amount calculated under this clause shall be reduced by an amount, as determined by Agent, equal to the aggregate amount of the fair market value of all Eligible Real Estate that has been disposed of by Obligors (other than in accordance with **Section 8.4.2(ii)**) since the date of the most recent fair market value appraisal of the Real Estate which has been accepted by and is satisfactory to Agent.

<u>Refinancing Conditions</u> — the following conditions, each of which must be satisfied before Refinancing Debt shall be permitted under **Section 10.2.3** of this Agreement: (i) the Refinancing Debt is in an aggregate principal amount that does not exceed the aggregate principal amount of the Debt being extended, renewed or refinanced (or in the case of the Indenture and Senior Notes, the original principal amount thereof), (ii) the Refinancing Debt has a later or equal final maturity and a longer or equal weighted average life than the Debt being extended, renewed or refinanced, (iii) the Refinancing Debt does not bear a rate of interest that exceeds a market rate (as determined in good faith by a Senior Officer) as of the date of such extension, renewal or refinancing, (iv) if the Debt being extended, renewed or refinanced is subordinate to the Obligations, the Refinancing Debt is subordinated to the same extent, (v) the covenants contained in any instrument or agreement relating to the Refinancing Debt are no less favorable to Obligors than those relating to the Debt being extended, renewed or refinancing, no Default or Event of Default shall exist.

<u>Refinancing Debt</u> — Debt for Money Borrowed that is permitted by **Section 10.2.3** and that is the subject or the result of an extension, renewal or refinancing.

<u>Regulation D</u> — Regulation D of the Board of Governors.

Register — the register maintained by Agent in accordance with Section 5.8.2 of this Agreement.

Reimbursement Date — as defined in Section 2.3.1(iii) of this Agreement.

Related Real Estate Documents — with respect to any Real Estate subject to a Mortgage, the following, in form and substance satisfactory to Agent and received by Agent for review at least 15 days prior to the effective date of the Mortgage (or most recent amendment thereto): (a) duly executed amendments or assignments of the existing Mortgages in favor of the Existing Agent, conveying to Agent for the benefit of the Secured Parties a first priority mortgage lien on such Real Estate (including the Real Estate located at 2801 SW 46th Avenue, Davie, Florida, SR540 Lakeland, Florida, 7221 Dr. Martin Luther King Boulevard East, Tampa, Florida, 209 Art Bryant Drive, Asheboro, North Carolina, Highway #2 East, Shevlin, Minnesota, and 2700, 2701 and 2716 E. 5th Street and 2808 Industrial Terrace, Austin, Texas); (b) fully paid mortgagee title insurance policies (or binding commitments to issue title insurance policies, marked to Agent's satisfaction to evidence the form of such policies to be delivered), in standard ALTA form, issued by a title insurance company satisfactory to Agent, each in an amount equal to not less than the fair market value of the Real Estate or leasehold interest, as the case may be, subject to the Mortgages, insuring the Mortgages to create a valid Lien on all Real Estate and valid Liens on the leasehold interest described therein with no exceptions which Agent shall not have approved in writing and no survey exceptions, which policies (and commitments therefor) shall have acceptable zoning endorsements; (c) a current, as-built survey with respect to each parcel of Real Estate subject to a Mortgage, which survey shall indicate the following: (i) an accurate metes and bounds or lot, block and parcel description of such Real Estate; (ii) the correct location of all buildings, structures and other improvements on such Real Estate, including all streets, easements, rights of way and utility lines; (iii) the location of ingress and egress from such Real Estate, and the location of any set-back or other building lines affecting such Real Estate; and (iv) a flood plain certification, and (v) a certification by a registered land surveyor in form and substance acceptable to Agent, certifying to the accuracy and completeness of such survey and to such other matters relating to such Real Estate and survey as Agent shall require; (d) the favorable, written opinions of Greenberg Traurig LLP and the respective local counsel to Borrowers in each state in which the Real Estate subject to a Mortgage is located covering, to Agent's satisfaction the matters set forth on Exhibit F attached hereto with respect to the Mortgages and the other documents to be executed and delivered in connection therewith; (e) such assignments of leases, estoppel letters, attornment agreements, consents, waivers and releases as Agent may require with respect to other Persons having an interest in the Real Estate; (f) flood insurance in an amount, with endorsements and by an insurer acceptable to Agent, if the Real Estate is within a flood plain; (g) a current appraisal of the Real Estate, prepared by an appraiser acceptable to Agent, and in form and substance satisfactory to Required Lenders; (h) an environmental assessment, prepared by environmental engineers acceptable to Agent, and accompanied by such reports, certificates, studies or data as Agent may reasonably require, which shall all be in form and substance satisfactory to Required Lenders; and (i) an Environmental Agreement and such other documents, instruments or agreements as Agent may reasonably require with respect to any environmental risks regarding the Real Estate.

<u>Rent Reserve</u> — on any date, the aggregate of (i) all past due rent, fees or other charges owing on such date by any Obligor to any landlord of any premises where any of the Collateral is located or to any processor, repairman, mechanic or other Person who is in possession of any Collateral or has asserted any Lien or claim thereto, and (ii) a reserve equal to 3 months rent or other charges with respect to any Collateral in the possession of, or at a location owned or

-34-

operated by, a Person other than an Obligor if such Person has not duly executed and delivered to Agent a Lien Waiver satisfactory to Agent.

<u>Reportable Event</u> — any of the events set forth in Section 4043(b) of ERISA or any similar provision of the PBA, including, with respect to a Canadian Obligor, any other event or condition which might constitute grounds for the termination of, winding up or partial termination or winding up or the appointment of trustee to administer any Plan.

<u>Required Lenders</u> — at any date of determination thereof, Lenders having Commitments representing more than 50% of the aggregate Commitments at such time (and in any event including BofA); <u>provided</u>, <u>however</u>, that if any Lender shall be in breach of any of its obligations hereunder to Borrowers or Agent, including any breach resulting from its failure to honor its Commitment in accordance with the terms of this Agreement, then, for so long as such breach continues, the term "<u>Required Lenders</u>" shall mean Lenders (excluding each Lender that is in breach of its obligations under this Agreement) having Commitments representing more than 50% of the aggregate Commitments (excluding the Commitments of each Lender that is in breach of its obligations under this Agreement) at such time (and in any event including BofA); <u>provided further</u>, <u>however</u>, that if all of the Commitments have been terminated, the term "<u>Required Lenders</u>" shall mean Lenders (excluding each Lender that is in breach of its obligations under this Agreement) holding Loans (including Swingline Loans) representing more than 50% of the aggregate principal amount of Loans (including Swingline Loans) outstanding at such time (and in any event including BofA).

<u>Restricted Investment</u> — any acquisition of Property by an Obligor or any of its Subsidiaries in exchange for cash or other Property, whether in the form of an acquisition of Equity Interests or Debt, or the purchase or acquisition by such Obligor or any of its Subsidiaries of any other Property, or a loan, advance, capital contribution or subscription (each of the foregoing, an "Investment"), except acquisitions of the following: (i) fixed assets to be used in the Ordinary Course of Business of such Obligor or any of its Subsidiaries so long as the acquisition costs thereof are Capital Expenditures permitted hereunder; (ii) goods held for sale or lease or to be used in the manufacture of goods or the provision of services by such Obligor or any of its Subsidiaries in the Ordinary Course of Business; (iii) Current Assets arising from the sale or lease of goods or the rendition of services in the Ordinary Course of Business; (iv) Investments in Subsidiaries to the extent existing on the Closing Date; (v) Cash Equivalents to the extent they are not subject to rights of offset in favor of any Person other than Agent or a Lender; (vi) loans and other advances of money to the extent not prohibited by **Section 10.2.2**; (vii) Permitted Acquisitions; (viii) those Investments of Borrowers described in <u>Schedule 1.1A</u>, to the extent existing or committed to (as described in <u>Schedule 1.1A</u>) on the Closing Date; (ix) Permitted DirectStar Investment, and (x) any other Investment (other than a Permitted DirectStar Investment) that does not result in an Acquisition, so long as Borrowers have demonstrated to Agent's satisfaction that both before or after giving effect to such Investment (A) no Default or Event of Default exists and (B) Availability is greater than \$25,000,000.

<u>Restrictive Agreement</u> — an agreement (other than any of the Loan Documents) that, if and for so long as an Obligor or any Subsidiary of such Obligor is a party thereto, would prohibit, condition or restrict such Obligor's or Subsidiary's right to incur or repay Debt for Money Borrowed (including any of the Obligations); grant Liens upon any of such Obligor's or Subsidiary's assets (including Liens granted in favor of Agent pursuant to the Loan Documents); declare or make Distributions; amend, modify, extend or renew any agreement evidencing Debt

-35-

for Money Borrowed (including any of the Loan Documents); or repay any Debt owed to another Obligor.

<u>Revolver Commitment</u> — at any date for any Lender, the obligation of such Lender to make Revolver Loans and to purchase participations in LC Obligations pursuant to the terms and conditions of this Agreement, which shall not exceed the principal amount set forth opposite such Lender's name under the heading "Revolver Commitment" on the signature pages of this Agreement or as described in the Assignment and Acceptance by which it became a Lender, as modified from time to time pursuant to the terms of this Agreement or to give effect to any applicable Assignment and Acceptance; and "Revolver Commitments" means the aggregate principal amount of the Revolver Commitments of all Lenders, the maximum amount of which on any date shall be \$210,000,000, as reduced from time to time pursuant to **Section 2.2**, or increased from time to time pursuant to **Section 2.1.6**.

<u>Revolver Loan</u> — a loan made by Lenders as provided in **Section 2.1** of this Agreement (including any Out-of-Formula Loan) or a Swingline Loan funded solely by BofA or a Protective Advance.

<u>Revolver Note</u> — a Revolver Note to be executed by Borrowers in favor of each Lender in the form of <u>Exhibit A</u> attached hereto, which shall be in the face amount of such Lender's Revolver Commitment and which shall evidence all Revolver Loans (other than Swingline Loans) made by such Lender to Borrowers pursuant to this Agreement.

<u>Royalties</u> — with respect to a License Agreement, all royalties, fees, expense reimbursement and other amounts at any time owing by an Obligor under such License Agreement.

<u>S&P</u> — Standard & Poor's Ratings Group, a division of McGraw-Hill, Inc.

<u>Schedule of Accounts</u> — as defined in **Section 8.2.1** of this Agreement.

SEC — Securities and Exchange Commission.

<u>Secured Parties</u> — Agent, Issuing Bank, Lenders (including BofA as the provider of Swingline Loans) and BofA or any other Lender (and any Affiliate of BofA or such Lender) as the provider of any Bank Products.

<u>Security Documents</u> — each Mortgage, each Guaranty, each Pledge Agreement, the Canadian Security Agreement, the Trademark Security Agreement, the Control Agreements, the Business Interruption Insurance Assignment and all other instruments and agreements now or at any time hereafter securing the whole or any part of the Obligations.

Senior Funded Debt — all Funded Debt other than Subordinated Debt.

<u>Senior Notes</u> — MasTec's Senior Notes having a maturity date of February 1, 2017 in the original principal amount of \$150,000,000, issued pursuant to the New Notes Indenture on January 31, 2007 on an unsecured basis and otherwise on terms satisfactory to Agent and Lenders; provided, that, the original principal amount set forth above may be increased by an amount of up to \$25,000,000 reflecting an oversubscription of the New Notes issued under the New Notes Indenture.

-36-

<u>Senior Officer</u> — the chairman of the board of directors, the president, the chief financial officer, the corporate controller, the treasurer or the cash manager of, or in-house legal counsel to, a Borrower.

<u>Settlement Date</u> — as defined in **Section 4.1.3(i)** of this Agreement.

<u>Settlement Report</u> — a report delivered by Agent to Lenders summarizing the amount of the outstanding Revolver Loans as of the Settlement Date and the calculation of the Borrowing Base as of such Settlement Date.

Software — shall have the meaning given to the term "software" in the UCC.

<u>Solvent</u> — as to any Person, such Person (i) owns Property whose fair salable value is greater than the amount required to pay all of such Person's debts (including contingent, subordinated, unmatured and unliquidated liabilities), (ii) owns Property whose present fair salable value (as defined below) is greater than the probable total liabilities (including contingent, subordinated, unmatured and unliquidated liabilities), of such Person as they become absolute and matured, (iii) is able to pay all of its debts as such debts mature, (iv) has capital that is not unreasonably small for its business and is sufficient to carry on its business and transactions and all business and transactions in which it is about to engage, (v) is not "insolvent" within the meaning of Section 101(32) of the Bankruptcy Code and the requisite section of the Bankruptcy and Insolvency Act (Canada), and (vi) has not incurred (by way of assumption or otherwise) any obligations or liabilities (contingent or otherwise) under any of the Loan Documents, or made any conveyance pursuant to or in connection therewith, with actual intent to hinder, delay or defraud either present or future creditors of such Person or any of its Subsidiaries. As used herein, the term "fair salable value" of a Person's assets means the amount that may be realized within a reasonable time, either through collection or sale of such assets at the regular market value, based upon the amount that could be obtained for such assets within such period by a capable and diligent seller from an interested buyer who is willing (but is under no compulsion) to purchase under ordinary selling conditions.

<u>Statutory Reserves</u> — on any date, the percentage (expressed as a decimal) established by the Board of Governors which is the then stated maximum rate for all reserves (including all basic, emergency, supplemental or other marginal reserve requirements and taking into account any transitional adjustments or other scheduled in reserve requirements) applicable to any member bank of the Federal Reserve System in respect to Eurocurrency Liabilities (or any successor category of liabilities under Regulation D). Such reserve percentage shall include those imposed pursuant to said Regulation D. The Statutory Reserve shall be adjusted automatically on and as of the effective date of any change in such percentage.

<u>Subordinated Debt</u> — unsecured Debt incurred by an Obligor that is expressly subordinated and made junior to the Full Payment of the Obligations and contains terms and conditions (including terms relating to interest, fees, repayment and subordination) satisfactory to Agent.

<u>Subsidiary</u> — any Person in which more than 50% of its outstanding Voting Securities or more than 50% of all Equity Interests is owned directly or indirectly by a Borrower, by one or more other Subsidiaries of such Borrower or by a Borrower and one or more other Subsidiaries.

-37-

<u>Subsidiary Guarantor</u> — any Subsidiary of MasTec that is not a Borrower and that has, at Agent's request or with its consent, executed and delivered the Subsidiary Guaranty or a joinder agreement in respect thereof and a Subsidiary Security Agreement.

<u>Subsidiary Guaranty</u> — a Guaranty of the Obligations substantially in the form of <u>Exhibit K</u> attached hereto or as otherwise acceptable to Agent and MasTec.

<u>Subsidiary Security Agreement</u> — one or more agreements substantially in the form of <u>Exhibit L</u> attached hereto or otherwise in form and substance satisfactory to Agent in its reasonable judgment, sufficient to create in favor of Agent a Lien on all of the assets of any Subsidiary Guarantor and proceeds thereof.

Supporting Obligation — shall have the meaning given to the term "supporting obligation" in the UCC.

<u>Swingline Loan</u> — as defined in **Section 4.1.3(ii)** of this Agreement.

<u>Taxes</u> — any present or future taxes, levies, imposts, duties, fees, assessments, deductions, withholdings or other charges of whatever nature, including income, receipts, excise, property, sales, use, transfer, license, payroll, withholding, social security and franchise taxes now or hereafter imposed or levied by the United States or any other Governmental Authority and all interest, penalties, additions to tax and similar liabilities with respect thereto, but excluding, in the case of each Lender, taxes imposed on or measured by the net income or overall gross receipts of such Lender.

<u>Term</u> — as defined in **Section 6.1** of this Agreement.

<u>Trademark Security Agreement</u> — the Trademark Security Agreement to be executed by Obligors in favor of Agent on or before the Closing Date and by which Obligors shall grant to Agent, for the benefit of Secured Parties, as security for the Obligations, a security interest in all of Obligors' right, title and interest in and to all of their trademarks.

<u>Transferee</u> — as defined in **Section 14.3.3** of this Agreement.

<u>Trigger Event</u> — as defined in Section 8.2.5(ii) of this Agreement.

<u>Type</u> — any type of a Loan determined with respect to the interest option applicable thereto, which shall be either a LIBOR Loan or a Base Rate Loan.

<u>UCC</u> — the Uniform Commercial Code (or any successor statute) as adopted and in force in the State of Georgia or, when the laws of any other state govern the method or manner of the perfection or enforcement of any security interest in any of the Collateral, the Uniform Commercial Code (or any successor statute) of such state.

<u>Undrawn Amount</u> — on any date and with respect to a particular Letter of Credit, the total amount then available to be drawn under such Letter of Credit in Dollars.

<u>Upstream Payment</u> — a payment or distribution of cash or other Property by a Subsidiary of a Borrower to such Borrower, whether in repayment of Debt owed by such Subsidiary to such Borrower, as a dividend or distribution on account of such Borrower's ownership of Equity Interests or otherwise.

-38-

<u>USA Patriot Act</u> — the Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, Pub. L. No. 107-56, 115 Stat. 272 (2001).

<u>Verizon</u> — Verizon Communications, Inc. and its Affiliates and Subsidiaries.

<u>Verizon Concentration Percentage</u> — (a) at any time that Verizon's corporate credit rating or senior debt rating (secured or unsecured) by Moody's is Baa3 or higher and by S&P BBB- or higher, 40%; or (b) at any time that Verizon's corporate credit rating or senior debt rating (secured or unsecured) by Moody's is lower than Baa3 or by S&P is lower than BBB-, 15%; or (c) irrespective of Verizon's rating, such lesser percentage as Agent may in its reasonable credit judgment determine from time to time.

<u>Voting Securities</u> — Equity Interests of any class or classes of a corporation or other entity the holders of which are ordinarily, in the absence of contingencies, entitled to elect a majority of the corporate directors or Persons performing similar functions.

1.2. <u>Accounting Terms</u>. Unless otherwise specified herein, all terms of an accounting character used in this Agreement shall be interpreted, all accounting determinations under this Agreement shall be made, and all financial statements required to be delivered under this Agreement shall be prepared in accordance with GAAP, applied on a basis consistent with the most recent audited Consolidated financial statements of Borrowers and their Subsidiaries heretofore delivered to Agent and Lenders and using the same method for inventory valuation as used in such audited financial statements, except for any change required by GAAP.

1.3. <u>Other Terms</u>. All other terms contained in this Agreement shall have, when the context so indicates, the meanings provided for by the UCC to the extent the same are used or defined therein.

1.4. Certain Matters of Construction. The terms "herein," "hereof" and "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular section, paragraph or subdivision. Any pronoun used shall be deemed to cover all genders. In the computation of periods of time from a specified date to a later specified date, the word "from" means "from and including" and the words "to" and "until" each means "to but excluding." The section titles, table of contents and list of exhibits appear as a matter of convenience only and shall not affect the interpretation of this Agreement. All references to statutes and related regulations shall include any amendments of same and any successor statutes and regulations; to any agreement, instrument or other documents (including any of the Loan Documents) shall include any and all modifications and supplements thereto and any and all restatements, extensions or renewals thereof to the extent such modifications, supplements, restatements, extensions or renewals of any such documents are permitted by the terms thereof; to any Person (including Agent, an Obligor, a Lender or BofA) shall mean and include the successors and permitted assigns of such Person; to "including" and "include" shall be understood to mean "including, without limitation" (and, for purposes of each Loan Document, the parties agree that the rule of *ejusdem generis* shall not be applicable to limit a general statement, which is followed by or referable to an enumeration of specific matters to matters similar to the matters specifically mentioned); or to the time of day shall mean the time of day on the day in question in Atlanta, Georgia, unless otherwise expressly provided in this Agreement. A Default or an Event of Default shall be deemed to exist at all times during the period commencing on the date that such Default or Event of Default occurs to the date on which such Default or Event of Default is waived in writing by Agent acting with the consent of or at the direction of the Lenders or the Required Lenders, as applicable pursuant to this Agreement or, in the case of a Default, is cured within any period of cure expressly provided in this Agreement; and an Event of Default shall "continue" or be "continuing" until such Event of Default has

-39-

been waived in writing by Agent acting with the consent of or at the direction of the Lenders or the Required Lenders, as applicable. All calculations of value shall be in Dollars, all Loans shall be funded in Dollars and all Obligations shall be repaid in Dollars. Whenever the phrase "to the best of Borrowers' knowledge" or words of similar import relating to the knowledge or the awareness of a Borrower are used in this Agreement or other Loan Documents, such phrase shall mean and refer to (i) the actual knowledge of a Senior Officer of any Borrower or (ii) the knowledge that a Senior Officer of a Borrower would have obtained if they had engaged in good faith and diligent performance of his duties, including the making of such reasonably specific inquiries as may be necessary of the employees or agents of a Borrower and a good faith attempt to ascertain the existence or accuracy of the matter to which such phrase relates. Any Lien referred to in this Agreement or any of the other Loan Documents, any payment made by or to, or funds received by, Agent pursuant to or as contemplated by any of the Loan Documents, or any other act taken or omitted to be taken by Agent shall, unless otherwise expressly provided, be created, entered into, made or received, or taken or omitted for the benefit or account of Agent and the Lenders.

SECTION 2. CREDIT FACILITIES

Subject to the terms and conditions of, and in reliance upon the representations and warranties made in, this Agreement and the other Loan Documents, Lenders severally agree to the extent and in the manner hereinafter set forth to make their respective shares of the Commitments available to Borrowers in the aggregate amount up to \$210,000,000, as set forth herein below:

2.1. Revolver Commitments.

2.1.1. <u>Revolver Loans</u>. Each Lender agrees, severally to the extent of its Revolver Commitment and not jointly with the other Lenders, upon the terms and subject to the conditions set forth herein, to make Revolver Loans to Borrowers on any Business Day during the period from the Closing Date through the Business Day before the last day of the Term, not to exceed in aggregate principal amount outstanding at any time such Lender's Revolver Commitment at such time, which Revolver Loans may be borrowed, prepaid, repaid and reborrowed in accordance with the provisions of this Agreement; <u>provided</u>, <u>however</u>, that Lenders shall have no obligation to Borrowers whatsoever to honor any request for a Revolver Loan on or after the Commitment Termination Date or if at the time of the proposed funding thereof the aggregate principal amount of all of the Revolver Loans then outstanding and Pending Revolver Loans exceeds, or would exceed after the funding of such Revolver Loans, the Borrowing Base. Each Borrowing of Revolver Loans shall be funded by Lenders on a Pro Rata basis in accordance with their respective Revolver Commitments (except for BofA with respect to Swingline Loans). The Revolver Loans shall bear interest as set forth in **Section 3.1.** hereof. Each Revolver Loan shall, at the option of Borrowers, be made or continued as, or converted into, part of one or more Borrowings that, unless specifically provided herein, shall consist entirely of Base Rate Loans or LIBOR Loans.

2.1.2. <u>Out-of-Formula Loans</u>. If the unpaid balance of Revolver Loans outstanding at any time should exceed the Borrowing Base at such time (an "Out-of-Formula Condition"), such Revolver Loans shall nevertheless constitute Obligations that are secured by the Collateral and entitled to all of the benefits of the Loan Documents. In the event that Lenders are willing in their sole and absolute discretion to make Out-of-Formula Loans or are required to do so by **Section 13.9.4** hereof, such Out-of-Formula Loans shall be payable **on demand** and shall bear interest as provided in **Section 3.1.5** of this Agreement. Notwithstanding the foregoing, the maximum amount of Out-of-Formula Loans outstanding at any time, when added to the aggregate of Revolver Loans, LC Obligations and Protective Advances outstanding at such time, shall not exceed the aggregate principal amount of the Revolver Commitments of all Lenders at such time.

-40-

2.1.3. Use of Proceeds. The proceeds of the Revolver Loans shall be used by Borrowers solely for one or more of the following purposes: (i) to satisfy the Debt owing on the Closing Date to the Existing Lenders under the Existing Loan Agreement; (ii) to pay the fees and transaction expenses associated with the closing of the transactions described herein; (iii) to pay any of the Obligations in accordance with this Agreement; (iv) to finance the ongoing working capital and other general corporate purposes of the Borrowers and their Subsidiaries; and (v) to make expenditures for other lawful corporate purposes of Borrowers to the extent such expenditures are not prohibited by this Agreement or Applicable Law. In no event may any Revolver Loan proceeds be used by any Borrower (x) to purchase or to carry, or to reduce, retire or refinance any Debt incurred to purchase or carry, any Margin Stock or for any related purpose that violates the provisions of Regulations T, U or X of the Board of Governors, (y) to fund any operations or finance any investments or activities in, or to make payments to, a Blocked Person, or (z) to defease, redeem or refinance the Senior Notes.

2.1.4. <u>Revolver Notes</u>. The Revolver Loans made by each Lender and interest accruing thereon shall be evidenced by the records of Agent and such Lender and by the Revolver Note payable to such Lender (or the assignee of such Lender), which shall be executed by Borrowers, completed in conformity with this Agreement and delivered to such Lender. All outstanding principal amounts and accrued interest under the Revolver Notes shall be due and payable as set forth in **Section 5.2** hereof.

2.1.5. Protective Advances. Agent shall be authorized, in its sole and absolute discretion, at any time or times that a Default or Event of Default exists or any of the conditions precedent set forth in **Section 11** hereof have not been satisfied, to make Revolver Loans that are Base Rate Loans to Borrowers in an aggregate amount outstanding at any time not to exceed \$5,000,000, but only to the extent that Agent, in the exercise of its business judgment, deems the funding of such Loans (herein called "Protective Advances") to be necessary or desirable (i) to preserve or protect the Collateral or any portion thereof, (ii) to enhance the likelihood, or increase the amount, of repayment of the Obligations or (iii) to pay any other amount chargeable to Borrowers pursuant to the terms of this Agreement, including costs, fees and expenses, all of which Protective Advances shall be deemed part of the Obligations and secured by the Collateral and shall be treated for all purposes of this Agreement (including **Sections 5.6.1** and **15.4**) as advances for the repayment to Agent and Lenders of Extraordinary Expenses; provided, however, that the Required Lenders may at any time revoke Agent's actual receipt thereof. Absent such revocation, Agent's determination that the making of a Protective Advance is required for any such purposes shall be conclusive. Each Revolver Lender shall participate in each Protective Advances outstanding at any time, when added to the aggregate of Revolver Loans, LC Obligations and Out-of-Formula Loans outstanding at such time, shall not exceed the total of the Revolver Commitments. Nothing in this **Section 2.1.6** shall be construed to limit in any way the amount of Extraordinary Expenses that may be incurred by Agent and that Borrowers shall be obligated to reimburse to Agent as provided in the Loan Documents.

2.1.6. Increase in Revolver Commitments.

(i) To the extent that any requested increase in the Revolver Commitments is permitted under and will not violate the Indenture, and provided that Agent has received evidence satisfactory to it from Borrowers that such requested increase is permitted under and will not violate the Indenture, then upon the terms and subject to the conditions set forth herein, on any Business Day during the period from July 31, 2007 through the Business Day immediately prior to the last calendar day of the Term, and so long as no Default or Event of Default exists, Borrowers may request that the Revolver

-41-

Commitments be increased and, upon such request, Agent shall use reasonable efforts in light of then-current market conditions to solicit additional Eligible Assignees to become Lenders for the purposes of this Agreement, or to encourage any Lender to increase its Revolver Commitment; provided, that (a) each Lender that is a party to this Agreement immediately prior to such increase shall have the first option, and may elect, to fund its Pro Rata share of the amount of the increase in the Revolver Commitment (or any such greater amount in the event that one or more Lenders does not elect to fund its respective Pro Rata share of the amount of the increase in the Revolver Commitment), thereby increasing its Revolver Commitment hereunder, but no Lender shall have any obligation to do so, (b) in the event that it becomes necessary to include a new Eligible Assignee to fund the amount of the requested increase in the Revolver Commitment, each such Eligible Assignee shall become a Lender hereunder and agree to become party to, and shall assume and agree to be bound by, this Agreement, subject to all terms and conditions hereof; (c) no Lender shall have an obligation to Borrowers, Agent or any other Lender to increase its Revolver Commitment or its Pro Rata share of the Revolver Commitments, which decision shall be made in the sole discretion of each Lender; and (d) in no event shall the addition of any Lender or Lenders or the increase in the Revolver Commitment of any Lender under this Section 2.1.6 increase the aggregate Revolver Commitments to an aggregate amount greater than \$260,000,000 less the amount of any voluntary reductions under Section 2.2 hereof. Upon the addition of any Lender, or the increase in the Revolver Commitment of any Lender, the dollar amount of the Revolver Commitment set forth opposite each Lender's name on the signature pages to this Agreement shall be amended by Agent and Borrowers to reflect such addition or such increase, and Agent shall deliver to Lenders and Borrowers a copy of such amendment. Lenders shall be entitled to receive and Borrowers shall be obligated to pay a mutually agreeable amendment fee to Agent for the Pro Rata benefit of those Lenders who increase their Revolver Commitment and any new Lenders, such fee to be based upon the increase in their Revolver Commitments only and not on their aggregate Revolver Commitments after giving effect to such increase.

(ii) If any requested increase in the Revolver Commitments is agreed to in accordance with subsection (i) above, Agent and Borrowers shall determine the effective date of such increase (the "Increase Effective Date"). Agent, with the consent and approval of Borrowers, shall promptly confirm in writing to Lenders the final allocation of such increase as of the Increase Effective Date, and each new Lender and each existing Lender that has increased its Revolver Commitment shall purchase Revolver Loans and LC Obligations from each other Lender in an amount such that, after such purchase or purchases, the amount of outstanding Revolver Loans and LC Obligations from each Lender shall equal such Lender's respective Pro Rata share of the Revolver Commitments, as modified to give effect to such increase, multiplied by the aggregate amount of Revolver Loans outstanding and LC Obligations from all Lenders. As condition precedents to the effectiveness of such increase, Borrowers shall deliver to Agent (i) a certificate dated as of the Increase Effective Date (in sufficient copies for each Lender) signed by the Chief Financial Officer of Borrower Agent on behalf of Borrowers, including a Compliance Certificate demonstrating compliance with the terms of this Agreement and certification that, both before and after giving effect to such increase, each representation and warranty contained in **Section 9** and each certification in **Section 15.18** of this Agreement is true and correct in all material respects on and as of the Increase Effective Date (except to the extent that any such representation or warranty is stated to relate solely to an earlier date), that the requested increase is permitted under and will not violate the Indenture, and that no Default or Event of Default exists, and (ii) legal opinions from counsel to the Borrowers in form and substance acceptable to Agent that the Loan Agreement and the requested increase in the Revolver Commitments provided for herein is permitted under and does not violate the Indentu

2.2. <u>Voluntary Reductions of Revolver Commitments</u>. Borrowers shall have the right to permanently reduce the amount of the Revolver Commitments, on a Pro Rata basis for each Lender, at

-42-

any time and from time to time upon written notice to Agent of such reduction, which notice shall specify the amount of such reduction, shall be irrevocable once given, shall be given at least 5 Business Days prior to the end of a month and shall be effective only upon Agent's receipt thereof. Agent shall promptly transmit such notice to each Lender. The effective date of any voluntary reduction of the Revolver Commitments shall be the first day of the month following the month in which such notice is timely received by Agent. If on the effective date of any such reduction in the Revolver Commitments and after giving effect thereto an Out-of-Formula Condition exists, then the provisions of **Section 5.2.1(iii)** shall apply, except that such repayment shall be due immediately upon such effective date without further notice to or demand upon Borrowers. If the Commitments are reduced to zero, then such reduction shall be deemed a termination of the Commitments by Borrowers pursuant to **Section 6.2.2**. The Revolver Commitments, once reduced, may not be reinstated without the written consent of all Lenders.

2.3. LC Facility.

2.3.1. <u>Issuance of Letters of Credit</u>. Subject to all of the terms and conditions hereof, Issuing Bank agrees to establish the LC Facility pursuant to which, during the period from the date hereof to (but excluding) the 30th day prior to the last day of the Term, Issuing Bank shall issue one or more Letters of Credit on Borrower Agent's request therefor from time to time, subject to the following terms and conditions:

(i) Each Borrower acknowledges that Issuing Bank's willingness to issue any Letter of Credit is conditioned upon Issuing Bank's receipt of (A) an LC Application with respect to the requested Letter of Credit and (B) such other instruments and agreements as Issuing Bank may customarily require for the issuance of a letter of credit of equivalent type and amount as the requested Letter of Credit. Issuing Bank shall have no obligation to issue any Letter of Credit unless (x) Issuing Bank receives an LC Request and LC Application at least 3 Business Days prior to the date of issuance of a Letter of Credit, and (y) each of the LC Conditions is satisfied on the date of Issuing Bank's receipt of the LC Request and at the time of the requested issuance of a Letter of Credit. Any Letter of Credit issued on the Closing Date shall be for an amount in Dollars that is greater than \$250,000.

(ii) Letters of Credit may be requested by a Borrower only if such Letters of Credit are to be used (a) to support obligations of such Borrower incurred in the Ordinary Course of Business of such Borrower, on a standby basis or (b) for such other purposes as Agent and Lenders may approve from time to time in writing.

(iii) Borrowers shall comply with all of the terms and conditions imposed on Borrowers by Issuing Bank that are contained in any LC Application or in any other agreement customarily or reasonably required by Issuing Bank in connection with the issuance of any Letter of Credit. If Issuing Bank shall honor any request for payment under a Letter of Credit, Borrowers shall pay to Issuing Bank, in Dollars on the first Business Day following the date on which payment was made by Issuing Bank (the "Reimbursement Date"), an amount equal to the amount paid by Issuing Bank under such Letter of Credit, together with interest from and after the Reimbursement Date until Full Payment is made by Borrowers at the Default Rate for Revolver Loans constituting Base Rate Loans. Until Issuing Bank has received payment from Borrowers in accordance with the foregoing provisions of this clause (iii), Issuing Bank, in addition to all of its other rights and remedies under this Agreement and any LC Application, shall be fully subrogated to the rights and remedies of each beneficiary under such Letter of Credit whose claims against Borrowers have been discharged with the proceeds of such Letter of Credit. Whether or not a Borrower submits any Notice of Borrowing to Agent, Borrowers shall be deemed to have requested from Lenders a Borrowing of Base Rate Loans in an amount

-43-

necessary to pay to Issuing Bank all amounts due Issuing Bank on any Reimbursement Date and each Lender agrees to fund its Pro Rata share of such Borrowing whether or not any Default or Event of Default has occurred or exists, the Commitments have been terminated, the funding of the Borrowing would result in (or increase the amount of) any Out-of-Formula Condition, or any of the conditions set forth in **Section 11** hereof are not satisfied.

(iv) Borrowers assume all risks of the acts, omissions or misuses of any Letter of Credit by the beneficiary thereof. The obligation of Borrowers to reimburse Issuing Bank for any payment made by Issuing Bank under a Letter of Credit shall be absolute, unconditional, irrevocable and joint and several and shall be paid without regard to any lack of validity or enforceability of any Letter of Credit or the existence of any claim, setoff, defense or other right which Borrowers may have at any time against a beneficiary of any Letter of Credit. In connection with the issuance of any documentary Letter of Credit, none of Agent, Issuing Bank or any Lender shall be responsible for the existence, character, quality, quantity, condition, packing, value or delivery of any goods purported to be represented by any Documents; any differences or variation in the character, quality, quantity, condition, packing, value or delivery of any goods from that expressed in the Documents; the validity, sufficiency or genuineness of any Documents or of any endorsements thereon, even if such Documents should in fact prove to be in any or all respects invalid, insufficient, fraudulent or forged; the time, place, manner or order in which shipment of goods is made; partial or incomplete shipment of, or failure or omission to ship, any or all of the goods referred to in a documentary Letter of Credit or Documents applicable thereto; or any breach of contract between the shipper or vendors and a Borrower. The rights, remedies, powers and privileges of Issuing Bank under this Agreement with respect to Letters of Credit shall be in addition to, and cumulative with, all rights, remedies, powers and privileges of Issuing Bank under any of the LC Documents. Nothing herein shall be deemed to release Issuing Bank from any liability or obligation that it may have in respect to any Letter of Credit arising out of and directly resulting from its own gross negligence or willful misconduct.

(v) No Letter of Credit shall be extended or amended in any respect that is not solely ministerial, unless all of the LC Conditions are met as though a new Letter of Credit were being requested and issued. With respect to any Letter of Credit that contains any "evergreen" or automatic renewal provision, each Lender shall be deemed to have consented to any such extension or renewal, unless any such Lender shall have provided to Agent written notice that it declines to consent to any such extension or renewal at least 30 days prior to the date on which the Issuing Bank is entitled to decline to extend or renew the Letter of Credit. If all of the LC Conditions are met and no Default or Event of Default exists, no Lender shall have the right to decline to consent to any such extension or renewal.

(vi) Unless otherwise provided in any of the LC Documents, each LC Application and each standby Letter of Credit shall be subject to the Uniform Customs and Practice for Documentary Credits (1993 Revision), International Chamber of Commerce No. 500, and any amendments or revisions thereto.

Each letter of credit issued or deemed issued and outstanding under the Existing Loan Agreement on the Closing Date, including any extension or renewal thereof (each, an "Existing Letter of Credit") shall constitute a "Letter of Credit" for all purposes of this Agreement, issued, for purposes of this **Section 2.3**, on the Closing Date.

2.3.2. Participations.

-44-

(i) Immediately upon the issuance of any Letter of Credit, each Lender shall be deemed to have irrevocably and unconditionally purchased and received from Issuing Bank, without recourse or warranty, an undivided interest and participation equal to the Pro Rata share of such Lender (a "Participating Lender") in all LC Obligations arising in connection with such Letter of Credit, but in no event greater than an amount which, when added to such Lender's Pro Rata share of all Revolver Loans and LC Obligations then outstanding, exceeds such Lender's Revolver Commitment; <u>provided</u>, <u>however</u>, that if Issuing Bank shall have received written notice from a Lender on or before the Business Day immediately prior to the date of Issuing Bank's issuance of a Letter of Credit that one or more of the conditions set forth in **Section 11** or **Section 2.3.1** has not been satisfied, Issuing Bank shall have no obligation to issue the requested Letter of Credit or any other Letter of Credit until such notice is withdrawn in writing by that Lender or until the Required Lenders shall have effectively waived such condition in accordance with this Agreement. In no event shall Issuing Bank be deemed to have notice or knowledge of any existence of any Default or Event of Default or the failure of any conditions in **Sections 11** or **2.3.1** hereof to be satisfied prior to its receipt of such notice from a Lender.

(ii) If Issuing Bank makes any payment under a Letter of Credit and Borrowers do not repay or cause to be repaid the amount of such payment on the Reimbursement Date, Issuing Bank shall promptly notify Agent, which shall promptly notify each Participating Lender, of such payment and each Participating Lender shall promptly (and in any event within 1 Business Day after its receipt of notice from Agent) and unconditionally pay to Agent, for the account of Issuing Bank, in immediately available funds, the amount of such Participating Lender's Pro Rata share of such payment, and Agent shall promptly pay such amounts to Issuing Bank. If a Participating Lender does not make its Pro Rata share of the amount of such payment available to Agent on a timely basis as herein provided, such Participating Lender agrees to pay to Agent for the account of Issuing Bank, forthwith on demand, such amount together with interest thereon at the Federal Funds Rate until paid. The failure of any Participating Lender to make available to Agent for the account of Issuing Bank such Participating Lender's Pro Rata share of the LC Obligations. No Participating Lender shall be responsible for the failure of any other Participating Lender of any other Participating Lender of any other Participating Lender to make available to Agent its Pro Rata share of the LC Obligations on the date such payment is to be made.

(iii) Whenever Issuing Bank receives a payment on account of the LC Obligations, including any interest thereon, as to which Agent has previously received payments from any Participating Lender for the account of Issuing Bank, Issuing Bank shall promptly pay to each Participating Lender which has funded its participating interest therein, through Agent, in immediately available funds, an amount equal to such Participating Lender's Pro Rata share thereof.

(iv) The obligation of each Participating Lender to make payments to Agent for the account of Issuing Bank in connection with Issuing Bank's payment under a Letter of Credit shall be absolute, unconditional and irrevocable, not subject to any counterclaim, setoff, qualification or exception whatsoever, and shall be made in accordance with the terms and conditions of this Agreement under all circumstances and irrespective of whether or not Borrowers may assert or have any claim for any lack of validity or unenforceability of this Agreement or any of the other Loan Documents; the existence of any Default or Event of Default; any draft, certificate or other document presented under a Letter of Credit having been determined to be forged, fraudulent, invalid or insufficient in any respect or any statement therein being untrue or inaccurate in any respect; the existence of any setoff or defense any Obligor may have with respect to any of the Obligations; or the termination of the Commitments.

-45-

(v) Neither Issuing Bank nor any of its officers, directors, employees or agents shall be liable to any Participating Lender for any action taken or omitted to be taken under or in connection with any of the LC Documents except as a result of actual gross negligence or willful misconduct on the part of Issuing Bank. Issuing Bank does not assume any responsibility for any failure or delay in performance or breach by a Borrower or any other Person of its obligations under any of the LC Documents. Issuing Bank does not make to Participating Lenders any express or implied warranty, representation or guaranty with respect to the Collateral, the LC Documents, or any Obligor. Issuing Bank shall not be responsible to any Participating Lender for any recitals, statements, information, representations or warranties contained in, or for the execution, validity, genuineness, effectiveness or enforceability of or any of the LC Documents; the validity, genuineness, enforceability, collectibility, value or sufficiency of any of the Collateral or the perfection of any Lien therein; or the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any Obligor or any Account Debtor. In connection with its administration of and enforcement of rights or remedies under any of the LC Documents, Issuing Bank shall be entitled to act, and shall be fully protected in acting upon, any certification, notice or other communication in whatever form believed by Issuing Bank, in good faith, to be genuine and correct and to have been signed, sent or made by a proper Person. Issuing Bank may consult with and employ legal counsel, accountants and other experts and to advise it concerning its rights, powers and privileges under the LC Documents and shall be entitled to act upon, and shall be fully protected in any action taken in good faith reliance upon, any advice given by such experts. Issuing Bank may employ agents and attorneys-in-fact in connection with any matter relating to the LC Documents and shall not be liable for the negligence, default or misconduct of any such agents or attorneysin-fact selected by Issuing Bank with reasonable care. Issuing Bank shall not have any liability to any Participating Lender by reason of Issuing Bank's refraining to take any action under any of the LC Documents without having first received written instructions from the Required Lenders to take such action.

(vi) Upon the request of any Participating Lender, Issuing Bank shall furnish to such Participating Lender copies (to the extent then available to Issuing Bank) of each outstanding Letter of Credit and related LC Documents as may be in the possession of Issuing Bank and reasonably requested from time to time by such Participating Lender.

2.3.3. <u>Cash Collateral Account</u>. If any LC Obligations, whether or not then due or payable, shall for any reason be outstanding (i) at any time that an Event of Default exists, (ii) on any date an Out-of-Formula Condition exists, (iii) on or at any time after the Commitment Termination Date, or (iv) on the day that is 30 days prior to the last Business Day of the Term, then Borrowers shall, on Issuing Bank's or Agent's request, forthwith pay to Issuing Bank the amount of any LC Obligations that are then due and payable and shall, upon the occurrence of any of the events described in clauses (i), (iii) and (iv) hereinabove, Cash Collateralize all outstanding Letters of Credit. If notwithstanding the occurrence of one or more of the events described in clauses (i), (iii) and (iv) in the immediately preceding sentence Borrowers fail to Cash Collateralize any outstanding Letters of Credit on the first Business Day following Agent's or Issuing Bank's demand therefor, Lenders may (and shall upon direction of Agent) advance such amount as Revolver Loans (whether or not the Commitment Termination Date has occurred or an Out-of-Formula Condition is created thereby). Such cash (together with any interest accrued thereon) shall be held by Agent in the Cash Collateral Account and may be invested, in Agent's discretion, in Cash Equivalents. Each Borrower hereby pledges to Agent and grants to Agent a security interest in, for the benefit of Agent in such capacity and for the Pro Rata benefit of Lenders, all Cash Collateral held in the Cash Collateral Account from time to time and all proceeds thereof, as security for the payment of all Obligations (including LC Obligations), whether or not then due or payable. From time to time after cash is deposited in the Cash Collateral Account, Agent may (and shall upon the direction of the Required Lenders) apply Cash Collateral then held in the Cash Collateral Account to the payment of any amounts,

-46-

in such order as Agent may elect, as shall be or shall become due and payable by Borrowers to Issuing Bank, Agent or any Lender with respect to the LC Obligations. Neither Borrowers nor any other Person claiming by, through or under or on behalf of Borrowers shall have any right to withdraw any of the Cash Collateral held in the Cash Collateral Account, including any accrued interest, <u>provided that</u> (A) upon termination or expiration of all Letters of Credit and the payment and satisfaction of all of the LC Obligations, or (B) if Agent has confirmed in writing to Borrowers that the events described in clauses (i) or (ii) of the initial sentence of this **Section 2.3.3** which required the Cash Collateralization of outstanding Letters of Credit, no longer exist, then any Cash Collateral remaining in the Cash Collateral Account shall be returned to Borrowers unless an Event of Default then exists (in which event Agent may (and shall upon the direction of the Required Lenders) apply such Cash Collateral to the payment of any other Obligations outstanding in accordance with the provisions of **Section 5.6**, with any surplus to be turned over to Borrowers).

2.3.4. Indemnifications.

(i) In addition to any other indemnity which Borrowers may have to any Indemnitees under any of the other Loan Documents and without limiting such other indemnification provisions, each Borrower hereby agrees to indemnify and defend each of the Indemnitees and to hold each of the Indemnitees harmless from and against any and all Indemnified Claims which any Indemnitee may (other than solely as the result of its own gross negligence or willful misconduct or its breach of any of the Loan Documents) incur or be subject to as a consequence, directly or indirectly, of (a) the issuance of, payment or failure to pay or any performance or failure to perform under any Letter of Credit, (b) any suit, investigation or proceeding as to which Agent or any Lender is or may become a party to as a consequence, directly or indirectly, of the issuance of any Letter of Credit or the payment or failure to pay thereunder or (c) Issuing Bank following any instructions of a Borrower with respect to any Letter of Credit or any Document received by Issuing Bank with reference to any Letter of Credit; provided that no Borrower shall be liable to any Indemnitee for any of the foregoing to the extent that they result solely from the willful misconduct or gross negligence of such Indemnitee.

(ii) Each Participating Lender agrees to indemnify and defend each of the Issuing Bank Indemnitees (to the extent the Issuing Bank Indemnitees are not reimbursed by Borrowers or any other Obligor, but without limiting the indemnification obligations of Borrowers under this Agreement), to the extent of such Lender's Pro Rata share of the Revolver Commitments, from and against any and all Indemnified Claims which may be imposed on, incurred by or asserted against any of the Issuing Bank Indemnitees in any way related to or arising out of Issuing Bank's administration or enforcement of rights or remedies under any of the LC Documents or any of the transactions contemplated thereby (including costs and expenses which Borrowers are obligated to pay under **Section 15.2** hereof), <u>provided that</u> no Participating Lender shall be liable to any of the Issuing Bank Indemnitees for any of the foregoing to the extent that they result solely from the willful misconduct or gross negligence of such Issuing Bank Indemnitees.

SECTION 3. INTEREST, FEES AND CHARGES

3.1. Interest.

3.1.1. <u>Rates of Interest</u>. Borrowers agree to pay interest in respect of all unpaid principal amounts of the Revolver Loans from the respective dates such principal amounts are advanced until paid (whether at stated maturity, on acceleration or otherwise) at a rate per annum equal to the applicable rate indicated below:

(i) for Revolver Loans made or outstanding as Base Rate Loans, the Applicable Margin plus the Base Rate in effect from time to time; or

(ii) for Revolver Loans made or outstanding as LIBOR Loans, the Applicable Margin <u>plus</u> the relevant Adjusted LIBOR Rate for the applicable Interest Period selected by Borrowers in conformity with this Agreement.

Upon determining the Adjusted LIBOR Rate for any Interest Period requested by Borrowers, Agent shall promptly notify Borrowers thereof by telephone and, if so requested by Borrowers, confirm the same in writing. Such determination shall, absent manifest error, be final, conclusive and binding on all parties and for all purposes. The applicable rate of interest for all Loans (or portions thereof) bearing interest based upon the Base Rate shall be increased or decreased, as the case may be, by an amount equal to any increase or decrease in the Base Rate, with such adjustments to be effective as of the opening of business on the day that any such change in the Base Rate becomes effective. Interest on each Loan shall accrue from and including the date on which such Loan is made, converted to a Loan of another Type or continued as a LIBOR Loan to (but excluding) the date of any repayment thereof; provided, however, that, if a Loan is repaid on the same day made, one day's interest shall be paid on such Loan. The Base Rate on the date hereof is 6.00% per annum and, therefore, the rate of interest in effect hereunder on the date hereof, expressed in simple interest terms, is 6.00% per annum with respect to any portion of the Revolver Loans bearing interest as a Base Rate Loan.

3.1.2. Conversions and Continuations.

(i) Borrowers may on any Business Day, subject to the giving of a proper Notice of Conversion/Continuation as hereinafter described, elect (A) to continue all or any part of a LIBOR Loan by selecting a new Interest Period therefor, to commence on the last day of the immediately preceding Interest Period, or (B) to convert all or any part of a Loan of one Type into a Loan of another Type; provided, however, during the period that any Default or Event of Default exists, Agent may (and shall at the direction of the Required Lenders) declare that no Loan may be made or continued as or converted into a LIBOR Loan. Any conversion of a LIBOR Loan into a Base Rate Loan shall be made on the last day of the Interest Period for such LIBOR Loan. Any conversion or continuation made with respect to less than the entire outstanding balance of the Loans must be allocated among Lenders on a Pro Rata basis, and the Interest Period for Loans converted into or continued as LIBOR Loans shall be coterminous for each Lender.

(ii) Whenever Borrowers desire to convert or continue Loans under **Section 3.1.2(i)**, Borrower Agent shall give Agent written notice (or telephonic notice promptly confirmed in writing) substantially in the form of **Exhibit C** (a "Notice of Conversion/Continuation"), signed by an authorized officer of Borrower Agent, or in other form required by Agent, at least 1 Business Day before the requested conversion date, in the case of a conversion into Base Rate Loans, and at least 3 Business Days before the requested conversion or continuation of LIBOR Loans. Promptly after receipt of a Notice of Conversion/Continuation, Agent shall notify each Lender in writing of the proposed conversion or continuation. Each such Notice of Conversion or continuation (which shall be irrevocable and shall specify the aggregate principal amount of the Loans to be converted or continued, the date of such conversion or continuation (which shall be a Business Day) and whether the Loans are being converted into or continued as LIBOR Loans (and, if so, the duration of the Interest Period to be applicable thereto and, in the absence of any specification by Borrowers of the Interest Period, an Interest Period of one month will be deemed to be specified) or Base Rate Loans. If, upon the expiration of any Interest Period in respect of any LIBOR

-48-

Loans, Borrowers shall have failed to deliver the Notice of Conversion/Continuation, Borrowers shall be deemed to have elected to convert such LIBOR Loans to Base Rate Loans.

3.1.3. <u>Interest Periods</u>. In connection with the making or continuation of, or conversion into, each Borrowing of LIBOR Loans, Borrowers shall select an interest period (each an "Interest Period") to be applicable to such LIBOR Loan, which interest period shall commence on the date such LIBOR Loan is made and shall end on a numerically corresponding day in the first, second, third or sixth month thereafter; <u>provided</u>, <u>however</u>, that:

(i) the initial Interest Period for a LIBOR Loan shall commence on the date of such Borrowing (including the date of any conversion from a Loan of another Type) and each Interest Period occurring thereafter in respect of such Revolver Loan shall commence on the date on which the next preceding Interest Period expires;

(ii) if any Interest Period would otherwise expire on a day that is not a Business Day, such Interest Period shall expire on the next succeeding Business Day, <u>provided</u> that, if any Interest Period in respect of LIBOR Loans would otherwise expire on a day that is not a Business Day but is a day of the month after which no further Business Day occurs in such month, such Interest Period shall expire on the immediately preceding Business Day;

(iii) any Interest Period that begins on a day for which there is no numerically corresponding day in the calendar month at the end of such Interest Period shall expire on the last Business Day of such calendar month; and

(v) no Interest Period shall extend beyond the last day of the Term.

3.1.4. <u>Interest Rate Not Ascertainable</u>. If Agent shall determine (which determination shall, absent manifest error, be final, conclusive and binding upon all parties) that on any date for determining the Adjusted LIBOR Rate for any Interest Period, by reason of any changes arising after the date of this Agreement affecting the London interbank market or any Lender's position in such market, adequate and fair means do not exist for ascertaining the applicable interest rate on the basis provided for in the definition of Adjusted LIBOR Rate, then, and in any such event, Agent shall forthwith give notice (by telephone promptly confirmed in writing) to Borrowers of such determination. Until Agent notifies Borrowers that the circumstances giving rise to the suspension described herein no longer exist, the obligation of Lenders to make LIBOR Loans shall be suspended, and such affected Loans then outstanding shall, at the end of the then applicable Interest Period or at such earlier time as may be required by Applicable Law, bear the same interest as Base Rate Loans.

3.1.5. <u>Default Rate of Interest</u>. Borrowers shall pay interest at a rate per annum equal to the Default Rate (i) with respect to the principal amount of any portion of the Obligations (and, to the extent permitted by Applicable Law, all past due interest) that is not paid on the due date thereof (whether due at stated maturity, on demand, upon acceleration or otherwise) until Full Payment; (ii) with respect to the principal amount of all of the Obligations (and, to the extent permitted by Applicable Law, all past due interest) upon the earlier to occur of (x) Borrower Agent's receipt of notice from Agent of the Required Lenders' election to charge the Default Rate based upon the existence of any Event of Default (which notice Agent shall send only with the consent or at the direction of the Required Lenders), whether or not acceleration or demand for payment of the Obligations has been made, or (y) the commencement by or against any Borrower of an Insolvency Proceeding whether or not under the circumstances described in clauses (i) or (ii) hereof Lenders elect to accelerate the maturity or demand payment of any of the Obligations; and (iii) with respect to the principal amount of any Out-of-Formula Loans (unless otherwise agreed in writing by the Required Lenders), whether or not demand for payment thereof has

-49-

been made by Agent. To the fullest extent permitted by Applicable Law, the Default Rate shall apply and accrue on any judgment entered with respect to any of the Obligations and to the unpaid principal amount of the Obligations during any Insolvency Proceeding of a Borrower. Each Borrower acknowledges that the cost and expense to Agent and each Lender attendant upon the occurrence of an Event of Default are difficult to ascertain or estimate and that the Default Rate is a fair and reasonable estimate to compensate Agent and Lenders for such added cost and expense. Interest accrued at the Default Rate shall be due and payable on demand.

3.2. <u>Fees</u>. In consideration of Lenders' establishment of the Commitments in favor of Borrowers, and Agent's agreement to serve as collateral and administrative agent hereunder, Borrowers agree to pay the following fees:

3.2.1. <u>Unused Line Fee</u>. Borrowers shall pay to Agent for the Pro Rata benefit of Lenders a fee equal to (i) 0.375% per annum of the amount by which the Average Revolver Loan Balance for any month (or portion thereof that the Commitments are in effect) is less than the aggregate amount of the Revolver Commitments; <u>provided that</u> if the Average Revolver Loan Balance for the immediately preceding Fiscal Quarter (or portion thereof) is greater than 50% of the aggregate amount of the Revolver Commitments, then such fee shall be 0.250% per annum of the amount by which the Average Revolver Loan Balance for such month (or portion thereof) is less than the aggregate amount of the Revolver Commitments, in each case such fee to be paid on the first day of the following month, <u>provided that</u>, if the Commitments are terminated on a day other than the first day of a month, then any such fee payable for the month in which termination occurs shall be paid on the effective date of such termination.

3.2.2. LC Facility Fees. Borrowers shall pay: (a)(i) to Agent, for the Pro Rata account of each Lender for all Letters of Credit, the Applicable Margin in effect for Revolver Loans that are LIBOR Loans on a per annum basis based on the average amount available to be drawn under Letters of Credit outstanding and all Letters of Credit that are paid or expire during the period of measurement (or, with respect to each Letter of Credit that is secured by cash deposited by Borrowers into the Cash Collateral Account on terms satisfactory to Agent, 0.65% on a per annum basis based on the average amount available to be drawn under all such cash-collateralized Letters of Credit outstanding and all such cash-collateralized Letters of Credit that are paid or expire during the period of measurement), in each case payable monthly, in arrears, on the first Business Day of the following month; (ii) to Agent, for its own account a Letter of Credit that are paid or expire during the average amount available to be drawn under all Letters of Credit of measurement, payable monthly, in arrears, on the first Business Day of the following month; (ii) to Agent, for its own account a Letters of Credit that are paid or expire during the period of measurement, payable monthly, in arrears, on the first Business Day of the following month; and (iii) to Issuing Bank for its own account all customary charges associated with the issuance, amending, negotiating, payment, processing and administration of all Letters of Credit.

3.2.3. <u>Audit and Appraisal Fees and Expenses</u>. Borrowers shall reimburse Agent for all reasonable costs and expenses incurred by Agent (including standard fees charged by Agent's internal appraisal department) in connection with (i) examinations and reviews of any Obligor's books and records and such other matters pertaining to any Obligor or any Collateral as Agent shall deem appropriate in the exercise of its reasonable credit judgment, up to three (3) times per Loan Year, unless a Default or Event of Default exists (in which event, there shall be no limit on the number of examinations and reviews for which Borrowers shall be obligated to reimburse Agent) and, in each case, shall pay to Agent the standard amount charged by Agent per day (\$850 per day as of the Closing Date) for each day that an employee or agent of Agent shall be engaged in an examination or review of any Obligor's books and records, (ii) appraisals of Equipment no more frequently than three (3) times per Loan Year (one of which may be a full physical appraisal and the other two of which shall be desktop appraisals performed by employees or agents of Agent), and (iii) appraisals of Eligible Real Estate no more frequently than one (1) time per Loan Year; provided, that, in the case of each of clauses (ii) and (iii), if a Default or

-50-

Event of Default exists, there shall be no limit on the number or type of appraisals for which Borrowers shall be obligated to reimburse Agent.

3.2.4. Other Fees. Borrowers shall pay to Agent the fees in the amounts and on the dates as provided in the Fee Letter.

3.2.5. <u>General Provisions</u>. All fees shall be fully earned by the identified recipient thereof pursuant to the foregoing provisions of this Agreement on the due date thereof (and, in the case of Letters of Credit, upon each issuance, renewal or extension of such Letter of Credit) and, except as otherwise set forth herein or required by Applicable Law, shall not be subject to rebate, refund or proration. All fees provided for in **Section 3.2** are and shall be deemed to be compensation for services and are not, and shall not be deemed to be, interest or any other charge for the use, forbearance or detention of money.

3.3. <u>Computation of Interest and Fees</u>. All fees and other charges provided for in this Agreement that are calculated as a per annum percentage of any amount and all interest shall be calculated daily and shall be computed on the actual number of days elapsed over a year of 360 days. For purposes of computing interest and other charges hereunder, each Payment Item and other form of payment received by Agent shall be deemed applied by Agent on account of the Obligations (subject to final payment of such items) on the first Business Day after the Business Day on which Agent receives such Payment Item in the Payment Account, and Agent shall be deemed to have received such Payment Item on the date specified in Section 5.7 hereof. Each determination by Agent of interest and fees hereunder shall be made in good faith and, except for manifest error, shall be presumptive evidence of the correctness of such interest and fees.

3.4. Reimbursement Obligations.

3.4.1. Borrowers shall reimburse Agent and (during any period that an Event of Default exists) each Lender for all legal, accounting, appraisal, consulting and other fees and expenses incurred by Agent or any Lender in connection with: (i) the negotiation and preparation of any of the Loan Documents, any amendment or modification thereto, any waiver of any Default or Event of Default thereunder, or any restructuring or forbearance with respect thereto; (ii) the administration of the Loan Documents and the transactions contemplated thereby; (iii) action taken to perfect or maintain the perfection or priority of any of Agent's Liens with respect to any of the Collateral; (iv) any inspection of or audits conducted by Agent with respect to any Borrower's books and records or any of the Collateral; (v) any effort by Agent to verify, protect, appraise, preserve, or restore any of the Collateral or to collect, sell, liquidate or otherwise dispose of or realize upon any of the Collateral; (vi) any litigation, contest, dispute, suit, proceeding or action (whether instituted by or against Agent, any Lender, any Obligor or any of the Loan Documents or the validity, perfection or priority of any of Agent's Liens thereon), any of the Loan Documents or the validity, allowance or amount of any of the Obligations; (vii) the protection or enforcement or any rights or remedies of Agent against any Obligor or Account Debtor or to enforce collection of any of the Collateral. All amounts chargeable to Borrowers under this **Section 3.4** shall constitute Obligations that are secured by all of the Collateral to the extent and in the manner provided in **Section 8** hereof or in any of the other Loan Documents. The foregoing shall be in addition to, and shall not be construed to limit, any other provision of any of the Loan Documents

-51-

regarding the reimbursement by Borrowers of costs, expenses or liabilities suffered or incurred by Agent or any Lender.

3.4.2. If at any time Agent or (with the prior consent of Agent) any Lender shall agree to indemnify any Person against losses or damages that such Person may suffer or incur in its dealings or transactions with Borrowers, or shall guarantee or otherwise assure payment of any liability or obligation of Borrowers to such Person, or otherwise shall provide assurances of Borrowers' payment or performance under any agreement with such Person, including indemnities, guaranties or other assurances of payment or performance given by Agent or any Lender with respect to Banking Relationship Debt, then the Contingent Obligation of Agent or any Lender providing any such indemnity, guaranty or other assurance of payment or performance, together with any payment made or liability incurred by Agent or any Lender in connection therewith, shall constitute Obligations that are secured by the Collateral and Borrowers shall repay, on demand, any amount so paid or any liability incurred by Agent or any Lender in connection with any such indemnity, guaranty or assurance. Nothing herein shall be construed to impose upon Agent or any Lender any obligation to provide any such indemnity, guaranty or assurance. The foregoing agreement of Borrowers shall apply whether or not such indemnity, guaranty or assurance is in writing or oral and regardless of any Borrower's knowledge of the existence thereof, shall survive termination of the Commitments and Full Payment of the Obligations and any other provisions of the Loan Documents regarding reimbursement or indemnification by Borrowers of costs, expenses or liabilities suffered or incurred by Agent or any Lender. Agent will endeavor to provide notice to Borrowers of any such indemnity, guaranty or assurance if the applicable Borrower is not otherwise a party thereto; <u>provided, that</u>, Agent's failure to so notify any Borrower shall not result in any liability to Agent or any Lender or in any way condition Agent's or any Lender's right to repayment or reimbursement hereunder.

3.5. <u>Bank Charges</u>. Borrowers shall pay to Agent, **on demand**, any and all fees, costs or expenses which Agent pays to a bank or other similar institution (including any fees paid by Agent or any Lender to any Participant) arising out of or in connection with (i) the forwarding to a Borrower or any other Person on behalf of Borrower by Agent of proceeds of Loans made by Lenders to a Borrower pursuant to this Agreement and (ii) the depositing for collection by Agent of any Payment Item received or delivered to Agent on account of the Obligations. Each Borrower acknowledges and agrees that Agent may charge such costs, fees and expenses to Borrowers based upon Agent's good faith estimate of such costs, fees and expenses as they are incurred by Agent.

3.6. <u>Illegality</u>. Notwithstanding anything to the contrary contained elsewhere in this Agreement, if (i) any change in any law or regulation or in the interpretation thereof by any Governmental Authority charged with the administration thereof shall make it unlawful for a Lender to make or maintain a LIBOR Loan or to give effect to its obligations as contemplated hereby with respect to a LIBOR Loan or (ii) at any time such Lender determines that the making or continuance of any LIBOR Loan has become impracticable as a result of a contingency occurring after the date hereof which adversely affects the London interbank market or the position of such Lender in such market, then such Lender shall give after such determination Agent and Borrowers notice thereof and may thereafter (1) declare that LIBOR Loans will not thereafter be made by such Lender, whereupon any request by a Borrower for a LIBOR Loan from such Lender shall be deemed a request for a Base Rate Loan unless such Lender's declaration shall be subsequently withdrawn (which declaration shall be withdrawn promptly after the cessation of the circumstances described in clause (i) or (ii) above); and (2) require that all outstanding LIBOR Loans made by such Lender be converted to Base Rate Loans, under the circumstances of clause (i) or (ii) of this **Section 3.6** insofar as such Lender determines the continuance of LIBOR Loans to be impracticable, in which event all such LIBOR Loans of such Lender shall be

-52-

converted automatically to Base Rate Loans as of the date of any Borrower's receipt of the aforesaid notice from such Lender.

3.7. <u>Increased Costs</u>. If, by reason of (a) the introduction after the date hereof of or any change (including any change by way of imposition or increase of Statutory Reserves or other reserve requirements) in or in the interpretation of any law or regulation, or (b) the compliance with any guideline or request from any central bank or other Governmental Authority or quasi-Governmental Authority exercising control over banks or financial institutions generally (whether or not having the force of law):

(i) any Lender shall be subject after the date hereof to any Tax, duty or other charge with respect to any LIBOR Loan or Letter of Credit or its obligation to make LIBOR Loans or to issue Letters of Credit or participate in the LC Obligations arising from the issuance of Letters of Credit, or a change shall result in the basis of taxation of payment to any Lender of the principal of or interest on its LIBOR Loans or its obligation to make LIBOR Loans, issue Letters of Credit or participate in the LC Obligations arising from the issuance of Letters of Credit (except for changes in the rate of Tax on the overall net income or gross receipts or franchise tax of such Lender imposed by the jurisdiction in which such Lender's principal executive office is located); or

(ii) any reserve (including any imposed by the Board of Governors), special deposits or similar requirement against assets of, deposits with or for the account of, or credit extended by, any Lender shall be imposed or deemed applicable or any other condition affecting its LIBOR Loans or Letters of Credit or its obligation to make LIBOR Loans or to issue Letters of Credit or participate in the LC Obligations arising from the issuance of Letters of Credit shall be imposed on such Lender or the London interbank market;

and as a result thereof there shall be any increase in the cost to such Lender of agreeing to make or making, funding or maintaining LIBOR Loans or issuing Letters of Credit (except to the extent already included in the determination of the applicable Adjusted LIBOR Rate for LIBOR Loans), or there shall be a reduction in the amount received or receivable by such Lender, then such Lender shall, promptly after determining the existence or amount of any such increased costs for which such Lender seeks payment hereunder, give any Borrower notice thereof and Borrowers shall from time to time, upon written notice from and demand by such Lender (with a copy of such notice and demand to Agent), pay to Agent for the account of such Lender, within 5 Business Days after the date specified in such notice and demand, an additional amount sufficient to indemnify such Lender against such increased costs. A certificate as to the amount of such increased costs, submitted to Borrowers by such Lender, shall be final, conclusive and binding for all purposes, absent manifest error.

If any Lender shall advise Agent at any time that, because of the circumstances described hereinabove in this **Section 3.7** or any other circumstances arising after the date of this Agreement affecting such Lender or the London interbank market or such Lender's or BofA's position in such market, the Adjusted LIBOR Rate, as determined by Agent, will not adequately and fairly reflect the cost to such Lender of funding LIBOR Loans or issuing Letters of Credit, then, and in any such event:

(i) Agent shall forthwith give notice (by telephone confirmed promptly in writing) to Borrowers and Lenders of such event;

(ii) Borrowers' right to request and such Lender's obligation to make LIBOR Loans or to issue Letters of Credit or participate in the LC Obligations arising from the issuance of Letters of Credit shall be immediately suspended and Borrowers' right to continue a LIBOR Loan as such beyond the then applicable Interest Period or to request a Letter of Credit shall also

-53-

be suspended, until each condition giving rise to such suspension no longer exists; and

(iii) such Lender shall make a Base Rate Loan as part of the requested Borrowing of LIBOR Loans, which Base Rate Loan shall, for all purposes, be considered part of such Borrowing.

For purposes of this **Section 3.7**, all references to a Lender shall be deemed to include any bank holding company or bank parent of such Lender. If any Lender provides notice that, due to the circumstances described in this **Section 3.7**, the Adjusted LIBOR Rate will not adequately and fairly reflect the cost to such Lender of funding LIBOR Loans or participating in LC Obligations arising from the issuance of Letters of Credit, then such Lender may be replaced pursuant to the provisions of **Section 13.17**.

3.8. <u>Capital Adequacy</u>. If any Lender determines that (i) the introduction after the date hereof of any Capital Adequacy Regulation, (ii) any change in any Capital Adequacy Regulation, (iii) any change in the interpretation or administration of any Capital Adequacy Regulation by any central bank or other Governmental Authority charged with the interpretation or administration thereof, or (iv) compliance by such Lender or any corporation or other entity controlling such Lender with any Capital Adequacy Regulation, affects or would affect the amount of capital required or expected to be maintained by such Lender or any corporation or other entity controlling such Lender and (taking into consideration such Lender's or such corporation's or other entity's policies with respect to capital adequacy and such Lender's desired return on capital) determines that the amount of such capital is increased as a consequence of its Commitments, loans, credits or obligations under this Agreement, then: (a) Agent shall promptly, after its receipt of a certificate from such Lender setting forth such Lender's determination of such occurrence, give notice thereof to Borrowers and Lenders; and (b) Borrowers shall pay to Agent, for the account of such Lender for such reduction. A certificate of such Lender claiming entitlement to compensation as set forth above will be conclusive in the absence of manifest error. Such certificate will set forth the nature of the occurrence giving rise to such compensation, the additional amount or amounts to be paid to such Lender (including the basis for such Lender's determination of such amount), and the method by which such amounts were determined. In determining such amount, such Lender may use any reasonable averaging and attribution method. For purposes of this **Section 3.8** all references to a Lender shall be deemed to include any bank holding company or bank parent of such Lender.

3.9. <u>Mitigation</u>. Each Lender agrees that, with reasonable promptness after such Lender becomes aware that such Lender is entitled to receive payments under **Section 3.6, 3.7 or 3.8**, or is or has become subject to U.S. withholding taxes payable by any Borrower in respect of its Loans, it will, to the extent not inconsistent with any internal policy of such Lender or any applicable legal or regulatory restriction, (i) use all reasonable efforts to make, fund or maintain the Commitment of such Lender or the Loans of such Lender through another lending office of such Lender, or (ii) take such other reasonable measures, if, as a result thereof, the circumstances which would relieve Borrowers from their obligations to pay such additional amounts (or reduce the amount of such payments), or such withholding taxes would be reduced, and if the making, funding or maintaining of such Commitment or Loans through such other lending office or in accordance with such other measures, as the case may be, would not otherwise adversely affect such Commitment or Loans or the interests of such Lender.

3.10. <u>Funding Losses</u>. If for any reason (other than due to a default by a Lender or as a result of a Lender's refusal to honor a LIBOR Loan request due to circumstances described in this Agreement) a Borrowing of, or conversion to or continuation of, LIBOR Loans does not occur on the date specified therefor in a Notice of Borrowing or Notice of Conversion/Continuation (whether or not withdrawn), or if

-54-

any repayment (including any conversions pursuant to **Section 3.1.2**) of any of its LIBOR Loans occurs on a date that is not the last day of an Interest Period applicable thereto, or if for any reason Borrowers default in their obligation to repay LIBOR Loans when required by the terms of this Agreement, then Borrowers shall pay to Agent an administrative fee of \$150 and to each Lender an amount equal to all losses and expenses which such Lender may sustain or incur as a consequence thereof, including any such loss or expense arising from the liquidation or redeployment of funds obtained by it to maintain its LIBOR Loans or from fees payable to terminate the deposits from which such funds were obtained. Borrowers shall pay all such amounts due to any Lender upon presentation by such Lender of a statement setting forth the amount and such Lender's calculation thereof, which statement shall be deemed true and correct absent manifest error. For purposes of this **Section 3.10**, all references to a Lender shall be deemed to include any bank holding company or bank parent of such Lender.

3.11. Maximum Interest. Regardless of any provision contained in any of the Loan Documents, in no contingency or event whatsoever shall the aggregate of all amounts that are contracted for, charged or received by Agent and Lenders pursuant to the terms of this Agreement or any of the other Loan Documents and that are deemed interest under Applicable Law exceed the highest rate permissible under any Applicable Law. No agreements, conditions, provisions or stipulations contained in this Agreement or any of the other Loan Documents or the exercise by Agent of the right to accelerate the payment or the maturity of all or any portion of the Obligations, or the exercise of any option whatsoever contained in any of the Loan Documents, or the prepayment by Borrowers of any of the Obligations, or the occurrence of any contingency whatsoever, shall entitle Agent or any Lender to charge or receive in any event, interest or any charges, amounts, premiums or fees deemed interest by Applicable Law (such interest, charges, amounts, premiums and fees referred to herein collectively as "Interest") in excess of the Maximum Rate and in no event shall Borrowers be obligated to pay Interest exceeding such Maximum Rate, and all agreements, conditions or stipulations, if any, which may in any event or contingency whatsoever operate to bind, obligate or compel Borrowers to pay Interest exceeding the Maximum Rate shall be without binding force or effect, at law or in equity, to the extent only of the excess of Interest over such Maximum Rate. If any Interest is charged or received with respect to the Obligations in excess of the Maximum Rate ("Excess"), each Borrower stipulates that any such charge or receipt shall be the result of an accident and bona fide error, and such Excess, to the extent received, shall be applied first to reduce the principal of such Obligations and the balance, if any, returned to Borrowers, it being the intent of the parties hereto not to enter into a usurious or otherwise illegal relationship. The right to accelerate the maturity of any of the Obligations does not include the right to accelerate any Interest that has not otherwise accrued on the date of such acceleration, and Agent and Lenders do not intend to collect any unearned Interest in the event of any such acceleration. Each Borrower recognizes that, with fluctuations in the rates of interest set forth in Section 3.1.1 of this Agreement or in the Notes, and the Maximum Rate, such an unintentional result could inadvertently occur. All monies paid to Agent or any Lender hereunder or under any of the other Loan Documents, whether at maturity or by prepayment, shall be subject to any rebate of unearned Interest as and to the extent required by Applicable Law. By the execution of this Agreement, each Borrower covenants that (i) the credit or return of any Excess shall constitute the acceptance by such Borrower of such Excess, and (ii) such Borrower shall not seek or pursue any other remedy, legal or equitable, against Agent or any Lender, based in whole or in part upon contracting for, charging or receiving any Interest in excess of the Maximum Rate. For the purpose of determining whether or not any Excess has been contracted for, charged or received by Agent or any Lender, all Interest at any time contracted for, charged or received from Borrowers in connection with any of the Loan Documents shall, to the extent permitted by Applicable Law, be amortized, prorated, allocated and spread in equal parts throughout the full term of the Obligations. Borrowers, Agent and Lenders shall, to the maximum extent permitted under Applicable Law, (i) characterize any non-principal payment as an expense, fee or premium rather than as Interest and (ii) exclude voluntary prepayments and the effects thereof. The provisions of this Section 3.11 shall be deemed to be incorporated into every Loan Document (whether or not any provision of this Section is

-55-

referred to therein). All such Loan Documents and communications relating to any Interest owed by Borrowers and all figures set forth therein shall, for the sole purpose of computing the extent of Obligations, be automatically recomputed by Borrowers, and by any court considering the same, to give effect to the adjustments or credits required by this **Section 3.11**.

SECTION 4. LOAN ADMINISTRATION

4.1. <u>Manner of Borrowing and Funding Revolver Loans</u>. Borrowings under the Commitments established pursuant to Section 2.1 hereof shall be made and funded as follows:

4.1.1. Notice of Borrowing.

(i) Whenever Borrowers desire to make a Borrowing under **Section 2.1** of this Agreement (other than a Borrowing resulting from a conversion or continuation pursuant to **Section 3.1.2**), Borrowers shall give Agent prior written notice (or telephonic notice promptly confirmed in writing) of such Borrowing request (a "Notice of Borrowing"), which shall be in the form of <u>Exhibit D</u> annexed hereto and signed by an authorized officer of Borrower Agent. Such Notice of Borrowing shall be given by Borrower Agent no later than 12:00 noon at the office designated by Agent from time to time (a) on the Business Day of the requested funding date of such Borrowing, in the case of Base Rate Loans, and (b) at least 2 Business Days prior to the requested funding date of such Borrowing, in the case of \$250,000 and shall be made as Base Rate Loans and thereafter may be made or continued as or converted into Base Rate Loans or LIBOR Loans. Each Notice of Borrowing (or telephonic notice thereof) shall be irrevocable and shall specify (a) the principal amount of the Borrowing, (b) the date of Borrowing (which shall be a Business Day), (c) whether the Borrowing is to consist of Base Rate Loans or LIBOR Loans, (d) in the case of LIBOR Loans, the duration of the Interest Period to be applicable thereto, and (e) the account of Borrowers to which the proceeds of such Borrowing are to be disbursed.

(ii) Unless payment is otherwise timely made by Borrowers, the becoming due of any amount required to be paid with respect to any of the Obligations (whether as principal, accrued interest, fees or other charges, including the repayment of any LC Obligations, and any amounts owed to any Affiliate of Agent) shall be deemed irrevocably to be a request by Borrower Agent (without any requirement for the submission of a Notice of Borrowing) for Revolver Loans on the due date of, and in an aggregate amount required to pay, such Obligations, and the proceeds of such Revolver Loans may be disbursed by way of direct payment of the relevant Obligation and shall bear interest as Base Rate Loans. Neither Agent nor any Lender shall have any obligation to Borrowers to honor any deemed request for a Revolver Loan on or after the Commitment Termination Date, when an Out-of-Formula Condition exists or would result therefrom, or when any applicable condition precedent set forth in **Section 11** hereof is not satisfied, but may do so in the discretion of Agent (or at the direction of the Required Lenders) and without regard to the existence of, and without being deemed to have waived, any Default or Event of Default and regardless of whether such Revolver Loan is funded after the Commitment Termination Date.

(iii) If Borrowers elect to establish a Controlled Account with BofA or any Affiliate of BofA, then the presentation for payment by Bank of any check or other item of payment drawn on the Controlled Account at a time when there are insufficient funds in such account to cover such check shall be deemed irrevocably to be a request by Borrower Agent (without any requirement for the submission of a Notice of Borrowing) for Revolver Loans on

-56-

the date of such presentation and in an amount equal to the aggregate amount of the items presented for payment, and the proceeds of such Revolver Loans may be disbursed to the Controlled Account and shall bear interest as Base Rate Loans. Neither Agent nor any Lender shall have any obligation to honor any deemed request for a Revolver Loan on or after the Commitment Termination Date or when an Out-of-Formula Condition exists or would result therefrom or when any condition precedent in Section 11 hereof is not satisfied, but may do so in the discretion of Agent (or at the direction of the Required Lenders) and without regard to the existence of, and without being deemed to have waived, any Default or Event of Default and regardless of whether such Revolver Loan is funded after the Commitment Termination Date.

4.1.2. Fundings by Lenders. Subject to its receipt of notice from Agent of a Notice of Borrowing as provided in Section 4.1.1(i) (except in the case of a deemed request by Borrower Agent for a Revolver Loan as provided in Sections 4.1.1(ii) or (iii) or 4.1.3(ii) hereof, in which event no Notice of Borrowing need be submitted), each Lender shall timely honor its Revolver Commitment by funding its Pro Rata share of each Borrowing of Revolver Loans that is properly requested and that Borrowers are entitled to receive under the Loan Agreement. Agent shall notify Lenders of each Notice of Borrowing (or deemed request for a Borrowing pursuant to Section 4.1.1(ii) or (iii) hereof), by 1:00 p.m. on the proposed funding date (in the case of Base Rate Loans) or by 1:00 p.m. at least 2 Business Days before the proposed funding date (in the case of LIBOR Loans). Each Lender shall deposit with Agent an amount equal to its Pro Rata share of the Borrowing requested or deemed requested by Borrowers at Agent's designated bank in immediately available funds not later than 3:00 p.m. on the date of funding of such Borrowing, unless Agent's notice to Lenders is received after 1:00 p.m. on the proposed funding date of a Base Rate Loan, in which event Lenders shall deposit with Agent their respective Pro Rata shares of the requested Borrowing on or before 11:00 a.m. of the next Business Day. Subject to its receipt of such amounts from Lenders, Agent shall make the proceeds of the Revolver Loans received by it available to Borrowers by disbursing such proceeds in accordance with Borrower Agent's disbursement instructions set forth in the applicable Notice of Borrowing. Neither Agent nor any Lender shall have any liability on account of any delay by any bank or other depository institution in treating the proceeds of any Revolver Loan as collected funds or any delay in receipt, or any loss, of funds that constitute a Revolver Loan, the wire transfer of which was initiated by Agent in accordance with wiring instructions provided to Agent. Unless Agent shall have been notified in writing by a Lender prior to the proposed time of funding that such Lender does not intend to deposit with Agent an amount equal such Lender's Pro Rata share of the requested Borrowing (or deemed request for a Borrowing pursuant to Section 4.1.1(ii) or (iii) hereof), Agent may assume that such Lender has deposited or promptly will deposit its share with Agent and Agent may in its discretion disburse a corresponding amount to Borrowers on the applicable funding date. If a Lender's Pro Rata share of such Borrowing is not in fact deposited with Agent, then, if Agent has disbursed to Borrowers an amount corresponding to such share, then such Lender agrees to pay, and in addition Borrowers agree to repay, to Agent forthwith on demand such corresponding amount, together with interest thereon, for each day from the date such amount is disbursed by Agent to or for the benefit of Borrowers until the date such amount is paid or repaid to Agent, (a) in the case of Borrowers, at the interest rate applicable to such Borrowing and (b) in the case of such Lender, at the Federal Funds Rate. If such Lender repays to Agent such corresponding amount, such amount so repaid shall constitute a Revolver Loan, and if both such Lender and Borrowers shall have repaid such corresponding amount, Agent shall promptly return to Borrowers such corresponding amount in same day funds. A notice from Agent submitted to any Lender with respect to amounts owing under this **Section 4.1.2** shall be conclusive, absent manifest error.

4.1.3. Settlement and Swingline Loans.

(i) In order to facilitate the administration of the Revolver Loans under this Agreement, Lenders and Agent agree (which agreement shall be solely between Lenders and

-57-

Agent and shall not be for the benefit of or enforceable by any Borrower) that settlement among them with respect to the Revolver Loans may take place on a periodic basis on dates determined from time to time by Agent (each a "Settlement Date"), which may occur before or after the occurrence or during the continuance of a Default or Event of Default and whether or not all of the conditions set forth in **Section 11** have been met. On each Settlement Date, payment shall be made by or to each Lender in the manner provided herein and in accordance with the Settlement Report delivered by Agent to Lenders with respect to such Settlement Date so that, as of each Settlement Date and after giving effect to the transaction to take place on such Settlement Date, each Lender shall hold its Pro Rata share of all Revolver Loans and participations in LC Obligations. Agent shall request settlement with the Lenders on a basis not less frequently than once every 5 Business Days.

(ii) Between Settlement Dates, Agent may request BofA to advance, and BofA may, but shall in no event be obligated to, advance to Borrowers out of BofA's own funds the entire principal amount of any Borrowing of Revolver Loans that are Base Rate Loans requested or deemed requested pursuant to this Agreement (any such Revolver Loan funded exclusively by BofA being referred to as a "Swingline Loan"). Each Swingline Loan shall constitute a Revolver Loan hereunder and shall be subject to all of the terms, conditions and security applicable to other Revolver Loans, except that all payments thereon shall be payable to BofA solely for its own account and all Swingline Loans shall be made as Base Rate Loans plus the Applicable Margin then in effect for Revolver Loans and shall not be entitled to be converted into LIBOR Loans. The obligation of Borrowers to repay such Swingline Loans to BofA shall be evidenced by the records of BofA and need not be evidenced by any promissory note. Unless a funding is required by all Lenders pursuant to Sections 2.1.5 or 13.9.4, Agent shall not request BofA to make any Swingline Loan if (A) Agent shall have received written notice from any Lender that one or more of the applicable conditions precedent set forth in Section 11 hereof will not be satisfied on the requested funding date for the applicable Borrowing and Agent has made a determination (without any liability to any Person) that such condition precedent will not be satisfied, (B) the requested Borrowing would exceed the amount of Availability on the funding date or (C) the aggregate amount of all Swingline Loans outstanding when added to the amount of the requested Borrowing, would exceed \$20,000,000. BofA shall not be required to determine whether the applicable conditions precedent set forth in Section 11 hereof have been satisfied or the requested Borrowing would exceed the amount of Availability on the funding date applicable thereto prior to making, in its sole discretion, any Swingline Loan. On each Settlement Date, or, if earlier, upon demand by Agent for payment thereof, the then outstanding Swingline Loans shall be immediately due and payable. As provided in Section 3.1.1(ii), Borrowers shall be deemed to have requested (without the necessity of submitting any Notice of Borrowing) Revolver Loans to be made on each Settlement Date in the amount of all outstanding Swingline Loans and to have Agent cause the proceeds of such Revolver Loans to be applied to the repayment of such Swingline Loans and interest accrued thereon. Agent shall notify the Lenders of the outstanding balance of Revolver Loans prior to 1:00 p.m. on each Settlement Date and each Lender (other than BofA) shall deposit with Agent (without setoff, counterclaim or reduction of any kind) an amount equal to its Pro Rata share of the amount of Revolver Loans deemed requested in immediately available funds not later than 3:00 p.m. on such Settlement Date. Each Lender's obligation to make such deposit with Agent shall be absolute and unconditional, without defense, offset, counterclaim or other defense, and without regard to whether any of the conditions precedent set forth in Section 11 hereof are satisfied, any Out-of-Formula Condition exists or the Commitment Termination Date has occurred. If, as the result of the commencement by or against Borrowers of any Insolvency Proceeding or otherwise, any Swingline Loan may not be repaid by the funding by Lenders of Revolver Loans, then each Lender (other than BofA) shall be deemed to have purchased a participating interest in any unpaid Swingline Loan in an amount equal to

-58-

such Lender's Pro Rata share of such Swingline Loan and shall transfer to BofA, in immediately available funds not later than the second Business Day after BofA's request therefor, the amount of such Lender's participation. The proceeds of Swingline Loans may be used solely for purposes for which Revolver Loans generally may be used in accordance with **Section 2.1.3** hereof. If any amounts received by BofA in respect of any Swingline Loans are later required to be returned or repaid by BofA to Borrowers or any other Obligor or their respective representatives or successors-in-interest, whether by court order, settlement or otherwise, the other Lenders shall, upon demand by BofA with notice to Agent, pay to Agent for the account of BofA, an amount equal to each other Lender's Pro Rata share of all such amounts required to be returned or repaid.

4.1.4. <u>Disbursement Authorization</u>. Each Borrower hereby irrevocably authorizes Agent to disburse the proceeds of each Revolver Loan requested by any Borrower, or deemed to be requested pursuant to **Section 4.1.1** or **Section 4.1.3(ii)**, as follows: (i) the proceeds of each Revolver Loan requested under **Section 4.1.1(i)** shall be disbursed by Agent in accordance with the terms of the written disbursement letter from Borrowers in the case of the initial Borrowing, and, in the case of each subsequent Borrowing, by wire transfer to such bank account of Borrowers as may be agreed upon by Borrowers and Agent from time to time or elsewhere if pursuant to a written direction from any Borrower that is approved by Agent; and (ii) the proceeds of each Revolver Loan requested under **Section 4.1.1(ii)** or **Section 4.1.3(ii)** shall be disbursed by Agent by way of direct payment of the relevant interest or other Obligation. Any Loan proceeds received by any Borrower or in payment of any of the Obligations shall be deemed to have been received by all Borrowers.

4.1.5. <u>Telephonic Notices</u>. Each Borrower authorizes Agent and Lenders to extend, convert or continue Loans, effect selections of Types of Loans and transfer funds to or on behalf of Borrowers based on telephonic notices or instructions from a Senior Officer or cash manager of MasTec whom Agent or any Lender in good faith believes to be acting on behalf of Borrower Agent or any Borrower. Borrowers shall confirm each such telephonic request for a Borrowing or conversion or continuation of Loans by prompt delivery to Agent of the required Notice of Borrowing or Notice of Conversion/Continuation, as applicable, duly executed by an authorized officer of Borrower Agent. If the written confirmation differs in any material respect from the action taken by Agent or Lenders, the records of Agent and Lenders shall govern. Neither Agent nor any Lender shall have any liability for any loss suffered by any Borrower as a result of Agent's or any Lender's acting upon its understanding of telephonic instructions or requests from a person believed in good faith by Agent or any Lender to be a person authorized by a Borrower to give such instructions or to make such requests on Borrowers' behalf.

4.2. <u>Defaulting Lender</u>. If any Lender shall, at any time, fail to make any payment to Agent or BofA that is required hereunder, Agent may, but shall not be required to, retain payments that would otherwise be made to such defaulting Lender hereunder and apply such payments to such defaulting Lender's defaulted obligations hereunder, at such time, and in such order, as Agent may elect in its sole discretion. With respect to the payment of any funds from Agent to a Lender or from a Lender to Agent, the party failing to make the full payment when due pursuant to the terms hereof shall, upon demand by the other party, pay such amount together with interest on such amount at the Federal Funds Rate. The failure of any Lender to fund its portion of any Loan or payment in respect of an LC Obligation shall not relieve any other Lender of its obligation, if any, to fund its portion of the Revolver Loan or payment in respect of an LC Obligation to be made by such Lender on the date of any Borrowing. Solely as among the Lenders and solely for purposes of voting or consenting to matters with respect to any of the Loan Documents, Collateral or any Obligations and determining a defaulting Lender's share of payments and proceeds of Collateral pending such defaulting Lender's cure of its defaults hereunder, a defaulting Lender shall not be deemed to be a "Lender" and such Lender's Commitment shall be deemed to be zero (0). The

-59-

provisions of this Section 4.2 shall be solely for the benefit of Agent and Lenders and may not be enforced by Borrowers.

4.3. Special Provisions Governing LIBOR Loans.

4.3.1. Number of LIBOR Loans. In no event may the number of LIBOR Loans outstanding at any time to any Lender exceed 7.

4.3.2. <u>Minimum Amounts</u>. Each Borrowing of LIBOR Loans pursuant to **Section 4.1.1(i)**, and each continuation of or conversion to LIBOR Loans pursuant to **Section 3.1.2** hereof, shall be in a minimum amount of \$1,000,000 and integral multiples of \$100,000 in excess of that amount.

4.3.3. <u>LIBOR Lending Office</u>. Each Lender's initial LIBOR Lending Office is set forth opposite its name on the signature pages hereof. Each Lender shall have the right at any time and from time to time to designate a different office of itself or of any Affiliate as such Lender's LIBOR Lending Office, and to transfer any outstanding LIBOR Loans to such LIBOR Lending Office. No such designation or transfer shall result in any liability on the part of Borrowers for increased costs or expenses resulting solely from such designation or transfer. Increased costs for expenses resulting from a change in Applicable Law occurring subsequent to any such designation or transfer shall be deemed not to result solely from such designation or transfer.

4.3.4. <u>Funding of LIBOR Loans</u>. Each Lender may, if it so elects, fulfill its obligation to make, continue or convert LIBOR Loans hereunder by causing one of its foreign branches or Affiliates (or an international banking facility created by such Lender) to make or maintain such LIBOR Loans; provided, however, that such LIBOR Loans shall nonetheless be deemed to have been made and to be held by such Lender, and the obligation of Borrowers to repay such LIBOR Loans shall nevertheless be to such Lender for the account of such foreign branch, Affiliate or international banking facility. The calculation of all amounts payable to Lender under **Section 3.7** and **3.10** shall be made as if each Lender had actually funded or committed to fund its LIBOR Loan through the purchase of an underlying deposit in an amount equal to the amount of such LIBOR Loans in any manner it deems fit and the foregoing presumption shall be utilized only for the calculation of amounts payable under **Section 3.7** and **3.7** and **3.10**.

4.4. Borrower Agent. Each Obligor hereby irrevocably appoints MasTec and MasTec agrees to act under this Agreement, as the agent and representative of itself and each other Obligor for all purposes under this Agreement (in such capacity, "Borrower Agent"), including requesting Borrowings, selecting whether any Loan or portion thereof is to bear interest as a Base Rate Loan or a LIBOR Loan, and receiving account statements and other notices and communications to Obligors (or any of them) from Agent. Agent may rely, and shall be fully protected in relying, on any Notice of Borrowing, Notice of Conversion/Continuation, disbursement instructions, reports, information, Borrowing Base Certificate or any other notice or communication made or given by Borrower Agent, whether in its own name, on behalf of any Obligor or on behalf of "the Borrowers" or "the Obligors," and Agent shall have no obligation to make any inquiry or request any confirmation from or on behalf of any other Borrower as to the binding effect on such Obligor of any such Notice of Borrowing, Notice of Conversion Continuation, instruction, report, information, Borrowing Base Certificate or other notice or communication, nor shall the joint and several character of Borrowers' liability for the Obligations be affected, provided that the provisions of this **Section 4.4** shall not be construed so as to preclude any Borrower from directly requesting Borrowings or taking other actions permitted to be taken by "a Borrower" hereunder. Agent may maintain a single Loan Account in the name of "MasTec" hereunder, and each Borrower expressly

-60-

agrees to such arrangement and confirms that such arrangement shall have no effect on the joint and several character of such Borrower's liability for the Obligations.

4.5. <u>All Loans to Constitute One Obligation</u>. The Loans and LC Obligations shall constitute one general obligation of Borrowers and (unless otherwise expressly provided in any Security Document) shall be secured by Agent's Lien upon all of the Collateral; <u>provided</u>, <u>however</u>, that Agent and each Lender shall be deemed to be a creditor of each Borrower and the holder of a separate claim against each Borrower to the extent of any Obligations owed by Borrowers to Agent or such Lender.

SECTION 5. PAYMENTS

5.1. <u>General Payment Provisions</u>. All payments (including all prepayments) of principal of and interest on the Loans, LC Obligations and other Obligations that are payable to Agent or any Lender shall be made to Agent in Dollars without any offset or counterclaim and free and clear of (and without deduction for) any present or future Taxes, and, with respect to payments made other than by application of balances in the Payment Account, in immediately available funds not later than 12:00 noon on the due date (and payment made after such time on the due date to be deemed to have been made on the next succeeding Business Day). All payments received by Agent shall be distributed by Agent in accordance with this Agreement, hereof, subject to the rights of offset that Agent may have as to amounts otherwise to be remitted to a particular Lender by reason of amounts due Agent from such Lender under any of the Loan Documents.

5.2. <u>Repayment of Revolver Loans.</u>

5.2.1. <u>Payment of Principal</u>. The outstanding principal amounts with respect to the Revolver Loans shall be repaid as follows:

(i) Any portion of the Revolver Loans consisting of the principal amount of Base Rate Loans shall be paid by Borrowers to Agent, for the Pro Rata benefit of Lenders (or, in the case of Swingline Loans, for the sole benefit of BofA) unless timely converted to a LIBOR Loan in accordance with this Agreement, immediately upon (a) each receipt by Agent, any Lender or Borrowers of any proceeds of any of the Accounts or other Collateral, to the extent of such proceeds, (b) the Commitment Termination Date, and (c) in the case of Swingline Loans, the earlier of BofA's demand for payment or on each Settlement Date with respect to all Swingline Loans outstanding on such date.

(ii) Any portion of the Revolver Loans consisting of the principal amount of LIBOR Loans shall be paid by Borrowers to Agent, for the Pro Rata benefit of Lenders, unless converted to a Base Rate Loan or continued as a LIBOR Loan in accordance with the terms of this Agreement, immediately upon (a) the last day of the Interest Period applicable thereto and (b) the Commitment Termination Date. In no event shall Borrowers be authorized to make a voluntary prepayment with respect to any Revolver Loan outstanding as a LIBOR Loan prior to the last day of the Interest Period applicable thereto unless (x) otherwise agreed in writing by Agent or Borrowers are otherwise expressly authorized or required by any other provision of this Agreement to pay any LIBOR Loan outstanding on a date other than the last day of the Interest Period applicable thereto, and (y) Borrowers pay to Agent, for the Pro Rata benefit of Lenders, concurrently with any prepayment of a LIBOR Loan, any amount due Agent and Lenders under **Section 3.10** hereof as a consequence of such prepayment. Notwithstanding the foregoing provisions of this **Section 5.2.1(ii)**, if, on any date that Agent receives proceeds of any of the Accounts or other Collateral, there are no Revolver Loans outstanding as Base Rate Loans, Agent may either hold such proceeds as cash security for the timely payment of the

-61-

Obligations or apply such proceeds to any outstanding Revolver Loans bearing interest as LIBOR Loans as the same become due and payable (whether at the end of the applicable Interest Periods or on the Commitment Termination Date).

(iii) Notwithstanding anything to the contrary contained elsewhere in this Agreement, if an Out-of-Formula Condition shall exist, Borrowers shall, on the sooner to occur of Agent's demand or the first Business Day after any Borrower has obtained knowledge of such Out-of-Formula Condition, repay the outstanding Revolver Loans that are Base Rate Loans in an amount sufficient to reduce the aggregate unpaid principal amount of all Revolver Loans by an amount equal to such excess; and, if such payment of Base Rate Loans is not sufficient to eliminate the Out-of-Formula Condition, then Borrowers shall immediately deposit with Agent, for the Pro Rata benefit of Lenders, for application to any outstanding Revolver Loans bearing interest as LIBOR Loans as the same become due and payable (whether at the end of the applicable Interest Periods or on the Commitment Termination Date) cash in an amount sufficient to eliminate such Out-of-Formula Condition, and Agent may (a) hold such deposit as cash security pending disbursement of same to Lenders for application to the Obligations, or (b) if a Default or Event of Default exists, immediately apply such proceeds to the payment of the Obligations, including the Revolver Loans outstanding as LIBOR Loans (in which event Borrowers shall also pay to Agent for the Pro Rata benefit of Lenders any amounts required by **Section 3.10** to be paid by reason of the prepayment of a LIBOR Loan prior to the last day of the Interest Period applicable thereto).

5.2.2. <u>Payment of Interest</u>. Interest accrued on the Revolver Loans shall be due and payable on (i) the first day of each month (for the immediately preceding month), computed through the last day of the preceding month, with respect to any Revolver Loan (whether a Base Rate Loan or LIBOR Loan) and (ii) the last day of the applicable Interest Period in the case of a LIBOR Loan. Accrued interest shall also be paid by Borrowers on the Commitment Termination Date. With respect to any Base Rate Loan converted into a LIBOR Loan pursuant to **Section 3.1.2** on a day when interest would not otherwise have been payable with respect to such Base Rate Loan, accrued interest to the date of such conversion on the amount of such Base Rate Loan so converted shall be paid on the conversion date.

5.2.3. <u>Mandatory Prepayments</u>. Borrowers shall pay the entire unpaid principal balance of the Loans, and all accrued but unpaid interest thereon, upon the Commitment Termination Date. Borrowers shall also prepay the Loans as follows:

(i) Borrowers shall prepay the Loans in the amount of Net Disposition Proceeds from Permitted Asset Dispositions <u>provided that</u> Net Disposition Proceeds received from a Canadian Obligor shall be held by Agent as security for the payment of the Canadian Obligor Guarantee of such Canadian Obligor;

(ii) Borrowers shall prepay the Loans from the proceeds of insurance or condemnation awards paid in respect of any Equipment or Real Estate unless and to the extent Borrowers are authorized to use such proceeds pursuant to **Section 8.1.2(ii)**; and

(iii) Borrowers shall prepay the Loans from the net proceeds received by any Borrower from any issuance of any Debt or Equity Interests in an amount sufficient to cause Availability at the time of such issuance of Debt or Equity Interests, and immediately after giving effect thereto, to equal or exceed \$50,000,000.

5.2.4. <u>Application of Prepayments</u>. All distributions of prepayments by Agent to Lenders shall be on a Pro Rata basis. Each Lender shall apply the portion of a prepayment that is to be applied to principal installments first to outstanding Base Rate Loans and then to any outstanding LIBOR Loans with the shortest Interest Periods remaining; but if application to any LIBOR Loans would cause the same to be paid prior to the end of an applicable Interest Period, then, by prior written notice to Agent, Borrowers may elect as to such LIBOR Loan to deliver cash to Agent in the amount of the required prepayment, to be held by Agent as cash collateral until the end of the applicable Interest Period, at which time Agent shall disburse such cash collateral to the affected Lenders for application to such LIBOR Loans.

5.3. Reserved.

5.4. <u>Payment of Other Obligations</u>. The balance of the Obligations requiring the payment of money, including the LC Obligations and Extraordinary Expenses incurred by Agent or any Lender, shall be repaid by Borrowers to Agent for allocation among Agent and Lenders as provided in the Loan Documents, or, if no date of payment is otherwise specified in the Loan Documents, **on demand**.

5.5. <u>Marshaling; Payments Set Aside</u>. None of Agent or Lenders shall be under any obligation to marshal any assets in favor of Borrowers or any other Obligor or against or in payment of any or all of the Obligations. To the extent that Borrowers make a payment to Agent or Lenders or Agent or any Lender receives payment from the proceeds of any Collateral or exercises its right of setoff, and such payment or the proceeds of such Collateral or setoff (or any part thereof) are subsequently invalidated, declared to be fraudulent or preferential, set aside or required to be repaid to a trustee, receiver or any other Person, then to the extent of any loss by Agent or Lenders, the Obligations or part thereof originally intended to be satisfied, and all Liens, rights and remedies therefor, shall be revived and continued in full force and effect as if such payment or proceeds had not been made or received and any such enforcement or setoff had not occurred. The provisions of the immediately preceding sentence of this **Section 5.5** shall survive any termination of the Commitments and Full Payment of the Obligations.

5.6. Post Default Allocation of Payments.

5.6.1. <u>Allocation</u>. For so long as an Event of Default exists, all monies to be applied to the Obligations, whether such monies represent voluntary or mandatory payments or prepayments by one or more Obligors or are received pursuant to demand for payment or realized from any disposition of Collateral and irrespective of any designation by Borrowers of the Obligations intended to be satisfied, shall be allocated among Agent and such of the Lenders as are entitled thereto (and, with respect to monies allocated to Lenders, on a Pro Rata basis unless otherwise provided herein): (i) <u>first</u>, to Agent to pay principal and accrued interest on any portion of the Revolver Loans (including Protective Advances) which Agent may have advanced on behalf of any Lender and for which Agent has not been reimbursed by such Lender or Borrowers, until Full Payment of all such Obligations; (ii) <u>second</u>, to BofA to pay the principal and accrued interest on any portion of the Swingline Loans outstanding, to be shared with Lenders that have acquired and paid for a participating interest in such Swingline Loans, until Full Payment of all such Obligations; (iii) <u>third</u>, to the extent that Issuing Bank has not received from any Participating Lender a payment as required by **Section 2.3.2** hereof, to Issuing Bank to pay all such required payments from each Participating Lender, until Full Payment of all such Obligations; (iv) <u>fourth</u>, to Agent to pay the amount of Extraordinary Expenses (including Protective Advances) that have not been reimbursed to Agent by Borrowers or Lenders, together with interest accrued thereon at the rate applicable to Revolver Loans that are Base Rate Loans, until Full Payment of all such Obligations; (v) <u>fifth</u>, to Agent to pay any Indemnified Amount that has not been paid pursuant to any indemnity of Agent Indemnitees by any Obligor or to pay amounts owing by Lenders to Agent Indemnities pursuant to **Section 13.6**, in each case, together with interest accrued thereon at the rate applicable to Revolver Loa

-63-

that are Base Rate Loans, until Full Payment of all such Obligations; (vi) <u>sixth</u>, to Agent to pay any fees due and payable to Agent, until Full Payment of all such Obligations; (vii) <u>seventh</u>, to each Lender, ratably, for any Indemnified Amount that such Lender has paid to Agent Indemnitees pursuant to its indemnity of Agent Indemnitees and any Extraordinary Expenses that such Lender has reimbursed to Agent or such Lender has incurred, to the extent that such Lender has not been reimbursed by Obligors therefor; (viii) <u>eighth</u>, to Issuing Bank to pay principal and interest with respect to LC Obligations (or to the extent any of the LC Obligations are contingent and an Event of Default then exists, deposited in the Cash Collateral Account to Cash Collateralize the LC Obligations), which payment shall be shared with the Participating Lenders in accordance with **Section 2.3.2(iii)** hereof; (ix) <u>ninth</u>, to Lenders in payment of the unpaid principal and accrued interest in respect of the Loans and other Obligations (excluding Banking Relationship Debt) then outstanding, in such order of application as shall be designated by Agent (acting at the direction or with the consent of the Required Lenders); and (x) <u>tenth</u>, to Lenders and any Affiliates of Lenders, ratably, in payment of any Banking Relationship Debt owed to such Persons and secured by the Collateral hereunder. For the avoidance of doubt, to the extent constituting an Indemnified Amount, any indemnity, guarantee or other payment made by Agent or any Lender as described in **Section 3.4.2** with respect to Banking Relationship Debt shall be allocated to <u>tenth</u> priority, above. The allocations set forth in this **Section 5.6** are solely to determine the rights and priorities of Agent and Lenders as among themselves and may be changed by Agent and Lenders without notice to or the consent or approval of any Borrower or any other Person.

5.6.2. <u>Erroneous Allocation</u>. Agent shall not be liable for any allocation or distribution of payments made by it in good faith and, if any such allocation or distribution is subsequently determined to have been made in error, the sole recourse of any Lender to which payment was due but not made shall be to recover from the other Lenders any payment in excess of the amount to which such other Lenders are determined to be entitled (and such other Lenders hereby agree to return to such Lender any such erroneous payments received by them).

5.7. <u>Application of Payments and Collateral Proceeds</u>. All Payment Items received by Agent by 12:00 noon, on any Business Day shall be deemed received on that Business Day. All Payment Items received by Agent after 12:00 noon, on any Business Day shall be deemed received on the following Business Day. Each Borrower irrevocably waives the right to direct the application of any and all payments and Collateral proceeds at any time or times hereafter received by Agent or any Lender from or on behalf of Borrowers, and each Borrower does hereby irrevocably agree that Agent shall have the continuing exclusive right to apply and reapply any and all such payments and Collateral proceeds received at any time or times hereafter by Agent or its agent against the Obligations, in such manner as Agent may deem advisable, notwithstanding any entry by Agent upon any of its books and records; provided, however, that any payments or proceeds of Collateral received by Agent on any date that an Event of Default does not exist shall be applied in accordance with any provisions of this Agreement that govern the application of such payment or proceeds. If, as the result of Agent's collection of proceeds of Accounts and other Collateral as authorized by **Section 8.2.6** a credit balance exists, such credit balance shall not accrue interest in favor of Borrowers, but shall be available to Borrowers at any time or times for so long as no Default or Event of Default exists. Agent may apply such credit balance against any of the Obligations upon and after the occurrence of an Event of Default in the manner specified in **Section 5.6.1**.

5.8. Loan Accounts; the Register; Account Stated.

5.8.1. Loan Accounts. Each Lender shall maintain in accordance with its usual and customary practices an account or accounts (a "Loan Account") evidencing the Debt of Borrowers to such Lender resulting from each Loan owing to such Lender from time to time, including the amount of principal and interest payable to such Lender from time to time hereunder and under each Note payable to such Lender. Any failure of a Lender to record in the Loan Account, or any error in doing so, shall not

-64-

limit or otherwise affect the obligation of Borrowers hereunder (or under any Note) to pay any amount owing hereunder to such Lender.

5.8.2. <u>The Register</u>. Agent shall maintain a register (the "Register"), which shall include a master account and a subsidiary account for each Lender and in which accounts (taken together) shall be recorded (i) the date and amount of each Borrowing made hereunder, the Type of each Loan comprising such Borrowing and any Interest Period applicable thereto, (ii) the effective date and amount of each Assignment and Acceptance delivered to and accepted by it and the parties thereto, (iii) the amount of any principal or interest due and payable or to become due and payable from Borrowers to each Lender hereunder or under the Notes, and (iv) the amount of any sum received by Agent from Borrowers or any other Obligor and each Lender's Pro Rata share thereof. The Register shall be available for inspection by Borrowers or any Lender at the offices of Agent at any reasonable time and from time to time upon reasonable prior notice. Any failure of Agent to record in the Register, or any error in doing so, shall not limit or otherwise affect the obligation of Borrowers hereunder (or under any Note) to pay any amount owing with respect to the Loans or provide the basis for any claim against Agent.

5.8.3. Entries Binding. The entries made in the Register and each Loan Account shall constitute rebuttably presumptive evidence of the information contained therein; provided, however, that if a copy of information contained in the Register or any Loan Account is provided to any Person, or any Person inspects the Register or any Loan Account, at any time or from time to time, then the information contained in the Register or the Loan Account, as applicable, shall be conclusive and binding on such Person for all purposes absent manifest error, unless such Person notifies Agent in writing within 30 days after such Person's receipt of such copy or such Person's inspection of the Register or Loan Account of its intention to dispute the information contained therein.

5.9. <u>Gross Up for Taxes</u>. If Obligors shall be required by Applicable Law to withhold or deduct any Taxes from or in respect of any sum payable under this Agreement or any of the other Loan Documents or Agent or any Lender shall be liable to remit any such Taxes otherwise payable by Obligors, in either case, (a) the sum payable to Agent or such Lender shall be increased as may be necessary so that, after making all required withholding, remittances or deductions, Agent or such Lender (as the case may be) receives an amount equal to the sum it would have received had no such withholding, remittances or deductions been made, (b) Obligors shall make such withholding or deductions, and (c) Obligors shall pay the full amount withheld or deducted to the relevant taxation authority or other authority in accordance with Applicable Law.

5.10. <u>Withholding Tax Exemption</u>. At least 5 Business Days prior to the first date on which interest or fees are payable hereunder for the account of any Foreign Lender, such Foreign Lender agrees that it will deliver to Borrowers and Agent 2 duly completed copies of United States Internal Revenue Service Form W-8BEN or W-8ECI (or any subsequent replacement substitute or form therefor), certifying in either case that such Lender is entitled to receive payment under this Agreement and its Note without deduction or withholding of any United States federal income taxes. Each Foreign Lender which so delivers a Form W-8BEN or W-8ECI further undertakes to deliver to Borrowers and Agent 2 additional copies of such form (or a successor form) on or before the date that such form expires (currently, 3 successive calendar years for Form W-8BEN and one calendar year for Form W-8ECI) or becomes obsolete or after the occurrence of any event requiring a change in the most form so delivered by it, and such amendments thereto or extensions or renewals thereof as may be reasonably requested by Borrowers or Agent, in each case, certifying that such Foreign Lender is entitled to receive payments under this Agreement and its Notes without deduction or withholding of any United States federal income taxes, unless an event (including any change in treaty, law or regulation) has occurred prior to the date on which any such delivery would otherwise be required that renders all such forms inapplicable or that would prevent such Foreign Lender from duly completing and delivering any such form with respect to it

-65-

and such Lender advises Borrowers and Agent that it is not capable of receiving payments without any deduction or withholding of United States federal income taxes.

5.11. Nature and Extent of Each Borrower's Liability.

5.11.1. Joint and Several Liability. Each Borrower shall be liable for, on a joint and several basis, and hereby guarantees the timely payment by all other Borrowers of, all of the Loans and other Obligations, regardless of which Borrower actually may have received the proceeds of any Loans or other extensions of credit hereunder or the amount of such Loans received or the manner in which Agent or any Lender accounts for such Loans or other extensions of credit on its books and records, it being acknowledged and agreed that Loans to any Borrower inure to the mutual benefit of all Borrowers and that Agent and Lenders are relying on the joint and several liability of Borrowers in extending the Loans and other financial accommodations hereunder. Each Borrower hereby unconditionally and irrevocably agrees that upon default in the payment when due (whether at stated maturity, by acceleration or otherwise) of any principal of, or interest owed on, any of the Loans or other Obligations, such Borrower shall forthwith pay the same, without notice or demand.

5.11.2. Unconditional Nature of Liability. Each Borrower's joint and several liability hereunder with respect to, and guaranty of, the Loans and other Obligations shall, to the fullest extent permitted by Applicable Law, be unconditional irrespective of (i) the validity, enforceability, avoidance or subordination of any of the Obligations or of any promissory note or other document evidencing all or any part of the Obligations, (ii) the absence of any attempt to collect any of the Obligations from any other Obligor or any Collateral or other security therefor, or the absence of any other action to enforce the same, (iii) the waiver, consent, extension, forbearance or granting of any indulgence by Agent or any Lender with respect to any provision of any instrument evidencing or securing the payment of any of the Obligations, or any other agreement now or hereafter executed by any other Borrower and delivered to Agent or any Lender, (iv) the failure by Agent to take any steps to perfect or maintain the perfected status of its security interest in or Lien upon, or to preserve its rights to, any of the Collateral or other security for the payment or performance of any of the Obligations or Agent's release of any Collateral or of its Liens upon any Collateral, (v) Agent's or Lenders' election, in any proceeding instituted under the Bankruptcy Code, for the application of Section 1111(b)(2) of the Bankruptcy Code or under the Bankruptcy and Insolvency Act (Canada) or the Companies Creditors Arrangement Act (Canada), (vi) any borrowing or grant of a security interest by any other Borrower, as debtor-in-possession under Section 364 of the Bankruptcy Code, (vii) the release or compromise, in whole or in part, of the liability of any Obligor for the payment of any of the Obligations, (viii) any amendment or modification of any of the Loan Documents or any waiver of a Default or Event of Default, (ix) any increase in the amount of the Obligations beyond any limits imposed herein or in the amount of any interest, fees or other charges payable in connection therewith, or any decrease in the same, (x) the disallowance of all or any portion of Agent's or any Lender's claims against any other Obligor for the repayment of any of the Obligations under Section 502 of the Bankruptcy Code or under the Bankruptcy and Insolvency Act (Canada) or the Companies Creditors Arrangement Act (Canada), or (xi) any other circumstance that might constitute a legal or equitable discharge or defense of any Borrower. After the occurrence and during the continuance of any Event of Default, Agent may proceed directly and at once, without notice to any Obligor, against any or all of Obligors to collect and recover all or any part of the Obligations, without first proceeding against any other Obligor or against any Collateral or other security for the payment or performance of any of the Obligations, and each Borrower waives any provision under Applicable Law that might otherwise require Agent to pursue or exhaust its remedies against any Collateral or Obligor before pursuing another Obligor. Each Borrower consents and agrees that Agent shall be under no obligation to marshal any assets in favor of any Obligor or against or in payment of any or all of the Obligations.

-66-

5.11.3. <u>No Reduction in Liability for Obligations</u>. No payment or payments made by an Obligor or received or collected by Agent from a Borrower or any other Person by virtue of any action or proceeding or any setoff or appropriation or application at any time or from time to time in reduction of or in payment of the Obligations shall be deemed to modify, reduce, release or otherwise affect the liability of any Borrower under this Agreement, each of whom shall remain liable for the payment and performance of all Loans and other Obligations until the Full Payment of the Obligations.

5.11.4. <u>Contribution</u>. Each Borrower is unconditionally obligated to repay the Obligations as a joint and several obligor under this Agreement. If, as of any date, the aggregate amount of payments made by a Borrower on account of the Obligations and proceeds of such Borrower's Collateral that are applied to the Obligations exceeds the aggregate amount of Loan proceeds actually used by such Borrower in its business (such excess amount being referred to as an "Accommodation Payment"), then each of the other Borrowers (each such Borrower being referred to as a "Contributing Borrower") shall be obligated to make contribution to such Borrower (the "Paying Borrower") in an amount equal to (A) the product derived by multiplying the sum of each Accommodation Payment of each Borrower by the Allocable Percentage (as defined below) of the Borrower from whom contribution is sought less (B) the amount, if any, of the then outstanding Accommodation Payment of such Contributing Borrower (such last mentioned amount which is to be subtracted from the aforesaid product to be increased by any amounts theretofore paid by such Contribution Berrower by way of contribution hereunder; provided, however, that a Paying Borrower's recovery of contribution hereunder); provided, however, that a Paying Borrower's recovery of contribution Payments then outstanding of all Borrowers. As used herein, the term "Allocable Percentage" shall mean, on any date of determination thereof, a fraction the denominator of which shall be equal to the number of Borrowers who are parties to this Agreement on such date and the numerator of which shall be 1; <u>provided</u>, <u>however</u>, that such percentages shall be modified in the event that contribution from a Borrower is not possible by reason of insolvency, bankruptcy or otherwise by reducing such Borrower's Allocable Percentage equitably and by adjusting the Allocable Percentage of the other Borrowers at all times equals 100%.

5.11.5. <u>Subordination</u>. Each Borrower hereby subordinates any claims, including any right of payment, subrogation, contribution and indemnity, that it may have from or against any other Obligor, and any successor or assign of any other Obligor, including any trustee, receiver or debtor-in-possession, howsoever arising, due or owing or whether heretofore, now or hereafter existing, to the Full Payment of all of the Obligations.

SECTION 6. TERM AND TERMINATION OF COMMITMENTS

6.1. Term of Commitments. Subject to each Lender's right to cease making Loans and other extensions of credit to Borrowers when any Default or Event of Default exists or upon termination of the Commitments as provided in **Section 6.2** hereof, the Commitments shall be in effect from the date hereof through the close of business on May 10, 2013 (the "Term") unless sooner terminated as provided in **Section 6.2**.

6.2. Termination.

6.2.1. <u>Termination by Agent</u>. Agent may (and upon the direction of the Required Lenders, shall) terminate the Commitments without notice at any time that an Event of Default exists; <u>provided</u>, <u>however</u>, that the Commitments shall automatically terminate as provided in **Section 12.2** hereof.

6.2.2. <u>Termination by Borrowers</u>. Upon at least 90 days prior written notice to Agent, Borrowers may, at their option, terminate the Commitments; <u>provided</u>, <u>however</u>, no such termination by Borrowers shall be effective until Full Payment of the Obligations. Any notice of termination given by Borrowers shall be irrevocable unless Agent otherwise agrees in writing. Borrowers may elect to terminate the Commitments in their entirety only provided that nothing contained herein shall affect Borrowers' right to voluntarily reduce the Revolver Commitments as provided in **Section 2.2**. No section of this Agreement, Type of Loan available hereunder or Commitment may be terminated by Borrowers singly.

6.2.3. Reserved

6.2.4. Effect of Termination. On the effective date of termination of the Commitments by Agent or by Borrowers, all of the Obligations shall be immediately due and payable, Lenders shall have no obligation to make any Loans and Issuing Bank shall have no obligation to issue any Letters of Credit, and Lenders may terminate any Bank Products (including any services or products under Cash Management Agreements). All undertakings, agreements, covenants, warranties and representations of Borrowers contained in the Loan Documents shall survive any such termination and Agent shall retain its Liens in the Collateral and all of its rights and remedies under the Loan Documents notwithstanding such termination until Full Payment of the Obligations. Notwithstanding the Full Payment of the Obligations, Agent shall not be required to terminate its Liens in any of the Collateral unless, with respect to any loss or damage Agent may incur as a result of the dishonor or return of any Payment Items applied to the Obligations, Agent shall have received either (i) a written agreement, executed by Borrowers and any Person deemed financially responsible by Agent whose loans or other advances to Borrowers are used in whole or in part to satisfy the Obligations, indemnifying Agent and Lenders from any such loss or damage; or (ii) such monetary reserves and Liens on the Collateral for such period of time as Agent, in its reasonable discretion, may deem necessary to protect Agent from any such loss or damage. The provisions of **Sections 3.4, 3.7, 3.8, 3.10, 5.5, 5.9** and this **Section 6.2.4** and all obligations of Borrowers to indemnify Agent or any Lender pursuant to this Agreement or any of the other Loan Documents, shall in all events survive any termination of the Commitments and Full Payment of the Obligations.

SECTION 7. COLLATERAL

7.1. <u>Grant of Security Interest</u>. To secure the prompt payment and performance of all of the Obligations, each Obligor hereby (i) confirms the mortgage, pledge and assignment to Agent, for the benefit of the Secured Parties of the Collateral under (and as defined in) the Existing Loan Agreement and the other Loan Documents (as defined in the Existing Loan Agreement), and the creation in favor of Agent, for the benefit of the Secured Parties, under the Existing Loan Agreement and such other Loan Documents of a continuing Lien in such Collateral, all as security for the Obligations, and (ii) grants to Agent, for the benefit of Secured Parties, a continuing security interest in and Lien upon the following Property and interests in Property of such Obligor, whether now owned or existing or hereafter created, acquired or arising and wheresoever located:

(i) all Accounts;

(ii) all Supporting Obligations

(iii) all Goods, including all Inventory and Equipment;

(iv) all Instruments;

-68-

(v) all Chattel Paper;

(vi) all Documents;

(vii) all General Intangibles, including Payment Intangibles, Software and Intellectual Property;

(viii) all Deposit Accounts (other than the Insurance Cash Collateral Accounts);

(ix) all Investment Property (but excluding any portion thereof that constitutes Margin Stock unless otherwise expressly provided in any Security Documents;

(x) all Letter-of-Credit Rights;

(xi) all monies now or at any time or times hereafter in the possession or under the control of Agent or a Lender or a bailee or Affiliate of Agent or a Lender, including any Cash Collateral in the Cash Collateral Account and cash, Cash Equivalents and other Property in the LC Collateral Account;

(xii) all accessions to, substitutions for and all replacements, products and cash and non-cash proceeds of (i) through (xi) above, including proceeds of and unearned premiums with respect to insurance policies insuring any of the Collateral and claims against any Person for loss of, damage to or destruction of any of the Collateral; and

(xiii) all books and records (including customer lists, files, correspondence, tapes, computer programs, print-outs, and other computer materials and records) of such Borrower pertaining to any of (i) through (xii) above.

Notwithstanding anything to the contrary set forth herein, Agent shall not be deemed to have been granted a security interest in, and the term "Collateral" shall not include, any Equity Interests owned or held by any Borrower in DirectStar.

7.2. <u>Lien on Deposit Accounts</u>. As additional security for the payment and performance of the Obligations, each Borrower hereby grants to Agent, for the benefit of Secured Parties, a continuing security interest in and Lien upon all of such Borrower's right, title and interest in and to each Deposit Account of such Borrower (other than the Insurance Cash Collateral Account) and in and to any deposits or other sums at any time credited to each such Deposit Account, including any sums in any blocked account or any special lockbox account and in the accounts in which sums are deposited. In connection with the foregoing, each Borrower hereby authorizes and directs each such bank or other depository to pay or deliver to Agent upon its written demand therefor made at any time that an Event of Default exists and without further notice to such Borrower (such notice being hereby expressly waived), all balances in each such Deposit Account maintained by such Borrower with such depository for application to the Obligations then outstanding, and the rights given Agent in this Section shall be cumulative with and in addition to Agent's other rights and remedies in regard to the foregoing Property as proceeds of Collateral. Each Borrower hereby irrevocably appoints Agent as such Borrower's attorney-in-fact to collect any and all such balances to the extent any such payment is not made to Agent by such bank or other depository after demand thereon is made by Agent pursuant hereto.

7.3. Real Estate Collateral.

-69-

7.3.1. Lien on Real Estate. The due and punctual payment and performance of the Obligations shall also be secured by the Lien created by the Mortgages upon all Real Estate of each Borrower described therein. The Mortgages shall be executed by Borrowers in favor of Agent (for the benefit of Secured Parties) as provided in this Agreement and shall be duly recorded, at Borrowers' expense, in each office where such recording is required to constitute a fully perfected Lien upon the Real Estate covered thereby. Promptly after any Borrower's obtaining any interest in any Real Estate, Borrowers shall execute and deliver to Agent, for the benefit of Secured Parties, a Mortgage sufficient to convey and perfect, upon recordation thereof in the proper real estate records, in favor of Agent a first priority Lien upon such interest as security for the payment or performance of the Obligations.

7.3.2. <u>Collateral Assignment of Leases</u>. To further secure to the prompt payment and performance of the Obligations, each Borrower hereby grants, transfers and assigns to Agent, for the benefit of Secured Parties, and hereby grants to Agent, for the benefit of Secured Parties, a security interest in and Lien upon all of such Borrower's right, title and interest in, to and under all now or hereafter existing leases of real Property to which such Borrower is a party, whether as lessor or lessee, and all extensions, renewals and modifications thereof.

7.4. Other Collateral.

7.4.1. <u>Cash Collateral</u>. In addition to the items of Property referred to in **Section 7.1** above, (i) the Obligations shall also be secured by the Cash Collateral (other than any Cash Collateral held in the LC Collateral Account) and all of the other items of Property from time to time described in any of the Security Documents as security for any of the Obligations, including all of the Collateral described in the Canadian Security Agreement, in each case to the extent provided herein and (ii) the LC Obligations relating to the Ace LC, and only such LC Obligation, shall be secured by all cash, Cash Equivalents and other Property from time to time in the LC Collateral Account.

7.4.2. <u>Commercial Tort Claims</u>. Obligors shall promptly notify Agent in writing upon any Obligor's obtaining a Commercial Tort Claim (other than, for so long as no Default or Event of Default exists, a Commercial Tort Claim that is less than \$100,000) after the Closing Date against any Person and, upon Agent's written request, promptly execute such instruments or agreements and do such other acts or things deemed appropriate by Agent to confer upon Agent (for the benefit of Secured Parties) a security interest in each such Commercial Tort Claim.

-70-

7.4.3. <u>Certain After-Acquired Collateral</u>. Obligors shall promptly notify Agent in writing upon any Obligor's obtaining any Collateral after the Closing Date consisting of Deposit Accounts, Investment Property, Letter-of-Credit Rights, Chattel Paper or Intellectual Property consisting of registered trademarks or patents, and, upon Agent's request, shall promptly execute such documents and do such other acts or things deemed appropriate by Agent to confer upon Agent a duly perfected, first priority Lien upon, and (to the extent applicable for perfection of a Lien) control with respect to such Collateral; promptly notify Agent in writing upon any Borrower's obtaining any Collateral after the Closing Date consisting of Documents or Instruments and, upon Agent's request, shall promptly execute such documents and do such other acts or things deemed appropriate by Agent to deliver to Agent possession of such Documents as are negotiable and such Instruments, and, with respect to non-negotiable Documents, to have such non-negotiable Documents issued in the name of Agent; and with respect to Collateral in the possession of a third party, other than certificated securities and Goods covered by a Document, such Borrower shall obtain an acknowledgment from the third party that is in possession of such Collateral that such third party holds the Collateral for the benefit of Agent.

7.5. <u>No Assumption of Liability</u>. The security interest granted pursuant to this Agreement is granted as security only and shall not subject Agent or any Lender to, or in any way alter or modify, any obligation or liability of Borrowers with respect to or arising out of the Collateral.

7.6. <u>Lien Perfection; Further Assurances</u>. Promptly after Agent's request therefor, Obligors shall execute or cause to be executed and deliver to Agent such instruments, assignments, title certificates or other documents as are necessary under the UCC, the PPSA or other Applicable Law (including any motor vehicle certificates of title act) to perfect (or continue the perfection of) Agent's Lien upon the Collateral and shall take such other action as may be requested by Agent to give effect to or carry out the intent and purposes of this Agreement. Unless prohibited by Applicable Law, each Obligor hereby irrevocably authorizes Agent to execute and file in any jurisdiction any financing statement or amendment thereto on such Obligor's behalf, including financing statements that indicate the Collateral (i) as all assets or all personal property of such Obligor also hereby ratifies its authorization for Agent to have filed in any jurisdiction any like financing statement (or equivalent document in any applicable jurisdiction) or amendment thereto if filed prior to the date hereof. The parties agree that, to the extent permitted by Applicable Law, a carbon, photographic or other reproduction of this Agreement shall be sufficient as a financing statement and may be filed in any appropriate office in lieu thereof.

7.7. <u>Foreign Subsidiary Stock</u>. Notwithstanding anything to the contrary set forth in **Section 7.1** above, the types or items of Collateral described in such Section shall include only sixty-six percent (66%) of the Equity Interests of any Foreign Subsidiary, and shall not include the Equity Interests in any Foreign Subsidiary in existence on the date of this Agreement that is organized under the laws of any jurisdiction other than Canada or a province of Canada and that is listed on **Schedule 7.7** hereto.

SECTION 8. COLLATERAL ADMINISTRATION

8.1. <u>General Provisions</u>.

8.1.1. Location of Collateral . All tangible items of Collateral, other than Inventory in transit, shall at all times be kept by Obligors at one or more of the business locations of Obligors set forth in **Schedule 8.1.1** hereto and shall not be moved therefrom, except that in the absence of an Event of Default and acceleration of the maturity of the Obligations in consequence thereof, Obligors may (i) make sales or other dispositions of any Collateral to the extent authorized by **Section 10.2.10** hereof and (ii) move Inventory or Equipment or any record relating to any Collateral to a

location in the United States or Canada other than those shown on **Schedule 8.1.1** hereto so long as (a) Obligors have given Agent written notice of each such new location by the last day of the Fiscal Quarter in which any Inventory or Equipment were moved to such location, (b) prior to moving any Inventory or Equipment to such location there have been filed any UCC or PPSA financing statements and any other appropriate documentation necessary to perfect or continue the perfection of Agent's first priority Liens with respect to such Inventory or Equipment, and (c) the aggregate fair market value or book value, whichever is more, of the Inventory and Equipment moved to Canada from the United States after the date of this Agreement does not exceed \$1,000,000. Notwithstanding anything to the contrary contained in this Agreement, no Obligor shall be permitted to keep, store or otherwise maintain any Collateral at any location (including any location described in **Section 8.1.1**), unless (i) an Obligor is the owner of such location, (ii) an Obligor leases such location and the landlord has executed in favor of Agent a Lien Waiver, or (iii) the Collateral consists of Inventory placed with a warehouseman, bailee or processor an acceptable Lien Waiver and an appropriate UCC or PPSA financing statement has been filed with the appropriate Governmental Authority in the jurisdiction where such warehouseman, bailee or processor is located in order to perfect, or to maintain the uninterrupted perfection of, Agent's security interest in such Inventory.

8.1.2. Insurance of Collateral; Condemnation Proceeds.

(i) Each Obligor shall maintain and pay for insurance upon all Collateral, wherever located, covering casualty, hazard, public liability, theft, malicious mischief, and such other risks in such amounts and with such insurance companies as are reasonably satisfactory to Agent. **Schedule 8.1.2** describes all insurance of Obligors in effect on the date hereof. All proceeds payable under each such policy shall be payable to Agent for application to the Obligations. Obligors shall deliver the certified copies of such policies to Agent with satisfactory lender's loss payable endorsements reasonably satisfactory to Agent naming Agent as sole loss payee, assignee or additional insured, as appropriate. Each policy of insurance or endorsement shall contain a clause requiring the insurer to give not less than 30 days prior written notice to Agent in the event of cancellation of the policy for any reason whatsoever and a clause specifying that the interest of Agent shall not be impaired or invalidated by any act or neglect of any Obligor or the owner of the Property or by the occupation of the premises for purposes more hazardous than are permitted by said policy. If any Obligor fails to provide and pay for such insurance, Agent may, at its option, but shall not be required to, procure the same and charge Obligors therefor. Each Obligor provide and pay for Obligor shall have the right to settle, adjust and compromise any claim with respect to any insurance maintained by such Obligor provide that all proceeds thereof are applied in the manner specified in this Agreement, and Agent shall promptly provide any necessary endorsement to any checks or drafts issued in payment of any such claim. At any time that an Event of Default exists, only Agent shall be authorized to settle, adjust and compromise such claims, Agent shall have all rights and remedies with respect to such policies of insurance as are provided for in this Agreement and the other Loan Documents.

(ii) Any proceeds of insurance referred to in this **Section 8.1.2** and any condemnation awards that are paid to Agent in connection with a condemnation of any of the Collateral shall be paid to Agent and applied to any Obligations outstanding as Agent shall determine (or, with respect to insurance proceeds paid to Agent in connection with Collateral of a Canadian Obligor, held by Agent as security for the payment of the Canadian Obligor Guarantee of such Canadian Obligor); <u>provided</u>, <u>however</u>, that if an Event of Default exists on the date of Agent's receipt thereof, Agent shall apply such proceeds (other than insurance proceeds with respect to Collateral of a Canadian Obligor) to the Obligations in the order of application

-72-

provided in **Section 5.6.1** (or, in the case of insurance proceeds with respect to Collateral of a Canadian Obligor Agent shall hold such proceeds as security for the payment of the Canadian Obligor Guarantee of such Canadian Obligor). Notwithstanding the foregoing, if requested by Borrowers in writing within 5 days after Agent's receipt of insurance proceeds relating to any loss or destruction of Equipment or Real Estate (during which 5-day period, Agent shall hold such proceeds as security for, but without application to, the Obligations unless a Default or Event of Default exists) and if no Material Adverse Effect could reasonably be expected to result from any insured loss with respect to such Collateral, Agent shall allow Borrowers to use proceeds of insurance to repair or replace any damaged or destroyed Equipment or Real Estate so long as (1) no Default or Event of Default exists, (2) such repair or replacement is promptly undertaken and concluded, (3) replacements of buildings are constructed on the sites of the original casualties and are of comparable size, and quality and utility to the destroyed buildings, (4) the repaired or replaced Property is at all times free and clear of Liens other than Permitted Liens that are not Purchase Money Liens, (5) Borrowers comply with such disbursement procedures for such proceeds as Agent may reasonably impose for repair or replacement, and (6) the amount of proceeds from any single casualty affecting Equipment or Real Estate does not exceed \$1,000,000.

8.1.3. <u>Protection of Collateral</u>. All expenses of protecting, storing, warehousing, insuring, handling, maintaining and shipping any Collateral, all Taxes imposed under any Applicable Law on any of the Collateral or in respect of the sale thereof, and all other payments required to be made by Agent to any Person to realize upon any Collateral shall be borne and paid by Obligors. Agent shall not be liable or responsible in any way for the safekeeping of any of the Collateral or for any loss or damage thereto (except for reasonable care in the custody thereof while any Collateral is in Agent's actual possession) or for any diminution in the value thereof, or for any act or default of any warehouseman, carrier, forwarding agency, or other Person whomsoever, but the same shall be at Borrowers' sole risk.

8.1.4. <u>Defense of Title to Collateral</u>. Each Obligor shall at all times defend such Obligor's title to the Collateral and Agent's Liens therein against all Persons and all claims and demands whatsoever other than Permitted Liens.

8.2. Administration of Accounts.

8.2.1. <u>Records and Schedules of Accounts</u>. Each Obligor shall keep accurate and complete records of its Accounts and all payments and collections thereon and shall submit to Agent on such periodic basis as Agent shall request a sales and collections report for the preceding period, in form satisfactory to Agent. Each Obligor shall also provide to Agent on or before the 25th day of each month, a summary aging of all Accounts existing and lists specifying the names, outstanding face amounts of Accounts of the 10 Account Debtors owing the largest amounts of Accounts and of the 10 largest Approved Account Debtors, all as of the last day of such preceding month (collectively, the "Schedule of Accounts"). Each Obligor shall also provide to Agent, upon Agent's request therefor, a detailed aged trial balance for all Accounts specifying the names, addresses, face value and dates of invoices and due dates for each Account Debtor on an Account so listed and copies of proof of delivery of Inventory and a copy of all documents, including repayment histories and present status reports relating to the Accounts on the Schedule of Accounts or the detailed aged trial balance and such other matters and information relating to the status of then existing Accounts as Agent shall reasonably request. In addition, if Accounts in an aggregate face amount in excess of \$5,000,000 shall cease to be Eligible Accounts in whole or in part, Obligors shall notify Agent of such occurrence promptly (and in any event within 2 Business Days) after any Obligor's having obtained knowledge of such occurrence and the Borrowing Base shall thereupon be adjusted to reflect such occurrence. Each Obligor shall deliver to Agent copies of invoices or invoice registers related to all of its Accounts.

-73-

8.2.2. <u>Discounts, Disputes and Returns</u>. If any Obligor grants any discounts, allowances or credits that are not shown on the face of the invoice for the Account involved, such Obligor shall report such discounts, allowances or credits, as the case may be to Agent as part of the next required Schedule of Accounts. If any amounts due and owing in excess of \$2,500,000 are in dispute between any Obligor and any Account Debtor, or if any returns are made in excess of \$2,500,000 with respect to any Accounts owing from an Account Debtor, such Obligor shall provide Agent with written notice thereof at the time of submission of the next Schedule of Accounts, explaining in detail the reason for the dispute or return, all claims related thereto and the amount in controversy. At any time an Event of Default exists, Agent shall have the right to settle or adjust all disputes and claims directly with the Account Debtor and to compromise the amount or extend the time for payment of any Accounts comprising a part of the Collateral upon such terms and conditions as Agent may deem advisable, and to charge the deficiencies, costs and expenses thereof, including attorneys' fees, to Borrowers.

8.2.3. <u>Taxes</u>. If an Account of any Obligor includes a charge for any Taxes payable to any Governmental Authority, Agent is authorized, in its sole discretion, to pay the amount thereof to the proper taxing authority for the account of such Obligor and to charge Borrowers therefor; provided, however, that neither Agent nor Lenders shall be liable for any Taxes that may be due by Obligors.

8.2.4. <u>Account Verification</u>. Whether or not a Default or an Event of Default exists, Agent shall have the right at any time, in the name of Agent, any designee of Agent or any Obligor to verify the validity, amount or any other matter relating to any Accounts of such Obligor by mail, telephone, telegraph or otherwise. Obligors shall cooperate fully with Agent in an effort to facilitate and promptly conclude any such verification process.

8.2.5. Maintenance of Dominion Account; Trigger Events.

(i) At all times after the occurrence of a Trigger Event (as defined below), Borrowers shall maintain with a Clearing Bank a Dominion Account pursuant to a Lockbox Agreement or other arrangement acceptable to Agent. Borrowers shall issue to each such Clearing Bank an irrevocable letter of instruction directing such bank to deposit all payments or other remittances received in the Lockbox to the Dominion Account after the occurrence of a Trigger Event. Borrowers shall enter into agreements, in form satisfactory to Agent, with each bank at which a Dominion Account is maintained by which such bank shall, after the occurrence of a Trigger Event immediately transfer to the Payment Account all monies deposited to the Dominion Account. All funds deposited in each Dominion Account shall be subject to Agent's Lien. Borrowers shall obtain the agreement (in favor of and in form and content satisfactory to Agent and Lenders) by each bank at which a Dominion Account is maintained to waive any offset rights against the funds deposited into such Dominion Account, except offset rights in respect of charges incurred in the administration of such Dominion Account. Neither Agent nor Lenders assume any responsibility to Borrowers for such lockbox arrangement or Dominion Account, including any claim of accord and satisfaction or release with respect to deposits accepted by any bank thereunder.

(ii) If at any time Availability is less than an amount equal to the greater of (A) \$20,000,000 or (B) ten percent (10%) of Availability (without requiring deduction of the LC Reserve in the calculation thereof) on any date, or if an Event of Default exists (each, a "Trigger Event"), without limiting the right of Agent and the Lenders to exercise other rights and remedies as a result thereof, Agent may require (and at the direction of the Required Lenders, shall require) that all collected balances in each Controlled Account be transferred to Agent not less often than each Business Day and that any or all of the Borrowers establish Lockboxes to which monies, checks, notes, drafts and other payments relating to or constituting proceeds of Collateral shall be sent.

-74-

8.2.6. <u>Collection of Accounts and Proceeds of Collateral</u>. To expedite collection of Accounts, each Obligor shall endeavor, at such Obligor's expense, in the first instance to make collection of such Obligor's Accounts for Agent and Lenders and, in connection therewith, shall use commercially reasonable efforts to keep in full force and effect any Supporting Obligation or collateral security relating to each such Account. Obligors shall (i) request in writing and otherwise take such reasonable steps to ensure that all Account Debtors forward payment, after the occurrence of a Trigger Event, directly to the Dominion Account or to Lockboxes related to the Dominion Account, and (ii) deposit and cause its Subsidiaries to deposit or cause to be deposited promptly, and in any event no later than the first Business Day after the date of receipt thereof, all cash, checks, drafts or other similar items of payment relating to or constituting payments made in respect of any and all Collateral (whether or not otherwise delivered to a Lockbox), after the occurrence of a Trigger Event, into the Dominion Account. Obligors shall issue to each Lockbox bank an irrevocable letter of instruction directing such bank to deposit all payments or other remittances received in the Lockbox, after the occurrence of a Trigger Event, to the Dominion Account, with respect to remittances received in a Borrower's Lockbox. All Payment Items received by any Obligor in respect of its Accounts, together with the proceeds of any other Collateral, shall be held by such Obligor as trustee of an express trust for Agent's and Lenders' benefit; such Obligor shall immediately deposit same in kind, after the oominion Account to Lenders for application to the Obligations in the manner authorized by this Agreement. Agent retains the right at all times that a Default or an Event of Default exists to notify Account Debtors of any Obligor that Accounts have been assigned to Agent and to collect Accounts directly in its own name and to charge to Obligors the c

8.3. Administration of Inventory.

8.3.1. <u>Records and Reports of Inventory</u>. Each Obligor shall keep accurate and complete records of its Inventory (including records showing the cost thereof and daily withdrawals therefrom and additions thereto) and shall furnish Agent inventory reports respecting such Inventory in form and detail satisfactory to Agent at such times as Agent may request, but so long as no Default or Event of Default exists, no more frequently than once each week. Each Obligor shall, at its own expense, conduct a physical inventory no less frequently than annually (and on a more frequent basis if requested by Agent when an Event of Default exists) and periodic cycle counts consistent with such Obligor's historical practices and shall provide to Agent a report based on each such physical inventory and cycle count promptly after completion thereof, together with such supporting information as Agent shall request. Agent may participate in and observe each physical count or inventory, which participation shall be at Borrowers' expense at any time that an Event of Default exists.

8.3.2. <u>Returns of Inventory</u>. No Obligor shall return any of its Inventory to a supplier or vendor thereof, or any other Person, whether for cash, credit against future purchases or then existing payables, or otherwise, unless (i) such return is in the Ordinary Course of Business of such Obligor and such Person; (ii) no Default or Event of Default exists or would result therefrom; (iii) such Obligor promptly notifies Agent thereof if the aggregate value (calculated on the basis of the lower of cost or market, with the cost calculated on a first-in, first-out basis in accordance with GAAP) of all Inventory returned in any month exceeds \$2,500,000; and (iv) any payments received by such Obligor in connection with any such return are promptly turned over to Agent for application to the Obligations.

8.3.3. <u>Acquisitions and Sale of Inventory</u>. No Obligor shall acquire or accept any Inventory on consignment or approval and will use its best efforts to insure that all Inventory that is produced in the United States of America will be produced in accordance with the FLSA. No Obligor

-75-

shall sell any Inventory to any customer on approval or any other basis upon which the customer has a right to return or obligates any Obligor to repurchase such Inventory.

8.3.4. <u>Maintenance of Inventory</u>. Obligors shall produce, use, store and maintain all Inventory with all reasonable care and caution in accordance with applicable standards of any insurance and in conformity with Applicable Law (including the requirements of the FLSA) and will maintain current rent payments (within applicable grace periods provided for in leases) at all locations at which any Inventory is maintained or stored.

8.4. Administration of Equipment.

8.4.1. <u>Records and Schedules of Equipment</u>. Each Obligor shall keep accurate records itemizing and describing the kind, type, quality, quantity and cost of its Equipment and all dispositions made in accordance with **Section 8.4.2** hereof, and shall furnish Agent with a current schedule containing the foregoing information on at least an annual basis and more often if requested by Agent. Promptly after request therefor by Agent, Obligors shall deliver to Agent any and all evidence of ownership, if any, of any of the Equipment.

8.4.2. <u>Dispositions of Equipment</u>. No Obligor shall sell, lease or otherwise dispose of or transfer any of the Equipment or any part thereof, whether in a single transaction or a series of related transactions, without the prior written consent of Agent (acting at the direction of the Required Lenders), other than (i) a disposition of Equipment that qualifies as a Permitted Asset Disposition and (ii) a replacement of Equipment that is substantially worn, damaged or obsolete with Equipment of like kind, function and value, provided that the replacement Equipment shall be acquired prior to or concurrently with any disposition of the Equipment that is to be replaced, the replacement Equipment shall be free and clear of Liens other than Permitted Liens that are not Purchase Money Liens, and Obligors shall have given Agent at least 10 days prior written notice of such disposition.

8.4.3. <u>Condition of Equipment</u>. The Equipment is in good operating condition and repair, and all necessary replacements of and repairs thereto shall be made so that the value and operating efficiency of the Equipment shall be maintained and preserved, reasonable wear and tear excepted. Each Obligor shall ensure that the Equipment shall be mechanically and structurally sound, capable of performing the functions for which the Equipment was originally designed, in accordance with the manufacturer's published and recommended specifications. No Obligor shall permit any of the Equipment to become affixed to any real Property leased to such Obligor so that an interest arises therein under the real estate laws of the applicable jurisdiction unless the landlord of such real Property has executed a Lien Waiver in favor of and in form acceptable to Agent, and no Obligor will permit any of the Equipment to become an accession to any personal Property that is subject to a Lien unless the Lien is a Permitted Lien.

8.4.4. <u>Certificated Equipment</u>. With respect to any Equipment that is covered by a certificate of title under a law requiring indication of a security interest on such certificate as a condition to the perfection of such security interest, each Obligor shall, at the time of acquisition of each such item of Equipment after the date hereof and on Agent's request with respect to all other such items of Equipment, execute and file with the Registrar of Motor Vehicles or other appropriate Governmental Authority an application or other document requesting the notation or other indication of the security interest created hereunder on such certificate of title with respect to such Equipment. All certificates of title with respect to such Equipment shall be (i) held at MasTec's chief executive office in the custody and control of an appropriate officer or senior employee of MasTec, acceptable to Agent, pursuant to a limited agency agreement with Agent in respect of such certificates of title in form and substance satisfactory to Agent (the "Agency Appointment as to Vehicle Titles") and shall be deemed held by Obligors as trustees

-76-

of an express trust for the benefit of Agent and Secured Parties, and (ii) if at any time Availability (A) either is less than the greater of (I) \$20,000,000 or (II) ten percent (10%) of Availability (without requiring deduction of the LC Reserve in the calculation thereof) on any date, or (B) averages less than \$15,000,000 for any period of five (5) consecutive days then ending, in each case, regardless of whether or not a Default or Event of Default then exists, then Agent shall take and Obligors shall immediately deliver or surrender to Agent, or its designee, possession of all such certificates of title. Upon the request of Agent at any time, each Obligor shall execute and deliver to Agent an irrevocable power of attorney naming Agent as such Obligor's agent and attorney-infact to take any action in such Obligor's name, and to sign such Obligor's name to any documents, to facilitate Agent's perfection of a Lien in the Equipment evidenced by such certificates of title and Agent's enforcement of such Lien, including any repossession of or foreclosure upon such Equipment.

8.5. <u>Administration of Deposit Accounts</u>. Each Obligor represents that, as of the Closing Date, **Schedule 8.5** sets forth all of the Deposit Accounts maintained by each Obligor, including Deposit Accounts into which all Payment Items relating to any Collateral will be deposited; each Obligor is the sole account holder of each such Deposit Account and is not aware of any Person (other than Agent) having either sole dominion or control (within the meaning of Section 9-104 of the UCC) over any such Deposit Account or any property deposited therein (other than any such control that has been released or terminated on or before the Closing Date) and control arising by operation of law in favor the depository bank in which such Deposit Account is maintained.

8.6. <u>Borrowing Base Certificates</u>. On the Closing Date and on or before the 25th day of each month after the Closing Date, Borrowers shall deliver to Agent (and Agent shall, promptly deliver to each Lender) a Borrowing Base Certificate prepared as of the close of business on the last day of the preceding month, and at such other times as Agent may request. All calculations of Availability in connection with any Borrowing Base Certificate shall originally be made by Borrowers and certified by a Senior Officer to Agent, provided that Agent shall have the right to review and adjust, in the exercise of its credit judgment, any such calculation (i) to reflect its reasonable estimate of declines in value of any of the Collateral described therein or any disposition of Collateral and (ii) to the extent that such calculation is not in accordance with this Agreement or does not accurately reflect the amount of the Availability Reserve. In no event shall the Borrowing Base on any date be deemed to exceed the amount of the Borrowing Base shown on the Borrowing Base Certificate last received by Agent prior to such date, as the calculation in such Borrowing Base Certificate may be adjusted from time to time by Agent as herein authorized.

SECTION 9. REPRESENTATIONS AND WARRANTIES

9.1. <u>General Representations and Warranties</u>. To induce Agent, Syndication Agent and Lenders to enter into this Agreement and to make available the Commitments, each Obligor warrants and represents to Agent and Lenders that:

9.1.1. <u>Organization and Qualification</u>. Each Obligor and each of its Subsidiaries is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization. Each Obligor and each of its Subsidiaries is duly qualified and is authorized to do business and is in good standing as a foreign corporation in each state or jurisdiction listed on **Schedule 9.1.1** hereto and in all other states and jurisdictions in which the failure of such Obligor or any of such Subsidiaries to be so qualified would have a Material Adverse Effect.

9.1.2. <u>Power and Authority</u>. Each Obligor and each of its Subsidiaries is duly authorized and has the corporate, limited liability or other entity power, as the case may be, to enter into, execute, deliver and perform this Agreement and each of the other Loan Documents to which it is a party.

-77-

The execution, delivery and performance of this Agreement and each of the other Loan Documents have been duly authorized by all necessary entity action and do not and will not (i) require any consent or approval of any of the holders of the Equity Interests of any Obligor or any of its Subsidiaries other than those obtained on or prior to the date hereof; (ii) contravene the Organic Documents of any Obligor or any of its Subsidiaries; (iii) violate, or cause any Obligor or any of its Subsidiaries to be in default under, any provision of any Applicable Law, order, writ, judgment, injunction, decree, determination or award in effect having applicability to any Obligor or any of its Subsidiaries, in each case where such violation or default could reasonably be expected to have a Material Adverse Effect; (iv) result in a breach of or constitute a default under any indenture or loan or credit agreement or any other agreement, lease or instrument to which any Obligor or any of its Subsidiaries is a party or by which it or its Properties may be bound or affected; or (v) result in, or require, the creation or imposition of any Lien (other than Permitted Liens) upon or with respect to any of the Properties now owned or hereafter acquired by any Obligor or any of its Subsidiaries.

9.1.3. Legally Enforceable Agreement. This Agreement is, and each of the other Loan Documents when delivered under this Agreement will be, a legal, valid and binding obligation of each Obligor and each of its Subsidiaries signatories thereto enforceable against them in accordance with the respective terms of such Loan Documents, except as the enforceability thereof may be limited by bankruptcy, insolvency or other similar laws of general application affecting the enforcement of creditors' rights.

9.1.4. <u>Capital Structure</u>. As of the date hereof, **Schedule 9.1.4** hereto states (i) the correct name of each Obligor, its jurisdiction of formation and the percentage of its Equity Interests having voting powers owned by each Person, (ii) the name of each Obligor's corporate Affiliates and the nature of the affiliation and (iii) the number of authorized and issued Equity Interests (and treasury shares) of each Obligor and its Subsidiaries. Each Obligor has good title to all of the shares it purports to own of the Equity Interests of each of its Subsidiaries, free and clear in each case of any Lien other than Permitted Liens. All such Equity Interests have been duly issued and are fully paid and non-assessable. Since the date of the financial statements of Borrowers referred to in **Section 9.1.9** hereof, no Borrower has made, or obligated itself to make, any Distribution. There are no outstanding options to purchase, or any rights or warrants to subscribe for, or any commitments or agreements to issue or sell, or any Equity Interests or obligations convertible into, or any powers of attorney relating to, shares of the capital stock of any Obligor or any of its Subsidiaries. Except as set forth on **Schedule 9.1.4** hereto, there are no outstanding agreements or instruments binding upon the holders of any Obligor's Equity Interests relating to the ownership of its Equity Interests.

9.1.5. <u>Corporate Names</u>. During the 5-year period preceding the date of this Agreement, no Obligor nor any of its Subsidiaries has been known as or used any corporate, fictitious or trade names except those listed on **Schedule 9.1.5** hereto. Except as set forth on **Schedule 9.1.5**, no Obligor nor any of its Subsidiaries has been the surviving corporation of a merger or consolidation or acquired all or substantially all of the assets of any Person.

9.1.6. <u>Business Locations; Agent for Process</u>. As of the date hereof, the chief executive office and other places of business of each Obligor and its Subsidiaries are as listed on **Schedule 8.1.1** hereto. During the 5-year period preceding the date of this Agreement, no Obligor nor any of its Subsidiaries has had an office, place of business or agent for service of process other than as listed on **Schedule 8.1.1**. Except as shown on **Schedule 8.1.1** on the date hereof, no Inventory of any Obligor or any of its Subsidiaries is stored with a bailee, warehouseman or similar Person, nor is any Inventory consigned to any Person.

-78-

9.1.7. <u>Title to Properties; Priority of Liens</u>. Each Obligor and each of its Subsidiaries has good and marketable title to and fee simple ownership of, or valid and subsisting leasehold interests in, all of its real Property, and good title to all of its personal Property, including all Property reflected in the financial statements referred to in **Section 9.1.9** or delivered pursuant to **Section 10.1.3**, in each case free and clear of all Liens except Permitted Liens. Each Obligor has paid or discharged, and has caused each of its Subsidiaries to pay and discharge, all lawful claims which, if unpaid, might become a Lien against any Properties of such Obligor or any such Subsidiary that is not a Permitted Lien. The Liens granted to Agent pursuant to this Agreement and the other Security Documents are duly perfected, first priority Liens, subject only to those Permitted Liens that are expressly permitted by the terms of this Agreement to have priority over the Liens of Agent.

9.1.8. <u>Accounts</u>. Agent may rely, in determining which Accounts are Eligible Accounts, on all statements and representations made by Obligors with respect to any Account. Unless otherwise indicated in writing to Agent, with respect to each Account, each Borrower warrants that:

(i) It is genuine and in all respects what it purports to be, and it is not evidenced by a judgment;

(ii) It arises out of a completed, *bona fide* sale and delivery of goods or rendition of services by an Obligor in the Ordinary Course of Business and substantially in accordance with the terms and conditions of all purchase orders, contracts or other documents relating thereto and forming a part of the contract between an Obligor and the Account Debtor;

(iii) It is for a sum certain maturing as stated in the duplicate invoice covering such sale or rendition of services, a copy of which has been furnished or is available to Agent on request;

(iv) Such Account, and Agent's security interest therein, is not, and will not (by voluntary act or omission of an Obligor) be in the future, subject to any offset (except with respect to Accounts for which DirecTV is the Account Debtor, to the extent such offset is adequately reserved for in the DirecTV Reserve), Lien, deduction, defense, dispute, counterclaim or any other adverse condition except for disputes resulting in returned goods where the amount in controversy is deemed by Agent to be immaterial, and each such Account is absolutely owing to an Obligor and is not contingent in any respect or for any reason;

(v) The contract under which such Account arose does not condition or restrict an Obligor's right to assign to Agent the right to payment thereunder unless such Obligor has obtained the Account Debtor's consent to such collateral assignment or complied with any conditions to such assignment (regardless of whether under the UCC or other Applicable Law any such restrictions are ineffective to prevent the grant of a Lien upon such Account in favor of Agent);

(vi) Such Obligor has not made any agreement with any Account Debtor thereunder for any extension, compromise, settlement or modification of any such Account or any deduction therefrom, except discounts or allowances which are granted by an Obligor in the Ordinary Course of Business for prompt payment and which are reflected in the calculation of the net amount of each respective invoice related thereto and are reflected in the Schedules of Accounts submitted to Agent pursuant to **Section 8.2.1** hereof;

(vii) To the best of such Obligor's knowledge, there are no facts, events or occurrences which are reasonably likely to impair the validity or enforceability of such Account

-79-

or reduce the amount payable thereunder from the face amount of the invoice and statements delivered to Agent with respect thereto;

(viii) To the best of such Obligor's knowledge, the Account Debtor thereunder (1) had the capacity to contract at the time any contract or other document giving rise to the Account was executed and (2) is Solvent; and

(ix) To the best of such Obligor's knowledge, there are no proceedings or actions which are threatened or pending against any Account Debtor thereunder and which are reasonably likely to result in any material adverse change in such Account Debtor's financial condition or the collectibility of such Account.

9.1.9. <u>Financial Statements</u>; <u>Fiscal Year</u>. The Consolidated and consolidating balance sheets of Borrowers and such other Persons described therein (including the accounts of all Subsidiaries of Borrowers for the respective periods during which a Subsidiary relationship existed) as of December 31, 2004, and the related statements of income, changes in stockholder's equity, and changes in financial position for the periods ended on such dates, have been prepared in accordance with GAAP, and present fairly the financial positions of Borrowers and such Persons at such dates and the results of Borrowers' operations for such periods. Since December 31, 2004, there has been no material change in the condition, financial or otherwise, of Borrowers and such other Persons as shown on the Consolidated balance sheet as of such date and no material change in the aggregate value of Equipment and Real Estate owned by Borrowers or such other Persons.

9.1.10. <u>Full Disclosure</u>. The financial statements referred to in **Section 9.1.9** hereof do not contain any untrue statement of a material fact and neither this Agreement nor any other written statement contains or omits any material fact necessary to make the statements contained herein or therein not materially misleading. There is no fact or circumstances in existence on the date hereof which any Borrower has failed to disclose to Agent in writing that may reasonably be expected to have a Material Adverse Effect.

9.1.11. <u>Solvent Financial Condition</u>. Each Obligor and its Subsidiaries is now Solvent and, after giving effect to the Loans to be made hereunder, the LC Obligations to be incurred in connection herewith and the consummation of the other transactions described in the Loan Documents, will be Solvent.

9.1.12. <u>Surety Obligations</u>. Except as set forth on **Schedule 9.1.12** hereto on the date hereof, no Borrower nor any of its Subsidiaries is obligated as surety or indemnitor under any surety or similar bond or other contract issued or entered into any agreement to assure payment, performance or completion of performance of any undertaking or obligation of any Person.

9.1.13. <u>Taxes</u>. The FEIN of each of each Borrower and each of its Subsidiaries is as shown on **Schedule 9.1.13** hereto. Each Borrower and each of its Subsidiaries has filed all federal, state and local tax returns and other reports it is required by law to file and has paid, or made provision for the payment of, all Taxes upon it, its income and Properties as and when such Taxes are due and payable, except to the extent being Properly Contested. The provision for Taxes on the books of each Borrower and each of its Subsidiaries are adequate for all years not closed by applicable statutes, and for its current Fiscal Year.

9.1.14. <u>Brokers</u>. There are no claims against any Obligor for brokerage commissions, finder's fees or investment banking fees in connection with the transactions contemplated by this Agreement or any of the other Loan Documents.

-80-

9.1.15. Intellectual Property. Each Obligor and each of its Subsidiaries owns or has the lawful right to use all Intellectual Property necessary except where such conflict could not reasonably be expected to have a Material Adverse Effect for the present and planned future conduct of its business without any conflict with the rights of others; there is no objection to, or pending (or, to Obligors' knowledge, threatened) Intellectual Property Claim with respect to, any Obligor's or any of its Subsidiaries' right to use any material Intellectual Property and such Obligor is not aware of any grounds for challenge or objection thereto; and, except as may be disclosed on **Schedule 9.1.15**, no Obligor nor any of its Subsidiaries pays any royalty or other compensation to any Person for the right to use any Intellectual Property. All such patents, trademarks, service marks, trade names, copyrights, licenses and other similar rights are listed on **Schedule 9.1.15** hereto, to the extent they are registered under any Applicable Law or are otherwise material to any Obligor's or any of its Subsidiaries' business.

9.1.16. <u>Governmental Approvals</u>. Each Obligor and each of its Subsidiaries has, and is in good standing with respect to, all Governmental Approvals necessary to continue to conduct its business as heretofore or proposed to be conducted by it and to own or lease and operate its Properties as now owned or leased by it except where such failure could not reasonably be expected to have a Material Adverse Effect.

9.1.17. <u>Compliance with Laws</u>. Each Obligor and each of its Subsidiaries has duly complied with, and its Properties, business operations and leaseholds are in compliance in all material respects with, the provisions of all Applicable Law (except to the extent that any such noncompliance with Applicable Law could not reasonably be expected to have a Material Adverse Effect) and there have been no citations, notices or orders of noncompliance issued to any Obligor or any of its Subsidiaries under any such law, rule or regulation. No Inventory has been produced in violation of the FLSA. With respect to matters arising under any Environmental Laws, the representations and warranties contained in the Environmental Agreement are true and correct on the date hereof.

9.1.18. <u>Burdensome Contracts</u>. No Obligor nor any of its Subsidiaries is a party or subject to any contract, agreement, or charter or other corporate restriction, which has or could be reasonably expected to have a Material Adverse Effect. No Obligor nor any of its Subsidiaries is a party or subject to any Restrictive Agreements, except for (i) the DirectStar Purchase Agreement and (ii) such other agreements as are set forth on **Schedule 9.1.18** hereto, none of which agreements referenced in (i) or (ii) prohibit the execution or delivery of any of the Loan Documents by any Obligor or the performance by any Obligor of its obligations under any of the Loan Documents to which it is a party, in accordance with the terms of such Loan Documents.

9.1.19. <u>Litigation</u>. Except as set forth on **Schedule 9.1.19** hereto, there are no actions, suits, proceedings or investigations pending or, to the knowledge of any Obligor, threatened on the date hereof against or affecting any Obligor or any of its Subsidiaries, or the business, operations, Properties, prospects, profits or condition of any Obligor or any of its Subsidiaries, (i) which relate to any of the Loan Documents or any of the transactions contemplated thereby or (ii) which, if determined adversely to any Obligor or any of its Subsidiaries, could reasonably be expected to have a Material Adverse Effect.

9.1.20. <u>No Defaults</u>. No event has occurred and no condition exists which would, upon or after the execution and delivery of this Agreement or any Obligor's performance hereunder, constitute a Default or an Event of Default. No Obligor nor any of its Subsidiaries is in default, and no event has occurred and no condition exists which constitutes or which with the passage of time or the giving of notice or both would constitute a default, under any Material Contract or in the payment of any Debt of a Obligor or a Subsidiary to any Person for Money Borrowed.

-81-

9.1.21. Leases. Schedule 9.1.21 hereto is a complete listing of each capitalized and operating lease of each Obligor on the date hereof that constitutes a Material Contract. Each Obligor is in substantial compliance with all of the terms of each of its respective capitalized and operating leases and there is no basis upon which the lessors under any such leases could terminate same or declare such Obligor or any of its Subsidiaries in default thereunder.

9.1.22. <u>ERISA; Plans</u>. Except as disclosed on **Schedule 9.1.22** hereto, no Obligor nor any of its Subsidiaries has any Plan on the date hereof. Each Obligor and each of its Subsidiaries is in full compliance with the requirements of ERISA or any similar provision of the PBA and the regulations promulgated thereunder with respect to each Plan except where such noncompliance could not reasonably be expected to have a Material Adverse Effect. No fact or situation that is reasonably likely to result in a Material Adverse Effect exists in connection with any Plan. No Obligor nor any of its Subsidiaries has any withdrawal liability in connection with a Multiemployer Plan.

9.1.23. <u>Trade Relations</u>. There exists no actual or (to Obligors' knowledge, threatened) termination, cancellation or limitation of, or any materially adverse modification or change in, the business relationship between any Obligor and any customer or any group of customers whose purchases individually or in the aggregate are material to the business of such Obligor, or with any material supplier or group of suppliers, and there exists no condition or state of facts or circumstances which is reasonably likely to have a Material Adverse Effect or prevent any Obligor from conducting such business after the consummation of the transactions contemplated by this Agreement in substantially the same manner in which it has heretofore been conducted.

9.1.24. <u>Labor Relations</u>. Except as described on **Schedule 9.1.24** hereto, no Obligor nor any of its Subsidiaries is on the date hereof a party to or bound by any collective bargaining agreement, management agreement or consulting agreement. On the date hereof, there are no material grievances, disputes or controversies with any union or any other organization of any Obligor's or any Subsidiary's employees, or, to any Obligor's knowledge, any threats of strikes, work stoppages or any asserted pending demands for collective bargaining by any union or organization.

9.1.25. Not a Regulated Entity. No Obligor is (i) an "investment company" or a "person directly or indirectly controlled by or acting on behalf of an investment company" within the meaning of the Investment Company Act of 1940; (ii) a "holding company," or a "subsidiary company" of a "holding company," or an "affiliate" of a "holding company" or of a "subsidiary company" of a "holding company," within the meaning of the Public Utility Holding Company Act of 1935; or (iii) subject to regulation under the Federal Power Act, the Interstate Commerce Act, any public utilities code or any other Applicable Law regarding its authority to incur Debt.

9.1.26. <u>Margin Stock</u>. No Obligor nor any of its Subsidiaries is engaged, principally or as one of its important activities, in the business of extending credit for the purpose of purchasing or carrying any Margin Stock.

9.1.27. Anti-Terrorism Laws.

(i) <u>General</u>. No Borrower nor any of its Affiliates is in violation of any Anti-Terrorism Law or engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law.

(ii) Executive Order No. 13224.

-82-

(a) No Borrower nor any of its Affiliates is any of the following (each a "Blocked Person"): (1) a Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224; (2) a Person owned or controlled by, or acting for or on behalf of, any Person that is listed in the annex to, or is otherwise subject to the provisions of, Executive Order No. 13224; (3) a Person or entity with which any bank or other financial institution is prohibited from dealing or otherwise engaging in any transaction by any Anti-Terrorism Law; (4) a Person or entity that commits, threatens or conspires to commit or supports "terrorism" as defined in Executive Order No. 13224; (5) a Person or entity that is named as a "specially designated national" on the most current list published by the U.S. Treasury Department Office of Foreign Asset Control ("OFAC") at its official website or any replacement website or other replacement official publication of such list; (6) a Person or entity who is affiliated with a Person or entity listed above; or (7) an agency of the government of, an organization directly or indirectly controlled by, or a Person resident in, a country on any official list maintained by OFAC.

(b) No Borrower nor any of its Affiliates (1) conducts any business or engages in making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person, (2) has any of its assets in a Blocked Person, (3) deals in, or otherwise engages in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224, or (4) derives any of its operating income from investments in or transactions with a Blocked Person.

9.1.28. <u>Payable Practices</u>. No Borrower nor any of its Subsidiaries has made any material change in its historical accounts payable practices from those in effect immediately prior to the Closing Date.

9.1.29. <u>Not the Holder of Plan Assets</u>. No Obligor is an entity deemed to hold "plan assets" within the meaning of 29 C.F.R. §2510.3-101 of an "employee benefit plan" (as defined in Section 3(3) of ERISA) that is subject to Title I of ERISA or any "plan" (within the meaning of Section 4975 of the Internal Revenue Code or any similar provision of the PBA), and neither the execution of this Agreement nor the funding of any Loans gives rise to a prohibited transaction within the meaning of Section 406 of ERISA or Section 4975 of the Internal Revenue Code or any similar provision of the PBA.

9.1.30. <u>Real Estate</u>. None of the Real Estate of any Borrower that is subject to the Lien of a Mortgage is located in an area that has been identified by the Secretary of Housing and Urban Development as an area having special flood hazards and in which flood insurance has been made available under the National Flood Insurance Act of 1968.

9.2. <u>Reaffirmation of Representations and Warranties</u>. Each representation and warranty contained in this Agreement and the other Loan Documents shall be deemed to be reaffirmed on each day that Borrowers request or are deemed to have requested any extension of credit hereunder, except for changes in the nature of an Obligor's or, if applicable, any Subsidiary's business or operations that may occur after the date hereof in the Ordinary Course of Business so long as Agent has consented to such changes or such changes are not violative of any provision of this Agreement. Notwithstanding the foregoing, representations and warranties which by their terms are applicable only as of a specific date shall be deemed made only at and as of such date.

9.3. <u>Survival of Representations and Warranties</u>. All representations and warranties of Obligors contained in this Agreement or any of the other Loan Documents shall survive the execution,

-83-

delivery and acceptance thereof by Agent, Lenders and the parties thereto and the closing of the transactions described therein or related thereto.

SECTION 10. COVENANTS AND CONTINUING AGREEMENTS

10.1. <u>Affirmative Covenants</u>. For so long as there are any Commitments outstanding and thereafter until Full Payment of the Obligations, each Obligor covenants that it shall and shall cause each Subsidiary to:

10.1.1. <u>Visits and Inspections</u>. Permit representatives of Agent, from time to time, as often as may be reasonably requested, but only during normal business hours and (except when a Default or Event of Default exists) upon reasonable prior notice to an Obligor, to visit and inspect the Properties of such Obligor and each of its Subsidiaries, inspect, audit and make extracts from such Obligor's and each Subsidiary's books and records, and discuss with its officers, its employees and its independent accountants, such Obligor's and each Subsidiary's business, financial condition, business prospects and results of operations. Representatives of each Lender shall be authorized to accompany Agent on each such visit and inspection and to participate with Agent therein, but at their own expense, unless a Default or Event of Default exists. Neither Agent nor any Lender shall have any duty to make any such inspection and shall not incur any liability by reason of its failure to conduct or delay in conducting any such inspection.

10.1.2. Notices. Notify Agent and Lenders in writing, promptly after an Obligor's obtaining knowledge thereof, (i) of the commencement of any litigation affecting any Obligor, whether or not the claims asserted in such litigation are considered by Obligors to be covered by insurance, and of the institution of any administrative proceeding, to the extent that such litigation or proceeding, if determined adversely to such Obligor, could reasonably be expected to have a Material Adverse Effect; (ii) of any material labor dispute to which any Obligor may become a party, any pending or threatened strikes or walkouts relating to any of its plants or other facilities, and the expiration of any labor contract to which it is a party or by which it is bound; (iii) of any material default by any Obligor under, or termination of, any Material Contract or any note, indenture, loan agreement, mortgage, lease, deed, guaranty or other similar agreement relating to any Debt of such Obligor exceeding \$2,500,000; (iv) of the existence of any Default or Event of Default; (v) of any default by any Person under any note or other evidence of Debt payable to an Obligor in an amount exceeding \$2,500,000; (vi) of any judgment against any Obligor in an amount exceeding \$1,000,000; (vii) of the assertion by any Person of any Intellectual Property Claim, the adverse resolution of which could reasonably be expected to have a Material Adverse Effect; (viii) of any violation or asserted violation by any Obligor of any Applicable Law (including ERISA, the PBA, OSHA, FLSA, or any Environmental Laws), the adverse resolution of which could reasonably be expected to have a Material Adverse Effect; (ix) of any Environmental Release by an Obligor or on any Property owned or occupied by an Obligor or of the receipt by any Obligor of an Environmental Notice; (x) of the discharge of Borrowers' independent accountants or any withdrawal or resignation by such independent accountants from their acting in such capacity; (xi) with respect to any surety that has issued a payment or performance bond for the benefit of one or more Obligors, (a) such surety's decision to cease considering requests by any Obligor for the issuance of bonds generally, to the extent MasTec has knowledge thereof, (b) such surety's decision to take possession or control of the work under any bonded contract in order to complete such bonded contract, (c) the making of any payment by such surety pursuant to a bond, and (d) any amendment to the indemnity agreement or any other agreement in effect at any time between such surety and one or more Obligors relating to the issuance of bonds, together with a copy of such amendment; and (xii) the receipt of any notice from or giving of any notice to the trustee under the Indenture, together with a copy of such notice. In addition, Borrowers shall give Agent written notice of any Obligor's opening of any new office or place of business. Furthermore, Borrowers shall notify Agent promptly upon any Obligor (x) being

-84-

required to file reports under Section 15(b) of the Securities Exchange Act of 1934, (y) registering securities under Section 12 of the Securities Exchange Act of 1934 or (z) filing a registration statement under the Securities Act of 1933.

10.1.3. <u>Financial and Other Information</u>. Keep adequate records and books of account with respect to its business activities in which proper entries are made in accordance with GAAP reflecting all its financial transactions; and cause to be prepared and furnished to Agent and Lenders the following (all to be prepared in accordance with GAAP applied on a consistent basis, unless Borrowers' certified public accountants concur in any change therein, such change is disclosed to Agent and is consistent with GAAP and, if required by the Required Lenders, the financial covenants set forth in **Section 10.3** are amended in a manner requested by the Required Lenders to take into account the effects of such change):

(i) as soon as available, and in any event within 90 days after the close of each Fiscal Year, audited consolidated balance sheet of Borrowers and their respective Subsidiaries prepared as of the end of such Fiscal Year and accompanied by the related statements of income, shareholders' equity and cash flow, and certified without an Impermissible Qualification by a firm of independent certified public accountants of recognized national standing selected by Borrowers but reasonably acceptable to Agent, and setting forth in each case in comparative form the corresponding Consolidated figures for the preceding Fiscal Year. The audited consolidated balance sheet and related statements of income, shareholders' equity and cash flow shall be accompanied by supplemental schedules showing, on a Consolidated and consolidating basis (including DirectStar), the balance sheet of Borrowers and their respective Subsidiaries, prepared as of the end of such Fiscal Year and accompanied by the related statements of income, shareholders' equity and cash flow;

(ii) as soon as available, and in any event within 30 days after the end of each month hereafter (but (a) for the last month in each of the first 3 Fiscal Quarters of any Fiscal Year, on the sooner to occur of 45 days after the end of such month or the date on which MasTec files with the SEC its Form 10-Q for the Fiscal Quarter ending on such date, and (b) within 45 days after the last month in a Fiscal Year), including the last month of Borrowers' Fiscal Year, unaudited Consolidated and consolidating balance sheets of Borrowers and their respective Subsidiaries (including DirectStar), in each case as of the end of such month, and the related unaudited Consolidated statements of income and cash flow for such month and for the portion of the Fiscal Year then elapsed (including DirectStar), on a Consolidated and consolidating basis, setting forth in each case in comparative form the corresponding figures for the preceding Fiscal Year and certified by the principal financial officer of Borrowers as prepared in accordance with GAAP and fairly presenting the Consolidated and consolidating financial position and results of operations of Borrowers and their Subsidiaries, in each case for such month and period subject only to changes from audit and year-end adjustments and except that such statements need not contain notes;

(iii) as soon as available, and in any event within 30 days after the end of each month hereafter, unaudited balance sheets of DirectStar, as of the end of such month, and the related unaudited Consolidated statements of income and cash flow for such month and for the portion of the Fiscal Year then elapsed, on a Consolidated and consolidating basis, setting forth in each case in comparative form the corresponding figures for the preceding Fiscal Year and certified by the principal financial officer of DirectStar, as prepared in accordance with GAAP and fairly presenting the Consolidated and consolidating financial position and results of operations of DirectStar, for such month and period subject only to changes from audit and year-end adjustments and except that such statements need not contain notes;

-85-

(iv) not later than 25 days after each month, a summary aging of each Obligor's trade payables as of the last Business Day of such month, and, at Agent's request, a listing of all of each Obligor's trade payables, specifying the name of and balance due each trade creditor and monthly detailed trade payable agings, each in form acceptable to Agent;

(v) promptly after the sending or filing thereof, as the case may be, copies of any proxy statements, financial statements or reports which any Obligor has made generally available to its shareholders; copies of any regular, periodic and special reports or registration statements or prospectuses which any Obligor files with the SEC or any Governmental Authority which may be substituted therefor, or any national securities exchange; and copies of any press releases or other statements made available by an Obligor to the public concerning material changes to or developments in the business of such Obligor;

(vi) promptly after the sending or filing thereof, copies of any annual report to be filed in accordance with ERISA or any similar provision of the PBA in connection with each Plan and such other data and information (financial or otherwise) as Agent, from time to time, may request, bearing upon or related to the Collateral or any Borrower's or any Subsidiary's financial condition or results of operations; and

(vii) quarterly within 60 days after the end of each Fiscal Quarter, a certificate of the senior officer of MasTec serving as agent of Agent, in such form as Agent and the Borrowers may agree, listing all title certificates for vehicles owned by Obligors as of the last day of such Fiscal Quarter, indicating vehicles sold or transferred and vehicles purchased since the date of the last such certificate and confirming that all title certificates for such purchased vehicles with a purchase price in excess of \$2,500 reflect Agent as the first lienor thereof (or that applications for new title certificates reflecting such Lien have been properly made).

Concurrently with the delivery of the financial statements described in clause (i) of this **Section 10.1.3**, Borrowers shall deliver to Agent a copy of the accountants' letter to Borrowers' management that is prepared in connection with such financial statements. Concurrently with the delivery of the financial statements described in clauses (i) and (ii) of this **Section 10.1.3**, or more frequently if requested by Agent during any period that a Default or Event of Default exists, Borrowers shall cause to be prepared and furnished to Agent a Compliance Certificate executed by the chief financial officer of Borrowers.

10.1.4. <u>Landlord and Storage Agreements</u>. Provide Agent with copies of all existing agreements, and promptly after execution thereof provide Agent with copies of all future agreements, between any Obligor and any landlord, warehouseman or bailee which owns any premises at which any Collateral may, from time to time, be kept.

10.1.5. <u>Projections</u>. Deliver to Agent the Projections of Borrowers as soon as available, and in any event no later than (i) 15 days prior to the end of each Fiscal Year of Borrowers in draft form for the forthcoming Fiscal Year, month by month, and (ii) January 31 of each year, in final form for the then current Fiscal Year.

10.1.6. <u>Taxes</u>. Pay and discharge all Taxes prior to the date on which such Taxes become delinquent or penalties attach thereto, except and to the extent only that such Taxes are being Properly Contested.

10.1.7. <u>Compliance with Laws</u>. Comply with all Applicable Law, including ERISA, all Environmental Laws, FLSA, OSHA, Anti-Terrorism Laws and all laws, statutes, regulations

-86-

and ordinances regarding the collection, payment and deposit of Taxes, and obtain and keep in force any and all Governmental Approvals necessary to the ownership of its Properties or to the conduct of its business, in each case to the extent that any such failure to comply, obtain or keep in force could be reasonably expected to have a Material Adverse Effect. Without limiting the generality of the foregoing, if any Environmental Release shall occur at or on any of the Properties of any Borrower or any of its Subsidiaries, Borrowers shall, or shall cause the applicable Subsidiary to, act promptly and diligently to investigate and report to Agent and all appropriate Governmental Authorities the extent of, and to make appropriate remedial action to eliminate, such Environmental Release, whether or not ordered or otherwise directed to do so by any Governmental Authority.

10.1.8. <u>Insurance</u>. In addition to the insurance required herein with respect to the Collateral, maintain with its current insurers or with other financially sound and reputable insurers having a rating of at least A or better by *Best's Ratings*, a publication of A.M. Best Company, (i) insurance with respect to its Properties and business against such casualties and contingencies of such type (including product liability, workers' compensation, larceny, embezzlement, or other criminal misappropriation insurance) and in such amounts and with such coverages, limits and deductibles as is customary in the business of such Borrower or such Subsidiary and (ii) business interruption insurance in an amount not less than \$3,000,000 per occurrence.

10.1.9. <u>Intellectual Property</u>. Promptly after applying for or otherwise acquiring any Intellectual Property, deliver to Agent, in form and substance acceptable to Agent and in recordable form, all documents necessary for Agent to perfect its Lien on such Intellectual Property.

10.1.10. <u>License Agreements</u>. Keep each License Agreement in full force and effect for so long as any Obligor has any Inventory, the manufacture, sale or distribution of which is in any manner governed by or subject to such License Agreement.

10.1.11. <u>Future Subsidiaries</u>. Each Borrower shall promptly notify Agent upon any Person becoming a Subsidiary, or upon an Obligor directly or indirectly acquiring additional Equity Interests of any existing Subsidiary, and

(i) Any such Person (other than DirectStar) shall, if it is a Domestic Subsidiary, (i) execute and deliver to Agent a Joinder Agreement or, if elected by Agent, a Subsidiary Guaranty and Subsidiary Security Agreement and (ii) to the extent such Domestic Subsidiary is required to pledge Equity Interests of a Subsidiary pursuant to clause (ii) of this Section, become a party to the Pledge Agreement, if not already a party thereto as a pledgor, in a manner reasonably satisfactory to Agent;

(ii) Each Borrower and each Domestic Subsidiary (other than DirectStar) shall, pursuant to the Pledge Agreement, pledge to Agent all of the outstanding Equity Interests of each such new Domestic Subsidiary (and 66% of the Equity Interests of each new Foreign Subsidiary, if requested by Agent or the Required Lenders) owned directly by such Borrower or such Domestic Subsidiary, and shall deliver to Agent undated stock powers for such certificates, executed in blank (or, if any such Equity Interests are uncertificated, confirmation and evidence reasonably satisfactory to Agent that the security interest in such uncertificated securities has been transferred to and perfected by Agent, for the benefit of the Secured Parties, in accordance with Sections 8-313, 8-321 and 9-115 of the UCC or any other similar or local or foreign law that may be applicable); and

(iii) Each Borrower and each Domestic Subsidiary (other than DirectStar) shall, pursuant to the Pledge Agreement, pledge to Agent for its benefit and that of the Lenders,

-87-

all intercompany notes evidencing Debt of such new Subsidiary in favor of such Borrower or such Domestic Subsidiary (which shall be in a form acceptable to Agent).

10.1.12. <u>Pledged Shares</u>. Pledge to Agent, for the benefit itself and Secured Parties, all of the Equity Interests of each of their respective Subsidiaries (other than DirectStar) from time to time pursuant to a Pledge Agreement.

10.1.13. Compliance with Indenture. Comply with the terms and provisions of the Indenture and the Senior Notes.

10.1.14. <u>Reserved</u>.

10.1.15. <u>SEC Filings</u>. Each Borrower shall file each regular, periodic and special report required to be filed by it with the SEC on or before the due dates thereof.

10.1.16. <u>DirecTV Agreement</u>. Agent shall maintain the DirecTV Reserve until such time as Obligors and DirecTV execute and deliver an agreement satisfactory to Agent, Obligors and DirecTV pursuant to which DirecTV shall waive its right of setoff with respect to Accounts owed by DirecTV to any Obligor for such Obligor's installation services performed under such Obligor's contracts with DirecTV.

10.1.17. <u>Intercreditor Agreement</u>. 10.1.18. Not more than 45 days after the Closing Date (or such later date as may be expressly agreed upon by Agent in its sole and absolute discretion), Borrowers shall deliver to Agent a satisfactory intercreditor agreement duly executed and delivered by each surety that has issued and has outstanding surety bonds for the account of any Obligor and by such Obligor, or other evidence satisfactory to Agent that no issuer of surety bonds for the account of any Obligor claims a Lien on property of any Obligor other than the bonded contract, proceeds thereof and materials used in performance of such bonded contract.

10.2. <u>Negative Covenants</u>. For so long as there are any Commitments outstanding and thereafter until Full Payment of the Obligations, each Obligor covenants that it shall not and shall not permit any Subsidiary to:

10.2.1. <u>Fundamental Changes</u>. (i) Merge, reorganize, consolidate or amalgamate with any Person, or liquidate, wind up its affairs or dissolve itself, in each case whether in a single transaction or in a series of related transactions, except that (a) any Obligor or any Subsidiary of an Obligor may merge or amalgamate with an Obligor (provided that an Obligor is the surviving or continuing entity of such merger or amalgamation); (b) any Obligor or Subsidiary of an Obligor may liquidate, wind up or dissolve itself (provided that all of its Property is distributed to an Obligor upon the effectiveness of such liquidation, winding up or dissolution and Agent's Liens in all Property of the liquidated, wound up or dissolved entity continue in uninterrupted effect with the same priority as prior to such liquidation, winding up or dissolution); and (c) the Property or Equity Interests of any Obligor or any Subsidiary of an Obligor may be purchased or otherwise acquired by any Obligor (provided that Agent's Liens in all Property or Equity Interests so purchased or acquired continue in uninterrupted effect with the same priority as prior to such purchase or acquisition). Nothing herein shall authorize any merger, consolidation or amalgamation or purchase of assets or Equity Interests if, after giving effect thereto, any Property of an Obligor would be subject to a Lien that is not a Permitted Lien; or (ii) change an Obligor's name or conduct business under any new fictitious name; or change an Obligor's FEIN, organizational identification number or state of organization unless, in each such case, Borrowers shall have given at least 30 days prior notice to Agent and shall have taken such action as Agent may

-88-

reasonably request to maintain the perfection and priority of any Liens of Agent that would otherwise be affected thereby.

10.2.2. Loans. Make any loans or other advances of money to any Person (including to any Subsidiary that is not an Obligor) other than to an officer or employee of an Obligor or a Subsidiary for salary, travel advances, advances against commissions and other similar advances in the Ordinary Course of Business.

10.2.3. Permitted Debt. Create, incur, assume, guarantee or suffer to exist any Debt, except:

(i) the Obligations;

(ii) the Senior Notes;

(iii) accounts payable by such Borrower or a Subsidiary to trade creditors that are not aged more than 90 days from billing date or more than 30 days from the due date, in each case incurred in the Ordinary Course of Business and paid within such time period, unless the same are being Properly Contested or are paid in accordance with Borrowers' customary payment practices and are not in default;

(iv) Debt for rental payments under operating leases incurred in the Ordinary Course of Business of such Borrower or Subsidiary and not secured by a Lien (unless such Lien is a Permitted Lien);

(v) Permitted Purchase Money Debt;

(vi) Debt for accrued payroll, Taxes and other operating expenses (other than for Money Borrowed) incurred in the Ordinary Course of Business of such Borrower or such Subsidiary, including Cash Management Obligations, in each case so long as payment thereof is not past due and payable unless, in the case of Taxes only, such Taxes are being Properly Contested;

(vii) Debt for Money Borrowed by such Obligor (other than the Obligations, Permitted Purchase Money Debt and Subordinated Debt permitted herein), but only to the extent that such Debt is outstanding on the date of this Agreement, as described on Schedule 10.2.3 and is not to be satisfied on or about the Closing Date from the proceeds of the initial Loans;

(viii) Permitted Contingent Obligations;

(ix) Debt of any Person that is in existence at the time that it becomes or is consolidated into or merged with a Subsidiary of such Borrower in a Permitted Acquisition or that is secured by any fixed asset acquired by any Borrower or any Subsidiary at the time of any Permitted Acquisition, provided that such Debt is not incurred in contemplation of such Person becoming a Subsidiary or such acquisition of such asset by any Borrower or any of its Subsidiaries, as the case may be;

(x) Debt that is not included in any of the preceding paragraphs of this **Section 10.2.3**, and is not secured by a Lien (unless such Lien is a Permitted Lien), so long as the sum of the aggregate amount of such Debt plus the aggregate amount of Permitted Purchase Money Debt plus the aggregate amount of Debt described in clause (ix) of **Section 10.2.3** does not exceed at any time, as to all Obligors and all of their Subsidiaries, the sum of (i) \$50,000,000 plus (ii) 5% of MasTec's Consolidated Net Assets (as defined under the Indenture); provided that, to the extent that any portion of the unsecured Debt permitted pursuant to this clause (x) of **Section 10.2.3** is not scheduled to amortize or mature on or prior to the last day of the Term, the Dollar limit set forth herein shall not apply to such portion of such unsecured Debt; and provided further that, in no event shall the aggregate amount of all such Debt exceed the maximum aggregate amount of Debt that any of the Obligors are permitted to incur under the Indenture;

(xi) Refinancing Debt so long as each of the Refinancing Conditions is met; and

(xii) Debt for Money Borrowed provided to Pumpco, by GE Capital Solutions Equipment Finance, a division of General Electric Capital Corporation, or its assignee, secured by equipment and vehicles owned by Pumpco on May 30, 2008, in an aggregate principal amount not to exceed \$25,000,000.

None of the provisions of this **Section 10.2.3** that authorize any Obligor to incur any Debt shall be deemed to (A) override, modify or waive any of the provisions of **Section 10.3**, which shall constitute an independent and separate covenant and obligation of each Borrower, or (B) permit any Obligor to incur any Debt in violation of any provision of the Indenture.

10.2.4. <u>Affiliate Transactions</u>. Enter into, or be a party to any transaction with any Affiliate, officer, director, employee, joint venturer, partner or equityholder (or the immediate family members, spouses and lineal descendants of any of the foregoing Persons that are individuals), except: (i) the transactions contemplated by the Loan Documents; (ii) payment of reasonable compensation to officers and employees for services actually rendered to Borrowers or their respective Subsidiaries; (iii) payment of customary directors' fees and indemnities; (iv) transactions with Affiliates that were consummated prior to the date hereof and have been disclosed to Agent prior to the Closing Date; and (v) transactions with Affiliates in the Ordinary Course of Business and pursuant to the reasonable requirements of such Borrower's or such Subsidiary's business and upon fair and reasonable terms that are fully disclosed to Agent and are no less favorable to such Borrower or such Subsidiary than such Borrower or such Subsidiary would obtain in a comparable arm's length transaction with a Person not an Affiliate or stockholder of such Borrower or such Subsidiary.

-90-

10.2.5. <u>Limitation on Liens</u>. Create or suffer to exist any Lien upon any of its Property, income or profits, whether now owned or hereafter acquired, except the following (collectively, "Permitted Liens"):

(i) Liens at any time granted in favor of Agent;

(ii) Liens for Taxes (excluding any Lien imposed pursuant to any of the provisions of ERISA or the PBA) not yet due or being Properly Contested;

(iii) statutory Liens (excluding any Lien for Taxes, including any Lien imposed pursuant to any of the provisions of ERISA or the PBA) arising in the Ordinary Course of Business of a Borrower or a Subsidiary, but only if and for so long as (x) payment in respect of any such Lien is not at the time required or the Debt secured by any such Liens is being Properly Contested and (y) such Liens do not materially detract from the value of the Property of such Borrower or such Subsidiary and do not materially impair the use thereof in the operation of such Borrower's or such Subsidiary's business;

(iv) Purchase Money Liens securing Permitted Purchase Money Debt;

(v) Liens arising by virtue of the rendition, entry or issuance against such Borrower or any of its Subsidiaries, or any Property of such Borrower or any of its Subsidiaries, of any judgment, writ, order, or decree for so long as each such Lien (a) is in existence for less than 20 consecutive days after it first arises or is being Properly Contested and (b) is at all times junior in priority to any Liens in favor of Agent;

(vi) Liens incurred or deposits made in the Ordinary Course of Business to secure the performance of tenders, bids, leases, contracts (other than for the repayment of Money Borrowed), statutory obligations and other similar obligations or arising as a result of progress payments under government contracts, provided that, to the extent any such Liens attach to any of the Collateral, such Liens are at all times subordinate and junior to the Liens upon the Collateral in favor of Agent;

(vii) Liens in favor of the issuer of any performance, payment or bid bond incurred by a Borrower in the Ordinary Course of Business to the extent that such Liens (a) affect exclusively Collateral consisting of (1) bonded contracts, (2) Inventory, materials and Equipment of such Borrower, wherever located, which are purchased for and installed in or located at projects relating to bonded contracts or (3) proceeds of the foregoing or (b) are subordinated to the Lien of Agent on terms and conditions and pursuant to documentation satisfactory to Agent in its sole discretion;

(viii) easements, rights-of-way, restrictions, covenants or other agreements of record and other similar charges or encumbrances on real Property of such Obligor or a any of its Subsidiaries that do not secure any monetary obligation and do not interfere with the ordinary conduct of the business of such Obligor or such Subsidiary;

(ix) normal and customary rights of setoff upon deposits of cash in favor of banks and other depository institutions and Liens of a collecting bank arising under the UCC on checks and other items of payment in the course of collection;

-91-

(x) Liens in existence immediately prior to the Closing Date that are satisfied in full and released on the Closing Date as a result of the application of such Borrower's cash on hand at the Closing Date or the proceeds of Loans to be made on the Closing Date;

(xi) the Lien of the trustee under the Indenture pursuant to the applicable section thereof on certain property in its possession as security for payment of fees and other amounts owing to it in its capacity as such trustee;

(xii) the Lien of ACE American Insurance Company on the Insurance Cash Collateral Account, so long as (a) the aggregate amount that, on or before the date of this Agreement, has been and after the date hereof shall be, deposited in the Insurance Cash Collateral Account by Obligors does not exceed \$18,500,000; and (b) such Insurance Cash Collateral Account is in lieu of Obligors' providing any letter of credit to such insurer to secure Obligors' obligations arising with respect to insurance policies issued by such insurer for Obligors' account;

(xiii) such other Liens as appear on **Schedule 10.2.5** hereto, to the extent provided therein;

(xiv) Liens in favor of the seller of DirectStar in the assets of DirectStar to the extent such Liens secure the DirectStar's obligations to make the Earn-Out Payments under (and as defined in) the DirectStar Purchase Agreement (as in effect on February 6, 2007);

(xv) any Lien under the Indenture (an "Indenture Lien") arising out of the existence of a Lien in the same assets granted or suffered to exist by MasTec or any of its Subsidiaries that constitutes a Permitted Lien hereunder, provided that such Indenture Lien is granted or suffered to exist in order to prevent a violation of the negative pledge provisions of the Indenture; and

(xvi) such other Liens as Agent and the Required Lenders in their sole discretion may hereafter approve in writing.

The foregoing negative pledge shall not apply to any Margin Stock to the extent that the application of such negative pledge to such Margin Stock would require filings or other actions by any Lender under Regulation U or other regulations of the Board of Governors, or otherwise result in a violation of any such regulations.

10.2.6. <u>Restrictions on Payment of Certain Debt.</u> Make any payment with respect to any Subordinated Debt or take or omit to take any other action or omit to take any other action with respect to any Subordinated Debt, except in accordance with the subordination agreement relative thereto; or amend or modify the terms of any agreement applicable to any Subordinated Debt, other than to extend the time of payment thereof or to reduce the rate of interest payable in connection therewith. To the extent that any payment is permitted to be made with respect to any Subordinated Debt pursuant to the provisions of the subordination agreement applicable thereto, as a condition precedent to Borrowers' authorization to make any such payment, Borrowers shall provide to Agent, not less than 5 Business Days prior to the scheduled payment, a certificate from a Senior Officer of Borrower Agent stating that no Default or Event of Default is in existence as of the date of the certificate or will be in existence as of the date of such payment (both with and without giving effect to the making of such payment), and specifying the amount of principal and interest to be paid.

10.2.7. Distributions. Declare or make any Distributions, other than (a) Upstream Payments, and (b) Permitted Distributions.

-92-

10.2.8. <u>Upstream Payments</u>. Create or suffer to exist any encumbrance or restriction on the ability of a Subsidiary to make any Upstream Payment, except for encumbrances or restrictions (i) pursuant to the Loan Documents, (ii) existing under Applicable Law and (iii) identified and fully disclosed in **Schedule 10.2.8**.

10.2.9. <u>Reserved.</u>

10.2.10. Disposition of Assets. Make any Asset Disposition other than a Permitted Asset Disposition.

10.2.11. <u>Subsidiaries</u>. Except as otherwise provided in **Section 10.1.11** hereof, form or acquire any Subsidiary after the Closing Date or permit any existing Subsidiary to issue any additional Equity Interests except director's qualifying shares.

10.2.12. <u>Bill-and-Hold Sales and Consignments</u>. Make a sale to any customer on a bill-and-hold, guaranteed sale, sale and return, sale on approval or consignment basis, or any sale on a repurchase or return basis.

10.2.13. Restricted Investments. Make or have any Restricted Investment.

10.2.14. <u>Reserved</u>.

10.2.15. Tax Consolidation. File or consent to the filing of any consolidated income tax return with any Person other than a Subsidiary.

10.2.16. <u>Accounting Changes</u>. Make any significant change in accounting treatment or reporting practices, except as may be required by GAAP, or establish a fiscal year different from the Fiscal Year.

10.2.17. <u>Organic Documents</u>. Amend, modify or otherwise change any of the terms or provisions in any of its Organic Documents as in effect on the date hereof (or on the day any Person hereafter becomes an Obligor), except for changes that do not affect in any way (i) such Borrower's or any of its Subsidiaries' right and authority to enter into and perform the Loan Documents to which it is a party (ii) the perfection of Agent's Liens in any of the Collateral, or (iii) the authority and obligation of such Borrower or Subsidiary to perform and to pay the Obligations.

10.2.18. <u>Restrictive Agreements</u>. Enter into or become a party to any Restrictive Agreement; provided that the foregoing shall not apply to (i) the DirectStar Purchase Agreement and those Restrictive Agreements existing on the date hereof and identified on **Schedule 9.1.18** (but shall apply to any amendment or modification expanding the scope of any restriction or condition contained in any such Restrictive Agreement), (ii) restrictions or conditions imposed by any Restrictive Agreement evidencing or governing secured Debt that is permitted by this Agreement if such restrictions or conditions apply only to the Properties securing such Debt, and (iii) provided for under the DirectStar Purchase Agreement with respect to Upstream Payments by DirectStar to Borrowers, or as otherwise identified and fully disclosed in **Schedule 10.2.8**.

10.2.19. <u>Hedging Agreements</u>. Enter into any Hedging Agreement, other than Hedging Agreements entered into in the Ordinary Course of Business to hedge or mitigate risks to which any Borrower or any Subsidiary is exposed in the conduct of its business or the management of its liabilities and not for any speculative purpose.

-93-

10.2.20. <u>Cancellation of Claims</u>. Cancel any claim or debt owing to it, except for reasonable consideration negotiated on an arms-length basis and in the Ordinary Course of Business.

10.2.21. <u>Anti-Terrorism Laws</u>. Conduct any business or engage in any transaction or dealing with any Blocked Person, including the making or receiving any contribution of funds, goods or services to or for the benefit of any Blocked Person; deal in, or otherwise engage in any transaction relating to, any property or interests in property blocked pursuant to Executive Order No. 13224; or engage in on conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in Executive Order No. 13224 or the USA Patriot Act. Borrowers shall deliver to Agent and Lenders any certification or other evidence requested from time to time by Agent or any Lender, in its sole discretion, confirming each Borrower's compliance with this **Section 10.2.21**.

10.2.22. <u>Conduct of Business</u>. Engage in any business other than the business engaged in by it on the Closing Date and any business or activities which are substantially similar, related, incidental or, in the case of DirectStar, complementary thereto.

10.2.23. Multiemployer Plans. Become, or permit any Subsidiary to become a party to a Multiemployer Plan.

10.2.24. <u>Amendments to Other Agreements</u>. Amend the interest rate or principal amount or schedule of payments of principal and interest with respect to any Debt (other than the Obligations), or any dividend rate or redemption schedule applicable to any preferred stock of an Obligor, other than to reduce the interest or dividend rate or to extend any such schedule of payments or redemption schedule, or amend or cause or permit to be amended in any material respect or in any respect that may be adverse to the interests of Agent or Lenders (i) the Indenture or any other agreement at any time governing or evidencing Subordinated Debt or (ii) the general indemnity agreement between any Obligor and any surety that has issued any outstanding surety bonds for the account of such Obligor or any related intercreditor agreement.

10.2.25. <u>Surety Bonds</u>. Incur or permit to exist obligations (including Contingent Obligations) to issuers of surety or performance bonds for the account of Obligors in excess of \$300,000,000 in the aggregate outstanding at any time. For the avoidance of doubt, the obligations described in the preceding sentence shall be calculated based on the Borrowers' estimated cost to complete the applicable projects, not the face value of the bonds themselves.

10.2.26. <u>Real Estate</u>. (i) Create or permit any Lien to exist against all or any portion of the Real Estate except to the extent otherwise expressly permitted in this Agreement, (ii) sell, convey, lease, transfer, assign, hypothecate or otherwise dispose of any interest in all or any portion of the Real Estate, or (iii) enter into any agreement with any Person (other than Agent) that restricts a Borrower's ownership or rights with respect to the Real Estate. Any such sale, conveyance, lease, transfer, encumbrance or restriction of the Real Estate by any Obligor in violation of the foregoing shall be void.

10.3. Financial Covenant.

10.3.1. For so long as there are any Commitments outstanding and thereafter until Full Payment of the Obligations, Borrowers covenant that Borrowers shall maintain a Fixed Charge Coverage Ratio of no less than 1.20 to 1.0, calculated as of the last day of each month for the immediately preceding twelve (12) calendar months for which financial statements and the corresponding Compliance Certificate have been received by Agent in accordance with **Section 10.1.3**.

-94-

SECTION 11. CONDITIONS PRECEDENT

11.1. <u>Conditions Precedent to Initial Credit Extensions</u>. Initial Lenders shall not be required to fund any requested Loan, procure any Letter of Credit, or otherwise extend credit to Borrowers, unless each of the following conditions has been satisfied:

11.1.1. Loan Documents. Each of the Loan Documents shall have been duly executed and delivered to Agent by each of the signatories thereto (and, with the exception of the Notes, in sufficient counterparts for each Lender) and accepted by Agent and Initial Lenders and each Obligor shall be in compliance with all of the terms thereof.

11.1.2. <u>Availability</u>. Agent shall have determined, and Initial Lenders shall be satisfied that, immediately after Initial Lenders have made the initial Revolver Loans to be made on the Closing Date, Issuing Bank has issued the Letters of Credit to be issued on the Closing Date (if any) and Borrowers have paid (or made provision for payment of) all fees and closing costs incurred in connection with the Commitments, and after increasing the Availability Reserve in the amount of any payables of Borrowers that are stretched beyond Borrowers' customary payment practices, Availability is not less than \$25,000,000.

11.1.3. <u>Evidence of Perfection and Priority of Liens</u>. Agent shall have received copies of all filing receipts or acknowledgments issued by any Governmental Authority to evidence any filing or recordation necessary to perfect the Liens of Agent in the Collateral and evidence in form satisfactory to Agent and Initial Lenders that such Liens constitute valid and perfected security interests and Liens, and that there are no other Liens upon any Collateral except for Permitted Liens.

11.1.4. <u>Organic Documents</u>. Agent shall have received copies of the Organic Documents of each Obligor, and all amendments thereto, certified by the Secretary of State or other appropriate officials of the jurisdiction of each Borrower's and each other Obligor's states of organization.

11.1.5. <u>Good Standing Certificates</u>. Agent shall have received good standing certificates or certificates of status for each Obligor, issued by the Secretary of State or other appropriate official of such Obligor's jurisdiction of organization and each jurisdiction where the conduct of such Obligor's business activities or ownership of its Property necessitates qualification.

11.1.6. <u>Opinion Letters</u>. Agent and Lenders shall have received a favorable, written opinion of Greenberg Traurig LLP and the respective local counsel to Obligors, covering, to Agent's satisfaction, the matters set forth on <u>Exhibit F</u> attached hereto, including opinions that the Existing Loan Agreement, as amended and restated by this Agreement, including the provisions of **Section 2.1.6** and the revolving credit facility provided for herein, does not violate the Indenture or any Material Contracts.

11.1.7. <u>Insurance</u>. Agent shall have received certified copies of the property and casualty insurance policies or binders of Obligors with respect to the Collateral, or certificates of insurance with respect to such policies in form acceptable to Agent, and loss payable endorsements on Agent's standard form of loss payee endorsement naming Agent as loss payee with respect to each such policy and certified copies of Obligors' liability insurance policies, including product liability policies, together with endorsements naming Agent as an additional insured, all as required by the Loan Documents.

-95-

11.1.8. Lockbox and Dominion Accounts. Agent shall have received the duly executed agreements establishing the Lockbox and each Dominion Account, in each case with a financial institution acceptable to Agent for the collection or servicing of the Accounts.

11.1.9. <u>Solvency Certificates</u>. Agent and Initial Lenders shall have received certificates satisfactory to them from one or more knowledgeable Senior Officers of each Obligor that, after giving effect to the financing under this Agreement and the issuance of the Letters of Credit, each Obligor is Solvent.

11.1.10. <u>No Labor Disputes</u>. Agent shall have received assurances satisfactory to it that there are no threats of strikes or work stoppages by any employees, or organization of employees, of any Obligor which Agent reasonably determines may have a Material Adverse Effect.

11.1.11. <u>Compliance with Laws and Other Agreements</u>. Agent shall have determined or received assurances satisfactory to it that none of the Loan Documents or any of the transactions contemplated thereby violate any Applicable Law, court order or agreement binding upon any Obligor.

11.1.12. <u>No Material Adverse Change</u>. No material adverse change in the financial condition of any Obligor or in the quality, quantity or value of any Collateral shall have occurred since December 31, 2004.

11.1.13. Accounts Payable. Agent shall have reviewed and found acceptable Obligors' accounts payable and vendor arrangements.

11.1.14. <u>Payment of Fees</u>. Borrowers shall have paid, or made provision for the payment on the Closing Date of, all fees and expenses to be paid hereunder to Agent and Lenders on the Closing Date.

11.1.15. <u>Due Diligence</u>. Agent shall have completed its business and legal due diligence, including a roll forward of its previous Collateral audit, with results acceptable to Agent.

11.1.16. LC Conditions. With respect to the procurement of any Letter of Credit on the Closing Date, each of the LC Conditions is satisfied.

11.1.17. <u>Environmental Matters</u>. Agent shall have received, reviewed and found satisfactory the representations, warranties and disclosures in the Environmental Agreement.

11.1.18. <u>Syndication Market</u>. There shall not have occurred any disruption or change that is materially adverse in the market for syndicated bank credit facilities, or a material disruption of or material adverse change in financial, banking or capital market conditions, in each case as determined by Agent.

11.1.19. <u>Title Certificates</u>. An appropriate officer or senior employee of MasTec, acceptable to Agent, shall have entered into the Agency Appointment as to Vehicle Titles with Agent in respect of all title certificates, and Agent shall be satisfied that certificates of title for all Equipment for which certificates of title have been issued are physically assembled in MasTec's headquarters in Coral Gables, Florida.

-96-

11.1.20. <u>Senior Notes</u>. Agent shall have received evidence satisfactory to Agent (which may be a legal opinion) that the Obligations will constitute "Designated Senior Debt" for purposes of (and as defined in) the Indenture.

11.1.21. Default Under Existing Loan Agreement. No Default or Event of Default exists under the Existing Loan Agreement as of the date hereof.

11.1.22. <u>Mortgage Amendments and Related Real Estate Documents</u>. Agent shall have received and found acceptable each of the Mortgage amendments and the other Related Real Estate Documents with respect to each parcel of Real Estate subject to a Mortgage.

11.2. <u>Conditions Precedent to All Credit Extensions</u>. Lenders shall not be required to fund any Loans or otherwise extend any credit to or for the benefit of Borrowers and Issuing Bank shall have no obligation to issue any Letter of Credit, unless and until each of the following conditions has been and continues to be satisfied:

11.2.1. <u>No Defaults</u>. No Default or Event of Default exists at the time, or would result from the funding, of any Loan or other extension of credit.

11.2.2. <u>Satisfaction of Conditions in Other Loan Documents</u>. Each of the conditions precedent set forth in any other Loan Document shall have been and shall remain satisfied.

11.2.3. <u>No Litigation</u>. No action, proceeding, investigation, regulation or legislation shall have been instituted, threatened or proposed before any court, governmental agency or legislative body to enjoin, restrain or prohibit, or to obtain damages in respect of, or which is related to or arises out of, this Agreement or any of the other Loan Documents or the consummation of the transactions contemplated hereby or thereby.

11.2.4. <u>No Material Adverse Effect</u>. No event shall have occurred and no condition shall exist which has or could be reasonably expected to have a Material Adverse Effect.

11.2.5. <u>Borrowing Base Certificate</u>. Agent shall have received each Borrowing Base Certificate then required by the terms of this Agreement or otherwise requested by Agent.

11.2.6. LC Conditions . With respect to the procurement of any Letter of Credit after the Closing Date, each of the LC Conditions is satisfied.

11.3. <u>Inapplicability of Conditions</u>. None of the conditions precedent set forth in Sections 11.1 or 11.2 shall be conditions to the obligation of (i) each Participating Lender to make payments to BofA pursuant to Section 2.3.2, (ii) each Lender to deposit with Agent such Lender's Pro Rata share of a Borrowing in accordance with Section 4.1.2, (iii) each Lender to fund its Pro Rata share of a Revolver Loan to repay outstanding Swingline Loans to BofA as provided in Section 4.1.3(ii), (iv) each Lender to pay any amount payable to Agent or any other Lender pursuant to this Agreement or (v) Agent to pay any amount payable to any Lender pursuant to this Agreement.

11.4. <u>Limited Waiver of Conditions Precedent</u>. If Lenders shall make any Loan or otherwise extend any credit to Borrowers under this Agreement at a time when any of the foregoing conditions precedent are not satisfied (regardless of whether the failure of satisfaction of any such conditions precedent was known or unknown to Agent or Lenders), the funding of such Loan or other extension of credit shall not operate as a waiver of the right of Agent and Lenders to insist upon the satisfaction of all conditions precedent with respect to each subsequent Borrowing requested by Borrowers or a waiver of

-97-

any Default or Event of Default as a consequence of the failure of any such conditions to be satisfied, unless Agent, with the prior written consent of the Required Lenders, in writing waives the satisfaction of any condition precedent, in which event such waiver shall only be applicable for the specific instance given and only to the extent and for the period of time expressly stated in such written waiver.

SECTION 12. EVENTS OF DEFAULT; REMEDIES ON DEFAULT

12.1. <u>Events of Default</u>. The occurrence or existence of any one or more of the following events or conditions shall constitute an "Event of Default" (each of which Events of Default shall be deemed to exist unless and until waived by Agent and Lenders in accordance with the provisions of **Section 13.9** hereof):

12.1.1. <u>Payment of Obligations</u>. Borrowers shall fail to pay any of the Obligations on the due date thereof (whether due at stated maturity, on demand, upon acceleration or otherwise).

12.1.2. <u>Misrepresentations</u>. Any representation, warranty or other written statement to Agent or any Lender by or on behalf of any Obligor, whether made in or furnished in compliance with or in reference to any of the Loan Documents (including any representation made in any Borrowing Base Certificate), proves to have been false or misleading in any material respect when made or furnished or when reaffirmed pursuant to **Section 9.2** hereof.

12.1.3. <u>Breach of Specific Covenants</u>. Any Obligor shall fail or neglect to perform, keep or observe any covenant contained in **Sections 7.4, 7.6, 7.7, 8.1.1, 8.1.2, 8.2.4, 8.2.5, 8.2.6, 8.5, 10.1.1, 10.1.2, 10.1.3, 10.1.6, 10.1.9, 10.2** or **10.3** hereof on the date that such Obligor is required to perform, keep or observe such covenant.

12.1.4. <u>Breach of Other Covenants</u>. Any Obligor shall fail or neglect to perform, keep or observe any covenant contained in this Agreement (other than a covenant which is dealt with specifically elsewhere in **Section 12.1** hereof) and the breach of such other covenant is not cured to Agent's and the Required Lender's satisfaction within 30 days after the sooner to occur of any Senior Officer's receipt of notice of such breach from Agent or the date on which such failure or neglect first becomes known to any Senior Officer; provided, however, that such notice and opportunity to cure shall not apply in the case of any failure to perform, keep or observe any covenant which is not capable of being cured at all or within such 30-day period or which is a willful and knowing breach by any Obligor.

12.1.5. <u>Default Under Security Documents/Other Agreements</u>. Any Obligor or any other Obligor shall default in the due and punctual observance or performance of any liability or obligation to be observed or performed by it under any of the Other Agreements or Security Documents.

12.1.6. <u>Other Defaults</u>. There shall occur any default or event of default on the part of any Obligor or any Subsidiary under (i) the Indenture, or (ii) under any other agreement, document or instrument to which such Obligor or such Subsidiary is a party or by which such Obligor or such Subsidiary or any of their respective Properties is bound, creating or relating to any Debt (other than the Obligations) in excess of \$2,500,000, in each case if the payment or maturity of such Debt may be accelerated in consequence of such default or event of default or demand for payment of such Debt may be made.

12.1.7. <u>Uninsured Losses</u>. Any loss, theft, damage or destruction of any of the Collateral not fully covered (subject to such deductibles as Agent shall have permitted) by insurance if the amount not covered by insurance exceeds \$2,000,000.

-98-

12.1.8. Material Adverse Effect. There shall occur any event or condition that has a Material Adverse Effect.

12.1.9. Solvency. Any Obligor shall cease to be Solvent.

12.1.10. <u>Insolvency Proceedings</u>. Any Insolvency Proceeding shall be commenced by any Obligor; an Insolvency Proceeding is commenced against any Obligor and any of the following events occur: such Obligor consents to the institution of the Insolvency Proceeding against it, the petition commencing the Insolvency Proceeding is not timely controverted by such Obligor, the petition commencing the Insolvency Proceeding is not dismissed within 60 days after the date of the filing thereof (provided that, in any event, during the pendency of any such period, Lenders shall be relieved from their obligation to make Loans or otherwise extend credit to or for the benefit of Borrowers hereunder), an interim trustee is appointed to take possession all or a substantial portion of the Properties of such Obligor or to operate all or any substantial portion of the business of such Obligor or an order for relief shall have been issued or entered in connection with such Insolvency Proceeding; or any Obligor shall make an offer of settlement extension or composition to its unsecured creditors generally.

12.1.11. <u>Business Disruption; Condemnation</u>. There shall occur a cessation of a substantial part of the business of any Obligor for a period which may be reasonably expected to have a Material Adverse Effect; or any Obligor shall suffer the loss or revocation of any license or permit now held or hereafter acquired by such Obligor which is necessary to the continued or lawful operation of its business; or any Obligor shall be enjoined, restrained or in any way prevented by court, governmental or administrative order from conducting all or any material part of its business affairs; or any material lease or agreement pursuant to which any Obligor leases or occupies any premises on which any Collateral is located shall be canceled or terminated prior to the expiration of its stated term and such cancellation or termination has a Material Adverse Effect or results in an Out-of-Formula Condition; or any material part of the Collateral shall be taken through condemnation or the value of such Property shall be materially impaired through condemnation.

12.1.12. <u>ERISA; Plans</u>. A Reportable Event shall occur which Agent, in its reasonable discretion, shall determine constitutes grounds for the termination of, including by the Pension Benefit Guaranty Corporation, any Plan or for the appointment by the appropriate United States district court of a trustee for any Multiemployer Plan (or similar remedy under the PBA), or any Plan shall be terminated or any such trustee shall be requested or appointed, or any Borrower, any Subsidiary or any other Obligor is in "default" (as defined in Section 4219(c)(5) of ERISA or any similar provision of the PBA) with respect to payments to a Multiemployer Plan resulting from such Borrower's, such Subsidiary's or such other Obligor's complete or partial withdrawal from such Multiemployer Plan.

12.1.13. <u>Challenge to Loan Documents</u>. Any Obligor or any of its Affiliates shall challenge or contest in any action, suit or proceeding the validity or enforceability of any of the Loan Documents, the legality or enforceability of any of the Obligations or the perfection or priority of any Lien granted to Agent, or any of the Loan Documents ceases to be in full force or effect for any reason other than a full or partial waiver or release by Agent and Lenders in accordance with the terms thereof.

12.1.14. <u>Judgment</u>. One or more judgments or orders for the payment of money in an amount that exceeds, individually or in the aggregate, \$1,000,000 shall be entered against any Borrower or any other Obligor and (i) enforcement proceedings shall have been commenced by any creditor upon such judgment or order, (ii) there shall be any period of 30 consecutive days during which a stay of enforcement of such judgment or order, by reason of a pending appeal or otherwise, shall not be in

-99-

effect or (iii) results in the creation or imposition of a Lien upon any of the Collateral that is not a Permitted Lien.

12.1.15. <u>Repudiation of or Default Under Guaranty</u>. Any Guarantor shall revoke or attempt to revoke the Guaranty signed by such Guarantor, shall repudiate such Guarantor's liability thereunder, or shall be in default under the terms thereof, or shall fail to confirm in writing, promptly after receipt of Agent's written request therefor, such Guarantor's ongoing liability under the Guaranty in accordance with the terms thereof.

12.1.16. <u>Criminal Forfeiture</u>. Any Obligor (or any of its Senior Officers) is criminally indicted or convicted for (i) a felony committed in the conduct of the business of such Obligor or (ii) any law that could lead to a forfeiture of any material (as determined by Agent in the exercise of its discretion) Collateral.

12.1.17. Change of Control. A Change of Control shall occur.

12.2. <u>Acceleration of Obligations; Termination of Commitments</u>. Without in any way limiting the right of Agent to demand payment of any portion of the Obligations payable on demand in accordance with this Agreement upon or at any time after the occurrence of an Event of Default (other than pursuant to **Section 12.1.10** hereof) and for so long as such Event of Default shall exist, Agent may in its discretion (and, upon receipt of written instructions to do so from the Required Lenders, shall) (a) declare the principal of and any accrued interest on the Loans and all other Obligations owing under any of the Loan Documents to be, whereupon the same shall become without further notice or demand (all of which notice and demand each Borrower expressly waives), forthwith due and payable and Borrowers shall forthwith pay to Agent the entire principal of and accrued and unpaid interest on the Loans and other Obligations plus reasonable attorneys' fees and court costs if such principal and interest are collected by or through an attorney-at-law and (b) terminate the Revolver Commitments; provided, however, that upon the occurrence of an Event of Default specified in **Section 12.1.10**, all of the Obligations shall become automatically due and payable without declaration, notice or demand by Agent to or upon any Borrower or any other Obligor and the Revolver Commitments shall automatically terminate as if terminated by Agent pursuant to **Section 6.2.1** and with the effects specified in **Section 6.2.4**.

12.3. <u>Other Remedies</u>. Upon and after the occurrence of an Event of Default and for so long as such Event of Default shall exist, Agent may in its discretion (and, upon receipt of written direction of the Required Lenders, shall) institute any Enforcement Action and exercise from time to time the following rights and remedies:

12.3.1. All of the rights and remedies of a secured party under the UCC, the PPSA, the Civil Code of Quebec or under other Applicable Law, and all other legal and equitable rights to which Agent may be entitled under any of the Loan Documents, all of which rights and remedies shall be cumulative and shall be in addition to any other rights or remedies contained in this Agreement or any of the other Loan Documents, and none of which shall be exclusive.

12.3.2. The right to collect all amounts at any time payable to an Obligor from any Account Debtor or other Person at any time indebted to such Obligor.

12.3.3. The right to take immediate possession of any of the Collateral, including all certificates of title with respect to Obligors' motor vehicles, trailers and other Property for which a certificate of title has been issued, and to (i) require Obligors to assemble the Collateral, at Borrowers' expense, and make it available to Agent at a place designated by Agent which is

-100-

reasonably convenient to both parties, and (ii) enter any premises where any of the Collateral shall be located and to keep and store the Collateral on said premises until sold (and if said premises be owned or leased Property of an Obligor, then Borrowers agree not to charge Agent for storage of any Collateral thereon).

12.3.4. The right to sell or otherwise dispose of all or any Collateral in its then condition, or after any further manufacturing or processing thereof, at public or private sale or sales, with such notice as may be required by Applicable Law, in lots or in bulk, for cash or on credit, all as Agent, in its sole discretion, may deem advisable. Each Borrower agrees that any requirement of notice to any Borrower or any other Obligor of any proposed public or private sale or other disposition of Collateral by Agent shall be deemed reasonable notice thereof if given at least 10 days prior thereto, and such sale may be at such locations as Agent may designate in said notice. Agent shall have the right to conduct such sales on any Borrower's or any other Obligor's premises, without charge therefor, and such sales may be adjourned from time to time in accordance with Applicable Law. Agent shall have the right to sell, lease or otherwise dispose of the Collateral, or any part thereof, for cash, credit or any combination thereof, and Agent may purchase all or any part of the Collateral at public or, if permitted by law, private sale and, in lieu of actual payment of such purchase price, may set off the amount of such price against the Obligations. The proceeds realized from the sale or other disposition of any Collateral may be applied, after allowing 2 Business Days for collection, first to any Extraordinary Expenses incurred by Agent and then to the remainder of the Obligations as specified in **Section 5.6.1**.

12.3.5. The right to obtain the appointment of a receiver, without notice of any kind whatsoever, to take possession of the Collateral and to exercise such rights and powers as the court appointing such receiver shall confer upon such receiver.

12.3.6. The right to exercise all of Agent's remedies under the Mortgages with respect to any Real Estate.

12.3.7. The right to require Borrowers to Cash Collateralize outstanding Letters of Credit, and, if Borrowers fail promptly to make such deposit, Lenders may (and shall upon the direction of the Required Lenders) advance such amount as a Revolver Loan (whether or not an Out-of-Formula Condition exists or is created thereby). Any such deposit or advance shall be held by Agent in the Cash Collateral Account to fund future payments on any Letter of Credit. At such time as all Letters of Credit have been drawn upon or expired, any amounts remaining in the Cash Collateral Account shall be applied against any outstanding Obligations, or, after Full Payment of all Obligations, returned to Borrowers.

Agent is hereby granted an irrevocable, non-exclusive license or other right to use, license or sub-license (exercisable without payment of royalty or other compensation to any Obligor or any other Person) any or all of each Obligor's Intellectual Property and all of each Obligor's computer hardware and software trade secrets, brochures, customer lists, promotional and advertising materials, labels, and packaging materials, and any Property of a similar nature, in advertising for sale, marketing, selling and collecting and in completing the manufacturing of any Collateral, and each Obligor's rights under all licenses and all franchise agreements shall inure to Agent's benefit, provided that the foregoing shall not be deemed to confer upon Agent the right of use or any sub-license under any License Agreement under which the Licensor has not consented to such right of use of sub-licenses.

12.4. <u>Setoff</u>. In addition to any Liens granted under any of the Loan Documents and any rights now or hereafter available under Applicable Law, Agent and each Lender (and each of their respective Affiliates) is hereby authorized by Borrowers at any time that an Event of Default exists, without notice to

-101-

Borrowers or any other Person (any such notice being hereby expressly waived), to set off and to appropriate and apply any and all deposits, general or special (including Debt evidenced by certificates of deposit whether matured or unmatured (but not including trust accounts)) and any other Debt at any time held or owing by such Lender or any of their Affiliates to or for the credit or the account of any Obligor against and on account of the Obligations of Obligors arising under the Loan Documents to Agent, such Lender or any of their Affiliates, including all Loans and LC Obligations and all claims of any nature or description arising out of or in connection with this Agreement, irrespective of whether or not (i) Agent or such Lender shall have made any demand hereunder, (ii) Agent, at the request or with the consent of the Required Lenders, shall have declared the principal of and interest on the Loans and other amounts due hereunder to be due and payable as permitted by this Agreement and even though such Obligations may be contingent or unmatured or (iii) the Collateral for the Obligations is adequate. Notwithstanding the foregoing, each of Agent and Lenders agree with each other that it shall not, without the express consent of the Required Lenders, and that it shall (to the extent that it is lawfully entitled to do so) upon the request of the Required Lenders, exercise its setoff rights hereunder against any accounts of any Obligor now or hereafter maintained with Agent, such Lender or any Affiliate of any of them, but no Obligor shall have any claim or cause of action against Agent or any Lender for any setoff made without the consent of the Required Lenders and the validity of any such setoff shall not be impaired by the absence of such consent. If any party (or its Affiliate) exercises the right of setoff provided for hereunder, such party shall be obligated to share any such setoff in the manner and to the extent required by **Section 13.5**.

12.5. Remedies Cumulative; No Waiver.

12.5.1. All covenants, conditions, provisions, warranties, guaranties, indemnities, and other undertakings of Borrowers contained in this Agreement, the other Loan Documents, or any other agreement between Agent or any Lender and any Obligor, heretofore, concurrently, or hereafter entered into, shall be deemed cumulative to and not in derogation or substitution of any of the terms, covenants, conditions, or agreements of Borrowers herein contained. The rights and remedies of Agent and Lenders under this Agreement and the other Loan Documents shall be cumulative and not exclusive of any rights or remedies that Agent or any Lender would otherwise have.

12.5.2. The failure or delay of Agent or any Lender to require strict performance by Obligors of any provision of any of the Loan Documents or to exercise or enforce any rights, Liens, powers or remedies under any of the Loan Documents or with respect to any Collateral shall not operate as a waiver of such performance, Liens, rights, powers and remedies, but all such requirements, Liens, rights, powers, and remedies shall continue in full force and effect until the Full Payment of all Loans and all other Obligations owing or to become owing from Obligors to Agent and Lenders. None of the undertakings, agreements, warranties, covenants and representations of Obligors contained in this Agreement or any of the other Loan Documents and no Event of Default by any Obligor under this Agreement or any other Loan Documents shall be deemed to have been suspended or waived by Agent or any Lender, unless such suspension or waiver is by an instrument in writing specifying such suspension or waiver and is signed by a duly authorized representative of Agent or such Lender and directed to Borrowers.

12.5.3. If Agent or any Lender shall accept performance by an Obligor, in whole or in part, of any obligation that such Obligor is required by any of the Loan Documents to perform only when a Default or Event of Default exists, or if Agent or any Lender shall exercise any right or remedy under any of the Loan Documents that may not be exercised other than when a Default or Event of Default exists, Agent's or Lender's acceptance of such performance by an Obligor or Agent's or Lender's exercise of any such right or remedy shall not operate to waive any such Event

-102-

of Default or to preclude the exercise by Agent or any Lender of any other right or remedy, unless otherwise expressly agreed in writing by Agent or such Lender, as the case may be.

SECTION 13. AGENTS

13.1. Appointment of Agent and Syndication Agent; Authority and Duties of Agent.

13.1.1. Each Lender hereby irrevocably appoints and designates BofA as Agent and GECC as Syndication Agent to act as herein specified. Agent may, and each Lender by its acceptance of a Note and becoming a party to this Agreement shall be deemed irrevocably to have authorized Agent to, enter into all Loan Documents to which Agent is or is intended to be a party and all amendments hereto and all Security Documents at any time executed by any Obligor, for its benefit and the Pro Rata benefit of Lenders and, except as otherwise provided in this Section 13, to exercise such rights and powers under this Agreement and the other Loan Documents as are specifically delegated to Agent by the terms hereof and thereof, together with such other rights and powers as are reasonably incidental thereto. Each Lender agrees that any action taken by Agent or the Required Lenders in accordance with the provisions of this Agreement or the other Loan Documents, and the exercise by Agent or the Required Lenders of any of the powers set forth herein or therein, together with such other powers as are reasonably incidental thereto, shall be authorized and binding upon all Lenders. Without limiting the generality of the foregoing, Agent shall have the sole and exclusive right and authority to (a) act as the disbursing and collecting agent for Lenders with respect to all payments and collections arising in connection with this Agreement and the other Loan Documents; (b) execute and deliver as Agent each Loan Document and each Lien Waiver and accept delivery of each such agreement delivered by any Obligor or any other Person; (c) act as collateral agent for Secured Parties for purposes of the perfection of all security interests and Liens created by this Agreement or the Security Documents and, subject to the direction of the Required Lenders, for all other purposes stated therein, provided that Agent hereby appoints, authorizes and directs each Lender to act as a collateral sub-agent for Agent and the other Lenders for purposes of the perfection of all security interests and Liens with respect to an Obligor's Deposit Accounts maintained with, and all cash and Cash Equivalents held by, such Lender; (d) subject to the direction of the Required Lenders, manage, supervise or otherwise deal with the Collateral; and (e) except as may be otherwise specifically restricted by the terms of this Agreement and subject to the direction of the Required Lenders, exercise all remedies given to Agent with respect to any of the Collateral under the Loan Documents relating thereto, Applicable Law or otherwise. The duties of Agent shall be ministerial and administrative in nature, and Agent shall not have by reason of this Agreement or any other Loan Document a fiduciary relationship with any Lender (or any Lender's participants). Unless and until its authority to do so is revoked in writing by Required Lenders, Agent alone shall be authorized to determine whether any Accounts, Fixed Assets or Real Estate constitute Eligible Accounts, Eligible Fixed Assets or Eligible Real Estate (basing such determination in each case upon the meanings given to such terms in Section 1), or whether to impose or release any reserve, and to exercise its own credit judgment in connection therewith, which determinations and judgments, if exercised in good faith, shall exonerate Agent from any liability to Lenders or any other Person for any errors in judgment.

13.1.2. Agent (which term, as used in this sentence, shall include reference to Agent's officers, directors, employees, attorneys, agents and Affiliates and to the officers, directors, employees, attorneys and agents of Agent's Affiliates) shall not: (a) have any duties or responsibilities except those expressly set forth in this Agreement and the other Loan Documents or (b) be required to take, initiate or conduct any Enforcement Action (including any litigation, foreclosure or collection proceedings hereunder or under any of the other Loan Documents) except to the extent directed to do so by the Required Lenders during the continuance of any Event of

-103-

Default. The conferral upon Agent of any right hereunder shall not imply a duty on Agent's part to exercise any such right unless instructed to do so by the Required Lenders in accordance with this Agreement.

13.1.3. Agent may perform any of its duties by or through its agents and employees and may employ one or more Agent Professionals and shall not be responsible for the negligence or misconduct of any such Agent Professionals selected by it with reasonable care. Borrowers shall promptly (and in any event, **on demand**) reimburse Agent for all reasonable expenses (including all Extraordinary Expenses) incurred by Agent pursuant to any of the provisions hereof or of any of the other Loan Documents or in the execution of any of Agent's duties hereby or thereby created or in the exercise of any right or power herein or therein imposed or conferred upon it or Lenders (excluding, however, general overhead expenses), and each Lender agrees promptly to pay to Agent, **on demand**, such Lender's Pro Rata share of any such reimbursement for expenses (including Extraordinary Expenses) that is not timely made by Borrowers to Agent.

13.1.4. The rights, remedies, powers and privileges conferred upon Agent hereunder and under the other Loan Documents may be exercised by Agent without the necessity of the joinder of any other parties unless otherwise required by Applicable Law. If Agent shall request instructions from the Required Lenders with respect to any act or action (including the failure to act) in connection with this Agreement or any of the other Loan Documents, Agent shall be entitled to refrain from such act or taking such action unless and until Agent shall have received instructions from the Required Lenders; and Agent shall not incur liability to any Person by reason of so refraining. Without limiting the foregoing, no Lender shall have any right of action whatsoever against Agent as a result of Agent acting or refraining from acting hereunder or under any of the Loan Documents pursuant to or in accordance with the instructions of the Required Lenders except for Agent's own gross negligence or willful misconduct in connection with any action taken by it. Notwithstanding anything to the contrary contained in this Agreement, Agent shall not be required to take any action that is in its opinion contrary to Applicable Law or the terms of any of the Loan Documents or that would in its reasonable opinion subject it or any of its officers, employees or directors to personal liability.

13.1.5. Agent shall promptly, upon receipt thereof, forward to each Lender (i) copies of any significant written notices, reports, certificates and other information received by Agent from any Obligor, including each Schedule of Accounts (but only if and to the extent such Obligor is not required by the terms of the Loan Documents to supply such information directly to Lenders) and (ii) copies of the results of any appraisals, or field audits by or other examinations made or prepared by or on behalf of Agent with respect to Obligors or the Collateral (each, a "Report" and collectively, "Reports"). Agent shall have no liability to any Lender for any errors in or omissions from any field audit or other examination of Obligors or the Collateral, unless such error or omission was the direct result of Agent's willful misconduct or gross negligence.

-104-

13.2. Agreements Regarding Collateral and Examination Reports.

13.2.1. Lenders hereby irrevocably authorize Agent to release any Lien with respect to any Collateral (i) upon the termination of the Commitments and Full Payment of the Obligations, (ii) that is the subject of an Asset Disposition which Borrowers certify in writing to Agent is a Permitted Asset Disposition (and Agent may rely conclusively on any such certificate without further inquiry), or (iii) with the written consent of all Lenders. Agent shall have no obligation whatsoever to any of the Lenders to assure that any of the Collateral exists or is owned by an Obligor or is cared for, protected or insured or has been encumbered, or that Agent's Liens have been properly, sufficiently or lawfully created, perfected, protected or enforced or entitled to any particular priority or to exercise any duty of care with respect to any of the Collateral.

13.2.2. Each Lender agrees that neither BofA nor Agent makes any representation or warranty as to the accuracy or completeness of any Report and shall not be liable for any information contained in or omitted from any such Report; agrees that the Reports are not intended to be comprehensive audits or examinations and that BofA or Agent or any other Person performing any audit or examination will inspect only specific information regarding Obligations or the Collateral and will rely significantly upon Borrowers' books and records as well as upon representations of Borrowers' officers and employees; agrees to keep all Reports confidential and strictly for its internal use and not to distribute the Reports (or the contents thereof) to any Person (except to its Participants, attorneys, accountants and other Persons with whom such Lender has a confidential relationship) or use any Report in any other manner; and, without limiting the generality of any other indemnification contained herein, agrees to hold Agent and any other Person preparing a Report harmless from any action that the indemnifying Lender may take or conclusion the indemnifying Lender may reach or draw from any Report in connection with any Loans or other credit accommodations that the indemnifying Lender has made or may make to Borrowers, or the indemnifying Lender's participation in, or its purchase of, a loan or loans of any Obligor, and to pay and protect, and indemnify, defend and hold Agent and each other such Person preparing a Report harmless from and against all claims, actions, proceedings, damages, costs, expenses and other amounts (including attorneys' fees incurred by Agent and any such other Person preparing a Report as the direct or indirect result of any third parties who might obtain all or any part of any Report through the indemnifying Lender.

13.3. <u>Reliance By Agent</u>. Agent shall be entitled to rely, and shall be fully protected in so relying, upon any certification, notice or other communication (including any thereof by telephone, telex, telegram, telecopier message or cable) believed by it to be genuine and correct and to have been signed, sent or made by or on behalf of the proper Person or Persons, and upon advice and statements of Agent Professionals selected by Agent. Without limiting the generality of the foregoing, Agent may rely upon any Notice of Borrowing, LC Request, Notice of Conversion/Continuance or any similar notice or request believed by Agent to be genuine. As to any matters not expressly provided for by this Agreement or any of the other Loan Documents, Agent shall in all cases be fully protected in acting or refraining from acting hereunder and thereunder in accordance with the instructions of the Required Lenders, and such instructions of the Required Lenders and any action taken or failure to act pursuant thereto shall be binding upon Lenders.

13.4. <u>Action Upon Default</u>. Agent shall not be deemed to have knowledge of the occurrence of a Default or an Event of Default unless it has received written notice from a Lender or any or all Borrowers specifying the occurrence and nature of such Default or Event of Default. If Agent shall receive such a notice of a Default or an Event of Default or shall otherwise acquire actual knowledge of any Default or Event of Default, Agent shall promptly notify Lenders in writing and Agent shall take such action and assert such rights under this Agreement and the other Loan Documents, or shall refrain from

-105-

taking such action and asserting such rights, as the Required Lenders shall direct from time to time. If any Lender shall receive a notice of a Default or an Event of Default or shall otherwise acquire actual knowledge of any Default or Event of Default, such Lender shall promptly notify Agent and the other Lenders in writing. As provided in Section 13.3 hereof, Agent shall not be subject to any liability by reason of acting or refraining to act pursuant to any request of the Required Lenders except for its own willful misconduct or gross negligence in connection with any action taken by it. Before directing Agent to take or refrain from taking any action or asserting any rights or remedies under this Agreement and the other Loan Documents on account of any Event of Default, the Required Lenders shall consult with and seek the advice of (but without having to obtain the consent of) each other Lender, and promptly after directing Agent to take or refrain from taking any such action or asserting any such rights, the Required Lenders will so advise each other Lender of the action taken or refrained from being taken and, upon request of any Lender, will supply information concerning actions taken or not taken. In no event shall the Required Lenders, without the prior written consent of each Lender, direct Agent to accelerate and demand payment of the Loans held by one Lender without accelerating and demanding payment of all other Loans or to terminate the Commitments of one or more Lenders without terminating the Commitments of all Lenders. Each Lender agrees that, except as otherwise provided in any of the Loan Documents or with the written consent of Agent and the Required Lenders, it will not take any legal action or institute any action or proceeding against any Obligor with respect to any of the Obligations or Collateral or accelerate or otherwise enforce its portion of the Obligations. Without limiting the generality of the foregoing, none of Lenders may exercise any right that it might otherwise have under Applicable Law to credit bid at foreclosure sales, UCC sales, PPSA sales or other similar sales or dispositions of any of the Collateral except as authorized by Agent and Required Lenders. Notwithstanding anything to the contrary set forth in this Section 13.4 or elsewhere in this Agreement, each Lender shall be authorized to take such action to preserve or enforce its rights against any Obligor where a deadline or limitation period is otherwise applicable and would, absent the taken of specified action, bar the enforcement of Obligations held by such Lender against such Obligor, including the filing of proofs of claim in any Insolvency Proceeding.

13.5. <u>Ratable Sharing</u>. If any Lender shall obtain any payment or reduction (including any amounts received as adequate protection of a bank account deposit treated as cash collateral under the Bankruptcy Code) of any Obligation of Borrowers (whether voluntary, involuntary, through the exercise of any right of set-off or otherwise) in excess of its Pro Rata share of payments or reductions on account of such Obligations obtained by all of the Lenders, such Lender shall forthwith (i) notify the other Lenders and Agent of such receipt and (ii) purchase from the other Lenders such participations in the affected Obligations as shall be necessary to cause such purchasing Lender to share the excess payment or reduction, net of costs incurred in connection therewith, on a Pro Rata basis, provided that if all or any portion of such excess payment or reduction is thereafter recovered from such purchasing Lender or additional costs are incurred, the purchase shall be rescinded and the purchase price restored to the extent of such recovery or such additional costs, but without interest. Each Borrower agrees that any Lender so purchasing a participation from another Lender pursuant to this **Section 13.5** may, to the fullest extent permitted by Applicable Law, exercise all of its rights of payment (including the right of set-off) with respect to such participation as fully as if such Lender were the direct creditor of Borrowers in the amount of such participation.

13.6. Indemnification of Agent Indemnitees.

13.6.1. Each Lender agrees to indemnify and defend the Agent Indemnitees (to the extent not reimbursed by Borrowers, but without limiting the indemnification obligations of Borrowers under any of the Loan Documents), on a Pro Rata basis, and to hold each of the Agent Indemnitees harmless from and against, any and all Indemnified Claims which may be imposed on, incurred by or asserted against any of the Agent Indemnitees in any way related to or arising out of

-106-

this Agreement or any of the other Loan Documents or any other document contemplated by or referred to herein or therein or the transactions contemplated hereby or thereby (including the costs and expenses which Borrowers are obligated to pay under **Section 15.2** hereof or amounts Agent may be called upon to pay in connection with any lockbox or Dominion Account arrangement contemplated hereby or under any indemnity, guaranty or other assurance of payment or performance given by Agent pursuant to **Section 3.4.2** or the enforcement of any of the terms hereof or thereof or of any such other documents, <u>provided</u> that no Lender shall be liable to an Agent Indemnitee for any of the foregoing to the extent that they result solely from the willful misconduct or gross negligence of such Agent Indemnitee.

13.6.2. Without limiting the generality of the foregoing provisions of this **Section 13.6**, if Agent should be sued by any receiver, trustee in bankruptcy, debtor-in-possession or other Person on account of any alleged preference or fraudulent transfer received or alleged to have been received from any Borrower or any other Obligor as the result of any transaction under the Loan Documents, then in such event any monies paid by Agent in settlement or satisfaction of such suit, together with all Extraordinary Expenses incurred by Agent in the defense of same, shall be promptly reimbursed to Agent by Lenders to the extent of each Lender's Pro Rata share.

13.6.3. Without limiting the generality of the foregoing provisions of this **Section 13.6**, if at any time (whether prior to or after the Commitment Termination Date) any action or proceeding shall be brought against any of the Agent Indemnitees by an Obligor or by any other Person claiming by, through or under an Obligor, to recover damages for any act taken or omitted by Agent under any of the Loan Documents or in the performance of any rights, powers or remedies of Agent against any Obligor, any Account Debtor, the Collateral or with respect to any Loans, or to obtain any other relief of any kind on account of any transaction involving any Agent Indemnitees under or in relation to any of the Loan Documents, each Lender agrees to indemnify, defend and hold the Agent Indemnitees harmless with respect thereto and to pay to the Agent Indemnitees such Lender's Pro Rata share of such amount as any of the Agent Indemnitees shall be required to pay by reason of a judgment, decree, or other order entered in such action or proceeding or by reason of any compromise or settlement agreed to by the Agent Indemnitees, including all interest and costs assessed against any of the Agent Indemnitees in defending or compromising such action, together with attorneys' fees and other legal expenses paid or incurred by the Agent Indemnitees in connection therewith; provided, however, that no Lender shall be liable to any Agent Indemnitee for any of the foregoing to the extent that they arise solely from the willful misconduct or gross negligence of such Agent Indemnitee. In Agent's discretion, Agent may also reserve for or satisfy any such judgment, decree or order from proceeds of Collateral prior to any distributions therefrom to or for the account of Lenders.

13.7. Limitation on Responsibilities of Agent. Agent shall in all cases be fully justified in failing or refusing to act hereunder unless it shall have received further assurances to its satisfaction from Lenders of their indemnification obligations under Section **13.6** hereof against any and all Indemnified Claims which may be incurred by Agent by reason of taking or continuing to take any such action. Agent shall not be liable to Lenders for any action taken or omitted to be taken under or in connection with this Agreement or the other Loan Documents except as a result and to the extent of losses caused by Agent's actual gross negligence or willful misconduct. Agent does not assume any responsibility for any failure or delay in performance or breach by any Obligor or any Lender of its obligations under this Agreement or any of the other Loan Documents. Agent does not make to Lenders, and no Lender makes to Agent or the other Lenders, any express or implied warranty, representation or guarantee with respect to any Obligations, the Collateral, the Loan Documents or any Obligor. Neither Agent nor any of its officers, directors, employees, attorneys or agents shall be responsible to Lenders, and no Lender nor any of its agents, attorneys or employees shall be responsible to Agent or the other Lenders, for: (i) any recitals,

-107-

statements, information, representations or warranties contained in any of the Loan Documents or in any certificate or other document furnished pursuant to the terms hereof; (ii) the execution, validity, genuineness, effectiveness or enforceability of any of the Loan Documents; (iii) the validity, genuineness, enforceability, collectibility, value, sufficiency or existence of any Collateral, or the perfection or priority of any Lien therein; or (iv) the assets, liabilities, financial condition, results of operations, business, creditworthiness or legal status of any Obligor or any Account Debtor. Neither Agent nor any of its officers, directors, employees, attorneys or agents shall have any obligation to any Lender to ascertain or inquire into the existence of any Default or Event of Default, the observance or performance by any Obligor of any of the duties or agreements of such Obligor under any of the Loan Documents or the satisfaction of any conditions precedent contained in any of the Loan Documents. Agent may consult with and employ legal counsel, accountants and other experts and shall be entitled to act upon, and shall be fully protected in any action taken in good faith reliance upon, any advice given by such experts.

13.8. Successor Agent and Co-Agents.

13.8.1. Subject to the appointment and acceptance of a successor Agent as provided below, Agent may resign at any time by giving at least 30 days written notice thereof to each Lender and Borrowers. Upon receipt of any notice of such resignation, the Required Lenders, after prior consultation with (but without having to obtain consent of) each Lender, shall have the right to appoint a successor Agent which shall be (i) a Lender, (ii) a United States based affiliate of a Lender, or (iii) a commercial bank that is organized under the laws of the United States or of any State thereof and has a combined capital surplus of at least \$200,000,000 and, provided no Default or Event of Default then exists, is reasonably acceptable to Borrowers (and for purposes hereof, any successor to BofA shall be deemed acceptable to Borrowers). If no successor agent is appointed prior to the effective date of the resignation of Agent, then Agent may appoint, after consultation with Lenders and Borrower Agent, a successor agent from among Lenders. Upon the acceptance by a successor Agent of an appointment to serve as an Agent hereunder, such successor Agent shall thereupon succeed to and become vested with all the rights, powers, privileges and duties of the retiring Agent without further act, deed or conveyance, and the retiring Agent shall be discharged from its duties and obligations hereunder but shall continue to enjoy the benefits of the indemnification set forth in **Sections 13.6** and **15.2** hereof. After any retiring Agent's resignation hereunder as Agent, the provisions of this **Section 13** (including the provisions of **Section 13.6** hereof) shall continue in effect for its benefit in respect of any actions taken or omitted to be taken by it while it was acting as Agent. Notwithstanding anything to the contrary contained in this Agreement, any successor shall resign in accordance with the provisions hereof.

13.8.2. It is the intent of the parties that there shall be no violation of any Applicable Law denying or restricting the right of financial institutions to transact business as agent or otherwise in any jurisdiction. In case of litigation under any of the Loan Documents, or in case Agent deems that by reason of present or future laws of any jurisdiction Agent might be prohibited from exercising any of the powers, rights or remedies granted to Agent or Lenders hereunder or under any of the Loan Documents or from holding title to or a Lien upon any Collateral or from taking any other action which may be necessary hereunder or under any of the Loan Documents, Agent may appoint an additional Person as a separate collateral agent or co-collateral agent which is not so prohibited from taking any of such actions or exercising any of such powers, rights or remedies. If Agent shall appoint an additional Person as a separate collateral agent or co-collateral agent as provided above, each and every remedy, power, right, claim, demand or cause of action intended by any of the Loan Documents to be exercised by or vested in or conveyed to Agent with respect thereto shall be exercisable by and vested in such separate collateral agent or co-collateral

-108-

agent, but only to the extent necessary to enable such separate collateral agent or co-collateral agent to exercise such powers, rights and remedies, and every covenant and obligation necessary to the exercise thereof by such separate collateral agent or co-collateral agent shall run to and be enforceable by either of them. Should any instrument from Lenders be required by the separate collateral agent or co-collateral agent so appointed by Agent in order more fully and certainly to vest in and confirm to him or it such rights, powers, duties and obligations, any and all of such instruments shall, on request, be executed, acknowledged and delivered by Lenders whether or not a Default or Event of Default then exists. In case any separate collateral agent or cocollateral agent, or a successor to either, shall die, become incapable of acting, resign or be removed, all the estates, properties, rights, powers, duties and obligations of such separate collateral agent or co-collateral agent, so far as permitted by Applicable Law, shall vest in and be exercised by Agent until the appointment of a new collateral agent or successor to such separate collateral agent or co-collateral agent.

13.9. Consents, Amendments and Waivers; Out-of-Formula Loans.

13.9.1. No amendment or modification of any provision of this Agreement or any of the other Loan Documents, nor any waiver of any Default or Event of Default, shall be effective without the prior written agreement or consent of the Required Lenders; provided, however, that

(i) without the prior written consent of Agent, no amendment or waiver shall be effective with respect to any provision in any of the Loan Documents (including Section 3.4 and this Section 13) to the extent such provision relates to the rights, duties, immunities, exculpation, indemnification or discretion of Agent;

(ii) without the prior written consent of Issuing Bank, no amendment or waiver with respect to any of the LC Obligations or the provisions of **Sections 2.3, 4.1.3 or 11.2.6** shall be effective;

(iii) without the prior written consent of each affected Lender, no amendment or waiver shall be effective that would (1) increase or otherwise modify any Commitment of such Lender (other than (x) to reduce such Lender's Commitment on a proportionate basis with the same Commitments of other Lenders, or (y) pursuant to **Section 2.1.6** hereof); (2) alter (other than to increase) the rate of interest or any fee payable in respect of any Obligations owed to such Lender; (3) waive or defer collection of any interest or fee payable to such Lender pursuant to **Section 3**; or (4) subordinate the payment of any Obligations owed to such Lender to the payment of any Debt (except as expressly provided in **Section 5.6.1**); and

(iv) without the prior written consent of all Lenders (except a defaulting Lender as provided in **Section 4.2** of this Agreement), no amendment or waiver shall be effective that would (1) waive any Default or Event of Default if the Default or Event of Default relates to any Borrower's failure to observe or perform any covenant that may not be amended without the unanimous written consent of Lenders (and, where so provided hereinafter, the written consent of Agent) as hereinafter set forth; (2) alter the provisions of **Sections 3.6, 3.7, 3.8, 3.10, 5.6, 5.7, 7.1** (except to add to the categories of Property of Borrowers constituting Collateral), **8.4.4, 8.6, 13.9, 15.2, 15.3, 15.4** or **15.16**; (3) amend the definitions (and the other defined terms used in such definitions) of "Pro Rata," "Required Lenders," "Eligible Assignee," "Obligations," "Availability Reserve" or "Borrowing Base", or any provision of this Agreement obligating Agent to take certain actions at the direction of the Required Lenders, or any provision of any of the Loan Documents regarding the Pro Rata treatment or obligations of Lenders; (4) subordinate the priority of any Liens granted to Agent under any of the Loan Documents (including a subordination in favor of the holders of

-109-

Permitted Liens that have priority over Agent's Liens under Applicable Law) and except for Liens granted by an Obligor to financial institutions with respect to amounts on deposit with such financial institutions to cover returned items, processing and analysis charges and other charges in the Ordinary Course of Business that relate to deposit accounts with such financial institutions, or (5) alter the time (other than to accelerate or make more frequent) or amount (other than to increase the amount to be repaid) of repayment of any of the Loans or waive any Event of Default resulting from nonpayment of the Loans on the due date thereof (or within any applicable period of grace), or (6) release any Obligor that is Solvent from liability for any of the Obligations.

Notwithstanding the foregoing, the consent or agreement of Borrowers shall not be necessary to the effectiveness of any amendment or waiver of any provision of this Agreement that deals solely with the rights and duties of Lenders and Agent as among themselves, including **Sections 5.6.1** and **15**. The making of any Loans hereunder by any Lender during the existence of a Default or Event of Default shall not be deemed to constitute a waiver of such Default or Event of Default. Any waiver or consent granted by Lenders hereunder shall be effective only if in writing and then only in the specific instance and for the specific purpose for which it was given.

13.9.2. No Borrower will, directly or indirectly, pay or cause to be paid any remuneration or other thing of value, whether by way of supplemental or additional interest, fee or otherwise, to any Lender (in its capacity as a Lender hereunder) as consideration for or as an inducement to the consent to or agreement by such Lender with any waiver or amendment of any of the terms and provisions of this Agreement or any of the other Loan Documents to the extent that the agreement of all Lenders to any such waiver or amendment is required, unless such remuneration or thing of value is concurrently paid, on the same terms, on a Pro Rata or other mutually agreed upon basis to all Lenders; <u>provided</u>, <u>however</u>, that Borrowers may contract to pay a fee only to those Lenders who actually vote in writing to approve any waiver or amendment of the terms and provisions of this Agreement or any of the other Loan Documents to the extent that such waiver or amendment may be implemented by vote of the Required Lenders and such waiver or amendment is in fact approved.

13.9.3. Any request, authority or consent of any Person who, at the time of making such request or giving such a authority or consent, is a Lender, shall be conclusive and binding upon any Transferee of such Lender.

13.9.4. Unless otherwise directed in writing by the Required Lenders, Agent may require Lenders to honor requests by Borrowers for Out-of-Formula Loans (in which event, and notwithstanding anything to the contrary set forth in **Section 2.1.1** or elsewhere in this Agreement, Lenders shall continue to make Revolver Loans up to their Pro Rata share of the Commitments) and to forbear from requiring Borrowers to cure an Out-of-Formula Condition, (1) when no Event of Default exists (or if an Event of Default exists, when the existence of such Event of Default is not known by Agent), if and for so long as (i) such Out-of-Formula Condition does not continue for a period of more than 10 consecutive Business Days, following which no Out-of-Formula Condition exists for at least 10 consecutive Business Days before another Out-of-Formula Condition exists, (ii) the amount of the Revolver Loans outstanding at any time does not exceed the aggregate of the Commitments at such time, and (iii) the Out-of-Formula Condition is not known by Agent at the time in question to exceed \$5,000,000; and (2) regardless of whether or not an Event of Default exists, if Agent discovers the existence of an Out-of-Formula Condition not previously known by it to exist, but Lenders shall be obligated to continue making such Revolver Loans as directed by Agent only (A) if the amount of the Out-of-Formula Condition is not increased by more than \$5,000,000 above the amount determined by Agent to exist on the date of discovery thereof and (B) for a period not to exceed 5 Business Days. In no event shall any Borrower or any other Obligor

-110-

be deemed to be a beneficiary of this Section 13.9.4 or authorized to enforce any of the provisions of this Section 13.9.4.

13.10. <u>Due Diligence and Non-Reliance</u>. Each Lender hereby acknowledges and represents that it has, independently and without reliance upon Agent, Syndication Agent or the other Lenders, and based upon such documents, information and analyses as it has deemed appropriate, made its own credit analysis of each Obligor and its own decision to enter into this Agreement and to fund the Loans to be made by it hereunder and to purchase participations in the LC Obligations pursuant to **Section 2.3.2** hereof, and each Lender has made such inquiries concerning the Loan Documents, the Collateral and each Obligor as such Lender feels necessary and appropriate, and has taken such care on its own behalf as would have been the case had it entered into the other Loan Documents without the intervention or participation of the other Lenders or Agent. Each Lender hereby further acknowledges and represents that the other Lenders, Syndication Agent and Agent have not made any representations or warranties to it concerning any Obligor, any of the Collateral or the legality, validity, sufficiency or enforceability of any of the Loan Documents. Each Lender also hereby acknowledges that it will, independently and without reliance upon the other Lenders, Syndication Agent or Agent, and based upon such financial statements, documents and information as it deems appropriate at the time, continue to make and rely upon its own credit decisions in making Loans and in taking or refraining to take any other action under this Agreement or any of the other Loan Documents. Except for notices, reports and other information expressly required to be furnished to Lenders by Agent hereunder, Agent shall not have any duty or responsibility to provide any Lender with any notices, reports or certificates furnished to Agent by any Obligor or any credit or other information concerning the affairs, financial condition, business or Properties of any Obligor (or any of its Affiliates) which may come into possession of Agent or any of Agent's Affiliates.

13.11. <u>Representations and Warranties of Lenders</u>. Each Lender represents and warrants to each Borrower, Agent and the other Lenders that it has the power to enter into and perform its obligations under this Agreement and the other Loan Documents, and that it has taken all necessary and appropriate action to authorize its execution and performance of this Agreement and the other Loan Documents to which it is a party, each of which will be binding upon it and the obligations imposed upon it herein or therein will be enforceable against it in accordance with the respective terms of such documents; and none of the consideration used by it to make or fund its Loans or to participate in any other transactions under this Agreement constitutes for any purpose of ERISA or Section 4975 of the Internal Revenue Code assets of any "plan" as defined in Section 3(3) of ERISA or Section 4975 of the Internal Revenue Code and the rights and interests of such Lender in and under the Loan Documents shall not constitute plan assets under ERISA.

13.12. The Required Lenders. As to any provisions of this Agreement or the other Loan Documents under which action may or is required to be taken upon direction or approval of the Required Lenders, the direction or approval of the Required Lenders shall be binding upon each Lender to the same extent and with the same effect as if each Lender joined therein. Notwithstanding anything to the contrary contained in this Agreement, Borrowers shall not be deemed to be a beneficiary of, or be entitled to enforce, sue upon or assert as a defense to any of the Obligations, any provisions of this Agreement that requires Agent or any Lender to act, or conditions their authority to act, upon the direction or consent of the Required Lenders; and any action taken by Agent or any Lender that requires the consent or direction of the Required Lenders as a condition to taking such action shall, insofar as Borrowers are concerned, be presumed to have been taken with the requisite consent or direction of the Required Lenders.

13.13. <u>Several Obligations</u>. The obligations and commitments of each Lender under this Agreement and the other Loan Documents are several and neither Agent nor any Lender shall be responsible for the performance by the other Lenders of its obligations or commitments hereunder or

-111-

thereunder. Notwithstanding any liability of Lenders stated to be joint and several to third Persons under any of the Loan Documents, such liability shall be shared, as among Lenders, Pro Rata.

13.14. <u>Agent in its Individual Capacity</u>. With respect to its obligation to lend under this Agreement, the Loans made by it and each Note issued to it, Agent shall have the same rights and powers hereunder and under the other Loan Documents as any other Lender or holder of a Note and may exercise the same as though it were not performing the duties specified herein; and the terms "Lenders," "Required Lenders," or any similar term shall, unless the context clearly otherwise indicates, include Agent in its capacity as a Lender. Agent and its Affiliates may each accept deposits from, maintain deposits or credit balances for, invest in, lend money to, act as trustee under indentures of, serve as financial advisor to, and generally engage in any kind of business with any Borrower or any other Obligor, or any affiliate of any Borrower or any other Obligor, as if it were any other bank and without any duty to account therefor (or for any fees or other consideration received in connection therewith) to the other Lenders. BofA or its affiliates may receive information regarding any Borrower or any of such Borrower's Affiliates and account debtors (including information that may be subject to confidentiality obligations in favor of Borrowers or any of their Affiliates) and Lenders acknowledge that neither Agent nor BofA shall be under any obligation to provide such information to Lenders to the extent acquired by BofA in its individual capacity and not as Agent hereunder.

13.15. <u>No Third Party Beneficiaries</u>. This Section 13 is not intended to confer any rights or benefits upon Borrowers or any other Person except Lenders and Agent, and no Person (including any or all Borrowers) other than Lenders and Agent shall have any right to enforce any of the provisions of this Section 13 except as expressly provided in Section 13.17 hereof. As between Obligors and Agent, any action that Agent may take or purport to take on behalf of Lenders under any of the Loan Documents shall be conclusively presumed to have been authorized and approved by Lenders as herein provided.

13.16. Notice of Transfer. Agent may deem and treat a Lender party to this Agreement as the owner of such Lender's portion of the Revolver Loans for all purposes, unless and until a written notice of the assignment or transfer thereof executed by such Lender has been received by Agent.

13.17. <u>Replacement of Certain Lenders</u>. If a Lender ("Affected Lender") shall have (i) failed to fund its Pro Rata share of any Loan requested (or deemed requested) by Borrowers which such Lender is obligated to fund under the terms of this Agreement and which such failure has not been cured, (ii) requested compensation from Borrowers under Section 3.7 or Section 3.8 to recover increased costs incurred by such Lender (or its parent or holding company) or compensation required by such Lender which are not being incurred or required generally by the other Lenders (or their respective parents or holding companies), (iii) delivered a notice pursuant to Section 3.6 hereof claiming that such Lender is unable to extend LIBOR Loans to Borrowers for reasons not generally applicable to the other Lenders, (iv) defaulted in paying or performing any of its obligations to Agent, or (v) failed or refused to give its consent to any amendment, waiver or action for which consent of all of the Lenders is required and in respect of which the Required Lenders have consented, then, in any such case and in addition to any other rights and remedies that Agent, any other Lender or any Borrower may have against such Affected Lender, any Borrower or Agent may make written demand on such Affected Lender (with a copy to Agent in the case of a demand by Borrowers and a copy to Borrowers in the case of a demand by Agent) for the Affected Lender to assign, and such Affected Lender swilling to accept such assignment or assignment and Acceptances within 5 Business Days after the date of such demand, to one or more Lenders willing to accept such assignment (including its Commitments and all Loans owing to it) in accordance with Section 14 hereof. Agent is hereby irrevocably authorized to execute one or more Assignment and Acceptances as attorney-in-fact for any Affected Lender which fails or refuses to execute and deliver the same within 5 Business Days after the

-112-

date of such demand. The Affected Lender shall be entitled to receive, in cash and concurrently with execution and delivery of each such Assignment and Acceptance, all amounts owed to the Affected Lender hereunder or under any other Loan Document, including the aggregate outstanding principal amount of the Loans owed to such Lender, together with accrued interest thereon through the date of such assignment (but excluding any prepayment penalty or termination charge). Upon the replacement of any Affected Lender pursuant to this **Section 13.17**, such Affected Lender shall cease to have any participation in, entitlement to, or other right to share in the Liens of Agent in any Collateral and such Affected Lender shall have no further liability to Agent, any Lender or any other Person under any of the Loan Documents (except as provided in **Section 13.6** hereof as to events or transactions which occur prior to the replacement of such Affected Lender), including any Commitment to make Loans or purchase participations in LC Obligations. Agent shall have the right at any time, but shall not be obligated to, upon written notice to any Lender and with the consent of such Lender (which may be granted or withheld in such Lender's sole discretion), to purchase for Agent's own account all of such Lender's right, title and interest in and to this Agreement, the other Loan Documents and the Obligations (together with such Lender's interest in the Commitments), for the face amount of the Obligations owed to such Lender (or such greater or lesser amount as Agent and Lender may mutually agree upon).

13.18. Remittance of Payments and Collections.

13.18.1. All payments by any Lender to Agent shall be made not later than the time set forth elsewhere in this Agreement on the Business Day such payment is due; provided, however, that if such payment is due on demand by Agent and such demand is made on the paying Lender after 11:00 a.m. on such Business Day, then payment shall be made by 11:00 a.m. on the next Business Day. Payment by Agent to any Lender shall be made by wire transfer, promptly following Agent's receipt of funds for the account of such Lender and in the type of funds received by Agent; provided, however, that if Agent receives such funds at or prior to 12:00 noon, Agent shall pay such funds to such Lender by 2:00 p.m. on such Business Day, but if Agent receives such funds after 12:00 noon, Agent shall pay such funds to such Lender by 2:00 p.m. on the next Business Day.

13.18.2. With respect to the payment of any funds from Agent to a Lender or from a Lender to Agent, the party failing to make full payment when due pursuant to the terms hereof shall, on demand by the other party, pay such amount together with interest thereon at the Federal Funds Rate. In no event shall Borrowers be entitled to receive any credit for any interest paid by Agent to any Lender, or by any Lender to Agent, at the Federal Funds Rate as provided herein.

If Agent pays any amount to a Lender in the belief or expectation that a related payment has been or will be received by Agent from an Obligor and such related payment is not received by Agent, then Agent shall be entitled to recover such amount from each Lender that receives such amount. If Agent determines at any time that any amount received by it under this Agreement or any of the other Loan Documents must be returned to an Obligor or paid to any other Person pursuant to any Applicable Law, court order or otherwise, then, notwithstanding any other term or condition of this Agreement or any of the other Loan Documents, Agent shall not be required to distribute such amount to any Lender.

13.19. <u>Syndication Agent</u>. For avoidance of doubt, it is expressly acknowledged and agreed by Agent and each Lender for the benefit of Syndication Agent that, other than any rights or obligations explicitly reserved to or imposed upon Syndication Agent under this Agreement, Syndication Agent, in such capacity, has no rights or obligations hereunder nor shall Syndication Agent, in such capacity, be responsible or accountable to any other party hereto for any action or failure to act hereunder, other than in connection with such explicitly reserved rights or such obligations and then only for claims, damages, losses (other than consequential losses) and other liabilities arising out of Syndication Agent's own gross negligence or willful misconduct.

-113-

SECTION 14. BENEFIT OF AGREEMENT; ASSIGNMENTS AND PARTICIPATIONS

14.1. <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of Obligors, Agent, Syndication Agent and Lenders and their respective successors and assigns (which, in the case of Agent, shall include any successor Agent appointed pursuant to **Section 13.8** hereof), except that (i) no Obligor shall have the right to assign its rights or delegate performance of any of its obligations under any of the Loan Documents and (ii) any assignment by any Lender must be made in compliance with **Section 14.3**. Agent may treat the Person which made any Loan or holds any Note as the owner thereof for all purposes hereof unless and until such Person complies with **Section 14.3** in the case of an assignment thereof or, in the case of any other transfer, a written notice of the transfer is filed with Agent. Any assignee or transferee of any rights with respect to any Note or Loan agrees by acceptance thereof to be bound by all the terms and provisions of the Loan Documents. Any request, authority or consent of any Person, who at the time of making such request or giving such authority or consent is the holder of a Note, shall be conclusive and binding on any subsequent holder, transferee or assignee of such Note or of any Note or Notes issued in exchange therefor.

14.2. Participations.

14.2.1. <u>Permitted Participants; Effect</u>. Any Lender may, in the ordinary course of its business and in accordance with Applicable Law, at any time sell to one or more banks or other financial institutions (each a "Participant") a participating interest in any of the Obligations owing to such Lender, any Commitment of such Lender or any other interest of such Lender under any of the Loan Documents. In the event of any such sale by a Lender of participating interests to a Participant, such Lender's obligations under the Loan Documents shall remain unchanged, such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, such Lender shall remain the holder of its Loans and Commitments for all purposes under the Loan Documents, all amounts payable by Borrowers under this Agreement and any of the Notes shall be determined as if such Lender had not sold such participating interests, and Borrowers and Agent shall continue to deal solely and directly with such Lender in connection with such Lender shall give prompt written notice thereof to Borrowers and the other Lenders. A Participant that would be a Foreign Lender if it were a Lender shall not be entitled to the benefits of **Section 5.9** unless Borrowers are notified of the participation sold to Participant and such Participant agrees, for the benefit of Borrowers, to comply with **Section 5.10** as though such Participant were a Lender.

14.2.2. <u>Voting Rights</u>. Each Lender shall retain the sole right to approve, without the consent of any Participant, any amendment, modification or waiver of any provision of the Loan Documents other than an amendment, modification or waiver with respect to any Loans or Commitment in which such Participant has an interest which forgives principal, interest or fees or reduces the stated interest rate or the stated rates at which fees are payable with respect to any such Loan or Commitment, postpones the Commitment Termination Date, or any date fixed for any regularly scheduled payment of interest or fees on such Loan or Commitment, or releases from liability any Borrower or any Guarantor or releases any substantial portion of any of the Collateral.

14.2.3. <u>Benefit of Set-Off</u>. Each Borrower agrees that each Participant shall be deemed to have the right of set-off provided in **Section 12.4** hereof in respect of its participating interest in amounts owing under the Loan Documents to the same extent and subject to the same requirements under this Agreement (including **Section 13.5**) as if the amount of its participating interest were owing directly to it as a Lender under the Loan Documents, provided that each Lender shall retain the right of set-off provided in **Section 12.4** hereof with respect to the amount of participating interests sold to each Participant. Lenders agree to share with each Participant, and each Participant by exercising the right of

-114-

set-off provided in **Section 12.4** agrees to share with each Lender, any amount received pursuant to the exercise of its right of set-off, such amounts to be shared in accordance with **Section 13.5** hereof as if each Participant were a Lender.

14.2.4. <u>Notices</u>. Each Lender shall be solely responsible for notifying its Participants of any matters relating to the Loan Documents to the extent that any such notice may be required, and neither Agent nor any other Lender shall have any obligation, duty or liability to any Participant of any other Lender. Without limiting the generality of the foregoing, neither Agent nor any Lender shall have any obligation to give notices or to provide documents or information to a Participant of another Lender.

14.3. Assignments.

14.3.1. <u>Permitted Assignments</u>. Subject to its compliance with **Section 14.3.2**, a Lender may, in accordance with Applicable Law, at any time assign to any Eligible Assignee all or any part of its rights and obligations under the Loan Documents, so long as (i) each assignment is of a constant, and not a varying, ratable percentage of all of the transferor Lender's rights and obligations under the Loan Documents with respect to the Loans and the LC Obligations and, in the case of a partial assignment, is in a minimum principal amount of \$5,000,000 (or \$1,000,000 in the case of an assignment between Lenders), in either case unless otherwise agreed by Agent in its sole discretion, and integral multiples of \$1,000,000 in excess of that amount; (ii) except in the case of an assignment in whole of a Lender's rights and obligations under the Loan Documents or an assignment by one original signatory to this Agreement to another such signatory, immediately after giving effect to any assignment, the aggregate amount of the Commitments retained by the transferor Lender shall in no event be less than \$5,000,000 (unless otherwise agreed by Agent in its sole discretion); and (iii) the parties to each such assignment shall execute and deliver to Agent, for its acceptance and recording, an Assignment and Acceptance. Nothing contained herein shall limit in any way the right of a Lender to pledge or assign all or any portion of its rights under this Agreement or with respect to any of the Obligations to (x) any Federal Reserve Bank or the United States Treasury as collateral security pursuant to Regulation A of the Board of Governors and any Operating Circular issued by such Federal Reserve Bank, or (y) direct or indirect contractual counterparties in swap agreements relating to the Loans, <u>provided that</u> any payment by Borrowers to the assigning Lender in respect of any assigned Obligations in accordance with the terms of this Agreement shall satisfy Borrowers' obligations hereunder in respect of such assigned Obligations to t

14.3.2. Effect; Effective Date. Upon (i) delivery to Agent of a notice of assignment substantially in the form attached as **Exhibit H** hereto, together with any consents required by **Section 14.3.1**, and (ii) payment of a \$3,500 fee to Agent for processing any assignment to an Eligible Assignee that is not an Affiliate of the transferor Lender, such assignment shall become effective on the effective date specified in such notice of assignment. The Assignment and Acceptance shall contain a representation and warranty by the Eligible Assignee that the assignment evidenced thereby will not result in a non-exempt "prohibited transaction" under Section 406 of ERISA. On and after the effective date of such assignment, such Eligible Assignee shall for all purposes be a Lender party to this Agreement and the other Loan Document executed by the Lenders and shall have all the rights and obligations of a Lender under the Loan Documents to the same extent as if it were an original party thereto, and no further consent or action by Borrowers, Lenders or Agent shall be required to release the transferor Lender with respect to the Commitment (or portion thereof) of such Lender and Obligations assigned to such Eligible Assignee. Without limiting the generality of the foregoing, such Eligible Assignee shall be subject to and bound by all of the Loan Documents. Upon the consummation of any assignment to an Eligible Assignee pursuant to this **Section 14.3**, the transferor Lender, Agent and Borrowers shall make

-115-

appropriate arrangements so that replacement Notes are issued to such transferor Lender and new Notes or, as appropriate, replacement Notes, are issued to such Eligible Assignee, in each case in principal amounts reflecting their respective Commitments, as adjusted pursuant to such assignment. If the transferor Lender shall have assigned all of its interests, rights and obligations under this Agreement pursuant to **Section 14.3.1** hereof, then (i) such transferor Lender shall no longer have any obligation to indemnify Agent with respect to any transactions, events or occurrences that transpire after the effective date of such assignment, (ii) each Eligible Assignee to which such transferor Lender shall make an assignment shall be responsible to Agent to indemnify Agent in accordance with this Agreement with respect to transactions, events and occurrences transpiring on and after the effective date of such assignment to it, and (iii) the transferor Lender shall continue to be entitled to the benefits of those provisions of the Loan Documents (including indemnities from Obligors) that survive Full Payment of the Obligations.

14.3.3. <u>Dissemination of Information</u>. Each Borrower authorizes each Lender and Agent to disclose to any Participant, any Eligible Assignee or any other Person acquiring an interest in the Loan Documents by operation of law (each a "Transferee"), and any prospective Transferee, any and all information in Agent's or such Lender's possession concerning each Borrower, the Subsidiaries of each Borrower or the Collateral, subject to appropriate confidentiality undertakings on the part of such Transferee.

14.4. <u>Tax Treatment</u>. If any interest in any Loan Document is transferred to any Transferee that is organized under the laws of any jurisdiction other than the United States or any State thereof, the transferor Lender shall cause such Transferee, concurrently with the effectiveness of such transfer, to comply with the provisions of **Section 5.10** hereof.

SECTION 15. MISCELLANEOUS

15.1. <u>Power of Attorney</u>. Each Obligor hereby irrevocably designates, makes, constitutes and appoints Agent (and all Persons designated by Agent) as such Obligor's true and lawful attorney (and agent-in-fact) and Agent, or Agent's designee, may, without notice to such Obligor and in either such Obligor's or Agent's name, but at the cost and expense of Borrowers:

15.1.1. At such time or times as Agent or said designee, in its sole discretion, may determine, endorse such Obligor's name on any Payment Item or proceeds of the Collateral which come into the possession of Agent or under Agent's control.

15.1.2. At any time that an Event of Default exists: (i) demand payment of the Accounts from the Account Debtors, enforce payment of the Accounts by legal proceedings or otherwise, and generally exercise all of such Obligor's rights and remedies with respect to the collection of the Accounts; (ii) settle, adjust, compromise, discharge or release any of the Accounts or other Collateral or any legal proceedings brought to collect any of the Accounts or other Collateral; (iii) sell or assign any of the Accounts and other Collateral upon such terms, for such amounts and at such time or times as Agent deems advisable; (iv) take control, in any manner, of any item of payment or proceeds relating to any Collateral; (v) prepare, file and sign such Obligor's name to a proof of claim in bankruptcy or similar document against any Account Debtor or to any notice of Lien, assignment or satisfaction of Lien or similar document in connection with any of the Collateral; (vi) receive, open and dispose of all mail addressed to such Obligor upon any Payment Item relating to any Collateral and deposit the same to the account of Agent for application to the Obligations; (vii) endorse the name of such Obligor upon any Chattel Paper, Document, Instrument, invoice, freight bill, bill of lading or similar document or agreement relating to any Accounts or Inventory of any Obligor

-116-

and any other Collateral; (ix) use such Obligor's stationery and sign the name of such Obligor to verifications of the Accounts and notices thereof to Account Debtors; (x) use the information recorded on or contained in any data processing equipment and computer hardware and software relating to any other Collateral; (xi) make and adjust claims under policies of insurance; (xii) sign the name of such Obligor to and file any proof of claim in an Insolvency Proceeding of any Account Debtor and on notices of Liens, claims of mechanic's Liens or assignments or releases of mechanic's Liens securing any Accounts; (xiii) take all action as may be necessary to obtain the payment of any letter of credit or banker's acceptance of which such Obligor is a beneficiary; and (xiv) do all other acts and things necessary, in Agent's determination, to fulfill such Obligor's obligations under any of the Loan Documents.

15.2. General Indemnity. Each Obligor hereby agrees to indemnify and defend the Indemnitees against and to hold the Indemnitees harmless from any Indemnified Claim that may be instituted or asserted against or incurred by any of the Indemnitees and that (i) arises out of or relates to this Agreement or any of the other Loan Documents (including any transactions entered into pursuant to any of the Loan Documents, Agent's Lien upon the Collateral, or the performance by Agent or Lenders of their duties or the exercise of any of their rights or remedies under this Agreement or any of the other Loan Documents), or (ii) results from an Obligor's failure to observe, perform or discharge any of such Obligor's covenants or duties hereunder. Without limiting the generality of the foregoing, this indemnity shall extend to any Indemnified Claims instituted or asserted against or incurred by any of the Indemnitees (x) under any Environmental Laws or (other similar laws by reason of a Borrower's or any other Person's failure to comply with laws applicable to solid or hazardous waste materials or other toxic substances) or (y) under any Anti-Terrorism Laws, including any civil fines assessed against Agent or any Lender by any Governmental Authority as a result of conduct of an Obligor. Additionally, if any Taxes (excluding Taxes imposed upon or measured solely by the net income of Agent and Lenders, but including any intangibles tax, stamp tax, recording tax or franchise tax) shall be payable by Agent or any Obligor on account of the execution or delivery of this Agreement, or the execution, delivery, issuance or recording of any of the other Loan Documents, or the creation or repayment of any of the Obligations hereunder, by reason of any Applicable Law now or hereafter in effect, Borrowers shall pay (or shall promptly reimburse Agent and Lenders for the payment of) all such Taxes, including any interest and penalties thereon, and will indemnify and hold Indemnitees harmless from and against all liability in connection the

15.3. <u>Survival of All Indemnities</u>. Notwithstanding anything to the contrary in this Agreement or any of the other Loan Documents, the obligation of each Obligor and each Lender with respect to each indemnity given by it in this Agreement, whether given by any or all Obligors to Agent Indemnitees, Lender Indemnitees or BofA Indemnitees or by any Lender to any Agent Indemnitees or BofA Indemnitees, shall survive the Full Payment of the Obligations, and the termination of any of the Commitments and the resignation of Agent.

15.4. <u>Modification of Agreement</u>. This Agreement may not be modified, altered or amended, except by an agreement in writing signed by Obligors, Agent and Lenders (or, where otherwise expressly allowed by **Section 13** hereof, the Required Lenders in lieu of Agent and Lenders); <u>provided</u>, <u>however</u>, that no consent, written or otherwise, of Obligors shall be necessary or required in connection with any amendment of any of the provisions of **Sections 2.3.2**, **4.1.3**, **5.6** or **13** (other than **Section 13.17**), or any other provision of this Agreement that affects only the rights, duties and responsibilities of Lenders and Agent as among themselves so long as no such amendment imposes any additional obligations on Borrowers.

15.5. <u>Severability</u>. Wherever possible, each provision of this Agreement shall be interpreted in such manner as to be effective and valid under Applicable Law, but if any provision of this Agreement

-117-

shall be prohibited by or invalid under Applicable Law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidating the remainder of such provision or the remaining provisions of this Agreement.

15.6. <u>Cumulative Effect; Conflict of Terms</u>. The provisions of the Other Agreements and the Security Documents are hereby made cumulative with the provisions of this Agreement. Without limiting the generality of the foregoing, the parties acknowledge that this Agreement and the other Loan Documents may use several different limitations, tests or measurements to regulate the same or similar matters and that such limitations, tests and measures are cumulative and each must be performed, except as may be expressly stated to the contrary in this Agreement. Except as otherwise provided in any of the other Loan Documents by specific reference to the applicable provision of this Agreement, if any provision contained in this Agreement is in direct conflict with, or inconsistent with, any provision in any of the other Loan Documents, the provision contained in this Agreement shall govern and control.

15.7. <u>Counterparts; Facsimile Signatures</u>. This Agreement and any amendments hereto may be executed in any number of counterparts and by different parties hereto in separate counterparts, each of which when so executed and delivered shall be deemed to be an original and all of which counterparts taken together shall constitute but one and the same instrument. Loan Documents may be executed by facsimile and the effectiveness of any such Loan Documents and signatures thereon shall, subject to Applicable Law, have the same force and effect as manually signed originals and shall be binding on all parties thereto. Agent may require that any such documents and signatures be confirmed by a manually-signed original thereof, provided that the failure to request or deliver the same shall not limit the effectiveness of any facsimile signature.

15.8. <u>Consent</u>. Whenever the consent of Agent or Lenders (or any combination of Lenders) is required to be obtained under this Agreement or any of the other Loan Documents as a condition to any action, inaction, condition or event, each party whose consent is required shall be authorized to give or withhold its consent in its sole and absolute discretion and to condition its consent upon the giving of additional collateral security for the Obligations, the payment of money or any other matter.

15.9. Notices and Communications.

15.9.1. Except as otherwise provided in **Section 4.1.5**, all notices, requests and other communications to or upon a party hereto shall be in writing (including facsimile transmission or similar writing) and shall be given to such party at the address or facsimile number for such party on the signature pages hereof (or, in the case of a Person who becomes a Lender after the date hereof, at the address shown on the applicable Assignment and Acceptance by which such Person became a Lender) or at such other address or facsimile number as such party may hereafter specify for the purpose by notice to Agent and Obligors in accordance with the provisions of this **Section 15.9**.

15.9.2. Except as otherwise provided in **Section 3.1.5**, each such notice, request or other communication shall be effective (i) if given by facsimile transmission, when transmitted to the facsimile number specified herein for the noticed party and confirmation of receipt is received, (ii) if given by mail, 3 Business Days after such communication is deposited in the U.S. Mail, with first-class postage pre-paid, addressed to the noticed party at the address specified herein, or (iii) if given by personal delivery or by overnight courier delivery, when duly delivered with receipt acknowledged in writing by the noticed party. In no event shall a voicemail message be effective as a notice, communication or confirmation under any of the Loan Documents. Notwithstanding the foregoing, no notice to or upon Agent or BofA pursuant to **Sections 2.2, 2.3, 3.1.2, 4.1** or **6.2.2** shall be effective until after actually received by the individual to whose attention at Agent such notice is

-118-

required to be sent. Any written notice, request or demand that is not sent in conformity with the provisions hereof shall nevertheless be effective on the date that such notice, request or demand is actually received by the individual to whose attention at the noticed party such notice, request or demand is required to be sent.

15.9.3. Electronic mail and intranet websites may be used only to distribute routine communications, such as financial statements, Borrowing Base Certificates and other information required by **Section 10.1.3**, and to distribute Loan Documents for execution by the parties thereto, and may not be used for any other purpose as effective notice under this Agreement or any of the other Loan Documents. Any notice received by Borrower Agent shall be deemed received by all Obligors.

Agent and Lenders shall be authorized to rely and act upon any notices (including telephonic communications) purportedly given by or on behalf of any Borrower even if such notices were made in a manner other than as specified herein, were incomplete or were not preceded or followed by any other form of notice specified or required herein, or the terms thereof, as understood by the recipient, varied from any confirmation thereof. Borrowers agree to indemnify and defend each Indemnitee from all losses, costs, expenses and liabilities resulting from the reliance by any such Indemnitee on each telephone communication purportedly given by or on behalf of any Borrower.

15.10. Performance of Obligors' Obligations. If any Obligor shall fail to discharge any covenant, duty or obligation hereunder or under any of the other Loan Documents, Agent may, in its sole discretion at any time or from time to time, for such Obligor's account and at Borrowers' expense, pay any amount or do any act required of Obligors hereunder or under any of the other Loan Documents or otherwise lawfully requested by Agent to (i) enforce any of the Loan Documents or collect any of the Obligations, (ii) preserve, protect, insure or maintain or realize upon any of the Collateral, or (iii) preserve, defend, protect or maintain the validity or priority of Agent's Liens in any of the Collateral, including the payment of any judgment against any Obligor, any insurance premium, any warehouse charge, any finishing or processing charge, any landlord claim, any other Lien upon or with respect to any of the Collateral (whether or not a Permitted Lien). All payments that Agent may make under this Section and all out-of-pocket costs and expenses (including Extraordinary Expenses) that Agent pays or incurs in connection with any action taken by it hereunder shall be reimbursed to Agent by Obligors on demand with interest from the date such payment is made or such costs or expenses are incurred to the date of payment thereof at the Default Rate applicable for Revolver Loans that are Base Rate Loans. Any payment made or other action taken by Agent under this Section shall be without prejudice to any right to assert, and without waiver of, an Event of Default hereunder and to without prejudice to Agent's right proceed thereafter as provided herein or in any of the other Loan Documents.

15.11. <u>Credit Inquiries</u>. Each Obligor hereby authorizes and permits Agent and Lenders (but Agent and Lenders shall have no obligation) to respond to usual and customary credit inquiries from third parties concerning such Obligor or any of its Subsidiaries; provided, however, that Agent and Lenders shall not disclose any confidential information of Obligors except in accordance with **Section 15.17** hereof.

15.12. Time of Essence. Time is of the essence of this Agreement, the Other Agreements and the Security Documents.

15.13. <u>Indulgences Not Waivers</u>. Agent's or any Lender's failure at any time or times hereafter, to require strict performance by Obligors of any provision of this Agreement shall not waive, affect or diminish any right of Agent or any Lender thereafter to demand strict compliance and performance therewith.

-119-

15.14. Entire Agreement; Annexes, Exhibits and Schedules. This Agreement and the other Loan Documents, together with all other instruments, agreements and certificates executed by the parties pursuant to any Loan Document, embody the entire understanding and agreement between the parties hereto and thereto with respect to the subject matter hereof and thereof and supersede all prior agreements, understandings and inducements, whether express or implied, oral or written, regarding the same subject matter. Each of the Annexes, Exhibits and Schedules attached hereto is incorporated into this Agreement and by this reference made a part hereof.

15.15. <u>Interpretation</u>. No provision of this Agreement or any of the other Loan Documents shall be construed against or interpreted to the disadvantage of any party hereto by any court or other governmental or judicial authority by reason of such party having, or being deemed to have, structured, drafted or dictated such provision. The paragraph and section headings are for convenience of reference only and shall not affect the substantive meaning of any provision of this Agreement.

15.16. Obligations of Lenders Several. The obligations of each Lender hereunder are several, and no Lender shall be responsible for the obligations or Commitment of any other Lender. Nothing contained in this Agreement and no action taken by Lenders pursuant hereto shall be deemed to constitute the Lenders to be a partnership, association, joint venture or any other kind of entity. The amounts payable at any time hereunder to each Lender shall be a separate and independent debt, and each Lender shall be entitled, to the extent not otherwise restricted hereunder, to protect and enforce its rights arising out of this Agreement and any of the other Loan Documents and it shall not be necessary for Agent or any other Lender to be joined as an additional party in any proceeding for such purpose.

15.17. Confidentiality. Agent, Syndication Agent and Lenders each agrees to take normal and reasonable precautions to maintain the confidentiality of any confidential information that is delivered or made available by Borrowers to it and that is marked or otherwise identified by Borrowers as confidential (including information made available to Agent, Syndication Agent or any Lender in connection with a visit or investigation by any Person contemplated in Section 10.1.1 hereof), for a period of 24 months following the Commitment Termination Date, except that Agent and any Lender may disclose such information (i) to their respective Affiliates and individuals employed or retained by Agent or such Lender who are or are expected to become engaged in evaluating, approving, structuring, administering or otherwise giving professional advice with respect to any of the Loans or Collateral, including any of their respective legal counsel, auditors or other professional advisors; (ii) to any party to this Agreement from time to time or any Participant, (iii) pursuant the order of any court or administrative agency, (iv) upon the request or demand of any regulatory agency or other Governmental Authority having jurisdiction over Agent, Syndication Agent or such Lender or in accordance with Agent's or Lender's regulatory compliance policies, (v) which has ceased to be confidential other than by an act or omission of Agent or any Lender except as permitted herein or which becomes available to Agent or any Lender on a nonconfidential basis from a source other than Obligors, (vi) to the extent reasonably required in connection with any litigation (with respect to any of the Loan Documents or any of the transactions contemplated thereby) to which Agent, Syndication Agent, any Lender or their respective Affiliates may be a party, (vii) to the extent reasonably required in connection with the exercise of any remedies hereunder, (viii) to any Agent Professionals or Syndication Agent's or any Lender's legal counsel or auditors, (ix) to any actual or proposed Participant, Assignee, counterparty or advisors to any swap or derivative transactions relating to Obligors and the Obligations, or any other Transferee of all or part of a Lender's rights hereunder so long as such Person has agreed in writing to be bound by the provisions of this Section, (x) to the National Association of Insurance Commissioners or any similar organization or to any nationally recognized rating agency that requires access to information about a Lender's portfolio in connection with ratings issued with respect to such Lender, (xi) to the extent required (on the advice of Agent's, Syndication Agent's or such Lender's counsel) by Applicable Law, or (xii) with the consent of Borrowers.

-120-

15.18. <u>Certifications Regarding Indentures</u>. Each Borrower hereby certifies to Agent and Lenders that neither the execution or performance of this Agreement by Borrowers nor the incurrence of any Debt pursuant to the terms of this Agreement or any of the other Loan Documents violates any provision of the Indenture. Each Borrower further certifies to Agent and Lenders that (i) all of the Commitments constitute a "Credit Facility" (or the equivalent term defined in the Indenture) under the Indenture, and (ii) that all Obligations collectively constitute "Senior Debt" (or the equivalent term defined in the Indenture.

15.19. <u>Governing Law</u>. This Agreement has been negotiated, executed and delivered, and shall be deemed to have been made, in Atlanta, Georgia, and shall be governed by and construed in accordance with the internal laws (but without regard to conflict of law provisions) of the State of Georgia, but giving effect to federal laws relating to national banks.

15.20. <u>Consent to Forum</u>. Each Obligor hereby consents to the non-exclusive jurisdiction of any United States federal court sitting in or with direct or indirect jurisdiction over the Northern District of Georgia or any Georgia state or superior court sitting in Cobb County, Georgia, in any action, suit or other proceeding arising out of or relating to this Agreement or any of the other Loan Documents and each Obligor irrevocably agrees that all claims and demands in respect of any such action, suit or proceeding may be heard and determined in any such court and irrevocably waives any objection it may now or hereafter have as to the venue of any such action, suit or proceedings against any Obligor in the courts of any other jurisdiction. Any judicial proceeding commenced by any Obligor against Agent, BofA, any Lender or any holder of any of the Obligations, or any Affiliate of Agent, BofA, any Lender or any holder of any of the Obligations, or any Affiliate of Agent, BofA, any Lender or any holder of any of the Northern District of Georgia or any Georgia state or superior court sitting in or with direct jurisdiction over the Northern District of Georgia or any Georgia state or superior court sitting in Cobb County, Georgia. Nothing in this Agreement shall be deemed or operate to affect the right of Agent to serve legal process in any other manner permitted by law or to preclude the enforcement by Agent of any judgment or order obtained in such forum or the taking of any action under this Agreement to enforce same in any other appropriate forum or jurisdiction.

15.21. <u>Waivers by Obligors.</u> To the fullest extent permitted by Applicable Law, each Obligor waives (i) the right to trial by jury (which Agent and each Lender hereby also waives) in any action, suit, proceeding or counterclaim of any kind arising out of or related to any of the Loan Documents, the Obligations or the Collateral; (ii) presentment, demand and protest and notice of presentment, protest, default, non payment, maturity, release, compromise, settlement, extension or renewal of any or all commercial paper, accounts, contract rights, documents, instruments, chattel paper and guaranties at any time held by Agent on which such Obligor may in any way be liable and hereby ratifies and confirms whatever Agent may do in this regard; (iii) notice prior to taking possession or control of the Collateral or any bond or security which might be required by any court prior to allowing Agent to exercise any of Agent's remedies; (iv) the benefit of all valuation, appraisement and exemption laws; (v) any claim against Agent, Syndication Agent or any Lender, on any theory of liability, for special, indirect, consequential, exemplary or punitive damages (as opposed to direct or actual damages) arising out of, in connection with, or as a result of, any of the Loan Documents, any transaction thereunder or the use of the proceeds of any Loans; and (vi) notice of acceptance hereof. Each Obligor acknowledges that the foregoing waivers are a material inducement to Agent's, Syndication Agent's and Lenders' entering into this Agreement and that Agent and Lenders are relying upon the foregoing waivers in its future dealings with Obligors. Each Obligor warrants and represents that it has reviewed the foregoing waivers with its legal counsel and has knowingly and voluntarily waived its jury trial rights following consultation with

-121-

legal counsel. In the event of litigation, this Agreement may be filed as a written consent to a trial by the Court.

15.22. <u>Amendment and Restatement</u>. The aggregate principal amount of loans (including the aggregate face amount of letters of credit) outstanding under the Existing Loan Agreement as of July 28, 2008 is \$94,332,526.98. This Agreement amends and restates the Existing Loan Agreement in its entirety, and is not intended to be or operate as a novation or an accord and satisfaction of the Existing Loan Agreement or the Obligations evidenced or secured thereby or provided for thereunder.

[Remainder of page intentionally left blank; signatures begin on following page]

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered in Atlanta, Georgia, on the day and year specified at the beginning of this Agreement.

BORROWERS:

MASTEC, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

Address: 800 Douglas Road North Tower, 12th Floor Coral Gables, Florida 33134 Attention: Chief Executive Officer Telecopier No.: (___)____

[CORPORATE SEAL]

MASTEC TC, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

MASTEC FC, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

MASTEC CONTRACTING COMPANY, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

MASTEC SERVICES COMPANY, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

MASTEC OF TEXAS, INC.

By: /s/ C. Robert Campbell

- Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

MASTEC NORTH AMERICA, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer
 - Chief Financial Officer

MASTEC ASSET MANAGEMENT COMPANY, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

CHURCH & TOWER, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

POWER PARTNERS MASTEC, LLC

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and
 - Chief Financial Officer

GLOBETEC CONSTRUCTION, LLC

By: /s/ C. Robert Campbell

- Name: C. Robert Campbell
- Title: Executive Vice President and
 - Chief Financial Officer

THREE PHASE LINE CONSTRUCTION, INC.

By:

Name:

Title:

PUMPCO, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Vice President

GUARANTORS:

PHASECOM SYSTEMS, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

INTEGRAL POWER & TELECOMMUNICATIONS CORPORATION, LTD.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

MASTEC NORTH AMERICA AC, LLC

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

THREE PHASE ACQUISITION CORP.

By:

Revolver Commitment: \$85,000,000

Revolver Commitment: \$75,000,000

Revolver Commitment: \$25,000,000

AGENT AND LENDER:

BANK OF AMERICA, N.A.

By: /s/ Dennis S. Losin Dennis S. Losin, Senior Vice President

Address:

Suite 800 300 Galleria Parkway, N.W. Atlanta, Georgia 30339 Attention: Senior Portfolio Manager Telecopier No.: (404) 607-3277

LENDERS:

GENERAL ELECTRIC CAPITAL CORPORATION, as Syndication Agent and a Lender

By: /s/ Thomas Costello Thomas Costello, Duly Authorized Signatory

Address:

201 Merritt 7 P. O. Box 5201 Norwalk, Connecticut 06856 Attention: Loan Servicing Manager Telecopier No.: (203) 229-5791

PNC BANK, NATIONAL ASSOCIATION

By: /s/ Alex Council Title: Vice President

Address:

4720 Piedmont Row Drive Suite 300 Charlotte, North Carolina 28210 Attention: Alex Council Telecopier No.: (704) 643-7918

SIEMENS FINANCIAL SERVICES, INC.

Title: ______

By:	/s/ Michael Kunst
	Title:

By:

Address: 170 Wood Avenue South 8th Floor Iselin, New Jersey 08830 Attention: Michael Kunst Telecopier No.: (732) 590-6648

IN WITNESS WHEREOF, this Agreement has been duly executed and delivered in Atlanta, Georgia, on the day and year specified at the beginning of this Agreement.

BORROWERS:

MASTEC, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

Address: 800 Douglas Road North Tower, 12th Floor Coral Gables, Florida 33134 Attention: Chief Executive Officer Telecopier No.: (___)____

[CORPORATE SEAL]

MASTEC TC, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

MASTEC FC, INC.

By: /s/ C. Robert Campbell

- Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

MASTEC CONTRACTING COMPANY, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

MASTEC SERVICES COMPANY, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

MASTEC OF TEXAS, INC.

By: /s/ C. Robert Campbell

- Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

MASTEC NORTH AMERICA, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer
 - Chief Fillancial Officer

MASTEC ASSET MANAGEMENT COMPANY, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

CHURCH & TOWER, INC.

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and Chief Financial Officer

POWER PARTNERS MASTEC, LLC

- By: /s/ C. Robert Campbell
 - Name: C. Robert Campbell
 - Title: Executive Vice President and
 - Chief Financial Officer

GLOBETEC CONSTRUCTION, LLC

By: /s/ C. Robert Campbell

- Name: C. Robert Campbell
- Title: Executive Vice President and Chief Financial Officer

THREE PHASE LINE CONSTRUCTION, INC.

By: /s/ Pete Johnson

Name: Pete Johnson Title: President

PUMPCO, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Vice President

GUARANTORS:

PHASECOM SYSTEMS, INC.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

INTEGRAL POWER & TELECOMMUNICATIONS CORPORATION, LTD.

By: /s/ C. Robert Campbell

Name: C. Robert Campbell Title: Executive Vice President and Chief Financial Officer

MASTEC NORTH AMERICA AC, LLC

By: /s/ C. Robert Campbell

Name: C. Robert Campbell

Title: Executive Vice President and Chief Financial Officer

THREE PHASE ACQUISITION CORP.

By: /s/ Robert Apple

Name: Robert Apple Title: President

Revolver Commitment: \$85,000,000

Revolver Commitment: \$75,000,000

Revolver Commitment: \$25,000,000

AGENT AND LENDER:

BANK OF AMERICA, N.A.

By: /s/ Dennis S. Losin Dennis S. Losin, Senior Vice President

Address:

Suite 800 300 Galleria Parkway, N.W. Atlanta, Georgia 30339 Attention: Senior Portfolio Manager Telecopier No.: (404) 607-3277

LENDERS:

GENERAL ELECTRIC CAPITAL CORPORATION, as Syndication Agent and a Lender

By: /s/ Thomas Costello Thomas Costello, Duly Authorized Signatory

Address:

201 Merritt 7 P. O. Box 5201 Norwalk, Connecticut 06856 Attention: Loan Servicing Manager Telecopier No.: (203) 229-5791

PNC BANK, NATIONAL ASSOCIATION

By: /s/ Alex Council Title: Vice President

Address:

4720 Piedmont Row Drive Suite 300 Charlotte, North Carolina 28210 Attention: Alex Council Telecopier No.: (704) 643-7918

SIEMENS FINANCIAL SERVICES, INC.

By: /s/ Michael Kunst Title: Vice President

By: /s/ Jim Fuller Title: Co-Head

Address:

170 Wood Avenue South 8th Floor Iselin, New Jersey 08830 Attention: Michael Kunst Telecopier No.: (732) 590-6648

CERTIFICATIONS REQUIRED BY SECTION 302(A) OF SARBANES-OXLEY ACT OF 2002

I, Jose R. Mas, certify that:

I have reviewed this quarterly report on Form 10-Q of MasTec, Inc.;

Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;

Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;

The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under my supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.

The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 30, 2008

/s/ Jose R. Mas

Jose R. Mas President and Chief Executive Officer (Principal Executive Officer)

CERTIFICATIONS REQUIRED BY SECTION 302(A) OF SARBANES-OXLEY ACT OF 2002

I, C. Robert Campbell, certify that:

I have reviewed this quarterly report on Form 10-Q of MasTec, Inc.;

Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this quarterly report;

Based on my knowledge, the financial statements, and other financial information included in this quarterly report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this quarterly report;

The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and have:

- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under my supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this quarterly report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this quarterly report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this quarterly report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting.

The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):

- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: July 30, 2008

/s/ C. Robert Campbell C. Robert Campbell

Chief Financial Officer (Principal Financial and Accounting Officer)

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of MasTec, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2008 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Jose R. Mas, President and Chief Executive Officer of MasTec, Inc., certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

(1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 30, 2008

/s/ Jose R. Mas Jose R. Mas President and Chief Executive Officer (Principal Executive Officer)

The certification set forth above is being furnished as an exhibit solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and is not being filed as part of the Quarterly Report on Form 10-Q for the period ended June 30, 2008, or as a separate disclosure documents of we or the certifying officers.

CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of MasTec, Inc. (the "Company") on Form 10-Q for the period ended June 30, 2008 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, C. Robert Campbell, Chief Financial Officer of MasTec, Inc., certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of my knowledge:

- (1) The Report fully complies with the requirements of section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: July 30, 2008

/s/ C. Robert Campbell

C. Robert Campbell Chief Financial Officer (Principal Financial and Accounting Officer)

The certification set forth above is being furnished as an exhibit solely pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and is not being filed as part of the Quarterly Report on Form 10-Q for the period ended June 30, 2008, or as a separate disclosure documents of we or the certifying officers.